

REFERENCE ONLY

UNIVERSITY OF LONDON THESIS

Degree Pho Year 2005 Name of Author Poccana. G.

COPYRIGHT

This is a thesis accepted for a Higher Degree of the University of London. It is an unpublished typescript and the copyright is held by the author. All persons consulting the thesis must read and abide by the Copyright Declaration below.

COPYRIGHT DECLARATION

I recognise that the copyright of the above-described thesis rests with the author and that no quotation from it or information derived from it may be published without the prior written consent of the author.

LOANS

Theses may not be lent to individuals, but the Senate House Library may lend a copy to approved libraries within the United Kingdom, for consultation solely on the premises of those libraries. Application should be made to: Inter-Library Loans, Senate House Library, Senate House, Malet Street, London WC1E 7HU.

REPRODUCTION

University of London theses may not be reproduced without explicit written permission from the Senate House Library. Enquiries should be addressed to the Theses Section of the Library. Regulations concerning reproduction vary according to the date of acceptance of the thesis and are listed below as guidelines.

- Before 1962. Permission granted only upon the prior written consent of the A. author. (The Senate House Library will provide addresses where possible).
- In many cases the author has agreed to permit copying upon Β. 1962 - 1974. completion of a Copyright Declaration.
- 1975 1988. Most theses may be copied upon completion of a Copyright C. Declaration.
- D. 1989 onwards. Most theses may be copied.

This thesis comes within category D.



This copy has been deposited in the Library of ______

This copy has been deposited in the Senate House Library, Senate House, Malet Street, London WC1E 7HU.

C:\Documents and Settings\lproctor\Local Settings\Temporary Internet Files\OLK8\Copyright - thesis (2).doc

Functional consequences of HSV-1 infection of dendritic cells

A thesis submitted by

Gabriele Pollara

for the degree of Doctor of Philosophy in the University of London 2005

Department of Immunology & Molecular Pathology University College London Gower Street London WC1T 4JF UMI Number: U593594

All rights reserved

INFORMATION TO ALL USERS

The quality of this reproduction is dependent upon the quality of the copy submitted.

In the unlikely event that the author did not send a complete manuscript and there are missing pages, these will be noted. Also, if material had to be removed, a note will indicate the deletion.



UMI U593594 Published by ProQuest LLC 2013. Copyright in the Dissertation held by the Author. Microform Edition © ProQuest LLC. All rights reserved. This work is protected against unauthorized copying under Title 17, United States Code.



ProQuest LLC 789 East Eisenhower Parkway P.O. Box 1346 Ann Arbor, MI 48106-1346

Abstract

Herpes simplex virus-1 (HSV-1) infects a significant proportion of the population, causing widespread morbidity and occasional mortality. An understanding of the interaction of this virus with the immune system will aid in the development of vaccines to reduce the prevalence of this infection, and in evaluating its potential role as a vector in immunotherapy. Dendritic cells (DC) are the critical initiators of adaptive immune responses to HSV-1 and are likely to encounter the virus early after infection of the skin or mucosa. This thesis aims to describe the functional consequences of the interaction between HSV-1 and DC.

DCs are readily infected by HSV-1. The function of these cells is affected in a multitude of ways. The capacity of infected DC to stimulate T cell proliferation is reduced, following changes in morphology, IL-12 secretion and surface phenotype. However, the infection of DC by HSV-1 also results in activation of both infected and uninfected cells. The secretion and paracrine activity of type I IFN is responsible for the activation of bystander DC and plays an important amplifying role in the secretion of IL-12 by these cells. In contrast, HSV-1 envelope structures bind to the surface of DC and induce maturation directly. The binding of HSV-1 glycoprotein D is critical to the acquisition of this mature phenotype.

In the context of natural HSV-1 infection, these findings suggest the hypothesis that the coupling of viral entry to the activation of DC signalling pathways is counterbalanced by viral disruption of DC maturation. This delays the generation of the immune responses to HSV-1. Nonetheless, the parallel release of type I IFN, resulting in paracrine activation, enables other DC to mount an antiviral immune response To my family, for always being there

Acknowledgments

The work presented in this thesis could not have been carried out without the kind and selfless assistance of a large number of individuals.

First and foremost, special thanks goes to my supervisors Prof David Katz and Prof Benny Chain for giving me the opportunity to carry out these studies in their lab, and for their advice and support at all times. Thank you also to Rob Coffin for his advice and allowing me to use his lab's facilities.

Special thanks goes to Katharina Speidel, my supervisor at the beginning of the project, and to my BSc student, Mansi Rajpopat, both of whom contributed to some of the experiments presented in this thesis. Thank you also to Brian Young for processing my samples for electron microscopy and Meleri Jones for carrying out the antiviral assays.

In addition to those mentioned above, many friends and colleagues have helped me throughout the years, including Charles Alderman, Ali Al-Shangiti, Cheryl Chiang, Ian Gerrard, Matthew Handley, Paul Kellam, Antonia Kwan, Luci MacCormac, Yvonne McGrath, Philippa Newton, Paul Kaye, Ariel Rad, Guy Simpson and Claire Swetman.

Final words of thank you go to all friends and volunteers who donated blood for these experiments, without whom none of this work would have been possible.

Publications

Data and concepts in this thesis have contributed to the following publications (attached at the end of the thesis):

- Pollara G, Jones M, Handley ME, Rajpopat M, Kwan A, Coffin RS, Foster G, • Chain B, Katz DR. Herpes simplex virus type-1-induced activation of myeloid dendritic cells: the roles of virus cell interaction and paracrine type I IFN secretion. J Immunol. 2004 Sep 15;173(6):4108-19.
- Pollara G, Katz DR, Chain BM. • The host response to herpes simplex virus infection. Curr Opin Infect Dis. 2004 Jun;17(3):199-203.
- Pollara G, Speidel K, Samady L, Rajpopat M, McGrath Y, Ledermann J, • Coffin RS, Katz DR, Chain B. Herpes simplex virus infection of dendritic cells: balance among activation, inhibition, and immunity.

J Infect Dis. 2003 Jan 15;187(2):165-78.

Table of contents

Abstract
Acknowledgments
Publications
Table of contents
List of figures15
List of tables
Abbreviations
Chapter 1
Introduction
1.1 Natural history of HSV-1 infection in man
1.2 Biology of HSV-1
1.2.1 Virus structure
1.2.2 Mechanism of viral infection
1.2.2.1 Heparan sulphate
1.2.2.2 Entry receptors
1.2.3 Mechanisms of HSV-1 replication
1.3 Role of immune system in neuronal HSV infection
1.4 Role of the immune system in the control of peripheral HSV-1
disease
1.4.1 Innate immune system
1.4.1.1 Epithelial cells
1.4.1.2 NK cells
1.4.1.3 Complement system
1.4.2 Adaptive immune system
1.4.2.1 Antibody response
1.4.2.2 CD4+ T cell response

38
40
tic cells
40
42
45
46
46
51
53
53
53
54
55
57
57
60
60
61
62
65
66
67
68

2.2 Cell and Tissue Culture	68
2.2.1 Safety and sterile conditions	68
2.2.2 Culture medium	68
2.2.3 Culture condition	68
2.2.4 Cell counting	68
2.3 Cell lines: characteristics and growth conditions	69
2.3.1 BHK cells	69
2.3.2 HeLa cells	69
2.3.3 Storage of cell lines	69
2.4 Primary cells	
2.4.1 Isolation of peripheral blood mononuclear cells	70
2.4.2 Derivation of monocyte-derived dendritic cells	70
2.4.3 Preparation of T cells	71
2.5 HSV-1 preparation	71
2.5.1 Virus growth	71
2.5.2 Viral infectivity assay	72
2.5.3 Viral inactivation	72
2.6 HSV-1 infection of DC	
2.7 HSV-1 serology	74
2.8 HSV-1 replication in DC and HeLa cells	
2.9 DC stimulation	
2.10 Assessment of DC morphology	
2.11 Microscopy	
2.11.1 Confocal microscopy	77
2.11.2 Electron microscopy	
2.12 Secreted cytokine measurement	
2.13 Supernatant transfer studies	
2.14 Detection of antigens	
2.14.1 Extracellular immunofluorescence staining	79

2.14.2 Intracellular Immunofluorescence staining	80
2.14.3 Intracellular cytokine staining	80
2.14.4 Flow cytometry	80
2.15 Proliferation assays	81
2.15.1 Autologous recall assay	81
2.15.2 Allogeneic mixed lymphocyte reaction	81
2.15.3 Autologous concanavalin-A dependent assay	81
2.16 Detection of cell viability	82
2.17 Detection of apoptosis	82
2.18 Neutralisation of DC surface receptors	83
2.19 Analysis of intracellular signalling	83
2.19.1 Western blotting	
2.19.2 SB203580 inhibitor studies	83
2.20 Statistical analysis	84
2.21 List of antibodies	86
2.22 List of buffers	89
Chapter 3	
Characterisation of HSV-1 infection of DC	
3.1 Introduction	
3.2 Results	
3.2.1 DC morphology	93
3.2.2 DC surface phenotype	94
3.2.2.1 Purity of DC population	
3.2.2.2 DC maturation state	95
3.2.3 DC susceptibility to HSV-1 infection	
•	100
3.2.3.1 HSV-1 entry receptor expression	
3.2.3.1 HSV-1 entry receptor expression3.2.3.2 Quantifying HSV-1 infection of DC	100
 3.2.3.1 HSV-1 entry receptor expression	

3.3.1 DC susceptibility to viral infection	109
3.3.1.1 Receptor expression	
3.3.1.2 Quantification of infection efficiency	110
3.3.2 DC permissiveness to viral replication	111
3.3.2.1 In vitro replication of HSV-1 replication in DC	
3.3.2.2 In vivo implications of HSV-1 replication in DC	112
3.3.2.3 Mechanisms and consequences of DC permissiveness	s to HSV-1
replication	
3.3.3 Role of replication in DC in viral pathogenesis	114
3.4 Conclusions	115
Chapter 4	
HSV-1 mediated impairment of DC function	117
4.1 Introduction	118
4.2 Results	119
4.2.1 T cell stimulation by HSV-1 infected DC	119
4.2.1.1 HSV recall responses	
4.2.1.2 PPD recall responses	119
4.2.1.3 Allogeneic MLR proliferation	120
4.2.1.4 ConA stimulated autologous proliferation	120
4.2.2 T cell infection by HSV-1	120
4.2.3 Morphological consequences of HSV-1 infection of DC	125
4.2.3.1 Morphological changes of non-adherent DC	125
4.2.3.2 Morphological changes of fibronectin-adherent DC	125
4.2.3.3 Changes in integrin expression after HSV-1 infection	126
4.2.4 DC viability following HSV-1 infection	131
4.2.5 IL-12 secretion following HSV-1 infection	137
4.2.6 HSV-1 induced changes in DC surface phenotype	137
4.2.6.1 HSV-1 effects on DC differentiation	137
4.2.6.2 HSV-1 effects on surface phenotype	
4.3 Discussion	141
4.3.1 HSV-1 effects on DC-mediated T cell proliferation	141

4.3.1.1 HSV recall responses141
4.3.1.2 Cross-presentation in anti-viral responses
4.3.1.3 HSV-1 mediated inhibition of T cell stimulation144
4.3.1.4 HSV-1 effects on T cell function145
4.3.2 Morphological changes after infection147
4.3.2.1 Morphological changes vs. cell death effects
4.3.2.2 Viral induced changes in DC cytoskeleton149
4.3.2.3 Integrin mediated change in morphology
4.3.2.4 Effects of HSV-1 infection on DC migration153
4.3.3 DC viability after HSV-1 infection
4.3.3.1 HSV-1 and necrotic cell death
4.3.3.2 HSV-1 prevention and induction of apoptosis154
4.3.3.3 HSV-1 induced apoptosis in DC155
4.3.3.4 Mechanisms responsible for HSV-1 induced apoptosis156
4.3.4 Impact on DC IL-12 secretion by HSV-1
4.3.5 Changes in DC surface phenotype163
4.3.5.1 HSV-1 changes in MHC class I expression
4.3.5.2 HSV-1 changes in CD86 expression
4.3.5.3 HSV-1 changes in MHC class II expression165
4.3.5.4 Responses of the uninfected GFP-ve population166
4.3.5.5 Relationship between phenotype changes and T cell stimulation167
4.4 Conclusions167
Chapter 5
Type I IFN: autocrine and paracrine priming factor 169
5.1 Introduction170
5.2 Results
5.2.1 Bystander DC activation by HSV-1 infection
5.2.1.1 Phenotype changes
5.2.1.2 Effects on IL-12 secretion
5.2.2 Characterisation of the soluble factor

5.2.2.1 Confirmation of type I IFN secretion by MDDC......175

5.2.2.2 Confirmation of antiviral activity of type I IFN	176
5.2.2.3 Effect of type I IFN on MDDC phenotype	176
5.2.2.4 Confirmation of type I IFN effect on MDDC secretion of IL-12	177
5.2.3 Mechanism of type I IFN activity on DC	183
5.2.3.1 Signalling consequences of type I IFN	183
5.2.3.2 Role of p38 MAPK activation in IFN α mediated increase in	CD86
expression	183
5.2.4 Role of type I IFN in mediating TLR-dependent DC activation.	184
5.2.4.1 Phenotype changes	184
5.2.4.2 Cytokine secretion	185
5.3 Discussion	189
5.3.1 DC secretion of type I IFN	189
5.3.2 Role of type I IFN in DC physiology	192
5.3.2.1 Type I IFN effects on immature DC phenotype	193
5.3.2.2 Type I IFN effects on immature DC IL-12 secretion	194
5.3.3 Role of type I IFN in TLR induced changes to DC	196
5.3.3.1 Type I IFN and phenotype changes to TLR3 and TLR4	196
5.3.3.2 Role of autocrine type I IFN in LPS induced cytokine secretion	198
5.3.4 Signalling consequences of type I IFN on DC	199
5.3.4.1 Regulation of IL-12 and TNF α secretion	199
5.3.4.2 Differential regulation of CD86 and IL-12	201
5.4 Conclusions	202

Chapter 6

Mechanisms of HSV-1 activation and inhibition of DC 204

6.1 Introduction	205
6.2 Results	207
6.2.1 Type I IFN independent HSV-1 maturation of DC	207
6.2.2 Role of viral gene expression in DC activation and inhibition	207
6.2.2.1 Efficiency of UV inactivation of HSV-1	207
6.2.2.2 Binding of UV-inactivated HSV-1	208
6.2.2.3 Infectivity of UV-inactivated HSV	208

6.2.3 Functional effects of UV-HSV infection of DC	209
6.2.3.1 T cell proliferation induced by UV-HSV infected DC	209
6.2.3.2 Cell viability of UV-HSV infected DC	209
6.2.3.3 Morphology of UV-HSV infected DC	209
6.2.3.4 Surface phenotype of UV-HSV infected DC	210
6.2.4 Role of HSV binding to DC surface	216
6.2.5 Signalling consequences of HSV-1 binding to DC surface	217
6.2.5.1 Activation of signalling pathways by HSV-1	217
6.2.5.2 Role of p38 MAPK in HSV-1 induced DC activation	217
6.2.6 Receptors involved in HSV-1 induced activation of DC	223
6.2.6.1 Role of TLR2 in DC activation by HSV-1	223
6.2.6.2 Role of nectin-1 in DC activation by HSV-1	224
6.2.6.3 Role of HVEM in DC activation by HSV-1	224
6.2.7 Receptor internalisation during infection process	225
6.2.8 Viral ligands involved in DC activation	230
6.2.8.1 Role of gD in upregulation of CD86	230
6.2.8.2 Role of gD in type I IFN secretion by DC	231
6.2.8.3 Role of gD in bystander DC activation	232
6.2.9 Direct activation of DC by gD	232
6.3 Discussion	240
6.3.1 HSV-1 factors that inhibit DC functions	240
6.3.2 Mechanism of HSV induced apoptosis	242
6.3.3 HSV-1 induced DC maturation	242
6.3.3.1 HSV-1 activation of DC independent of type I IFN	242
6.3.3.2 Viral interaction with the cell surface	244
6.3.3.3 HSV-1 activation of NF-kB and p38 pathways	244
6.3.4 DC receptors involved in HSV-1 induced activation	246
6.3.5 Viral ligands responsible for DC activation	248
6.3.5.1 Role of gD in DC activation	249
6.3.5.2 Role of non-gD viral components in DC activation	251
6.3.6 HSV-1 induced cytokine secretion	253
6.3.6.1 Secretion of type I IFN	253

6.3.6.2 Secretion of other cytokines	254
6.3.7 Functional outcome of HSV-1 triggering of host signalling pa	thways
	254
6.3.7.1 Pathogen recognition: advantage to host survival	255
6.3.7.2 Receptor triggering: advantage to viral survival	255
6.4 Conclusions	257

Chapter 7

General discussion 24	59
7.1 Introduction2	60
7.2 Model for establishment and resolution of peripheral HSV-1 lesi	on
2	60
7.2.1 Role of DC impairment2	260
7.2.2 Role of bystander DC2	261
7.2.3 Role of type I IFNs in resolution of peripheral HSV-1 infection2	264
7.2.3.1 Type I IFN in the periphery2	264
7.2.3.2 Type I IFN in secondary lymphoid organs	264
7.3 HSV-DC interactions: implications from laboratory to clinic2	65
7.3.1 Impingement on pathogenesis of HIV-1 infection2	265
7.3.2 Development of a vaccine to HSV-12	266
7.3.3 Use of HSV-1 as a gene delivery vector2	268
7.4 Central role of DC in the pathogenesis of viral infections2	69
7.4.1 Herpesviruses2	270
7.4.2 Paramyxoviruses2	271
7.4.3 Retroviruses2	272
7.4.4 Other viruses2	274
7.5 Future work2	74
References	77

List of figures

Chapter 2 figures

Figure 2.1 Neutralisation of HSV-1 by LP2 clone anti-gD mAb.	75
Figure 2.2 Effect of mock viral preparation on DC phenotype.	76
Figure 2.3 Standard curves for TNFa, IL-12 p40 and p70.	85

Chapter 3 figures

Figure 3.1 Morphology of non-adherent DC.	97
Figure 3.2 DC morphology on a FN substratum.	98
Figure 3.3 Phenotype and purity of the DC population.	99
Figure 3.4 DC expression of HSV-1 entry receptors.	103
Figure 3.5 HSV-1 infection efficiency of DC.	104
Figure 3.6 Quantification and mechanism of GFP expression.	105
Figure 3.7 Assessment of GFP expression as marker of DC infection.	106
Figure 3.8 Quantification of HSV-1 replication in DC.	108

Chapter 4 figures

Figure 4.1 Autologous T cell responses in the presence of HSV-1	122
infected DC.	

Figure 4.2 Memory and primary T cell proliferation elicited by HSV-	123
1 infected DC.	
Figure 4.3 Infection of T cells by HSV-1.	124
Figure 4.4 Morphology of HSV-1 infected DC in suspension.	128
Figure 4.5 Morphology of FN-adherent DC infected with HSV-1.	129
Figure 4.6 Integrin expression of DC infected with HSV-1.	130
Figure 4.7 Change in DC size following HSV-1 infection.	133
Figure 4.8 Changes in DC viability following HSV-1 infection.	134
Figure 4.9 Ultrastructural changes in DC morphology after HSV-1	135
infection.	
Figure 4.10 Induction of apoptosis following HSV-1 infection of DC.	136
Figure 4.11 IL-12 secretion by DC infected with HSV-1.	139
Figure 4.12 Phenotypic changes following HSV-1 infection of DC.	140

Chapter 5 figures

Figure 5.1 Effects of soluble factors released following infection on	173
DC phenotype.	
Figure 5.2 Effects of soluble factors released following HSV-1 infection on DC IL-12 secretion.	174
Figure 5.3 Type I IFN secreted by MDDC infected with HSV-1.	178
Figure 5.4 Antiviral activity of IFNa on HSV-1 infection.	179
Figure 5.5 Effect of IFNa on DC phenotype.	180

Figure 5.6 Role of type I IFN in supHSV-DC in inducing DC upregulation of CD86.	181
Figure 5.7 Effect of IFNa on DC IL-12 secretion.	182
Figure 5.8 Intracellular signals activated by IFNa in DC.	186
Figure 5.9 Role of autocrine/paracrine type I IFN on DC phenotype changes by LPS and Poly(I:C).	187
Figure 5.10 Role of autocrine/paracrine type I IFN on LPS induced secretion of TNFa and IL-12.	188

Chapter 6 figures

-

Figure 6.1 Role of autocrine type I IFN in upregulation of CD86 by HSV-1.	211
Figure 6.2 Assessment of gD on UV-HSV and WT-HSV binding to DC surface.	212
Figure 6.3 Assessment of infectivity and inactivation of UV-HSV.	213
Figure 6.4 Effects of viral gene expression on HSV-1 mediated changes in DC function, viability, and morphology.	214
Figure 6.5 Effects of viral gene expression on HSV-1 mediated changes in DC phenotype.	215
Figure 6.6 Inactivation and cell surface binding capacity of FIX-HSV.	219
Figure 6.7 Surface phenotype changes on DC exposed to FIX-HSV.	220
Figure 6.8 Effects of virus inactivation protocol on DC CD86 expression.	221

Figure 6.9 Signalling consequences of HSV-1 infection of DC.	222
Figure 6.10 Effects of TLR2 ligation on DC stimulation by HSV-1.	226
Figure 6.11 Effects of nectin-1 ligation on DC stimulation by HSV-1.	227
Figure 6.12 Effects of HVEM ligation on DC stimulation by HSV-1.	228
Figure 6.13 Internalisation of TLR2, Nectin-1 and HVEM on DC.	229
Figure 6.14 Role of gD interaction with DC on CD86 expression.	234
Figure 6.15 Specificity of gD interaction with DC.	235
Figure 6.16 Attachment of HSV-1 to DC surface.	236
Figure 6.17 Role of gD in HSV-1 induced type I IFN secretion by DC.	237
Figure 6.18 Role of cell surface gD in bystander DC activation.	238
Figure 6.19 Changes in CD86 expression by recombinant gD.	239

.

List of tables

Chapter 1 tables

Table 1.1 TLR ligands.	47
Table 1.2 Functional consequences of viral infection of MDDC.	64

Chapter 2 tables

Table 2.1 Non-conjugated antibodies to human antigens.	86
Table 2.2 Conjugated antibodies to human antigens.	87
Table 2.3 Antibodies to viral antigens.	88
Table 2.4 Control antibodies.	88
Table 2.5 Antibodies to western blot targets.	89

Abbreviations

· · · ·

AAF	IFNalpha-associated factor
AdV	Adenovirus
APC	Antigen presenting cell
ARE	AU-rich elements
ATL	Adult T cell leukaemia
ConA	ConcanavalinA
DC	Dendritic cell
DRG	Dorsal root ganglia
dsRNA	double stranded RNA
E	Early
EBV	Epstein Barr Virus
EM	Electron microscopy
ER	Endoplasmic reticulum
FAK	Focal adhesion kinase
FCS	Foetal calf serum
FIX-HSV	Formaldehyde-inactivated HSV
FN	Fibronectin
FSC	Forward scatter

20

GAS	IFNgamma-activated-site
gB	Glycoprotein B
HBS\$ /	Hank's balanced salt solution
HCMV	Human cytomegalovirus
HCV	Hepatitis C virus
HEV	High endothelial venules
HIgR	Herpesvirus Ig receptor
HIV	Human immunodeficiency virus
HS	Heparan sulphate
HSP	Heat shock protein
HSV-1	Herpes simplex virus 1
HTLV-1	Human T-cell lymphotropic virus type 1
HUVEC	Human umbilical vein endothelial cells
HVEM	Herpesvirus entry mediator
IE	Immediate early
IFN	Interferon
IFNAR	Type I IFN receptor
IRF	Interferon response factor
IRFE	IRF-1 elements
ISG	Interferon-stimulated gene

.

ISGF3	IFN-stimulated gene factor-3
ISRE	Interferon-stimulated regulatory elements
JAK,	Janus kinase
L	Late
LAM	Lipoarabinomannan
LN	Lymph node
LPS	Lipopolysaccharide
LTA	Lipoteichoic acid
mAb	monoclonal antibody
ManLAM	Mannose-capped lipoarabinomannan
МАРК	Mitogen-Activated Protein Kinase
MCM	Monocyte conditioned medium
MCMV	Murine cytomegalovirus
MDC	Myeloid dendritic cell
MDDC	Monocyte derived dendritic cell
MF F	MV fusion glycoprotein
MLR	Mixed lymphocyte reaction
MMTV	Murine mammary tumour virus
MTT	3-(4,5-dimethylthiazol-2-yl)-2,5-diphenyltetrazolium bromide
MV	Measles virus

MV HA	MV haemagluttinin glycoprotein
MV NP	MV nucleoprotein
MVA /	Modified vaccinia virus Ankara
MyD88	Myeloid differentiation factor 88
PAMP	Pathogen associated molecular pattern
РВМС	Peripheral blood mononuclear cells
PDC	Plasmacytoid dendritic cell
pfu	plaque forming units
PGN	Peptidoglycan
PI	Propidium iodide
PIV	Parainfluenza virus
Poly(I:C)	Polyriboinosinic polyribocytidylic acid
PPD	Purified protein derivative
PRD	Positive regulatory domain
PRR	Pattern recognition receptors
PS	Phosphatidylserine
RSV	Respiratory syncitial virus
SDS	Sodium dodecyl sulphate
SEM	Standard error of the mean
siRNA	short interfering RNA

SSC	Side scatter
ssRNA	single stranded RNA
STAT	Signal transducer and activator of transcription
supCON-DC	supernatant from uninfected DC
supHSV-DC	supernatant from HSV-1 infected DC
ТАР	Transporter associated with antigen processing
TIR	Toll-interleukin 1 receptor-resistance
TIRAP	TIR domain-containing adaptor protein
TLR	Toll like receptor
TNF	Tumour necrosis factor
TNFR	Tumour necrosis factor receptor
TRAF	TNFR-associated factor
TRAIL	Tumour necrosis factor-related apoptosis-inducing ligand
TRIF	TIR-domain-containing adaptor inducing interferon-beta
UV-HSV	UV-inactivated HSV
VSV	Vesicular stomatitis virus
VZV	Varicella zoster virus
WT-HSV	Wild type HSV

Introduction

Chapter 1

Introduction

1.1 Natural history of HSV-1 infection in man

Primary infection with HSV-1 occurs predominantly in children. It is often asymptomatic. However, HSV-1 infection can result in a wide spectrum of clinical presentation, ranging from asymptomatic infection, to ulcerative and vesicular lesions of the skin, and to systemic infection of large organs. Furthermore, symptoms differ between primary and recurrent lesions.

When symptoms do occur, vesicular lesions are usually intraoral, affecting the gingival and buccal mucosa (gingivostomatitis and pharyngitis). The vesicles formed then ulcerate, exposing infectious particles to the environment and therefore other individuals. Over the following 1-2 weeks, any symptomatic peripheral lesion in the mucosa usually resolve. During primary infection, some viral particles enter the endings of peripheral sensory nerves (predominantly c-type nerve fibres) and the virus travels retrogradely to the dorsal root ganglion (DRG) of the innervating dermatome. For oral infection this is the trigeminal ganglion, and it is here that a latent infection occurs, a common ability of all herpesviruses. Oral HSV-1 infection can result in latent infection of up to 40-60% of sensory nerves innervating the infection site (Jones, 2003).

Periodically, the virus reactivates from latency. The factors responsible for this reactivation are still unclear, but include other febrile illnesses, direct sunlight (predominantly UV rays), psychological stress, and systemic immunocompromised states (both iatrogenic and infectious). After reactivation, viral particles travel down the same peripheral nerves and reinfect the innervated area of skin. The most common sites for reactivation include the lips and adjoining skin (herpes labialis). Clinically, secondary infection is characterised by a prodrome of pain, burning, tingling, or itching, followed by macular lesion formation within 24 hours, which later become vesicular and then pustular. The lesion is then most often cleared within 2 weeks.

In the majority of cases, local peripheral HSV infections are not life threatening, although the morbidity can be both physically and psychologically debilitating. This is particularly the case in the instance of complications of HSV-1 infection, such as

herpes keratitis (the second most common cause of non-traumatic corneal blindness), herpetic whitlow and eczema herpeticum, which may result from auto-inoculation from the oral lesion. HSV can also rarely result in life-threatening encephalitis, although the cause is not clear. It should also be noted that HSV-1 infection is not restricted to the oral region. An increasing number (up to 40%) of genital herpes infections, particularly in women under 25 years old, are now caused by HSV-1 (Malkin, 2004), and shedding genital herpes during child birth increases the risk of neonatal infection during birth. This has a high mortality rate and incidence of neurologic sequelae in survivors (Whitley, 2004).

Therefore, the clinical manifestation of HSV-1 disease demonstrate that despite being non-life threatening in the majority of immunocompetent individuals, this virus infection still carries considerable disease burden to the population. A more detailed understanding of its pathogenesis and immunobiology will aid towards successful vaccine design.

1.2 Biology of HSV-1

The virus name comes from the Greek *'herpein'* which means 'to creep', referring either to the chronic, latent or recurrent infections that occur or to the spreading nature of the visualised skin lesions. The epidemiology of HSV infections puzzled clinicians for many years and it was not until 1950 that Burnet and Buddingh first showed that HSV could become latent after a primary infection and reactivate after later provocation. Subsequent studies with other members of the herpesvirus family confirmed that latency is a shared property amongst them.

1.2.1 Virus structure

The HSV virion consists of a large double-stranded DNA genome (152 kbp long, with 85 open reading frames) wound around a proteinaceous core. The capsid surrounds the core and outside the capsid is the tegument, a protein-filled region that appears amorphous in electron micrographs. On the outside of the viral particle is the envelope,

which is derived from the outer leaflets of the Golgi apparatus of infected cells and contains numerous glycoproteins.

1.2.2 Mechanism of viral infection

Like other members of the herpesvirus family, HSV-1 is an enveloped virus and enters target cell cytoplasms following envelope fusion with the cell surface membrane. The initial interaction is attachment of virions to glycosaminoglycans (preferentially heparan sulphate) on the cell surface, followed by binding to and entry via specific entry receptors. The viral factors involved in this process have been well characterised and are described in more detail below.

1.2.2.1 Heparan sulphate

The role of heparan sulphate (HS) in HSV binding and entry has been studied extensively. Several lines of evidence have alluded to the role of HS as an initial receptor for HSV infection. Firstly, HS proteoglycans are frequently found on the surface of many cell types, including ones susceptible to HSV infection (WuDunn and Spear, 1989). Secondly, removal of HS from the cell surface, enzymatically or in HS deficient cells, renders the cells partially resistant to HSV infection by reducing surface binding of the virus (Shieh et al., 1992). Thirdly, heparin, a molecule chemically similar to HS (Kjellen and Lindahl, 1991), has been shown to prevent HSV infection by masking HS binding domains on the virus envelope (Herold et al., 1995).

The viral mediators of HS attachment are glycoprotein B (gB) and gC, as they can bind to immobilised heparin columns (Herold et al., 1991), and are the only HS binding proteins encoded by HSV-1 (Laquerre et al., 1998). Deletion mutants for gB and gC have demonstrated reduced virus binding to the cell surface (Herold et al., 1994), to a similar degree as infection of cell lines deficient for HS (Gruenheid et al., 1993). Furthermore, gB and gC deletion mutants can still penetrate cells, albeit at a reduced rate (Herold et al., 1991; Laquerre et al., 1998), suggesting that HS attachment may only be required to increase the efficiency of binding to a cellular surface receptor or to induce changes in the bound particle envelope that facilitates the process of envelope binding to surface receptors and fusion with the cell surface membrane.

1.2.2.2 Entry receptors

There are currently three classes of entry receptors known, a member of the tumour necrosis factor receptor (TNFR) family, immunoglobulin superfamily members and specific enzyme modified moieties in heparan sulphate.

- *Herpesvirus entry mediator (HVEM)* Located on human chromosome 1 (Kwon et al., 1997), this molecule is a member of the large TNFR superfamily. The expression of HVEM, also known as HveA, on non-permissive cells allows HSV-1 entry (Montgomery et al., 1996) after interaction with HSV-1 gD (Whitbeck et al., 1997). Like many other members of the TNFR family, the cytoplasmic tail of this receptor can bind the TNFR-associated factor (TRAF) family members TRAF1, 2, 3 and 5, but not 6 (Marsters et al., 1997). As a result, HVEM is capable of activating NF-kB and AP-1 signalling pathways (Marsters et al., 1997). Interestingly, abrogation of the cytoplasmic tail of HVEM, such that no TRAF binding sequences are present, does not interfere with viral entry, excluding the role of intracellular signalling via this receptor as being necessary for viral entry (Montgomery et al., 1996). However, this does not rule out that signalling induced by viral ligation of the receptor may be important in conditioning the cell for viral replication.
- Nectin-1 this molecule is a member of the immunoglobulin superfamily of nectins, that function as intercellular adhesion molecules by co-localising with cadherins in adherens junctions. HSV-1 gD can also bind this receptor (also known as poliovirus receptor-related protein 1 or HveC) (Geraghty et al., 1998; Krummenacher et al., 1998). This receptor is expressed on both epithelial and nervous tissue cells, and thus could be an important in vivo mechanism for viral infection of the local tissues and establishment of latency in the nervous system. Indeed, blocking viral-nectin-1 interaction can prevent HSV mucosal infection totally in a murine model of vaginal infection (Linehan et al., 2004). Related members of the nectin family can also bind gD; these include nectin-2

(Eberle et al., 1995; Connolly et al., 2001) and the herpesvirus Ig receptor (HIgR), which shares the same ectodomain as nectin-1 (Cocchi et al., 1998b; Cocchi et al., 1998a). However, HSV-1 has been shown not to require nectin-2 for cell entry (Warner et al., 1998; Krummenacher et al., 2004) and the role of HIgR in HSV-1 entry has not been defined.

The availability of nectins for viral binding may determine the relative susceptibility of epithelial cell layers to HSV-1 infection (Spear, 2002). Nectin-1 co-localises with E-cadherin at adherens junctions in epithelial cells and therefore, is relatively inaccessible to HSV-1 virions. Indeed, disruption of these junctions renders epithelial cells more susceptible to infection (Yoon and Spear, 2002; Marozin et al., 2004). As a result, it is likely that HSV-1 can only infect the host efficiently in the event of a previous break in the epithelial cell layer. The consequential advantage to the virus is that it can infect the more susceptible basal side of epithelial layers (Schelhaas et al., 2003; Marozin et al., 2004), as well as less differentiated epithelial cells in the wound site, on which most nectins are not sequestered in adherens junctions and thus, are available for viral entry (Spear, 2002)

• **3-0-sulfotransferases** - Specific glucosamine residues in heparan sulphate can be sulphated by 3-0-sulfotransferases to provide binding sites for gD, which are of similar affinity as the above entry receptors, and are alone sufficient to permit viral entry (Shukla et al., 1999). However, the necessity for enzyme modification of heparan sulphate prior to infection makes it difficult to determine the relative importance of this receptor in vivo.

Although the receptors described above are structurally unrelated, they can all bind gD directly (Whitbeck et al., 1997; Krummenacher et al., 1998; Nicola et al., 1998; Shukla et al., 1999). Therefore, coupled with the near ubiquitous distribution of heparan sulphate for viral attachment, the relative susceptibility of different cells to HSV-1 infection is likely to be dependent on gD receptor expression. An extension of this hypothesis is that polymorphisms in the sequence of some of the HSV entry receptors may determine infection susceptibility and disease outcome, but studies that have

attempted to address this question have not yet observed such a correlation (Struyf et al., 2002).

Following binding to the entry receptor(s), HSV-1 envelope fuses with the cell membrane and the tegument and capsid are released into the cytoplasm. In the majority of cells, this can occur at neutral pH on the surface of the cell (Wittels and Spear, 1991), whereas in a few cell types, acidification in endosomes is required to enable envelope-membrane fusion (Nicola et al., 2003). Envelope-membrane fusion requires gB, gH and gL, as well as gD binding to entry receptors. However, the mechanisms employed by these glycoproteins are still unclear. One possibility is that conformational changes in the entry receptors allow interaction with gB, gH or gL (Krummenacher et al., 2002). Alternatively there may be receptors for gB, gH and gL on the cell surface that are as yet unspecified.

1.2.3 Mechanisms of HSV-1 replication

After penetrating the surface membrane, tegument proteins disperse into the cytoplasm, whereas viral capsids travel along microtubules via dynein motors to the nucleus (Sodeik et al., 1997). This occurs by direct HSV-1 interaction with the motor ATPase dynein complex responsible for the transport of the virus to the nucleus (Ye et al., 2000; Mabit et al., 2002). The core enters via a nuclear pore (Dohner et al., 2002). The genome is subsequently circularised and can begin transcription using host RNA polymerase II. Like all other herpesvirus genomes, HSV-1 has a unique long (UL) and a unique short (US) region, bounded by inverted repeats. The repeats allow rearrangements of the unique regions and, as a result, herpesvirus genomes exist as a mixture of 4 isomers. Gene expression occurs in a cascade function, with immediate early genes (IE), important for activating early (E) genes, which are involved in viral DNA replication (e.g. thymidine kinase). Late (L) genes encode structural proteins and this occurs in tandem with E gene expression (Weir, 2001).

All IE promoters contain a common *cis*-acting sequence (TAATGARAT) that is required for VP16 mediated trans-activation. VP16 is a tegument protein that is delivered to the nucleus, where it interacts with two cellular proteins, Oct-1 and HCF,

to induce IE promoter activity efficiently (Wysocka and Herr, 2003). Although HCF is expressed in the nucleus of most cells, it is mostly found in the cytoplasm of neurones (Kristie et al., 1999), suggesting a mechanism for reduced gene expression in neurones and allowing the establishment of latency. However, other host factors may also play a role (Jones, 2003). One of these may be the role of inhibitory transcription factors. One such family is Oct-2, which is present as five different isoforms, two of which are neurone specific, Oct 2.4 and 2.5. In vitro transfection studies have shown that these factors can repress Oct-1-VP16 mediated expression of IE genes (Lillycrop et al., 1993). However, the expression of these inhibitory factors remains controversial. Some studies have detected their expression in rat adult DRG (Wood et al., 1992), but this has not been reproduced by all groups (Hagmann et al., 1995b; Turner et al., 1996), suggesting a role for other proteins that can regulate VP16 mediated transcription. It remains to be seen whether host mediated repression of IE gene expression plays a significant role in vivo in preventing the lytic replication cycle in neurones infected by HSV, and subsequently promoting latency.

One notable exception to the repression of HSV gene transcription in latently infected neurones is the LAT genes, whose RNA transcript accumulates in these cells. Although a functional protein product has not been detected (Stevens et al., 1987), one important function of these genes could be their ability to prevent apoptosis in latently infected neurones (Perng et al., 2000; Ahmed et al., 2002). Maintaining cell viability may be critical in maintaining the virus' ability to reactivate from latency (Inman et al., 2001).

1.3 Role of immune system in neuronal HSV infection

The lack of viral gene expression, other than the elusive LAT genes, led investigators to conclude that immunological recognition of viral proteins during latency did not occur. However, this concept is in disagreement with histological data of both animal models of HSV latency and human studies, showing an inflammatory cell infiltrate, including antigen specific CD8+ T cells, in latently infected ganglia (Gebhardt and Hill, 1990; Liu et al., 1996; Theil et al., 2003). This has been hypothesised to signify that low grade viral antigen expression occurs in latently infected neurones, and it has

been confirmed in several studies that have used RT-PCR to detect IE gene transcripts (Kramer and Coen, 1995; Chen et al., 1997; Chen et al., 2002). However, the assay does not give an estimate of the number of neurones expressing these genes and the RT-PCR may mask high-grade expression by a small number of cells. Indeed, recent studies using in situ hybridisation and immunofluorescence has described that one neurone in every 10-17 trigeminal ganglia expressed levels of IE and E transcripts and proteins at levels equivalent to that of lytically infected cells (Feldman et al., 2002; Sawtell, 2003). Furthermore, it was these neurones that were surrounded by an inflammatory cell infiltrate, suggesting that immune mediated control of latency reactivation may not be occurring globally (e.g. by cytokines), but rather in a focal antigen-specific mechanism on reactivating neurones (Feldman et al., 2002).

Currently, it is still unclear how the infiltrating inflammatory cells can regulate activation from latency. Significant levels of interferon γ (IFN γ) and tumour necrosis factor α (TNF α) are expressed in TG, suggesting that T cells present are activated following viral DNA replication (i.e. from the expression of the required proteins) (Cantin et al., 1995; Halford et al., 1996; Halford et al., 1997). The infiltrating T cells are CD8+ T cells and can block viral reactivation in an ex vivo model (Liu et al., 2000). They are antigen specific, although this study was only carried out in mice models where the HSV response is highly focused towards one epitope of gB (Khanna et al., 2003). The mechanism involves direct contact with infected neurons and IFN γ secretion (Khanna et al., 2003). The prevention of reactivation by T cell derived cytokines is supported by latency studies carried out in IFN γ and TNF α knockout mice, which demonstrated increased levels of HSV reactivation, though it is not possible to exclude that the immune defect in the periphery allowed a greater number of virions to enter latency per neurone, predisposing to increased frequency of reactivation (Cantin et al., 1999; Minami et al., 2002).

It is likely that the immunological control described in animal models can be extended to the human system. Histology of human DRG sections shows a CD8 T cell infiltrate and the presence of inflammatory cytokines (Theil et al., 2003). Thus, it seems that regulated HSV latency is not only an intrinsic property of the virus, but is also actively promoted by the host immune system. This might well explain the increased frequency of
reactivations in immunocompromised individuals (Schacker et al., 1998b). Furthermore, the buildup of IE gene transcript may explain the advantage of LAT gene expression, to prevent this stimulus promoting apoptosis of the neurone (Inman et al., 2001; Sanfilippo et al., 2004).

1.4 Role of the immune system in the control of peripheral HSV-1 disease

The immunological control of peripheral infection has been studied in greater detail than that of the central nervous system. The host response to infection occurs in two phases, the innate and adaptive immune responses. The prior is likely to be critical in limiting the spread of infection, although only the cellular adaptive immune responses is more strongly associated with resolution of HSV lesions (Siegal et al., 1981).

1.4.1 Innate immune system

1.4.1.1 Epithelial cells

In the context of a cutaneous/mucosal model of HSV infection, the first cell type to be infected is likely to be an epithelial cell or keratinocyte. Therefore, the signals secreted from this cell type will determine the nature and kinetics of the ensuing immune response. Indeed the cytokines found in herpetic lesion fluid within the first day of infection in humans, namely IL-1 α , IL-1 β , IL-6 and IL-10, are also secreted by keratinocytes following HSV-1 infection in vitro (Mikloska et al., 1998). Infected keratinocytes also secrete CCL3, CCL4 and CCL5, which are important in recruiting other inflammatory cells, including dendritic cell (DC) precursors and activated T cells (Taub et al., 1993; Yoneyama et al., 2004). In turn, infiltrating T cells can secrete IFN γ that upregulates HLA-DR expression on keratinocytes (Mikloska et al., 1996).

1.4.1.2 NK cells

NK cells may be protective in primary HSV infections (Pereira et al., 2001). Mouse models have also shown that they are an early source of IFN γ in the lesion, although

this is not critical for the clearance of vaginal infection with an attenuated HSV-2 strain (Milligan and Bernstein, 1997). Their role in humans has been extrapolated from a child with a deficiency in NK cells who succumbed to early and systemic herpesvirus infections (Biron et al., 1989), and an adult with disseminated HSV-2 disease in the absence of systemic NK cells (Dalloul et al., 2004).

Recently, NKT cells have also been suggested to play a protective role against HSV infection (Grubor-Bauk et al., 2003). However protection by this cell type, as determined by viral infection of the nervous system, was only apparent 7 days after infection. The delay in protection implies that this cell type, although sharing surface molecules with NK cells, should be considered part of the anti-HSV adaptive immune response (Grubor-Bauk et al., 2003).

1.4.1.3 Complement system

Complement can enhance the humoral immune response to HSV (Verschoor et al., 2001). A chimaeric mouse model demonstrated that local C3 production by myeloid BM-derived cells was sufficient to restore the humoral response to infectious HSV-1 in C3-deficient mice (Verschoor et al., 2003). The HSV strategy to avoid complement activation (section 1.5.1) may therefore result not only from an attempt to neutralise the complement activation cascade directly, but also to abolish the generation of an antibody response.

1.4.2 Adaptive immune system

1.4.2.1 Antibody response

The presence of neutralising antibodies to HSV has been observed for many years. The clearest example of their functionality comes in neonatal infections, whereby maternal antibodies protect from transmission to the newborn during delivery (Brown et al., 1997). The antibody targets are very broad, including envelope glycoproteins and viral capsid components (Ashley et al., 1985; Ashley et al., 1994), demonstrating that a wide variety of viral antigens are accessible to the immune system to sample.

The efficacy of the in vivo humoral response in adults is less clear. Murine studies have shown that antibodies can offer protection in the early stages of infection, when viral titres are low (Kuklin et al., 1998). Natural IgM antibodies may play a role in limiting low dose inocula spreading to the CNS (Deshpande et al., 2000), although the relevance to humans is unclear. Passive immunisation with HSV-specific IgG can also reduce lesion severity after vaginal infection (Parr and Parr, 1997), emphasising that antibodies are important in limiting the spread of infection in the early stages (the first 20 hours), whereas T cell responses peak later, independent of antibody activity (Parr and Parr, 2000). However, the role of B cells and antibodies remains controversial, as immunised B cell deficient mice can clear HSV infection equally well as wild type mice (Kuklin et al., 1998). Furthermore, large-scale clinical trials using recombinant glycoprotein subunit vaccines have showed that seroconversion did not correlate with subsequent disease severity (Corey et al., 1999). It should also be noted that the lack of correlation between antibody and protection from infection in man may be related to one of the immune evasion mechanisms of HSV-1 and -2, the gE-gI dimer that binds human, but not mouse, IgG, and neutralises its function (Johansson et al., 1985) (see section 1.5.2).

1.4.2.2 CD4+ T cell response

Approximately 0.2% of all CD4+ T cells in immune individuals are specific for HSV antigens, as detected by intracellular IFN γ staining (Asanuma et al., 2000). CD4+ T cells are the initial T cells infiltrating into herpetic lesions and can secrete a variety of cytokines (Mikloska et al., 1998). The most important of these is IFN γ and CD4+ T cells are the predominant source of this cytokine in peripheral blood mononuclear cells (PBMC) from patients with recurring herpetic lesions (Cunningham and Merigan, 1984; Cunningham et al., 1985a). This cytokine plays a role in partially reversing the HSV-induced downregulation of MHC class I and inducing MHC class II expression in keratinocytes, rendering the cells more detectable by both CD4+ and CD8+ T cells (Mikloska et al., 1996).

In mouse studies, CD4+ T cells confer greatest protection in vaginal challenge with HSV-2 (Kuklin et al., 1998). In humans, although increased IFN γ secretion by HSV

specific PBMC correlates weakly with reduced frequency of herpes labialis (Spruance et al., 1995), there has not been a study definitively showing a correlation between the strength and breadth of CD4+ responses and clinical severity (Posavad et al., 1997). This may relate to technical limitations of the assays used to detect CD4+ T cells. Alternatively CD4+ responses may be more important locally, either in the lymph nodes (LNs) for providing help to CD8+ T cells, or in the periphery for IFN γ secretion. Furthermore, HSV specific CD4+ T cells can be cytotoxic (Mikloska et al., 1996; Koelle et al., 1998b) and thus differences in circulating antigen specific CD4+ T cells, as detected by IFN γ secretion upon restimulation, may not necessarily correlate with their true role and distribution in vivo.

Recent focus has been directed on defining the antigenic epitopes of the host immune response, in order to identify desired vaccine targets. These studies observed that while keratinocytes can be induced to express HLA-DR after IFN γ stimulation (Mikloska et al., 1996), CD4+ T cell clones identified are also restricted by HLA-DP and –DQ (Koelle et al., 1998a). These are likely to be stimulated by professional APC in the draining LN, rather than keratinocytes in the lesion site. This underlies the importance of professional APC, such as DC, in the immune response to HSV.

1.4.2.3 CD8+ T cell response

Although CD8+ cells follow CD4+ T cells into lesions (Cunningham et al., 1985b), the role of CD8+ T cells in the control of HSV infections has been most clearly demonstrated in a cross-sectional study demonstrating an inverse correlation between CD8+ T cell number and the severity of genital herpes lesions (Posavad et al., 1997). Furthermore, serial histological studies of lesion biopsies demonstrated that the clearance of virus correlates best with the infiltration of CD8+ T cells (Koelle et al., 1998b). It should be noted that these studies were carried out on biopsies of HSV-2 infection of genital mucosa, and although likely to be similar, no direct studies of HSV-1 lesions have yet been performed. Nevertheless, a recent study demonstrated that CD8+ T cell responses could only prevent infection of peripheral nerve endings within 24 hours of epidermal inoculation (van Lint et al., 2004), emphasising that the

predominant role of CD8+ T cells in vivo is likely to be to resolve the lesion rather than prevent HSV spread to the nervous system.

Similarly to CD4+ T cells, recent studies have focused on understanding the breadth of the CD8+ T cell repertoire. Tetramer binding can give an estimate of individual TCR frequencies, and in vitro restimulation of T cells can generate clones that can identify antigen targets. However, the latter approach involves two rounds of antigen stimulation, which may bias the relative frequency of epitope-specific T cells. Recent studies using only one round of antigen stimulation have identified subpopulations of HSV-specific CD8 T cells with a broader range of reactivity than previously observed (Koelle et al., 2003). The relative immunodominance of the novel epitopes is still unspecified, but the data demonstrates clearly that despite the immune evasion mechanisms the virus utilises (see section 1.5), the host is still able to mount a potent and broad CD8+ T cell response.

1.5 HSV immune evasion

The suggested importance of the role of the immune system in the control of HSV infections has been based upon histological analysis of lesions and knockout mouse studies. However, the evolution of proteins that enable the virus to evade specific components of the immune system also implies that immunity plays an important role in the HSV life cycle. Below are the components of the immune response that the virus is known to interfere with.

1.5.1 Complement

HSV-1 gC is a receptor for C3b (Friedman et al., 1984) and prevents host activation of the alternative complement pathway (Fries et al., 1986). In part this occurs through accelerated decay of C3 convertase, but gC also has a domain that prevents the binding of C3b to C5 and properdin (Fries et al., 1986). It is the C3 binding domain that constitutes most of the virulence in vivo (Lubinski et al., 1999), as demonstrated by infections using gC mutant viruses (Lubinski et al., 1998; Lubinski et al., 1999). Importantly, this occurs largely independent of the presence of neutralising antibodies

(Friedman et al., 1996), demonstrating a specific evasion of the activation of the complement cascade by this virus and the effectiveness of this complement in control of HSV infection.

1.5.2 Antibody responses

A heterodimer formed between HSV-1 gE and gI functions as an IgG Fc receptor (Johnson et al., 1988). Virus specific antibodies can form antibody bridges between the target antigen via the F(ab')2 domain and the gE/gI receptor via the Fc domain of the antibody (Frank and Friedman, 1989). This can occur both on the surface of infected cells as well as on the viral envelope, preventing C1q binding and IgG binding to Fc γ R, inhibiting antibody-dependent cellular cytotoxicity (Dubin et al., 1991; Van Vliet et al., 1992). It is worth noting that as gE can prevent C1q binding, this is another mechanism of preventing complement activation, distinct from the actions of gC, which inhibits complement independent of antibody (section 1.5.1).

Murine models of infection with HSV-1 gE mutants possessing abrogated FcR activity suggest that this evasion mechanism is translated into a viral advantage in vivo (Nagashunmugam et al., 1998). Furthermore, similar IgG binding glycoproteins are encoded by HSV-2, varicella zoster virus (VZV) and human cytomegalovirus (HCMV), suggesting a conserved property within this virus family that confers a survival advantage (Keller et al., 1976; Para et al., 1982; Litwin et al., 1992).

1.5.3 Antigen processing

As noted earlier, it is the cellular immune response that correlates with resolution of the infection. Therefore, it is unsurprising that HSV-1 has also evolved mechanisms to prevent recognition by CD8+ T cells, by reducing the expression of (viral peptide presenting)-MHC class I molecules on the surface of infected cells.

The US12 gene of HSV-1 encodes for the ICP47 protein that can interfere with peptide loading of MHC class I molecules by binding to the cytosolic side of the transporter associated with antigen processing (TAP) molecules on the ER. The result is a lower expression of MHC class I molecules (including ones presenting viral antigens) on the surface of infected cells. Indeed, deletion of ICP47 renders infected cells more susceptible to CTL mediated lysis (York et al., 1994; Tomazin et al., 1996). It is difficult to determine the role of ICP47 in vivo, as this protein does not bind and inhibit the function of murine TAP as efficiently as its human counterpart (Ahn et al., 1996; Tomazin et al., 1996; Tomazin et al., 1998). Nevertheless, ICP47 mutant viruses still display decreased neurovirulence in mice, a property lost in the absence of a CD8 T cell response, demonstrating the important role this viral protein can have in masking CD8 T cell targets in vivo (Goldsmith et al., 1998).

1.5.4 Cytotoxic T cell activity

HSV has evolved several mechanisms to prevent apoptosis (Koyama and Miwa, 1997) (see chapter 4), presumably with the intent of keeping the cell alive a sufficiently long time to complete an efficient replication cycle and produce ample viral progeny. One possible apoptosis inducing-signal is that originating from CTL, either by Fas ligation on the surface of infected cells or by granzyme activity. Indeed, HSV infected cells are resistant to CTL induced apoptosis (Jerome et al., 1998) owed in part to the expression of the US5 gene product, gJ (Jerome et al., 2001). However, the modulation of CTL activity is more complex, as HSV infected cells expressing the US3 protein kinase can also inactivate CTL that make direct contact with the infected cell (Sloan et al., 2003). Furthermore, in the event of an activated T cell being infected, presumably after coming in contact with an infected non-lymphocyte, it renders them more susceptible to die by Fas-mediated fratricide mediated by other HSV-specific CTL (Raftery et al., 1999).

1.6 Generation of adaptive immune responses: the role of dendritic cells

As a result of the data pointing to a role of cellular immunity in the control of HSV infection, recent efforts have concentrated on understanding the process that generates these responses. It has been appreciated for some time now that the generation of adaptive cellular immune responses is dependent on the stimulation of T cells by

professional antigen presenting cells (APC). Specifically, the initial activation of antigen specific naïve T cells in vivo is carried out predominantly by DC. This section will review briefly the major aspects of DC biology, and then consider DC in the context of viral immunity, focusing specifically on the role of immunity to HSV-1.

What makes a mature DC a more potent stimulator of T cells? This has been an important question, not only in understanding DC physiology, but also for providing clues to host responses and mechanisms that pathogens may disturb. Originally identified on the basis of their morphology (Steinman and Cohn, 1973), these cells are now accepted as the most potent APC, particularly in their capacity to activate naïve T cells. Several of their cell biological properties, such as the stability of MHC-peptide complexes, explain this role (Cella et al., 1997; Zehn et al., 2004). Other important factors include the cytokine cocktail secreted at the time of T cell contact, and the motile and plastic DC cytoskeleton. Recent intravital dynamic imaging in lymph nodes has shown that rapid dendrite extensions by DC permit scanning of around 5000 T cells per hour (Miller et al., 2004). The interaction time between DC and T cells is increased both by the presence of antigen bearing DC and the maturation state of DC, such that mature DC induced formation of stable immunological synapses that last for >8 minutes at a time (Bousso and Robey, 2003; Stoll et al., 2002; Miller et al., 2002; Benvenuti et al., 2004b). The increased length of interaction ensures that T cells receive appropriate stimulatory signals from the DC, both in the form of surface molecule interaction and via the paracrine activity of DC-secreted cytokines.

However, a snapshot of their interaction with T cells belittles their central role in the generation of antigen specific immune responses. DC exist in two functional states, immature and mature, although this simple model has recently been expanded on (see section 1.9). In the immature state, DC form a network in the body's periphery: skin, mucosa and most other organs where pathogen entry can occur. In this state, DC are relatively poor T cell stimulators. Their predominant role is sampling the antigenic environment surrounding them. This occurs through non-specific uptake in the fluid phase, by macropinocytosis, or through phagocytosis of molecules binding to specific receptors on the DC surface, such as c-type lectins or FcR. The antigenic contents of this uptake is degraded and processed for presentation on MHC molecules. The model

predicts that in the steady state, in the absence of a surrounding infection, the DC remains in the periphery and continues to sample its microenvironment.

However, upon encounter of certain (combinations of) stimuli, DC undergo a process of maturation, which involves rapid downregulation of antigen sampling and a parallel increase in presentation of the endocytosed antigens. This increase in peptide loaded MHC molecules is coupled to an increase in expression of co-stimulatory molecules, such as CD80 or CD86, critical in the activation of T cells. In parallel, DC migrate out of the peripheral tissues and into draining LNs. However, there is increasing evidence that the maturation response to a variety of pathogens occurs in a non-overlapping fashion by different DC subsets. Therefore, their origin and function will be discussed below before considering the ligands that stimulate DC.

1.7 Dendritic cell subsets

It was known for many years that two types of antigen presenting cells reside in lymphoid organs, one with a pronounced dendritic morphology, and another with a more round shape. By electron microscopy (EM), this second cell type has plasma cell like features, with a smooth outline and a large amount of ER in the cytoplasm (Grouard et al., 1997) and was thus termed the "plasmacytoid monocyte" or "plasmacytoid T cell". Concurrently, it was known that a small proportion of cells in peripheral blood could produce large amounts of type I IFN when stimulated with viruses and these were named "natural interferon producing cells" (NIPC) (Sandberg et al., 1989). In the last few years, it has been established that the plasmacytoid cells are in fact the NIPC (Siegal et al., 1999) and therefore the precursor of plasmacytoid DC (PDC) (Siegal et al., 1999). It has been hypothesised that the extensive ER is necessary for production of large quantities of type I IFN after viral stimulation, and that this exposure to virus promotes their differentiation into mature DC, with dendritic morphology and capacity to activate T cells (Grouard et al., 1997; Cella et al., 2000; Bauer et al., 2001; Yonezawa et al., 2003). It is important to note that the nomenclature used in this field is still inconsistent. For the purpose of this thesis these cells will be referred to as PDC at all stages of their differentiation.

In humans, the DC subsets circulating in blood have been described most extensively and have been classified as 'classical' myeloid DC (MDC) and PDC. The terminology comes from the definitive myeloid origin of classical DC (e.g. CD13+), whereas PDC originally were believed to derive from lymphoid progenitors, as they lack myeloid markers, and expressed T and B cell molecules, such as the preTCR α and Spi-B (Blom et al., 2002). However, the developmental origin appears to be not so rigid, as thymic MDC and PDC can arise from progenitors distinct from those committed to T cell development (Fohrer et al., 2004). In the mouse, easier access to lymphoid organs has allowed the definition of a large number of subsets (Shortman and Liu, 2002). However, the ontogeny of many of these cell types is also still unspecified. Mouse DC subsets had originally been defined on the basis of CD8a expression, and this was also believed to signify a distinction between those of myeloid and lymphoid origin. However, as both subsets could be reconstituted by either myeloid or lymphoid progenitor cells, this hypothesis is probably incorrect (Akashi et al., 2000; Traver et al., 2000). Furthermore, tissue-derived migrating myeloid CD11b+ DC may differentiate into $CD8\alpha$ + DC in the LN (Moron et al., 2002). As the murine equivalent to human PDC can also be reconstituted from both myeloid- and lymphoid-committed progenitors (Shigematsu et al., 2004), a more plastic model of DC differentiation is now accepted in both mouse and man.

Irrespective of their ontogeny, the presence of DC subsets with divergent functions is likely to represent an evolutionary advantage to the host. Different organs are populated with varying proportions of DC subsets, and therefore their function must be analysed in the correct anatomical context in relation to the viral route of entry/transmission, in order to understand the relative roles of these two cell types. Infections can be subclassified broadly into those that enter via mucosae or skin and those that enter directly into the blood stream. In the resting state, MDC are ubiquitous in the skin, respiratory and gut mucosa, as well as peripheral blood and lymphoid organs. PDC, on the other hand, have only been found in peripheral blood and lymphoid organs (Ebner et al., 2004; Yoneyama et al., 2004). Few, if any, PDC are present in the skin or mucosae in the resting state (Jahnsen et al., 2000; Farkas et al., 2001; Wollenberg et al., 2002; Bangert et al., 2003). Furthermore, their migration pattern is different to that of MDC. Whereas the latter circulate from peripheral sites

via afferent lymphatics to draining lymphoid organs (Banchereau and Steinman, 1998), PDC enter lymph nodes predominantly from blood through high endothelial venules (HEV) (Cella et al., 1999a). Like MDC, they can also migrate into areas of inflammation in the periphery (Farkas et al., 2001; Wollenberg et al., 2002; Bangert et al., 2003; Zhao et al., 2003). Therefore, it is likely that viral interaction with both subsets is relevant, but only at appropriate times after infection.

In the initial stages of a skin/mucosal infection, MDC are likely to be the predominant DC subtype to be exposed to the invading virus. PDC could come into contact with the virus at later time points, in the periphery following inflammation-mediated migration (Jahnsen et al., 2000; Farkas et al., 2001; Wollenberg et al., 2002) and in secondary lymphoid organs, where cells, such as MDC, could be productively infected with a virus (Fugier-Vivier et al., 1997; Ho et al., 2001). However, for viruses that are transmissible via the blood route, PDC interaction may be more relevant at the earliest stages of infection. The in vitro observations that this cell type can secrete very large quantities of type I IFN may reflect such an in vivo scenario. The large quantities of this cytokine that are secreted may compensate for the dilutional effect of many litres of blood, achieving a sufficient functional concentration of type I IFN systemically. The importance of a PDC responses in human systemic viral infections has recently been proposed (Pichyangkul et al., 2003; Hishizawa et al., 2004), emphasising the critical role this DC subset plays in control of viral infections in the appropriate context.

When PDC are matured with virus, CD40L or a Toll like receptor (TLR) ligand, they differentiate into potent APC capable of stimulating memory responses (Fonteneau et al., 2003; Krug et al., 2003). However, the relative potency of mature PDC in stimulating naïve T cells remains controversial. Murine studies demonstrate that mature PDC are inferior to MDC in this capacity (Krug et al., 2003), whereas others observed that mature human PDC are only inferior in stimulating naïve T cells to exogenous, but not endogenous antigens (Salio et al., 2004). This implies an inefficiency in cross-presentation by this subset. Thus, in the early stages of infection, the main role of uninfected PDC may not be to cross-present antigen, but rather to amplify MDC activation, most probably through the secretion of type I IFN (Dalod et

al., 2003), which in turn may amplify cross-priming to exogenous antigens by MDC (Le Bon et al., 2003).

1.8 Dendritic ceil activating ligands

DC are likely to encounter multiple stimulatory ligands in their microenvironment, and this may be important in determining the degree of DC activation. Lipopolysaccharide (LPS)-containing Gram negative bacteria may activate DC fully, whereas other pathogens may not possess as much stimulating material per se, but may induce the secretion of cytokines that also promote the maturation of DC (e.g. $TNF\alpha$ (Sallusto and Lanzavecchia, 1994)). Thus, several relatively weak stimuli may add up to increase the maturation, or activation, state of the DC. In recent years, the number of ligands described that activate DC has greatly increased. Initial studies examined the ability of these ligands to upregulate the expression of costimulatory molecules on DC, thus inducing 'maturation'. However, recent work has demonstrated that this process is more complex and that the ligands the DC encounters can affect DC physiology differentially, particularly reflected in the cytokines secreted, and subsequently tailor the immune response appropriately for the pathogen encountered. The presence of DC subsets that detect pathogen structures in non-overlapping fashion has added further complexity to the field of DC recognition of the environment. However, in the context of ligands that activate DC, a unifying hypothesis has been proposed to explain how the innate immune system, including DC, is alerted to pathogen infection.

Charles Janeway first suggested that the immune system is activated via receptors, which he named pattern recognition receptors (PRRs), that recognise conserved pathogen associated molecular patterns (PAMPs) (Janeway, Jr., 1992). Much evidence has accrued to back up this theory, particularly following the identification of TLRs and c-type lectins that recognise specific components of a wide variety of pathogens. This hypothesis was extended to suggest that the adaptive immune system, for example via DC, can also be activated by mediators released during inflammation and invasion induced by pathogen infection (Ibrahim et al., 1995; Matzinger, 1998). In this way, the ligands that activate DC could also be endogenous and host derived, rather than unique to the pathogen.

There is now experimental data to support the function of both exogenous and endogenous activators, which may act to complement each other. However for the purpose of this introduction, PAMPs and endogenous ligands that activate DC will be discussed in separate sections.

1.8.1 Pattern recognition receptors

1.8.1.1 TLR family

This family of receptor was identified initially on the basis of its homology to the drosophila Toll protein (Medzhitov et al., 1997). There are currently 11 known TLR, and the number of ligands for each receptor is constantly growing. Notably, not all the structures recognised are pathogen specific, with a growing list of endogenous ligands being described, which will be discussed in section 1.8.2.2. These are summarised in table 1.1.

TLR	Pathogen associated ligand	Endogenous ligand
1	Mycobacterial lipoproteins, triacylated lipopeptides (Takeuchi et al.,	
	2002)	-
2	PGN, LTA(Schwandner et al., 1999)	
	MV + HCMV envelope protein (Bieback et al., 2002; Compton et	Heat shock proteins* (Vabulas et al., 2002)
	al., 2003)	
3	dsRNA (Alexopoulou et al., 2001)	siRNA (Kariko et al., 2004a)
		mRNA (Kariko et al., 2004b)
4	LPS (Poltorak et al., 1998b)	Heat shock proteins* (Palliser et al., 2004)
	RSV envelope protein (Kurt-Jones et al., 2000)	Hyaluronan* (Termeer et al., 2002)
5	Flagellin (Hayashi et al., 2001)	
6	Cooperates with TLR2 recognition of PGN and other diacylated	
	lipopeptides (Ozinsky et al., 2000; Takeuchi et al., 2001)	-
7	ssRNA (Diebold et al., 2004; Heil et al., 2004)	
8	ssRNA (Heil et al., 2004)	
9	CpG DNA (Hemmi et al., 2000)	Chromatin IgG complexes (Leadbetter et al., 2002)
10	-	
11	Uropathogenic bacteria (Zhang et al., 2004)	

(*) The complete absence of microbial contaminants that could bind TLR2 and TLR4 in the preparation of these proteins could not be confirmed, and therefore, the role of these proteins as endogenous TLR ligands remains controversial.

Table 1.1 TLR ligands. Peptidoglycan (PGN), Lipoteichoic acid (LTA), Measles virus (MV), Respiratory syncytial virus

(RSV), double stranded RNA (dsRNA), single stranded RNA (ssRNA), short interfering RNA (siRNA)

1.

One effect of TLR ligation on DC may be the regulation of antigen uptake and processing. Two recent studies have demonstrated that TLR ligation, through p38 activation, promote uptake of bacteria in macrophages. This occurs through upregulation of expression of scavenger receptors, such as SR-1 and MARCO, and the mannose receptor, that ultimately increases uptake efficiency of bacteria (Doyle et al., 2004) and through maturation of the phagosome (Blander and Medzhitov, 2004). In DC, this TLR signalling effect may induce the delivery of degraded antigen to MHC class II loading compartments. It is interesting that MARCO is also involved in cytoskeletal rearrangement in DC (Granucci et al., 2003), suggesting that TLR4 ligation may also prime DC for emigration from the periphery.

Despite both activation of NF-kB after ligation of all TLR and the close association between the activation of this pathway with DC maturation (Verhasselt et al., 1999; Neumann et al., 2000; Yoshimura et al., 2001), it has become evident that the parallel activation of other signalling pathways by different TLR results in diverse functional outcomes. The initial TLR ligation can condition the DC for the signals it will pass on to engaging T cells, in the forms of cytokines and surface molecule expression. An example of the latter includes the Th1 promoting ICAM-1 (Salomon and Bluestone, 1998), whereas OX40 binding on T cells by OX40L on DC promotes Th2 responses (Ohshima et al., 1998). However, the most important DC derived signals include cytokines secreted, particularly IL-12 and type I IFNs.

IL-12 is made up of a 35kDa and a 40kDa chain (IL-12 p35 and p40 respectively) that heterodimerise to form the functional IL-12 p70 (Trinchieri, 2003). IL-12 exerts its actions on the cell through the IL-12 receptor (IL-12R), which activates the Janus Kinase (JAK)-STAT (signal transducer and activator of transcription) pathway, and through the activation of STAT4, can induce Th1 differentiation (Thierfelder et al., 1996; Kaplan et al., 1996; Cho et al., 1996). Recently, other members of the IL-12 family of cytokines have been described, IL-23 (Oppmann et al., 2000) and IL-27 (Pflanz et al., 2002). These cytokines utilise the IL-12 p40 chain as a common component. IL-23 binds to one of the IL-12R chains and induces similar downstream signals, including STAT4 activation and as a result also favours Th1 responses, albeit not as efficiently as IL-12 (Oppmann et al., 2000). IL-27 can also promote Th1

differentiation of T cells, although its receptor and signalling consequences are still less well defined (Pflanz et al., 2002).

Among the many actions of **type I IFNs** that will be discussed in greater detail in chapter 5, 'this family of cytokines can also skew T cell responses towards Th1 (Brinkmann et al., 1993; Kadowaki et al., 2000). Type I IFNs signal through the type I IFN receptor (IFNAR) and also activate the JAK-STAT pathway, specifically STAT4 required for Th1 differentiation (Cho et al., 1996; Rogge et al., 1997), although recent studies suggest that this effect is less potent than that of IL-12 (Athie-Morales et al., 2004).

DC can secrete cytokines that prevent Th1 activation, such as the chemokine CCL2 that can promote Th2 differentiation (Gu et al., 2000) and IL-10, which has regulatory functions irrespective of the Th bias (Zeller et al., 1999). However, most TLR ligands promote DC to induce a Th1 phenotype, as demonstrated by TLR3 (de Jong et al., 2002), TLR4 (Boonstra et al., 2003), TLR7 (Ito et al., 2002) and TLR9 (Hemmi et al., 2000). It is notable that the expression of these TLR is not ubiquitous on all DC subsets (Krug et al., 2001), and as a result, each cell responds to the stimulus differently. This is exemplified by TLR7, which induces IL-12 secretion upon ligation in MDC, whereas type I IFN is secreted in PDC. However, the end result is that that both TLR7 stimulated MDC and PDC promote Th1 responses (Ito et al., 2002; Lore et al., 2003).

In contrast, the effects of TLR2 ligation on DC are different, resulting in little IL-12 p70 secretion (Weigt et al., 2003) and skewing the immune response towards Th2 (Qi et al., 2003). The recent observation that HSV-1 can elicit TLR2 signalling raises interesting questions relating to the relevance of this interaction in the in vivo pathogenesis of HSV infection (Kurt-Jones et al., 2004), and is discussed further in chapter 6. Mechanistically, the TLR2 capacity to induce Th2 responses stems from the increased ERK activation after receptor ligation, relative to other Th1-inducing TLRs (Agrawal et al., 2003; Dillon et al., 2004). Human in vitro studies suggest that the consequential absence of IL-12 secretion is solely responsible for the Th2 shift (Agrawal et al., 2003), whereas murine studies suggest that the increased ERK

activation also results in increased IL-10 secretion (Dillon et al., 2004), which in turn can suppress IL-12 secretion (Xia and Kao, 2003).

Although many of the conclusions described above have been corroborated in both the mouse and human system, there are also some notable discrepancies. For example, TLR5 ligation in human cells induces IL-12 secretion and Th1 development (Agrawal et al., 2003), whereas in the mouse no IL-12 secretion and Th2 skewed responses follow (Didierlaurent et al., 2004). Secondly, some mouse TLRs may be non-functional (e.g. TLR8 (Jurk et al., 2002)). The issue of disparate TLR signalling between the two species will be revisited in chapter 5.

The difficulty in comparing the effects of different TLRs should also be noted, as the signal from the TLR can be not only qualitative, but also quantitative. For example, low dose LPS can promote Th2 development (Eisenbarth et al., 2002), whereas more commonly used larger amounts promote Th1 (Boonstra et al., 2003). Furthermore, Th decisions may also stem from the antigen dose, as low doses again promote Th2 development, whereas Th1 skew is supported at higher doses (Boonstra et al., 2003). The physiological relevance of this non-pathogen driven Th decision is currently still unclear and may relate to genetic background (Guler et al., 1997) or in vitro conditions that promote Th2 responses artificially even in the steady state (e.g. IL-4 in the culture medium). It is also important to note that different Th responses can be induced by both the two main DC subsets, MDC and PDC (Boonstra et al., 2003). Originally, it was believed that the two subsets induced Th1 and Th2 responses respectively (Rissoan et al., 1999). However, this observation may have been related to differences in antigen dose required to elicit Th1 responses by the different subsets (Boonstra et al., 2003). It is also important to remember that single ligand stimulation presents an artificial snapshot of the immune response. For example, ligation of TLR4 on DC can upregulate expression of TLR2, TLR4 and TLR9 (Visintin et al., 2001; An et al., 2002; Nilsen et al., 2004), and thus the regulation of expression of Th1 and Th2 biasing TLR may be either amplificatory or immunoregulatory depending on the appropriate context (i.e. in the presence of pathogens that possess multiple TLR stimulating ligands).

Studies analysing the association between polymorphisms in TLR sequences and disease susceptibility have demonstrated the protective role of TLRs in vivo. For example, a role for TLR5 is demonstrated by the increase in *Legionella pneumophilia* infection in individuals with a common polymorphism that introduces a stop codon into the TLR5 gene (Hawn et al., 2003). Polymorphisms in the TLR4 gene have also been associated with increased susceptibility to RSV infection and gram negative bacterial infections (Tal et al., 2004; Agnese et al., 2002). It is also possible that pathogens could exploit TLRs to generate non-resolving Th2 biased responses. Some murine studies suggest that infection with *Yersinia enterocolitica*, a pathogen that can activate TLR2, is cleared more efficiently in the absence of TLR2 (Sing et al., 2002), although no human data to date is available to backup this observation and is in contradiction with the protective effect of TLR2 in *Borrelia burgdorferi* and *Staphylococcus aureus* infection (Wooten et al., 2002; Takeuchi et al., 2000). The concept of pathogen exploitation of innate immune recognition will be discussed further in chapter 6.

1.8.1.2 c-type lectin family

This family of receptors is important for the recognition of a wide range of carbohydrate structures. These receptors play a role in facilitating phagocytosis and endocytosis, and hence also enhancing antigen processing and presentation. However, recent evidence suggests that the downstream signalling events they induce may also play a significant role in shaping the ensuing immune response. The receptors include the mannose receptor (MR), DEC-205, DC-SIGN, BDCA-2, Dectin-1 and the LC specific receptor, Langerin. Most of these are expressed on MDDC, although this does not correlate with the expression on circulating MDC, suggesting differential expression at various stages of DC development. Furthermore, some are specific to certain DC subsets, such as Langerin on Langerhans cells, whereas others are noted by their absence, such as the lack of expression of DC-SIGN on PDC. They all possess at least one carbohydrate recognition domain and recognise sugars in a variety of secondary and tertiary structures (Geijtenbeek et al., 2004).

Like for TLRs, c-type lectin ligands can be endogenous, such as ICAM-2 and ICAM-3 for DC-SIGN (Geijtenbeek et al., 2000b), but recent work has focused on the field of pathogen recognition. Initial studies utilised soluble sugars, such as mannan, to compete for binding to the MR, but this approach is not selective, notably also preventing 'interaction with DC-SIGN (Geijtenbeek et al., 2003). As a result, only studies using neutralising antibodies to these receptors can be interpreted as showing an interaction definitively.

Many bacteria and fungi can interact with c-type lectins (Geijtenbeek et al., 2004). For virus recognition, the MR may play a role in binding to HSV-1 (Milone and Fitzgerald-Bocarsly, 1998b; Rong et al., 2003) and influenza virus (Reading et al., 2000), although the interaction with either of these viruses has not been proved definitively. On the other hand, DC-SIGN binds to envelope glycoproteins of human immunodeficiency virus (HIV) (Geijtenbeek et al., 2000a), Ebola virus (Alvarez et al., 2002), HCMV (Halary et al., 2002), hepatitis C virus (HCV) (Lozach et al., 2003) and DV (Tassaneetrithep et al., 2003). Notably, a HSV-1 clinical isolate tested did not bind to DC-SIGN (Halary et al., 2002), although the sensitivity of this receptor to variability in glycosylation suggests that different strains of HSV-1 may yet bind DC-SIGN. This is because the interaction is dependent on different envelope glycosylation patterns that may also be affected by the cell type in which the virus is passaged (Lin et al., 2003). Functionally, it has been suggested that viral DC-SIGN binding will protect viruses from degradation and promotes efficient trans infection of other target cells (Geijtenbeek et al., 2000a; Halary et al., 2002).

The signalling events downstream of c-type lectin receptors have attracted interest recently. Dectin-1 and TLR2 can co-operate to increase zymosan induced signalling and cytokine secretion (Gantner et al., 2003). Cross-linking the mannose receptor, despite inducing upregulation of co-stimulatory molecules, inhibits IL-12 secretion and induces secretion of IL-10 (Nigou et al., 2001; Chieppa et al., 2003). Cross-linking BDCA-2 also inhibits IFN α production by PDC (Dzionek et al., 2001), whereas *Mycobacterium tuberculosis* capsid sugar, mannose-capped lipoarabinomannan (ManLAM), binding to DC-SIGN can inhibit IL-12 secretion by DC stimulated with LPS (Geijtenbeek et al., 2003). This demonstrates how pathogens may have evolved to

bind c-type lectin receptors with the purpose of subverting host immune responses. This issue will be explored further in chapter 6.

1.8.1.3 Other pattern recognition receptors

Recent studies have identified a pentraxin protein, PTX3, which plays an important role in the clearance of Aspergillus infection in mice and is secreted by MDC. Interestingly, PTX3 associates with various TLR and this may be a host mechanism to expand the repertoire of microbial structures that are recognised (Garlanda et al., 2002; Doni et al., 2003).

TLRs and c-type lectin receptors recognise PAMPs in the extracellular space, but many pathogens reach the cell cytoplasm so that hosts have had to evolve mechanisms to detect this process. For virus infections, a good example of this is the detection of dsRNA in the cytoplasm by protein kinase R (PKR) activation. The importance of this host recognition is exemplified by the presence of the influenza virus protein NS1 that can mask the viral dsRNA genome and prevent PKR activation (Diebold et al., 2003). For bacterial infections, the NOD family of proteins are intracellular receptors of bacterial lipids (Girardin et al., 2003a; Girardin et al., 2003b) and may play a role in shaping the cell response to cytoplasmic invasion by *Listeria monocytogenes* (McCaffrey et al., 2004). NOD proteins may have evolved to complement the detection of pathogens through intracellular recognition.

1.8.2 Endogenous ligands

1.8.2.1 Cytokines

The immature DC's immediate environment is the critical factor in determining the functional outcome of the cellular response. As well as to pathogens, DC can also respond to non-pathogen derived molecules. These include cytokines secreted by other cells, including non-immune cells (Lebre et al., 2003). Upregulation of MHC and costimulatory molecules is induced by $TNF\alpha$ (Sallusto and Lanzavecchia, 1994) and type I IFN (Padovan et al., 2002). However, it should be noted that, although most responses are dose-dependent, cytokine stimulation of DC is generally a weaker signal than TLR stimulation, but that the activation signals are likely to be additive (or synergistic), and cytokines can amplify the responses to other maturation stimuli. For example, cytokines secreted by monocytes adherent to human IgG coated-plastic, known as monocyte conditioned medium (MCM), can amplify the response to trace amounts of LPS to induce full DC maturation (Nersting et al., 2003). Despite the uniform effect on phenotype, cytokines can also skew T cell responses to vards Th1 or Th2. For example, IFN α can enhance IL-12 secretion in response to CD40L (Luft et al., 2002b), aiding the generation of Th1 responses (Brinkmann et al., 1993), whereas PGE₂, one of the active components of MCM, can skew towards Th2 responses (Kalinski et al., 1998). Alternatively, cytokines may have no effect on DC phenotype, yet prime for increased IL-12 secretion and Th1 development, as seen for IL-1 β (Luft et al., 2002a).

DC can also respond to signals emanating from the T cell interaction. These include T cell surface ligands, such as CD40L (Sallusto and Lanzavecchia, 1994), LIGHT (Morel et al., 2003) or 4-1BBL (Futagawa et al., 2002; Wilcox et al., 2002)). Cytokines secreted by T cells can also affect DC. Interestingly, both IFN γ and IL-4 can synergise with CD40L to increase IL-12 secretion by the DC (Snijders et al., 1998; Ebner et al., 2001). In order to prevent permanent Th1 switching, it is likely that other cytokines are secreted to prevent excess IL-12 secretion.

1.8.2.2 Molecules derived from tissue damage

There is evidence that the immune system can respond to molecules released from areas of inflammation. These include lipid mediators, such as prostaglandins, which can skew responses to other TLR ligands towards a Th2 bias (Kalinski et al., 1998; Gosset et al., 2003). Similarly, extracellular adenosine (Panther et al., 2003) and its derivative ATP (Wilkin et al., 2002) can also have Th2 inducing effects. Intriguingly both these molecules and PGE₂ can also promote DC migration (Panther et al., 2003; Scandella et al., 2002; Schnurr et al., 2004). There is also some evidence that DC can respond to proteins released from cell injury, such as hyaluronan (Termeer et al., 2002), uric acid (Shi et al., 2003) and heat shock proteins (HSP) (Palliser et al., 2004) to skew immune responses towards Th1. It should be noted that stringent precaution to

ensure the absence of microbial contaminants in the preparation of these products are necessary, as it may be difficult to exclude that they are merely potentiating the effects on DC of microbial molecules.

1.9 Regulation of dendritic cell migration

An important aspect of the DC lifecycle is the migration from periphery to LN. This emigration from tissues is tightly regulated by chemotactic gradients, in particular by the chemokines CCL19 and CCL21, which are constitutively expressed in LNs (Gunn et al., 1998; Ngo et al., 1998). The regulatory step is at the DC level, specifically the expression of the receptor for these ligands, CCR7 (Dieu et al., 1998). CCR7 is a G-protein coupled receptor with 7-transmembrane domains and induces Ca2+ signalling upon ligation by appropriate ligands (Sato et al., 2001). Mice deficient in CCR7 expression display impaired DC migration to lymph nodes, although it should be noted that lymph node architecture is also disrupted in these mice (Forster et al., 1999).

Recent work has identified that CCR7 mediated signalling has important functional effects on DC other than just inducing migration. CCR7 mediated intracellular signals activate the Rho GTPases, Rac and Cdc42 (Yanagawa and Onoe, 2003) and induce membrane ruffling, although curiously this only occurred in response to CCL19, not CCL21, and at concentrations greater than those needed to induce migration (Yanagawa and Onoe, 2002). The morphological changes in that study were interpreted as dendrite extension, but they bore no resemblance to dendrites observed on a fibronectin (FN) substratum (Swetman et al., 2002), highlighting how the morphological changes in response to these stimuli may be dependent on the substratum the cell is in contact with. Nevertheless, changes in morphology following activation of Rho GTPase are consistent with the regulation of the DC cytoskeleton by these molecules. The high concentration of CCL19 required to induce these changes may reflect the response of mature DC in the lymph nodes to assume a dendritic shape during antigen presentation to T cells (Benvenuti et al., 2004a). The recent findings that CCL19 and CCL21 transmit anti-apoptotic signals, via NF-kB and PI3K activation, are consistent with the notion that chemokine ligation of mature DC not

only promotes their migration to lymph nodes, but also aids their survival during antigen presentation (Sanchez-Sanchez et al., 2004).

Intriguingly, parallel studies have also revealed that CCL19 and CCL21 stimulation of mature DC' induced a rapid increase in endocytosis (Yanagawa and Onoe, 2003). The physiological relevance of this study may relate to an increase in antigen uptake prior to emigration from the tissue. In that way, a DC leaving the periphery is induced to make one last efficient 'sweep' of its environment before presenting this content in the LN. Interestingly, CCL3, responsible for migration of DC precursors into the dermis (Yoneyama et al., 2004), also induces DC endocytosis, indicating again how chemokines can govern DC antigen uptake as well as migration (Yanagawa and Onoe, 2003).

The regulation of DC migration may be important both in the induction of immunity and the maintenance of peripheral tolerance. Recent data has shown that immature DC can also acquire CCR7 expression and migrate to draining LN (Geissmann et al., 2002). Thus, the expression of CCR7 and the migration potential of DC are independent of the maturation state of the cell (Llodra et al., 2004). The implication of these findings is that both immature and mature DC can enter draining LNs. Immature DC still express low levels of antigen loaded MHC molecules and co-stimulatory molecules, and this has been proposed as one way to maintain peripheral tolerance. T cells that encounter antigen on immature DC do not receive sufficient signal to differentiate into effectors and are either deleted or become anergic. In this way T cells reactive to the host's peripheral microenvironment are excluded from the T cell repertoire and autoimmunity avoided (Hawiger et al., 2001; Bonifaz et al., 2002).

A further layer of complexity to the regulation of DC migration is observed by the fact that CCR7 expression is not predictive of the migration capacity of the DC, as the signalling machinery can be uncoupled from the receptor (Penna et al., 2001). Important factors that may link the two systems include inflammatory mediators, such as PGE2 (Scandella et al., 2002) and adenosine (Robbiani et al., 2000) and ectoenzymes, such as CD38, that enhance extracellular derived Ca2+ fluxes, that are critical for the migration of both DC precursors into skin and mature DC emanating

from areas of inflammation into LN (Partida-Sanchez et al., 2004). This regulation may enable DC to migrate more appropriately in response to inflammatory environmental cues.

Human PDC are predominantly found in the blood and their migration pattern is likely to differ from that of MDC. Although recruited to some areas of inflammation in the periphery, the chemokines required for this have not been characterised. Mouse footpad infection with HSV recruits MDC, but not PDC, into the dermis. Concurrently, both MDC and PDC are recruited to LN. PDC enter in a process dependent on Eselectin expression on HEV and the chemokine CXCL9, acting via CXCR3 expressed on PDC (Yoneyama et al., 2004). Thus, this in vivo study demonstrates how the migration pattern of the two subsets is diverse and may be suited to their function: recruitment of MDC into area of inflammation and PDC into LN, possibly to amplify immune responses through type I IFN secretion (Dalod et al., 2003).

1.10 Signalling pathways involved in dendritic cell physiology

The regulation of DC maturation and migration lies at the level of intracellular signalling. Different PAMPs, as well as viruses can activate these pathways and alter DC function. Furthermore, viruses may regulate the activation of these pathways for their own survival, making them critical to the understanding of the functional outcome to the DC-virus interaction. Owing to the cell-specific nature of signalling regulation of cellular responses, the literature review below will focus specifically on studies carried out on DC. Data relating to signalling regulation in other cell types is also discussed where that may prove insightful for extrapolating function in DC.

1.10.1 NF-kB

The family of NF-kB transcription factors is made up of 5 members: p65 (RelA), p50/p105, c-Rel, p52/p100, and RelB. The classical regulation of these members is quite similar. They are retained in the cytoplasm by the inhibitory proteins IkB α and IkB β . Phosphorylation of these proteins results in ubiquithation and rapid protessomal

degradation. Phosphorylation of the IkB proteins is in turn regulated by IKK family (Karin and Ben Neriah, 2000).

DC maturation is preceded by an increase in DNA binding activity of RelB, p50, p52, and c-Rel (Verhasselt et al., 1999; Neumann et al., 2000), suggesting that NF-kB activation positively regulates DC maturation. Furthermore, transfecting DC with IkBa expressing adenoviral vectors prevented both the upregulation of molecules associated with maturation and the secretion of inflammatory cytokines, including IL-12, IL-6 and TNFa (Yoshimura et al., 2001). The role of the p50 subunit in IL-12 secretion by DC in response to LPS has been formally demonstrated using siRNA (Laderach et al., 2003). It should be noted that the pathways become increasingly divergent and stimulus specific upstream of the effector molecules/transcription factors, as exemplified by the role of IKK2 in CD40L but not LPS induced maturation of DC (Andreakos et al., 2003). In this respect, it is interesting that PKCE is phosphorylated rapidly (within 15 min) after LPS stimulation of DC, and promotes TNFa and IL-12 secretion by facilitating degradation of IkB and therefore NF-kB access to the nucleus. However, this enzyme does not play a role in the upregulation of costimulatory molecules by LPS and subsequent inhibition of its activity only prevented Th1 skewing of co-cultured T cells, but not their proliferation, demonstrating how NF-kB can differentially activate effector pathways (Aksoy et al., 2002). Indeed, NF-kB subunits are differentially recruited upon DC maturation by various stimuli (Neumann et al., 2000; Hofer et al., 2001). The role of individual subunits has only been addressed in the murine system using knockout mice. Interestingly, large degree of redundancy has been observed. Mice lacking c-Rel showed no impairment in IL-12 secretion (Mason et al., 2002), whereas mice deficient in both p50 and c-Rel demonstrated reduced IL-12 secretion, despite DC from these mice expressing normal levels of MHC and costimulatory molecules (Ouaaz et al., 2002). It should also be noted that the role of the NF-kB subunits may vary between DC subsets, as c-Rel is important in IL-12 p35 expression in CD8 α + DC (Grumont et al., 2001), but not in BMDC (Mason et al., 2002).

NF-kB is also involved in maintaining DC viability, such as after Fas ligation, an event that is pro-apoptotic in most cell types. Fas induced NF-kB activation in DC prevents

these cells from dying when DC encounter FasL on T cells (Rescigno et al., 2000; Guo et al., 2003). Interestingly, this is a pathway that may be targeted by viruses, as MV infection renders DC susceptible to T cell mediated induced apoptosis by FasL (Servet-Delprat et al., 2000a).

1.10.2 p38 MAPK

DC stimulation with LPS or CD40L induces the phosphorylation of p38 Mitogen-Activated Protein Kinase (MAPK) (Aicher et al., 1999; Ardeshna et al., 2000). This occurs rapidly (within 30 minutes) and is transient (phosphorylation is absent after 2 hours) (Handley M, submitted for publication). Studies using inhibitors of p38 MAPK have shown that this activation is necessary for important physiological changes during DC maturation. The upregulation of co-stimulatory molecules, CD80 and CD86, is p38 dependent, as is the transcription and secretion of IL-12, and both these factors contribute to enhancing T cell stimulation (Aicher et al., 1999; Lu et al., 1999; Ardeshna et al., 2000; Vidalain et al., 2000b; Puig-Kroger et al., 2001). It is worth noting that there are inconsistencies in the literature on the relative contribution of the p38 pathway in the regulation of MHC class II expression (Ardeshna et al., 2000), which may stem from the varying length of exposure to the inhibitor and the concentrations of inhibitor used. The latter is particularly important as excessive amounts of inhibitor may also affect pathways other than the one intended.

Although p38 MAPK can phosphorylate and activate several transcription factors directly (e.g. ATF2, CREB), it is also important in positively regulating DNA binding activity of NF-kB during DC maturation (Puig-Kroger et al., 2001). A model for the role of p38 in DC maturation was proposed, whereby p38 is important for phosphorylating histones, inducing nucleosomal rearrangement, and thus exposing promoter sites of some genes to greater access by NF-kB subunits (Saccani et al., 2002). This model can apply to genes whose promoters lie within nucleosomes and may be regulated in this way, such as IL-12p40 (Weinmann et al., 1999). However, some genes, such as TNF α , are not regulated in this way, yet the secretion of TNF α and presence of its mRNA is dependent on p38 MAPK activity (Arrighi et al., 2001), suggesting that different mechanisms may be responsible, such as stabilisation of the mRNA transcript (Dean et al., 1999).

1.10.3 ERK MAPK

The ERK signalling pathway is antagonistic to the activity of p38. It prevents the upregulation of MHC and costimulatory molecules (Arrighi et al., 2001; Puig-Kroger et al., 2001). With regards to IL-12, ERK induced IL-10 secretion inhibits IL-12 production (Xia and Kao, 2003). This is likely to be an endogenous negative feedback mechanism that limits IL-12 secretion to avoid immunopathology, and some pathogens, such as *Leishmania* amastigotes, may exploit this regulation by skewing towards ERK activation, instead of p38 MAPK, after CD40 stimulation (Mathur et al., 2004). Furthermore, different TLR ligands differentially activate p38 and ERK to varying degrees. Specifically, TLR2 ligation induces greater degree of ERK activation, which in turn activates c-Fos that is responsible for inhibiting IL-12 and increasing IL-10 secretion (Agrawal et al., 2003; Dillon et al., 2004). Therefore, the relative activation of the ERK pathway, through concomitant activation of c-Fos, is an important regulator of IL-12 secretion in DC.

Like NF-kB, the ERK pathway is also involved in maintaining DC viability after stimulation. Studies in murine DC suggested that ERK promotes DC survival (Rescigno et al., 1998), and experiments using an ERK inhibitor in human DC support a role for this MAPK in maintaining DC survival after maturation (Yu et al., 2004).

1.10.4 PI3K

PI3K is activated by most TLRs and members of the TNFR family that signal via TRAF6. By activating the pro-survival kinase, Akt, PI3K prevents cell death after DC maturation (Yu et al., 2004; Ardeshna et al., 2000) but also plays an important role in IL-12 secretion, although with disparate effect between mouse and man. PI3K potentates IL-12 secretion in humans (Re and Strominger, 2001; Yu et al., 2004), whereas it negatively regulates its secretion in the mouse (Fukao et al., 2002). Although possibly intrinsic to species differences, the differential role of PI3K may

also relate to the DC models used in the two systems. Nevertheless, if PI3K does play an important role in the regulation of IL-12 secretion, then it complicates the comparison of studies between mice and humans (see chapter 5).

1.11 Role of dendritic cells in the control of HSV infections

Sections 1.6-1.10 demonstrate the evolving understanding of DC physiology, and how this involves many layers of complexity in order to shape an appropriate effective immune response in the face of maintaining immune tolerance to self. Several pieces of evidence have indicated that DCs play an important role in vivo in the context of HSV infection. Mice deficient in the costimulatory molecules, CD80 and CD86 control HSV infections poorly (Thebeau and Morrison, 2002). Conversely, vaccination of mice with either DC loaded with HSV antigens (Schon et al., 2001), or DNA encoding CCL19 or CCL21 (Lee et al., 2003b), offer protection against HSV infection in these mice.

These studies serve to highlight the potency of DCs in generating T cell responses that are protective for HSV. They do not specifically highlight the role that DCs and their subsets play in the natural course of infection. This has been more directly approached by studies demonstrating that the severity of cutaneous HSV infection inversely correlates with the number of CD1a+ LC in the epidermis (Sprecher and Becker, 1986; Sprecher and Becker, 1989; Yasumoto et al., 1986). It is interesting that this DC subset may not directly play an important role in antigen presentation following cutaneous/mucosal infection (Zhao et al., 2003; Allan et al., 2003), but may be important in the role of antigen transfer to LN resident DC subsets, such as the CD8 α + DC that can activate CD8+ T cells (Iyoda et al., 2002; Schulz and Reis e Sousa, 2002).

Further in vivo studies have attempted to study the role of different DC subsets in the initiation of anti-HSV T cell responses by isolating different subpopulations from LN. However, in none of these models were any DC subsets depleted prior to HSV infection. Thus, while $CD8\alpha$ + DC may stimulate anti-HSV CD8 T cell responses and CD4 T cell responses initiated by submucosal CD11b+ DC (Allan et al., 2003; Zhao et al., 2003), it is not clear if either of these subsets are essential, or whether redundancy

exists within the DC system. Such an interpretation would be further complicated by the unresolved questions regarding ontogeny of DC subsets (Shortman and Liu, 2002) and the possibility that DC subsets may differentiate into each other, as defined by the expression of surface molecules. For example, it is possible that 'migrating' myeloid CD11b+ DC may differentiate into 'resident' CD8 α + DC in the LN (Moron et al., 2002). Despite their accumulation in lymph nodes after peripheral infection (Smith et al., 2003b; Yoneyama et al., 2004), PDC were excluded from playing a role in T cell activation after peripheral HSV infection (Zhao et al., 2003), so their precise function in vivo remains unclear. Recently, a blunted PDC response was associated with more severe dengue virus (DV) disease (Pichyangkul et al., 2003), but such a correlation may not be present for a more peripheral infection, such as that caused by HSV-1.

There has not as yet been any evidence to show that DCs are infected by HSV-1 in vivo in humans. However, a combination of the evidence from mice studies, the critical function of DC in initiating any primary T cell response and the similar susceptibility of human DC in vitro to HSV-1 infection as epithelial cells (see chapter 3), suggest that the interaction between this cell and the virus is a relevant one in vivo.

1.12 Role of dendritic cells in the control of viral infections

The role of DC in inducing immunity suggests that, in general, it would be an evolutionary advantage for viruses to subvert this system. Preventing or delaying the presentation of viral antigens by DC may permit the virus sufficient time to establish in the host and replicate to transmissible titres, and perhaps even to establish latency. As a result, prior to, during and after the completion of the work presented in this thesis, the infection of DC by many other viruses has also been studied and the functional outcomes analysed.

The frequency of all DC subsets in peripheral blood is between 0.1% and 2%, depending on criteria used (Vuckovic et al., 2004), but it is possible to obtain large numbers of MDC in vitro from monocyte precursors (Sallusto and Lanzavecchia, 1994), a differentiation event that also occurs in vivo (Randolph et al., 1999).

Therefore, as summarised in table 1.2, the vast majority of studies have analysed the interaction of viruses with monocyte derived dendritic cells (MDDC).

1

.

Enhanced	No change	Impaired
Dengue virus (Ho et al., 2001)	HIV-1* (Sapp et al., 1999)	Varicella Zoster Virus (Abendroth et al., 2001)
Hantavirus (Raftery et al., 2002)	HCV* (Longman et al., 2004)	Human Cytomegalovirus (Moutaftsi et al., 2002)
Influenza virus (Cella et al., 1999b)		HHV-6 (Kakimoto et al., 2002)
		Measles Virus (Grosjean et al., 1997)
		Parainfluenza Virus (Plotnicky-Gilquin et al., 2001)
		Respiratory Syncytial Virus (Bartz et al., 2003)
		Adenovirus (Tuettenberg et al., 2004)
		Vaccinia virus (Engelmayer et al., 1999)
		Ebola virus (Mahanty et al., 2003)
		Lassa fever (Mahanty et al., 2003)
		HIV-1* (Donaghy et al., 2003)
		HCV* (Kanto et al., 1999)

1

(*) – Refers to the function of MDDC from patients with chronic or long term infection, not MDDC infected in vitro with the relevant virus.

Table 1.2 Functional consequences of viral infection of MDDC. The capacity of virus infected DC to activateT cells is compared to uninfected DC.

It is evident that some viruses do not impair DC function following infection, whereas many others hinder the DC's ability to stimulate T cells. A wide variety of DC parameters are affected that may impinge on the ability to stimulate T cells. These include the morphology of the DC, the cytokine secretion profile, the viability and the expression of surface molecules, such as MHC and costimulatory ones. These will all be analysed in relation to HSV-1 in this thesis, and comparison between the effect of HSV-1 and other viruses on DC will be made in the appropriate sections.

The absence of an in vitro differentiation model, and the paucity of ex vivo derived PDC have limited the number of studies detailing their interaction with viruses. One parameter that has been studied is the number of PDC circulating in peripheral blood. However, this may not give a clear indication of the direct effects of the viral infection, as it may result from peripheral or central migration of these cells to areas of infection or antigen presentation respectively. Nevertheless, impaired PDC IFN α secretion or T cell stimulation has been observed in patients with adult T cell leukaemia (ATL) caused by human T-cell lymphotropic virus type 1 (HTLV-1) infection (Hishizawa et al., 2004), chronic HCV infection (Goutagny et al., 2004) and HIV-1 infection (Donaghy et al., 2003), suggesting that viruses may also alter the function of this subset of DC.

1.13 Hypotheses of thesis

In this introduction, the main features of the cell biology of DC and the central role they play in the interaction with viruses, and in the generation of antiviral responses have been discussed. On this basis, two main hypotheses will be investigated in this thesis:

- 1. HSV-1 infects DC and disrupts their function as APC.
- 2. The host has evolved mechanisms to bypass this DC-based viral evasion and is therefore able to generate a strong antiviral response capable of resolving the infection.

1.14 Aims and objectives of thesis

The aim of this thesis will be to test these two hypotheses using an in vitro culture system of human MDDC and a laboratory strain of HSV-1. The investigation will be divided into four main objectives, each discussed in a separate chapter:

- 1. Characterisation of the interaction between HSV-1 and DC, defining the normal range of parameters that may be affected by the virus.
- 2. Functions that define DC, namely changes in T cell stimulation, cytokine secretion and surface phenotype, will be assessed following HSV-1 infection.
- 3. Host mechanisms to enhance immune responses after HSV-1 infection will be investigated.
- 4. The functional consequence of interaction between HSV-1 and the DC surface will be investigated with a view to understanding the co-evolution between the virus and the host that permits both these species to co-exist.

In the concluding chapter of this thesis, these findings will be discussed in the broader context of the role of DC in the control of human viral infections, along with a model of a peripheral herpetic lesion, with DC playing a central role both in its establishment and resolution. Finally, the data acquired will be placed into a clinical setting, and its application to the fields of HSV vaccinology and HSV-based gene therapy will be considered.

Materials & Methods

Chapter 2

Materials & Methods

2.1 Materials

All reagents were of the highest grade commercially available. Unless otherwise stated, all chemicals and reagents were obtained from Sigma (UK) Ltd. (Poole, Dorset, UK). Disposable pipettes were from Philip Harris (Leicestershire, UK). Pipette tips were from Elkay (Hampshire, UK). All culture plates, flasks and centrifuge tubes were from Nunc (Rochester, NY, USA), unless otherwise stated.

2.2 Cell and Tissue Culture

2.2.1 Safety and sterile conditions

All work involving human tissues were performed in a M.D.H. InterMed class 2 safety cabinet, using sterile techniques. All short and long term cell cultures were carried out at a temperature of 37°C in an atmosphere compromising 95% air and 5% CO₂.

2.2.2 Culture medium

Unless otherwise stated, all cell cultures were conducted in Roswell Park Memorial Institute medium-1640 (RPMI-1640) (Gibco BRL, Paisley, UK) that was supplemented with 100U/ml penicillin, 100µg/ml streptomycin, 2mM L-glutamine (all from Clare Hall Laboratories, Cancer Research UK, London, UK) and 10% v/v heat inactivated foetal calf serum (FCS) (PAA Laboratories, Linz, Austria). The combined medium was denoted as complete medium (CM).

2.2.3 Culture condition

Unless otherwise stated, all incubations at 37° C were carried out in an incubator at 5% CO₂. Incubations at 4° C were carried out on ice.

2.2.4 Cell counting

All cell samples were counted using a Neubauer haemocytometer. 10μ l of cell suspension was mixed thoroughly with an equal volume of 0.04% trypan blue stain

(Gibco). This technique allowed a rapid assessment of cell viability, since trypan blue is excluded from live cells.

2.3 Cell lines: characteristics and growth conditions

2.3.1 BHK cells

Baby Hamster Kidney 21 clone 13 cells (BHK) were a gift from Y. McGrath and grown in 1x Dulbecco's modified Eagle's medium (DMEM) containing 10% (v/v) FCS, 100 IU/ml penicillin and 100 μ g/ml streptomycin. The combined medium was denoted as BHK complete medium (BCM).

BHK cells were grown in 175 cm² flasks and were passaged when they were 80-90% confluent. Cells were washed with Hank's balanced salt solution (HBSS) (Gibco) at room temperature, detached from the surface of the flask with trypsin/versene (1:10) (Gibco) at room temperature, and disaggregated. The reaction was stopped by adding BCM and cells were aliquoted in 1:10 ratio in new flasks containing BCM.

2.3.2 HeLa cells

HeLa cells were a gift from A. Kwan and grown in CM in 80 cm^2 flasks and passaged in the same way as BHK cells.

2.3.3 Storage of cell lines

For long term storage, cells were resuspended in 10% (v/v) DMSO in FCS at an approximate concentration of 5×10^5 cells/ml. The freezing vials were slowly cooled in a polystyrene box to -70°C and then transferred to liquid nitrogen for longer term storage. For recovery, cells from one vial were thawed rapidly and transferred into a 25 cm² flask containing the appropriate selection medium. Cells were passaged within 2 days depending on confluence.

All cell lines were regularly tested and shown to be negative for the presence of both mycoplasma antigens and DNA.
2.4 Primary cells

2.4.1 Isolation of peripheral blood mononuclear cells

PBMC were isolated from healthy volunteers by density gradient centrifugation. Peripheral blood (60 or 120ml) was obtained by venesection using a 21g butterfly needle into a heparanised syringe (original heparin concentration 5000IU/ml - C.P. Pharmaceuticals Ltd. Rexam, UK). This was diluted into an equal volume of HBSS. 30 ml of this diluted solution were layered carefully over 17.5 ml of Lymphoprep 1077 (Nycomed Pharma, Oslo, Norway) in a 50 ml tissue culture tube (Sterilin) and centrifuged for 30 minutes at 600g (brake off) at room temperature. The interface over the Lymphoprep contained the PBMC and was aspirated and washed three times with HBSS.

2.4.2 Derivation of monocyte-derived dendritic cells

The PBMC fraction from 120ml of blood was resuspended in 36 ml of CM and incubated (3ml per well) in 6 well tissue culture plates (Falcon, Becton-Dickinson, Mountain view, CA, USA) for 2 hours. Non-adherent cells were removed and the adherent cells were cultured in fresh CM supplemented with 100 ng/ml human recombinant GM-CSF and 50 ng/ml IL-4 (both gifts from Schering-Plough Research Institute, Kenilworth, New Jersey). Monocytes subsequently differentiated into MDDC.

MDDC were purified from any other non-adherent cells on day 4 of cell culture, according to the protocol described previously (Alderman et al., 2002). Loosely adherent cells were collected and layered carefully over 5ml of Lymphoprep. Cells were centrifuged at 600g for 30 minutes at room temperature (brake off) to remove dead cells and debris. The Lymphoprep interface was removed and washed three times in HBSS before being resuspended in 2ml ice cold CM and incubated on ice for 30 minutes in the presence of mouse anti-human monoclonal antibodies (mAb) that recognised the major contaminating cell populations: CD19 mAb removed B cells, CD3 T cells, and CD2 T cells and NK cells. Excess antibody was washed off twice in

ice cold HBSS and the cells were resuspended in ice cold CM. 10μ l of immunomagnetic beads coated with sheep anti-mouse Ig (Dynal, Merseyside, UK) per 10^6 non-DC (as determined by cell size by microscopy). The cell-bead mixture was rotated for 45 minutes at 4°C. Contaminating cells and excess beads were removed by placing sample tubes adjacent to a magnet for 2 minutes. Supernatants were transferred to fresh tubes and placed adjacent to a magnet for a further 2 minutes. The supernatants, which contained the purified DC, were aspirated, and the cells cultured for a further 3 days in CM with fresh GM-CSF and IL-4 at a concentration of 5×10^5 DC/ml. They were used as a pure population of immature DC.

2.4.3 Preparation of T cells

Following 2 hour culture, non-adherent PBMC were removed and resuspended at approximately $5x10^6$ cells/ml in 10% (v/v) DMSO in FCS. Samples were transferred to 1ml cryotubes (Nunc) and frozen at -80°C in a polystyrene freezing box.

When required, frozen vials were thawed rapidly in a water bath at 37° C and then washed three times in HBSS. T cells were purified by negative immunomagnetic depletion as described for DC in section 2.4.2. The only difference was that the antibodies used to deplete contaminating cells were HLA-DR (monocytes, DC, B cells, activated T cells), CD14 (monocytes) and CD19 (B cells). This procedure removed about 10% of T cells that express HLA-DR, and therefore considered to be activated. At the end of purification the T cells were >95% viable as determined by trypan blue exclusion.

2.5 HSV-1 preparation

2.5.1 Virus growth

The HSV-1 construct used in this study was derived from HSV-1 strain 17+, and contained a cassette consisting of the CMV IE gene promoter driving expression of green fluorescent protein (GFP) (Clontech, Hampshire, UK). This cassette was inserted

into the UL43 gene as reported previously (Coffin et al., 1996). This virus was a gift from R.S. Coffin.

For large-scale viral culture, $10x175 \text{ cm}^2$ tissue culture flasks containing 90% confluent BHK cells were infected at a multiplicity of infection of MOI 0.01-0.03 and grown in fresh media for up to 48 hours until complete cytopathic effects were observed. Flasks were frozen immediately at -80°C. After thawing, the cellular debris was removed by centrifugation at 3500 rpm for 45 minutes at 4°C. The supernatant was removed immediately, filtered through a 5µm, followed by a 0.45µm filter, and spun at 12000 rpm for 2 hours at 4°C in a GS12 rotor. The supernatant was discarded and the viral pellet was resuspended gently in 1ml HBSS. The resuspended pellet was then sonicated 3 times for 10 seconds in a water bath sonicator and stored at -70°C. The virus stock was titred by the viral infectivity assay (see section 2.5.2). The usual titre achieved was 10° plaque forming units (pfu)/ml in 1 ml. All virus stocks used were tested and shown to be negative for the presence of both mycoplasma antigens and DNA.

2.5.2 Viral infectivity assay

Viral infectivity, and therefore titre, was determined from the ability of HSV-1 to form plaques on cell monolayer. Serial dilutions $(10^{-3} \text{ to } 10^{-8} \text{ pfu})$ of virus in DMEM without FCS were prepared and added to 80% confluent BHK cells in 6 well plates in triplicates. After 1 hour incubation at 37°C, 2.5ml of 1:2 (v/v) of 1.6% (w/v) carboxymethyl cellulose:DMEM supplemented with 10% FCS were added to each well. Cells were then incubated for 48 hours at 37°C and the number of plaques in each well were counted using GFP fluorescence in order to determine the virus titre in pfu/ml.

2.5.3 Viral inactivation

UV-inactivation was carried out by exposing the HSV-1 preparation (10^7 pfu/ml) to UV light at a distance of 6 cm, with a short wavelength UV-B light source delivering 20 mW/cm² (R-52 Grid Lamp, UVP, Upland, USA). A titration of time of exposure to

UV light and GFP expression by infected DC is shown in fig. 3.6C. For the experiments in chapter 6, 20 minutes of UV-inactivation were used.

Formaldehyde fixation was carried out by neutralising HSV-1 (10^9 pfu/ml) with equal volume of 2% formaldehyde (BDH, Poole, UK). The virus stock was diluted to 10^7 pfu/ml with RPMI and excess formaldehyde neutralised by adding 0.175% (w/v) sodium bisulphite solution (final concentration). Exposure of DC to the equivalent concentrations of fixative and neutralising agent (formaldehyde and sodium bisulphite) did not alter DC surface phenotype, excluding indirect effects on DC by the viral treatment protocol (fig. 6.8). Neutralisation of viral gB, gC or gD was carried out by co-incubating equal volumes of virus (@ $2x10^7$ pfu/ml) and LP2 or AP7 antibody +/-12.5 µg/ml heparin (C.P. Pharmaceuticals Ltd.) at 37° C for 15 minutes, and then used to infect DC (see section 2.6). A titration of heparin addition is shown in fig. 6.2B. A titration of LP2 titration is shown in fig. 2.1. The antibody concentration of the neat supernatant of AP7 was approximately four times lower than that of LP2 (S. Bell, personal communication). Therefore, LP2 antibody was still sufficient to neutralise the virus (fig. 2.1) and also controlled for the antibody concentration of AP7.

2.6 HSV-1 infection of DC

For all experiments other than infection of FN-adherent DC (see below), appropriate numbers of DC were centrifuged in a 15 ml tube at 1400 rpm for 5 minutes. Then the cell pellet was infected with appropriate pfu of HSV-1, depending on the required MOI. The virus-DC mixture was incubated at 37° C for 1 h. The cells were washed once and re-plated at $5x10^{5}$ cells/ml in CM with fresh GM-CSF and IL-4.

For all virus infections, uninfected groups received RPMI treated in the same manner as the virus. Previous experiments had determined that mock viral preparations had equivalent effects on DC as RPMI (fig. 2.2). The presence of significant stimulating non-viral contaminants in the preparation was excluded by neutralisation with anti-gD mAb (LP2 clone) in fig. 6.14. For infections of FN-adherent DC, $5x10^4$ DC were allowed to adhere to FN for 2 hours in CM supplemented with GM-CSF and IL-4 (see section 2.10). RPMI or HSV-1 was then added at a MOI of 1 for 1 hour. Any unbound virus was then removed by washing twice with 1ml CM, before adding 500µl CM supplemented with GM-CSF + IL-4. The DCs were incubated at 37°C for 8 hours and the morphology analysed by confocal microscopy.

2.7 HSV-1 serology

This method was adapted from that devised by G. Simpson (PhD thesis). To test for the presence of neutralising antibodies to HSV-1, sera were collected from volunteers at the same time as venepuncture for DC and T cell isolation, heat inactivated to remove the effect of complement components and incubated at 37°C with 10⁵ pfu/ml HSV-1 for 15 minutes. DMEM supplemented with 10% FCS was used as a negative control. The resulting HSV-sera mixture was titrated on BHK in a viral infectivity assay as described in section 2.5.2. A greater than 3 log decrease in plaque formation was interpreted as reflecting the presence of neutralising anti-HSV antibodies in the serum sample. Neutralising antibodies can be type common, and this assay does not therefore distinguish between previous infection with HSV-1 and HSV-2 (Gorander et al., 2003).



Figure 2.1 Neutralisation of HSV-1 by LP2 clone anti-gD mAb. HSV-1 was neutralised with a range of concentrations of mAb and used to infect DC. After 16 hours of culture, expression of GFP was assessed. Representative of two independent experiments.



Figure 2.2 Effect of mock viral preparation on DC phenotype. DC were infected with RPMI (uninfected group), a mock viral preparation, or UV-inactivated HSV (UV-HSV), and cultured for a further 16 hours. Representative of at least two independent experiments.

2.8 HSV-1 replication in DC and HeLa cells

 $5x10^5$ DC and HeLa cells were infected at a MOI of 1 and aliquots of the supernatants and the infected cells were harvested at 0, 24, 48, 72 and 96 hours post-infection (at the same cell concentration) and stored at -80°C until use. Harvested samples were thawed, disrupting the cells, and the virus yield was assessed by plaque assay

2.9 DC stimulation

In some experiments control and infected DC were stimulated on day 7 by addition to the wells of LPS, polyriboinosinic polyribocytidylic acid (Poly(I:C)), IFN α 2c (Roche, Welwyn Garden City, UK) or gD(285t) (gift from Dr. C. Krummenacher, University of Philadelphia, Philadelphia, USA) and cultured for 16 hours. All stimuli were added in a 1:100 dilution, so as not to disturb the total volume and cytokine concentrations in the CM.

2.10 Assessment of DC morphology

In some experiments DC were plated on FN-coated glass cover slips, which had been pre-coated overnight in 20 μ g/ml FN (Sigma) in HBSS. After 2 hours of adhesion, DC were stimulated with 100 ng/ml LPS or infected with HSV-1 as described in section 2.6.

2.11 Microscopy

2.11.1 Confocal microscopy

All confocal microscopy was carried out by a confocal microscope (Bio-Rad Confocal microscope, Hercules, California, USA). GFP fluorescence was recorded by a 488nm excitation laser and detected in the fluorescence channel by a 522 ± 32 nm emission filter. The images were analysed using Confocal Assistant and Adobe Photoshop software.

2.11.2 Electron microscopy

Uninfected or HSV-1 infected DC were harvested 16 hours after infection, and pelleted by centrifugation at 4500 RPM in a bench top centrifuge in sterile eppendorf tubes. Cell pellets were fixed carefully in 3% buffered gluteraldehyde (0.1 M, pH 7.4) while avoiding disruption or dispersion of cells from the pellet. Pellets were embedded in resin and sections were cut on a Leica Ultracut E ultramicrotome. Ultra-thin sections were cut using a Diatome diamond knife and collected on naked 300 mesh copper grids. These were stained with a solution of 25% uranyl acetate in 100% methyl alcohol for 20 minutes, followed by Reynold's lead citrate staining for 20 minutes. The grids were examined in the Jeol 1200EX and Jeol 120CX electron microscopes at a magnification specified on the micrographs.

2.12 Secreted cytokine measurement

After 16 hours of treatment or infection, supernatants were harvested and used to assay for various cytokines. Secreted type I IFN was measured by a collaborating group using a bioassay as described (Foster et al., 2000). Secreted IL-12 was measured by ELISA: IL-12 p40 (R&D Systems Europe, Abingdon, UK) and IL-12 p70 (eBioscience, San Diego, USA). Secreted TNF α was measured by ELISA (eBioscience). Representative standard curves for ELISAs are shown in fig. 2.3.

2.13 Supernatant transfer studies

In some experiments, the effect of supernatant of HSV-1 infected DC on autologous uninfected DC was tested. Supernatants from uninfected and infected DC were taken at 16 hours incubation, centrifuged in a minifuge at 16000g for 30 minutes and passed through a 0.2-µm-pore-size filter (Sartorius AG, Goettingen, Germany), to remove any viral particles and cell debris, before adding to autologous DC. Direct assay of these supernatants found no remaining live virus, both by plaque assay (data not shown) and by GFP expression (fig. 5.1). Supernatant-treated DC were then incubated for 16h and examined for surface phenotype changes and IL-12 secretion.

2.14 Detection of antigens

2.14.1 Extracellular immunofluorescence staining

Within individual experiments, the number of cells stained in all samples was the same to ensure a constant cell: antibody ratio. The usual number of cells stained was 5×10^4 – 10^5 cells. These cells were harvested and resuspended in 50µl of blocking buffer for 15 minutes at 4°C. The blocking buffer was HBSS containing 0.1% NaN₃ and 10% rabbit or foetal calf serum depending on whether mouse or rabbit primary antibodies were used respectively. 50µl of primary mAb (supernatants or determined optimum concentrations of purified antibody) were added to the cell suspension in a 96-well round-bottom plate (Nunc), and then incubated for a further 30 min at 4°C. For experiments utilising primary antibodies not directly conjugated to fluorochromes, cells were washed twice with blocking buffer, to remove any unbound antibody, and then incubated for a further 30 minutes at 4°C with the relevant secondary antibody. For mouse primary antibodies, either a fluorescein isothiocyanate (FITC) conjugated rabbit anti-mouse IgG (Dako, Glostreup, Denmark) diluted 1:20 in blocking buffer or a R-phycoerythrein (PE)-conjugated goat anti-mouse IgG (Dako) diluted 1:25 in blocking buffer were used. For rabbit primary antibodies, a FITC- or R-PE-conjugated goat anti-rabbit IgG (Southern Biotech Associates, Birmingham, USA) diluted 1:20 in blocking buffer were used. Finally, cells were washed twice in HBSS containing 0.1% NaN3, agitated and fixed by adding 50µl of HBSS containing 0.1% NaN3, followed by 100µl of HBSS containing 3.7% (v/v) formaldehyde (BDH, Poole, UK). Samples were stored at 4°C in the dark and analysed within 24 hours.

Detection of opsonised HSV-1 particles was achieved by harvesting infected cells and blocking non-specific antibody binding with blocking buffer for 15 minutes at 4°C. R-PE-conjugated goat anti-mouse IgG (Southern Biotech) or R-PE goat anti-rabbit IgG (Southern Biotech) was added subsequently to the cell suspension for 30 minutes at 4°C. The cells were then washed and fixed as detailed above.

2.14.2 Intracellular Immunofluorescence staining

For the staining of intracellular targets, cells were harvested and fixed in 4% formaldehyde for 10 min at 4°C at a concentration of $\leq 10^6$ cells/ml. This was followed by a wash in HBSS supplemented with 2% FCS and 0.1% NaN₃ (FACS buffer). All subsequent washes and incubations were carried out in FACS buffer supplemented with 0.5% (w/v) saponin (PERM buffer). Non-specific antibody binding was prevented by incubating cell suspensions with 10% (v/v) goat serum for 15 min at 4°C. Subsequently, the primary antibody was added and incubated for a further 30 min at 4°C. Cells were washed twice in PERM buffer and then incubated for a further 30 min at 4°C with the secondary antibody, R-PE-conjugated goat anti-mouse IgG (Dako). Cells were washed twice more with PERM buffer and once with FACS buffer prior to immediate examination by flow cytometry.

2.14.3 Intracellular cytokine staining

After HSV-1 infection (MOI = 3), DC were cultured in CM supplemented with $2\mu M$ monensin (Sigma) for a maximum of 12 hours and then fixed, permeabilised and stained as detailed in section 2.14.2.

2.14.4 Flow cytometry

Stained cells were acquired on a FACScan (Becton-Dickinson) using CellQuest software version 3.3 (Becton-Dickinson). For each sample at least 5000 events were acquired. Analysis of the data was carried out using WinMDI software (Joseph Trotter, Scripps Research Institute). A measure of the amount of fluorescence emitted by the cells in the appropriate channels was determined by the median fluorescence intensity (MFI). Where appropriate, a marker was set such that <2% of negative control cells gave a fluorescence signal beyond this level. The percentage fluorescence (% +ve) refers to the percentage of cells with fluorescence above this marker.

2.15 Proliferation assays

All proliferation assays were conducted using 10^5 purified T cells incubated with increasing numbers of DC. Experiments were performed in 96-well flat bottom plates (Nunc). Quantification of cell proliferation was by [methyl-³H] thymidine (ICN Pharmaceuticals Inc., CA, USA) incorporation. Cells were pulsed with 10µl of 100μ Ci/ml [methyl-³H] thymidine for the final 16 hours of incubation (1µCi/well). All proliferation assays were performed in triplicate wells.

Cells were harvested, and DNA was transferred from wells on to glass fibre filters (Wallac, Turkum Finland) using a Tomtec cell harvester. "Melt-on" scintillant (Wallac) was added to the filters, and incorporation of radiolabel into cells was quantified using a 1450 Microbeta liquid scintillant counter (Wallac). Results are expressed as counts per minute (cpm).

2.15.1 Autologous recall assay

Titrations of DC from each condition were incubated with autologous T cells with or without 500 U/ml (final concentration) purified protein derivative (PPD) of M. Tuberculosis (Evans Medical Limited, Leatherhead, UK). The cell cultures were incubated for 6 days and then pulsed with 1 μ Ci ³H thymidine (ICN Biomedical, High Wycombe, UK) for a further 16 hours.

2.15.2 Allogeneic mixed lymphocyte reaction

Titrations of DC from each condition were incubated with allogeneic T cells for 6 days and then pulsed with 1 μ Ci ³H thymidine (ICN Biomedical, High Wycombe, UK) for a further 16 hours.

2.15.3 Autologous concanavalin-A dependent assay

Titrations of DC from each condition were incubated with autologous T cells with or without 1.25 μ g/ml (final concentration) of concanavalinA (ConA). The cell cultures

were incubated for 2 days and then pulsed with 1 μ Ci ³H thymidine (ICN Biomedical, High Wycombe, UK) for a further 16 hours.

2.16 Detection of cell viability

DC were infected and cultured for 16h and 40h at 37° C. 10^{5} DC per treatment condition were cultured in triplicates in a 96 well plate with 20µl 5mg/ml 3-(4,5-dimethylthiazol-2-yl)-2,5-diphenyltetrazolium bromide (MTT) (Sigma) at 37° C for 4 hours. Following this, 100µl of 10% sodium dodecyl sulphate (SDS) (Sigma) was added and the plate incubated at 37° C/5% CO₂ overnight. Subsequently, the plate was read in a microplate reader at a wavelength of 570 nm with wavelength correction at 630 nm. Results are expressed as mean optical density.

Historically, this assay has been accepted as measuring the ability of the cell to reduce the MTT dye by mitochondrial enzymes (Mosmann, 1983). Recently this notion has been challenged by studies demonstrating that MTT reduction occurs in endosomes (Liu et al., 1997) and that increased exocytosis reduces MTT reduction (Liu and Schubert, 1997). However, endocytosis would still be reduced in dead cells, and succinate dehydrogenase inhibitors still prevent some MTT reduction (Abe and Saito, 1998). Therefore, despite the controversy surrounding its precise scientific basis, this assay is still a suitable assessment of cell viability.

2.17 Detection of apoptosis

To identify whether or not the cells had undergone apoptosis, DC were washed once and resuspended in HBSS to give 10° cells/ml. Aliquots were fixed by dropwise addition of an equal volume of 70% ethanol (stored at -20 °C) to give a final concentration of 35% ethanol. The cells were then left on ice for 30 min, washed twice, and treated with DNAse-free RNAse A (final concentration, 500 µg/ml) at 37°C for 10 min. Propidium iodide (PI) (Sigma) was added (final concentration, 50 µg/ml) at 37°C for 30 min and the cells were analysed immediately by flow cytometry.

2.18 Neutralisation of DC surface receptors

DC were incubated with an optimum final concentration of antibody or antisera for 30 min on ice before being infected with virus (section 2.6) or appropriately stimulated (section 2.9). The antibody concentrations used were extrapolated from saturating amount of antibody required for extracellular staining.

2.19 Analysis of intracellular signalling

2.19.1 Western blotting

DC were incubated in CM that contained 0.5% FCS v/v for at least 24 hours prior to treatment. Total cell extracts were prepared by resuspending the cells in 50µl reducing sample buffer (2% SDS, 10% glycerol (BDH), 2% β-mercaptoethanol (Gibco), 60mM Tris-HCl (pH 6.8; Calbiochem CA, USA), and bromophenol blue) and stored at -20°C. Before loading, lysates were sonicated and boiled for 5 minutes, and then resolved by running 25µl on a SDS/12.5% PAGE gel and transferred to ECL nitrocellulose membrane (Amersham Pharmacia Biotech, Bucks, UK). Immunoblotting was performed by standard procedures. HRP-conjugated rabbit anti-mouse IgG or swine anti-rabbit IgG detection antibodies (both Dako) were used in combination with ECL detection reagent according to manufacturer's guidelines (Amersham Pharmacia Biotech). Densitometry analysis of the blots was performed using GeneSnap software (Syngene). The ratios of the densities of the phospho-p38 and total-p38 were calculated to give an indication of the proportion of p38 phosphorylated in each condition.

Where necessary, membranes were stripped of bound antibodies by incubating at 65°C for 1 hour in stripping buffer (see section 2.22). The membranes were then washed 5 times in TTBS before immunoblotting was carried out as above.

2.19.2 SB203580 inhibitor studies

In some studies, the role of p38 MAPK in the maturation of DC was determined. This was carried out using the p38 MAPK inhibitor, SB203580 (Sigma) (Tong et al., 1997).

83

Prior to FIX-HSV, IFN α or LPS stimulation, 3 μ M of SB203580 was added to half the DC culture on day 6 and the cells cultured for a further 24 hours. On day 7, DC from the two groups were removed from the culture dish with no additional wash steps, spun down and resuspended in fresh CM containing GM-CSF and IL-4 in the presence or absence of 3 μ M SB203580. This approach permitted counting of the two groups, without changing the concentration of SB203580 inside DC significantly. There was no difference in DC yield or cell viability in the presence of SB203580 (data not shown). The DC were subsequently plated in a 24 well plate at 5x10⁵ DC/ml in the appropriate concentration of inhibitor and stimulated with FIX-HSV (equivalent MOI of 3), IFN α or LPS, and cultured for a further 16 hours prior to assessment of changes in surface phenotype. In this assay, the virus was added directly to DC plated out, and not washed off after 1 hour as for normal protocol. This was again to avoid affecting the extracellular concentration of SB203580.

2.20 Statistical analysis

Where appropriate, the means of paired groups were analysed by a 2-tailed Student's t test, using Microsoft Excel software.





2.21 List of antibodies

All antibodies are mouse monoclonal unless otherwise stated.

(AF) = Ascites fluid

(P) = Purified

(SN) = Supernatant

Table 2.1	Non-conjugated	antibodies to	human	antigens
	rion vonjugutou		mannan	anngens

Specificity	Clone	Isotype	Source
CD1a	NA1/34	IgG2a (SN)	Gift from Prof. A. McMichael,
			John Radcliffe Hospital, Oxford,
			UK
CD2	RPA-2.10	IgG2b (P)	eBioscience
CD3	UCHT1	IgG1 (SN)	Gift from Prof P.C.L. Beverley,
			the Edward Jenner Institute for
			Vaccine Research, Newbury,
			UK .
CD11a	ICRF38	IgG1 (P)	Gift from Prof. N. Hogg, Cancer
			Research UK, London, UK
CD11b	ICRF44	IgG1 (P)	Gift from Prof. N. Hogg
CD11c	ICRF319	IgG1 (P)	Gift from Prof. N. Hogg
CD14	HB246	IgG2b (SN)	Gift from Prof P.C.L. Beverley
CD18	B1/18	IgG1 (P)	Gift from Prof. N. Hogg
CD19	BU12	IgG1 (SN)	Gift from D. Hardie,
			Birmingham University,
			Birmingham, UK
CD29	P5D2	IgG1 (P)	Gift from Prof. N. Hogg
CD49d	7.2R	IgG1 (P)	Gift from Prof. N. Hogg
CD86	BU63	IgG1 (SN)	Gift from D. Hardie
HLA-ABC	W6/32	IgG2a (P)	Serotec UK, Oxford, UK

HLA-DQ	Ia3	IgG2a (SN)	Gift from Prof R. Winchester, New York University School of
			Medicine, NY, USA
HLA-D'R	L243	IgG2a (SN)	Gift from Prof P.C.L. Beverley
Cathepsin E	CE1.1	IgM (SN)	Prof B. Chain
IFNAR2	MMHAR-2	IgG2a (P)	EMD Biosciences, San Diego,
			USA
IFNα	MMHA-11	IgG1 (P)	IDS Ltd., Boldon, UK
(recognises largest			
number of IFNa			
	TI 2 1	InC2a (P)	eRioscience
	11.2.1	IgOza (I)	Chioselence
HVEM	122	IgG1 (P)	Labvision, Fremont, USA
HVEM	-	Rabbit	Gift from P. Spear,
		antiserum	Northwestern University,
			Pittsburgh, USA
Nectin-1	CK41	IgG1 (P)	Gift from Dr. C Krummenacher,
			University of Philadelphia,
			Philadelphia, USA

Table 2.2 Conjugated antibodies to human antigens

Specificity	Clone	Isotype	Source
CD3-APC	UCHT1	IgG1 (P)	BD Pharmingen, Oxford, UK
CD13-FITC	WM15	IgG1 (P)	DPC Biermann, Bad Nauheim,
			Germany
CD86-PE	HA5.2B7	IgG2b (P)	Beckman Coulter, High
			Wycombe, UK
HLA-ABC-PE	W6/32	IgG2a (P)	eBioscience
HLA-DR-PE	B8.12.2	IgG2b (P)	Beckman Coulter, High
			Wycombe, UK

Specificity	Clone	Isotype	Source
VP16	LP1	IgG1 (AF)	Gift from Prof T Minson,
,			Cambridge University,
			Cambridge, UK
gD	LP2	IgG2a (SN)	Gift from Prof T Minson
(neutralising)			
gD	AP7	IgG2a (SN)	Gift from Prof T Minson
(non-neutralising)			

Table 2.3 Antibodies to viral antigens

Table 2.4 Control antibodies

-

-

Specificity	Clone	Isotype	Source
Aspergillus niger	DAK-GO1	IgG1 (P)	DakoCytomation, Ely, UK
glucose oxidase			
	679.1Mc7	IgG1 (P)	Beckman Coulter, High
	(PE conjugated)		Wycombe, UK.
Mouse MHC	TIB92	IgG2a (P)	ATCC, Manassas, USA
class I			
	HB38	IgM (SN)	Prof B.Chain
_	-	Pre-immune	Gift from P. Spear, Northwestern
		rabbit serum	University, Pittsburgh, USA

Specificity	Product no.	Isotype	Source
Phospho-p38	9211	Rabbit	Cell Signalling Technology,
МАРК		polyclonal (P)	Beverly, MA, USA
Total-p38 MAPK	9212	Rabbit	Cell Signalling Technology,
		polyclonal (P)	Beverly, MA, USA
IkBα	Sc-371	Rabbit	Santa Cruz Biotechnology, Santa
		polyclonal (P)	Cruz, CA, USA

Table 2.5 Antibodies to western blot targets

2.22 List of buffers

.

Buffer	Components		
TBS	8g NaCl, 0.2 g KCl, 3g Tris-Base in 1L of distilled water -		
120	adjusted to pH 7.5		
TTBS	TBS supplemented with 0.1% Tween 20		
Running buffer 14.4g Glycine, 3g Tris-base in 1L of distilled wate			
Transfer buffer 800 ml running buffer, 200 ml methanol (Analar, I			
Blocking buffer for	2.5g milk powder (Tesco, Cheshunt, Hertfordshire, UK) in		
immunoblotting	50 ml TTBS		
Antibody staining	0.1.g milk powder (Tesco) in 10 ml TTBS		
buffer			
Strinning huffer	40 ml 10% SDS solution, 12.5 ml 1M Tris pH 6.7, 1.45ml		
Sark Lang Carro	2-ME		

Characterisation of HSV-1 infection of DC

Chapter 3

Characterisation of HSV-1 infection of DC

3.1 Introduction

As outlined in chapter 1, several lines of evidence allude to the important role of DC in the pathogenesis of HSV infections. As a result, the interaction between myeloid DC and HSV-1, was investigated in this thesis. This chapter will establish the models that will be used to carry out the functional studies later in the thesis.

It is possible to isolate myeloid DC directly from peripheral blood using commercially available bead systems that involve both positive and negative antibody selection criteria. However, the yield of DC is relatively poor, as all DC subtypes in blood make up between 0.1-2% of all PBMC, depending on the criteria used (Vuckovic et al., 2004). In order to study the DC-HSV interaction in sufficient detail, larger yields of DC are needed. Therefore, in this thesis a well-characterised DC model system was used, based on the differentiation of DC from monocyte precursors in the presence of GM-CSF and IL-4 for 7 days (Sallusto and Lanzavecchia, 1994). In the last few years, this protocol has been modified to ensure purity of the population, eliminating contamination with other non-DC PBMC (Alderman et al., 2002).

There is evidence that this in vitro DC model is relevant to the in vivo scenario. Tissue resident myeloid DC, which are the most relevant DC in models of a peripheral herpetic lesion, have been shown to be differentiated from monocytes in models of transendothelial migration (Randolph et al., 1998). Furthermore, the function of monocyte-derived DC (MDDC) is comparable to that of myeloid DC isolated directly from blood, in terms of the ability to stimulate T cell proliferation (Leon et al., 2004). Therefore, MDDC were used as the DC population in this study to assess the functional consequences of HSV-1 infection of DC.

The HSV-1 strain available for these studies was 17+, which has been wellcharacterised and used extensively by collaborating laboratories. Furthermore, the virus used has been engineered to contain an expression cassette, consisting of a CMV promoter driving expression of green fluorescent protein (GFP), inserted into the UL43 gene, as described previously (Coffin et al., 1996). UL43 is a late viral gene of unclear function that may play a role in the assembly of viral capsids in the nucleus (Ward et al., 1996; Carter et al., 1996). HSV-1 lacking this gene show the same growth characteristics both in vivo and in vitro as the wild-type virus (MacLean et al., 1991). No role has been attributed to UL43 with regards to the in vivo pathogenesis of HSV-1, but the possibility remains that its deletion may affect the functional consequences of DC infection in unexpected ways. Nevertheless, the advantages of utilising GFP to track and gate on infected cells outweighed these known drawbacks, and this virus was used for all the studies.

The objectives of this chapter are:

- To define the purity of the DC population as well as its "normal" morphology and phenotype, in order to determine the parameters that HSV-1 infection will affect in chapters 4, 5 and 6.
- To assess the susceptibility of DC to HSV-1 infection and permissiveness for viral replication.
- To evaluate the suitability of using GFP transgene expression as an accurate quantification of DC infection efficiency.

3.2 Results

3.2.1 DC morphology

As detailed, in chapter 1, one of the defining features of DC is their morphology (Steinman and Cohn, 1973). After 7-day differentiation from monocytes, DC were non-adherent to plastic and assumed a "veiled" appearance, and possessed a characteristic ruffled appearance, with many fine membrane protrusions (fig. 3.1A), as previously described in vivo (Lens et al., 1983). This was confirmed by EM (fig. 3.1B).

Despite their non-adherence to plastic, immature DC can adhere to a FN substratum via β 1 integrins (Swetman, C., submitted for publication) and assume an elongated dendritic shape (fig 3.2A). In contrast to the fine processes seen on DC in suspension, the number of processes seen on FN-adherent DC were low (on average 2 or 3 per cell) but they extended to several times the length of the cell body (fig 3.2), as described recently (Swetman et al., 2002). Two hours after seeding on FN, DC adherence to FN reached a state of equilibrium, with a significant proportion of cells adhering to and spreading on the substratum at any one time.

In order to be able to gauge whether HSV-1 infection would have a maturation-like effect on DC morphology, it was important to characterise the DC changes in response to a maturation stimulus. This was modelled by exposing DC to 100 ng/ml LPS. Although mature DC adhere poorly to fibronectin (Brand et al., 1998; Shutt et al., 2000; Puig-Kroger et al., 2000; Swetman et al., 2002), addition of LPS to DC cultures already adherent to FN had the opposite effect. 8 hours after the addition of LPS, the proportion of DC adherent to the FN increased dramatically (fig. 3.2B; quantification in fig. 4.5). There was also an increase in dendrite length of the adherent DC (fig 3.2B). This phenomenon of increased DC adherence and dendrite length was not unique to LPS, as Poly(I:C) stimulation of DC induced similar changes (data not shown). Therefore, from these experiments, one can conclude that the most dramatic consequence of exposing FN-adherent DC to maturation stimuli was the increase in the

proportion of DC adherent to a FN substratum and the length of the dendrites extended.

3.2.2 DC surface phenotype

3.2.2.1 Purity of DC population

To characterise the DC further, the surface expression of several molecules was assessed. Despite many years of study on DC biology, a single surface molecule that defines these cells conclusively, in a manner analogous, for example, to CD3 for T ceils or CD19 for B cells, has not yet been identified. This ambiguity has been further exacerbated by different in vitro DC model systems described and by the increasing number of DC subsets that have been proposed (Shortman and Liu, 2002).

Nevertheless, DC can still be characterised by the surface expression of a combination of surface molecules. Immature MDDC express CD1a at high levels (Pickl et al., 1996), a non-classical MHC molecule involved in non-protein antigen presentation (Porcelli et al., 1998). In conjunction, MDDC express CD14, the LPS-LPS binding protein receptor, at low levels on the cell surface (Sallusto and Lanzavecchia, 1994; Pickl et al., 1996). The combination of high CD1a and low CD14 expression in the MDDC cultures in these studies confirmed the purity of this cell population (fig. 3.3A) and excluded the presence of macrophages, that possess a similar FSC/SSC profile on the flow cytometer and can also be derived from monocytes in vitro under modified cytokine conditions (Chapuis et al., 1997).

To confirm the absence of contamination from other non-DC PBMC, DC cultures were stained for the expression of CD3, CD14 and CD19 (T cells, monocytes and B cells respectively). The expression of these molecules was analysed without a FSC/SSC gate, so as not to exclude any smaller contaminating cells. In comparison to the percentage of these cell types in untreated PBMC, the DC cultures were devoid of CD3+ T cells, CD19+ B cells, and CD14^{hi} monocytes (fig. 3.3B). T and B cells were removed by immunodepletion, whereas the monocytes had either differentiated into

DC or died and therefore been removed on the ficoll gradient during the second round of DC purification on day 4.

In the light of the recent literature describing non-myeloid DC subsets (Shortman and Liu, 2002), it was important to exclude the contamination with these cells, namely PDC, from peripheral blood. The presence of these cells was unlikely, as their survival in culture is abolished by IL-4 (ubiquitous in MDDC cultures) (Rissoan et al., 1999) and their low expression of CD2 (Cella et al., 1999a), would have resulted in their depletion in the DC purification protocol used. Nevertheless, the myeloid derivation of the entire DC population was confirmed by the expression of CD13 and CD11c (fig. 3.3A). CD13 is an aminopeptidase which is expressed exclusively on cells of myeloid origin and is absent on PDC (Cella et al., 1999a). Furthermore, CD11c is the αX chain of β 2 integrins, that has been used as a specific marker of DC in vivo. Although it is expressed at lower levels on murine PDC, it is absent on human PDC (Grouard et al., 1997; Asselin-Paturel et al., 2001). Flow cytometry revealed that >99% of MDDC expressed both CD13 and CD11c. Furthermore, CD11b, the α M chain of β 2 integrin, absent on both mouse and human PDC (Grouard et al., 1997; Asselin-Paturel et al., 2001), was expressed on >99% of MDDC in these cultures (fig. 4.6). These data confirmed the homogenous nature of the MDDC population used for the subsequent studies in this thesis, which was devoid of contaminating non-DC and PDC.

3.2.2.2 DC maturation state

Although the in vitro derivation of DC from monocytes has been described to generate cells with an immature phenotype, this resting state can easily be disturbed by a cytokine imbalance (Alderman, C., PhD thesis), contamination with DC maturation stimuli (Xie et al., 2003; Rotta et al., 2003), or even excess mechanical agitation in the preparation protocol. As the immature to mature transition of DC is regarded as a terminal differentiation, the loss of an immature phenotype could result in insensitivity to further DC maturation. This was assessed by the addition of 100 ng/ml LPS to day 7 immature DC for a further 16 hours. The resultant changes in surface expression of markers of DC maturation were determined by flow cytometry (fig. 3.3C). In agreement with many other studies, MHC class I and II, as well as the co-stimulatory

molecule CD86 were upregulated in response to LPS (fig. 3.3C), confirming the immature state of DC used in this study.

- 7





Immature DC were fixed in 3% gluteraldehyde, prepared as described in materials & methods section 2.11, and visualised by confocal microscopy (A) or electron microscopy (B). Representative fields of two independent experiments.



Figure 3.2 DC morphology on a FN substratum.

(A) DC were allowed to adhere to FN-coated glass coverslips for 2 hours and morphology assessed 8 hours later. (B) 100 ng/ml LPS added to DC adherent to FN and morphology assessed 8 hours later. Representative of at least 3 independent experiments. Scale bar represents $50\mu m$.



Figure 3.3 Phenotype and purity of the DC population.

(A) The expression of DC and myeloid markers by the MDDC was assessed by flow cytometry. (B) Day 7 DC cultures were stained for CD3, CD19 and CD14, to detect contaminating T cells, B cells and monocytes. Data were analysed ungated to avoid excluding smaller contaminating cells. (C) 100 ng/ml LPS was added to immature DC and changes in the expression of MHC class I, HLA-DR and CD86 were assessed 16 hours later. Representative of at least 3 independent experiments.

3.2.3 DC susceptibility to HSV-1 infection

3.2.3.1 HSV-1 entry receptor expression

The susceptibility to HSV-1 infection is determined by the expression of receptors required for viral entry (Spear et al., 2000). The two most critical for the entry of HSV-1 strain 17+ into cells are HVEM and nectin-1 (Krummenacher et al., 2004). Using mouse mAb, >75% of DC expressed these receptors (fig. 3.4). The expression in this homogenous population of cells is likely to be ubiquitous, and the <100% expression is most likely to reflect non-specific background staining by the isotype control mAb, rather than DC that did not express any nectin-1 or HVEM. This phenomenon was demonstrated clearly when anti-HVEM rabbit sera was used to stain DC and only 12% of DC appeared positive for the expression of this molecule, relative to the staining of pre-immune rabbit sera control (fig. 3.4). These staining data were in agreement with the expression observed in a previous study (Salio et al., 1999), predicting that DC would be susceptible to HSV-1 infection.

3.2.3.2 Quantifying HSV-1 infection of DC

After viral exposure, infected DC could be detected by the intracellular expression of GFP. This fluorescence could be detected not only by fluorescence microscopy (fig. 3.5A), but also by flow cytometry for quantification (fig. 3.5B). Infection of DC at a low MOI range (0.3 to 3) resulted in high infection efficiency, increasing in a dose dependent manner (31% GFP+ve DC at MOI of 0.3, compared with 78% GFP+ve DC at MOI of 3). However, it was important to ensure that the GFP expressed in DC was a direct consequence of viral gene expression and was, therefore, an accurate assessment of infection in DC. Despite the small volumes of concentrated virus used to infect DC, it was possible that the GFP detected was derived from the endocytosis of soluble GFP in the viral preparation, as opposed to the consequence of viral gene expression. This possibility was excluded in two ways.

Firstly, the kinetics of GFP expression was assessed over a 24-hour time course (fig. 3.6A). Endocytic uptake of GFP would have predicted a rapid increase of GFP to levels near those seen after overnight infection (Neil, S., unpublished observation), in

manner analogous to rapid macropinocytosis in vitro (Sallusto et al., 1995). However, in contrast, GFP was not detected until 4 hours post-infection and steadily increased until 16 hours post-infection, after which GFP expression plateaued (fig 3.6A+B). The kinetics of this slow increase in GFP expression were consistent with the immediate early CMV promoter driving the GFP transgene and subsequent protein accumulation. The clear distinction between the GFP-ve and GFP+ve population further excluded a pan-uptake of soluble GFP by all DC.

Secondly, UV inactivation of HSV-1 was used to block GFP transgene expression. UV light is able to inhibit DNA expression by generating dimers between adjacent pyrimidines, with relative sparing of protein structure (Tornaletti and Pfeifer, 1996). Consistent with the kinetic data above, GFP expression was almost totally abrogated after exposure to UV light for >5 min, excluding the possibility that the GFP inside DC derived from exogenous protein in the viral preparation (fig. 3.6C).

Although GFP expression was confirmed as a good index of viral infection in the cells expressing this transgene, it was also important to exclude the presence of infected DC not expressing GFP. The transcription of the CMV IE promoter driving the GFP gene is dependent on host cell factors, such as NF-kB activation and cAMP concentration (Sambucetti et al., 1989; Stamminger et al., 1990), although this was likely to be less of an issue in the homogenous, non-dividing DC population used in this study. Nevertheless, infected DC were counter-stained for two late viral proteins to confirm that GFP accumulation correlated with HSV-1 protein production.

VP16 is a viral tegument protein that functions as a transactivator of HSV-1 immediate early gene expression (Wysocka and Herr, 2003). As it is also an important structural component of the virus, it is incorporated into the viral progeny inside cells and can be detected intracellularly (fig. 3.7B). Glycoprotein D is an integral envelope component important for viral entry (Spear et al., 2000). As a result, infected cells express this protein during viral replication and it can be detected extracellularly (fig. 3.7C).

As the target molecules in these experiments ought to have been co-expressed only on GFP expressing DC and were not expressed in uninfected DC, positive staining

controls were needed for uninfected DC to establish the correct compensation settings on the flow cytometer. For intracellular staining, cathepsin E was used, as DC contain large quantities of this enzyme that can be detected by flow cytometry (Medd, P., PhD thesis) (fig. 3.7A). The data clearly demonstrate that all DC expressing GFP expressed both VP16 and gD (fig. 3.7B+C). An important observation was the absence of the two single positive populations, confirming that GFP was indeed an accurate marker of infection efficiency, with all cells infected expressing late HSV proteins.



Figure 3.4 DC expression of HSV-1 entry receptors.

DC were stained for the presence of HVEM using a mouse monoclonal antibody *(lower left panel)* or rabbit anti-HVEM antiserum *(upper left panel)*, and for nectin-1 *(right panel)*. Gates and percentages relate to <2% staining of isotype control or non-immune sera. Representative of three independent experiments.



Figure 3.5 HSV-1 infection efficiency of DC.

DC were infected with HSV-1 at MOI of 1 in (A) or at the specified MOI in (B), and cultured for a further 16 hours. The expression of GFP was assessed by light microscopy (A) or flow cytometry (B). Scale bar represents $50\mu m$. Representative of at least 3 independent experiments.



Figure 3.6 Quantification and mechanism of GFP expression.

(A) DC were infected at MOI of 1 for various times and expression of GFP assessed by flow cytometry. This was quantified in (B) both for MFI of GFP expression of GFP+ve cells, and percentage of DC expressing GFP at appropriate time points. (C) HSV-1 was exposed to UV light for various times, prior to infection of DC. DC were cultured for a further 16 hours and GFP expression assessed by flow cytometry. Representative of at least 3 independent experiments.


Figure 3.7 Assessment of GFP expression as marker of DC infection. (A) DC were stained for the intracellular expression of cathepsin E *(right panel)* and the relevant isotype control *(left panel)*. (B+C) DC were infected with HSV-1 at MOI of 1 and stained for intracellular VP16 (B) and extracellular gD (C). Representative of at least 3 independent experiments.

3.2.4 HSV-1 replication in DC

The ability to express viral genes is not always followed by the production of infectious particles in all cell types. Therefore, it was important to relate the susceptibility to infection observed in section 3.2.3, to the replication competence of HSV-1 in DC. Furthermore, production of significant quantities of infectious particles may result in secondary infection of DC in long term cultures.

The production of infectious virus particles was measured by viral infectivity assay, as described in section 2.5.2. HSV-1 has evolved its fusogenic properties to allow intracellular replication and intracytosplasmic infection of neighbouring cells, excluding the need for the extracellular route of infection (Spear et al., 2000). As a result, both virus in the supernatant and intracellular infectious virus (by harvesting cell lysates) were assayed for 4 days following infection. HSV-1 is tropic for neuronal and epithelial cells, but it is in the latter that the greatest productive infection is seen, as evidenced by oral and mucosal herpetic lesions. Therefore, the ability of HSV-1 to replicate in an epithelial cell line (HeLa) was assessed in parallel to gauge the efficiency of replication in DC relative to a cell that permits efficient viral replication.

To ensure a comparable assessment of replication efficiency, both DC and HeLa were infected with the same viral load (MOI of 1), which resulted in similar infection efficiency, as detected by confocal microscopy and flow cytometry (fig. 3.8A vs. fig. 3.5). The data showed that virus production was different in the two cell types (fig. 3.8B). After the first day of infection, there was little difference between virus secreted from DC and HeLa cells. However, from 2 days post infection, there was significantly more virus in the supernatant of HeLa cells than DC, with nearly a 3 log difference after 4 days. This suggested strongly that HSV-1 could replicate more efficiently in HeLa than DC. The data for the production of intracellular virus reflected similar differences between HeLa cells and DC. Noticeable was that the disparity between the two cell types occurred one day earlier for intracellular virus as extracellular virus, as expected. Therefore, from fig. 3.8, it was concluded that expression of late viral genes seen in fig. 3.7 was indeed indicative of a productive viral replication in DC but that this was less efficient than that seen in epithelial cells.





(A) HeLa cells were infected with HSV-1 at MOI of 1 and GFP expression was assessed by confocal microscopy (upper panel) and flow cytometry (lower panel). Scale bar represents $50\mu m$. (B) DC and HeLa cells were infected with HSV-1 at MOI of 1 and supernatant and cell lysates were collected for 4 days following infection. Infectious virus was quantified by viral infectivity assay. Representative of 3 independent experiments.

3.3 Discussion

This chapter has set out to establish important setpoint conditions from which the effects of HSV on DC can be gauged in the later chapters. These include:

- 1. Detailing the "normal" DC parameters in terms of morphology and phenotype. Deviation from these descriptions will be interpreted as virally induced changes, which could occur either directly or indirectly as a result of the infection.
- 2. Confirming that DC are susceptible to HSV-1 infection and that the GFP transgene expressed by the infecting virus is an accurate assessment of viral infectivity. This will be useful for the delineation between changes in infected and uninfected cells in later analyses.
- 3. Confirming that HSV-1 can replicate in DC, albeit at low levels. Consequently, HSV-1 can execute its full gene expression profile in DC, and therefore candidate viral proteins that affect DC function could be predicted from this. Secondly, the low viral production means that only a small number of DC will be significantly infected by viral progeny emanating from DC and therefore this will not be a major obstacle in interpreting the effects on DC of the initial HSV-1 infection.

3.3.1 DC susceptibility to viral infection

3.3.1.1 Receptor expression

As detailed in chapter 1, HSV-1 infection occurs in a stepwise process. Viral attachment via gB/gC to surface HS is followed by gD interaction with entry receptors, which then induces the envelope-cell membrane fusion process (Spear et al., 2000). As HS is ubiquitous on all cell types, it is the expression of the entry receptors that determines the susceptibility of cells to HSV-1 infection. HSV-1 utilises HVEM and nectin-1 as entry receptors (Montgomery et al., 1996; Geraghty et al., 1998), but not nectin-2 (Warner et al., 1998). There are however strain differences in receptor usage. Recent studies have confirmed that the 17+ strain used in these studies does indeed use HVEM and nectin-1, and not nectin-2, to any significant degree (Krummenacher et al., 2004).

Consistent with the receptor expression data, it was confirmed that HSV-1 does indeed infect DC. This was not a novel finding, as DC susceptibility has been described in studies carried out both prior to and in parallel with the data above (Coffin et al., 1998; Salio et al., 1999; Mikloska et al., 2001).

Herpesviruses use different cellular entry receptors, but it is interesting to note that many of them can infect DC with similar efficiency (MOI of 1 can infect around 50% of cells (fig. 3.5) (Moutaftsi et al., 2002; Abendroth et al., 2001)). This may reflect one aspect of herpesvirus infection that is well conserved: attachment to surface heparan sulphate prior to interaction with entry receptors (Compton et al., 1993; Zhu et al., 1995; Laquerre et al., 1998). HS binding may increase the proximity of the virus to the entry receptors and form the basis for the similar infection efficiency seen for viruses that exploit this entry strategy. The importance of HS interaction for HSV-1 infection is considered further in chapter 6. Studies detailing the relative role of the entry receptors, both in the infection of DC and any downstream signalling consequences, will be described in chapter 6.

3.3.1.2 Quantification of infection efficiency

The advantage of utilising a GFP expressing virus was that it allowed rapid assessment of infectivity. Furthermore, it permits the gating of infected cells by flow cytometry without additional steps to stain for viral proteins. However, it was important in this setting, to confirm that GFP expression was an accurate correlate of infection efficiency. This was done by staining infected DC for VP16, the viral IE gene transactivator, and gD (fig. 3.7). The data confirmed that GFP expression closely correlated with the expression of both these proteins in DC and that the transgene was therefore an accurate gauge of infection efficiency.

In these studies, the linear correlation between GFP and VP16/gD expression was noticeable. This was consistent with the notion that the expression of these proteins was dependent on the number of HSV virions infecting the cells, and thus the number of viral gene copies inside DC. Furthermore, the correlation between early (GFP) and

late (VP16/gD) protein accumulation confirmed that infected DC are able to express the full repertoire of viral genes, excluding that the infection is abortive at the early gene stage, as is the case in vaccinia virus (VV) infection (Engelmayer et al., 1999).

The data in fig. 3.7 also illustrated the important absence of certain populations. A GFP+ VP16- or GFP+ gD- population DC would have represented DC infected with HSV-1 but in the absence of significant viral gene expression despite transgene transcription. This would have suggested that only the CMV IE promoter was being expressed and that HSV-1 IE gene transcription was not initiated. Such an abortive infection has been described for HSV-1 infection of some cell types (Kemp et al., 1990; Bruun et al., 1998). Secondly, the absence of GFP- VP16+ or GFP- gD+ DC confirmed that GFP was expressed in all cells that underwent HSV-1 replication, ensuring that GFP-ve DC in culture were not infected, and therefore under intracellular control by the virus.

3.3.2 DC permissiveness to viral replication

3.3.2.1 In vitro replication of HSV-1 replication in DC

Measuring virus in the supernatants and the cell lysates of the cells, allowed the assessment of both the quantity of infectious viral particles and consequently the replicative capacity of HSV-1 in DC. The data demonstrated that although DC permit viral replication, this happens to a low degree, compared both to the production in HeLa cells and relative to the input of virus (fig. 3.8). Generally, a productive viral infection can be defined as one that yields more virus than the input amount in one round of replication. At a MOI of 1, $5x10^5$ pfu of HSV-1 were added to DC, but viral production never rose above 10^4 pfu (the experiments were carried out in 1 ml volumes). This figure is also important as it can be deduced that insufficient virus was present in the supernatant after 1 day of infection of DC at an MOI of 1, with approximately 40% of DC remaining uninfected (fig 3.5), there would be 10^4 pfu of virus to infect $0.2x10^6$ DC (60% of initial $0.5x10^6$ DC are already infected). This results in a net MOI of 0.05, which would infect <2.5% of DC (data not shown). The

infection efficiency is likely to be even lower as the infection would occur in the larger volume of a DC culture, rather than the concentrated conditions used for 1 hour infection in all experiments. This calculation is important, because it confirms that the population of GFP+ve cells observed after one day of infection, is derived almost exclusively from the inoculum of the initial infection, and not as a result of a secondary infection from new virions made following replication in DC.

3.3.2.2 In vivo implications of HSV-1 replication in DC

The low permissiveness for viral replication in DC means that this cell type is unlikely to contribute significantly to the generation of a transmissible titre of the virus to infect a new host. However, the low viral production may still be relevant and significant in the DC's own microenvironment, as the LC and dermal DC network is surrounded by epithelial cells or fibroblasts. Perhaps more significant is the close anatomical relationship between DC and peripheral nerve endings in the skin, especially between LC and c-fibre nerve endings in the epidermis, that may provide a convenient entry point into the nervous system for LC-derived virus (Hosoi et al., 1993). The implications of this hypothesis are discussed further in the general discussion chapter 7.

3.3.2.3 Mechanisms and consequences of DC permissiveness to HSV-1 replication

The reasons behind the low replication permissiveness in DC relative to HeLa cells are unclear. As HSV-1 gene expression occurs in a cascade fashion of viral gene groups (Weir, 2001), a critical initiation step in replication is transcription of immediate early genes. The viral protein VP16 is important in this process, as it can form multi-protein transcriptional regulatory complex (the VP16 complex) with Oct-1 and HCF-1, two host cell factors, on the viral IE promoters. This process requires first the interaction with HCF-1, priming the dimer for interaction with Oct-1 transcription factor which targets 'TAATGARAT' sequences on the viral IE promoters (Wysocka and Herr, 2003). Therefore, viral gene expression and replication is dependent on the host cell environment.

One factor that may play a role is the presence of inhibitory transcription factors that can prevent Oct-1 binding to the target promoters. These include some Oct-2 isoforms that prevent IE gene expression, a strategy the virus may exploit to prevent initiation of a lytic cycle in neurones and consequently establish latency (Lillycrop et al., 1993). HeLa cells do not express Oct-2 transcription factors (Hagmann et al., 1995a), whereas immature DC only do so at low levels (Neumann et al., 2000). The presence of Oct-2 may quantitatively reduce VP16 complex binding to viral promoters in DC and result in lower gene expression and replication. However, while both monocytes and macrophages express high levels of Oct-2 (Neumann et al., 2003), HSV-1 replication is only observed in macrophages (Bruun et al., 1998), suggesting that other host factors may be more important in determining the expression of viral genes.

An important difference between DC and HeLa cells is that the latter are dividing cells. HCF-1 promotes proliferation (Wysocka and Herr, 2003) and may therefore be more abundant in HeLa cells. Indeed, in non-proliferating sensory neurons, HCF-1 is sequestered in the cytoplasm and only translocates VP16 to the nucleus upon induction of lytic infection (Kristie et al., 1999). However, recently an important role for the PKR pathway in determining the permissiveness of HSV-1 infected cells has also been described. Phosphorylation of PKR results in the phosphorylation of the translation initiation factor eIF-2 α , which reduces viral protein translation (Clemens and Elia, 1997). The HSV-1 gene ICP34.5 can prevent this process (Chou et al., 1995), but was not a factor in these experiments as the same virus was used for infecting both DC and HeLa cells (i.e. no ICP34.5 mutants were used). However, an important endogenous negative regulator of the phosphorylation of PKR is the Ras pathway, whereby Ras activation prevents the phosphorylation of PKR. The net result is that in cells where the Ras pathway is most active, viral replication is enhanced (Farassati et al., 2001). Therefore, the difference in HSV-1 replication may lie at the level of Ras and PKR activation.

Being a transformed, replicating cell line, HeLa cells are likely to express elevated levels of active Ras (Dobrowolski et al., 1994) and this may have played a major role in permitting greater viral replication in these cells by preventing PKR activity.

However, it is also possible that PKR activation in DC is quantitatively greater and the subsequent inhibition of viral protein synthesis more potent in DC than in HeLa cells. Unfortunately, few studies have compared replication of other viruses in DC relative to other cells directly, and therefore it is difficult to determine the relative role of these two pathways. This could only be carried out by measuring Ras activity and PKR phosphorylation in parallel after infection of cells. However, the observation that maturation of DC renders the cells less susceptible to viral induced changes in phenotype (Salio et al., 1999) supports the hypothesis that DC have an innate protection to viral replication. Therefore, if HSV-1 infection were to elicit DC maturation, the cells may be able to switch to a more efficient antiviral state than HeLa cells and intrinsically hinder viral replication more effectively.

3.3.3 Role of replication in DC in viral pathogenesis

With regards to the general permissiveness of DC to viral infections, it should be noted that measuring the growth of virus in individual cells might mask the true events that occur in vivo. Although replication of MV and HIV in DC cultures alone is low or negligible, upon interaction with T cells (e.g. CD40L), this can be elevated to extremely high levels (Fugier-Vivier et al., 1997; Granelli-Piperno et al., 1999; MacDougall et al., 2002). This "inducible" form of viral replication demonstrates how some viruses may have evolved to exploit DC to serve as a reservoir of large virus production at appropriate time points in the virus lifecycle and pathogenesis, so as to produce a large progeny in close proximity to its target cells (e.g. CD4+ T cells for HIV).

The consequences of viral infection on DC's ability to initiate T cell activation will be discussed at greater length in the next chapter. Rapid viral infection, replication and lysis of DC is one potential strategy to disrupt the APC function of these cells. However, replication in DC does not correlate with the effects on DC induction of T cell proliferation. For example, HSV-1 and VV impair DC equally well despite the latter being an abortive infection (figs. 3.8 + 4.1) (Salio et al., 1999; Engelmayer et al., 1999). DC infected with strains of HCMV that differ in their ability to replicate, still result in equal inhibition of T cell stimulation (Beck et al., 2003). Furthermore, the

time necessary for viral replication varies. For example, HCMV requires up to four days to undergo a full replicative cycle, by which time the infected DC would have encountered responding T cells (Riegler et al., 2000). Therefore, viral replication per se is unlikely to be a predominant strategy to disable DC function, but rather reflects an in-built programme of the viral lifecycle in human cells. It is more likely that other aspects of DC physiology are specifically targeted by viral immunomodulatory genes, and these are discussed in the following chapter.

It is also worth mentioning that a critical factor in the study of viral egress from DC may relate more to the timing and context rather than the quantity of virus production. In recent years, the ability of DC to "capture" viruses by binding it to the c-type lectin, DC-SIGN, has generated much interest, with HIV-1 being the prototypic example (Geijtenbeek et al., 2000a). The pathogenesis model proposes that following viral interaction with the DC in the periphery, the cell "carries" the virus to the lymph node, where it can infect T cells in trans after interaction between the two cell types. Therefore, although this scenario proposes that no viral replication has taken place, the DC has facilitated the infection of a neighbouring target cell, albeit with a low virus inoculum, demonstrating an extra layer of complexity in the in vivo relevance of viral egress from infected DC.

3.4 Conclusions

This chapter has characterised the DC that will be used for the remainder of the studies described in this thesis. Determining the morphology and phenotype of DC has established the "normal" conditions which virus infection might offset. Data in this chapter have also characterised the HSV-1 strain that will be used for future study. Notably, it was concluded that GFP is an efficient method to quantify infection efficiency in vitro and will aid in discerning the effects of uninfected and infected DC. The degree of HSV-1 replication in DC was also assessed, and it excluded that DC infection results in the production of large quantities of infectious virus, which should not hinder the interpretation of assays used in the remainder of the thesis. The rest of the work presented will now build on these findings, by analysing in greater detail the impact of HSV-1 infection on the ability of DC to function as an APC, focusing on

Characterisation of HSV-1 infection of DC

viral mechanisms that affect the cell and the host's response to detect and counter the effects of the infection.

1

HSV-1 mediated impairment of DC function

Chapter 4

HSV-1 mediated impairment of DC function

4.1 Introduction

Chapter 1 detailed how DC play an important role in the generation of antiviral T cells to HSV infection. It would be in the virus' interest to prevent DC from stimulating resolving T cell responses. The ubiquitous location of immature DC at the sites of potential viral entry, including the blood, provide a hypothetical motive for the virus to disrupt the ability of DC to function as APC. Subsequent hindrance to the onset of immunity may delay clearance of viral infection, allowing the virus to establish itself, replicate and reach transmissible titres in target organs. Some viruses could exploit this scenario to gain time to establish themselves in a latent state.

However, DC activation and the resultant enhancement of the immune response may promote and even exacerbate immunopathology, and this may generate outlets for viral transmission. Therefore, the viral-DC interaction can have important functional outcomes in the context of the natural history of the disease.

The most important function of DC that can be affected by HSV-1 is their ability to stimulate naïve T cell proliferation. Disruption of many aspects of DC physiology could impinge on the outcome of this interaction between DC and T cells. Changes on DC morphology would affect both the ability to migrate to secondary lymphoid organs (Swetman et al., 2002) and to present antigen to responding T cells (Kobayashi et al., 2001; Benvenuti et al., 2004a). Interference with the DC maturation process, including secretion of relevant cytokines, would prevent the generation of potent mature APC, possibly skewing towards a tolerogenic immune response (Hawiger et al., 2001). Finally, by inducing DC death, the virus would reduce the time of interaction between the DC and the responding T cell, preventing necessary activation for generating a potent multi-epitope anti-viral T cell response.

In this chapter, the aim is to determine how HSV-1 influences the DC ability to stimulate T cells and to define which DC functions are affected that could account for these changes. The focus will be on the effect on DC morphology, cytokine secretion, cell viability and surface phenotype.

4.2 Results

4.2.1 T cell stimulation by HSV-1 infected DC

The ultimate function of a DC is to stimulate T cell proliferation. Therefore, this was the first DC property measured, and evaluated to see how it was affected by HSV-1 infection.

4.2.1.1 HSV recall responses

Resolution of HSV infections requires a cellular immune response. This presence of HSV-specific T cells correlates with serostatus in most cases (Posavad et al., 2003). Therefore, it should be possible to measure memory responses in those individuals who had been previously infected with HSV-1. Indeed, HSV-1 infected DC stimulated autologous T cell proliferation, in the absence of any other antigen, in a proportion of individuals tested (fig. 4.1). In all, 10 individuals were tested, both for the presence of an HSV-specific T cell proliferative response, and for the presence of neutralising antibodies to HSV in serum (6 individuals shown in fig. 4.1). All the 5 responders were seropositive, while all 5 non-responders were seronegative. There were no discrepancies between responder status and seropositivity. This close relationship between serostatus to HSV and T cell proliferation supported the interpretation that the T cell response represented a memory response to the virus.

A noteworthy feature of the HSV recall responses was the reverse dose-dependent nature of the responses observed, with the greatest proliferation observed at the lowest MOI (i.e. in the conditions where the greatest number of uninfected DC were present). However, this response diminished with increasing numbers of DC infected (fig. 4.1).

4.2.1.2 PPD recall responses

All individuals tested had been vaccinated with BCG. The ability of infected DC to activate memory T cells was assessed by a recall antigen assay to PPD. As a proportion of individuals tested possessed HSV-1 recall responses (fig. 4.1), proliferation to PPD could only be analysed in those individuals that showed no such

responses (fig. 4.2A). This showed that the ability to elicit PPD responses was impaired in infected DC populations in a dose dependent manner, with the greatest inhibition observed at the highest MOI.

4.2.1.3 Allogeneic MLR proliferation

The capacity of HSV-1 infected DC to elicit primary T cell proliferation was tested in an allogeneic mixed lymphocyte reaction (MLR). Fig. 4.2B shows that HSV-1 infection resulted in an impaired ability to induce proliferation of naïve allogeneic T cells, with the greatest inhibition again seen at the highest infection rates. The data shown are a mean of six individuals, three with an HSV recall response and three without (see chapter 5). The ability to elicit an autologous HSV response did not correlate with the degree of inhibition observed in the allogeneic response.

4.2.1.4 ConA stimulated autologous proliferation

The impaired stimulation of naïve T cells in section 4.2.1.3 was assessed by 6 day assays. During this time, the viability of infected DC could have been affected (see section 4.2.4). Using a mitogen reduces the contribution of DC death in a shorter assay, excludes variability introduced due to differing degrees of allogeneic recognition, and also avoids HSV recall responses (chapter 5) and potential fratricide within the T cell population (Raftery et al., 1999).

The ability of HSV-1 infected DC to stimulate the proliferation of ConA-primed autologous naïve T cells in this assay was severely impaired. The degree of inhibition again correlated with the number of DC infected (fig 4.2C).

4.2.2 T cell infection by HSV-1

One possible explanation for the reduced T cell stimulating capacity was direct infection of T cells by virus emanating from infected DC in the coculture. T cell proliferation could be hindered by this infection. T cells could be unable to respond to DC stimuli, or susceptible to CTL lysis (Raftery et al., 1999). T cell infection by DC-

derived herpesvirus has been described (Abendroth et al., 2001), and it has been proposed that activated, not resting, T cells are permissive to HSV (Ito et al., 1997; Raftery et al., 1999). One of these previous studies used ConA stimulation to render the T cells susceptible to infection (Raftery et al., 1999). Therefore, T cells were pretreated with $1.25\mu g/ml$ ConA (the optimum concentration used in the proliferation assays in fig. 4.2C), prior to HSV-1 infection. Only (1.2%) of T cells expressed the GFP transgene, in comparison to 0.8% in the absence of ConA. This result questioned the efficiency of ConA to render T cells susceptible to HSV-1 infection, but also confirmed both that the CMV-driven GFP transgene could be expressed in T cells (Barry et al., 2000) and also that some T cells could be infected by HSV-1.

The number of infected T cells was also assessed in an allogeneic MLR. No significant numbers of CD3+ GFP+ cells could be detected 1, 3 or 5 days from the beginning of the co-culture (fig. 4.3). These data excluded the possibility that the inability for HSV-1 infected DC to stimulate T cell proliferation was secondary to trans infection of T cells.

Thus, based upon the results in sections 4.2.1 and 4.2.2, HSV-1 infection impairs DC ability to stimulate both naïve and memory T cell proliferation. The absence of T cell infection implies that the inhibitory effect was dependent predominantly on the viral effects on DC function. Hence, the remainder of this chapter will investigate the cellular processes affected in greater detail. The viral mechanisms responsible are analysed further in chapter 6.



Figure 4.1 Autologous T cell responses in the presence of HSV-1 infected DC. DC were infected with HSV-1 and cultured for 16 hours prior to addition of autologous T cells (10^5) and proliferation assessed by incorporation of ³H-thymidine. Results expressed as the mean \pm SEM for triplicate cultures from individual experiments. HSV serostatus of individuals indicated in box inserts.



Figure 4.2 Memory and primary T cell proliferation elicited by HSV-1 infected DC. (A) DC were infected and cultured for 16 hours prior to addition of autologous T cells (10⁵) in the presence of 500 U/ml PPD. Results expressed as mean \pm SEM (n = 7). (B) DC were infected and cultured for 16 hours prior to addition of allogeneic T cells (10⁵). Results expressed as mean \pm SEM (n = 6). (C) DC were infected and cultured for 16 hours prior to addition of allogeneic T cells (10⁵). Results expressed as mean \pm SEM (n = 6). (C) DC were infected and cultured for 16 hours prior to addition of autologous T cells (10⁵) in the presence of 1.25 µg/ml ConA. Results expressed as mean \pm SEM (n = 4). Proliferation was assessed by incorporation of ³H-thymidine. DC only proliferation 56 \pm 10 cpm (mean \pm SEM; n = 6). T cell only proliferation 57 \pm 11 (mean \pm SEM; n = 6). * = p<0.05 is the proliferation in presence of uninfected DC relative to all other conditions. Statistical analysis by Student's *t* test.



Figure 4.3 Infection of T cells by HSV-1.

(A) T cells were infected with HSV-1 (MOI = 1) and cultured for 16 hours. (B) DC were infected with HSV-1 (MOI = 1) and cultured for 16 hours prior to addition of allogeneic T cells (10^5). DC-T mixtures were co-cultured for the times indicated and GFP expression by CD3+ T cells assessed. Representative of three independent experiments.

4.2.3 Morphological consequences of HSV-1 infection of DC

An intact and flexible DC cytoskeleton is important in maintaining DC morphology during antigen sampling in the periphery, DC migration to secondary lymphoid organs and for the presentation of antigen to T cells. Disruption of this cytoskeleton, and therefore cell shape, could affect the ability to stimulate T cells (Kobayashi et al., 2001; Benvenuti et al., 2004a). Indeed, after HSV-1 infection, the most striking changes seen following infection were in the morphology of the DC.

4.2.3.1 Morphological changes of non-adherent DC

Following HSV-1 infection, the veiled DC appearance was lost and the cells assumed a more rounded morphology. Infected cells displayed a heterogeneous morphology. Many of the DC lost the membrane processes observed in chapter 3 (fig. 4.4A). Another feature was the large number of non-adherent homotypic clusters. GFP fluorescence revealed that these comprised both uninfected and infected DC (fig. 4.4B). Homotypic clustering is observed commonly in vitro in response to some maturation stimuli (e.g. LPS), but these are often of different character, being adherent and larger in size than those observed after HSV-1 infection. HSV-induced clusters were 10-20 cells or less (fig. 4.4B). Occasionally, multi-nucleate giant cells were visible, particularly within or near DC homotypic clusters. The multicellular origin of these structures was confirmed by EM, as determined by the presence of two or more nuclei in cells significantly larger than normal DC (fig. 4.4C)

4.2.3.2 Morphological changes of fibronectin-adherent DC

In the context of a cutaneous/mucosal infection in vivo, HSV-1 is unlikely to come into contact with non-adherent DC, but rather DC adhered and spread in their surrounding tissue microenvironment. Therefore, to model this scenario, DC were adhered to a FN substratum for 2 hours, prior to infection with HSV-1. After 1 hour of infection, any unbound virus was washed off and the morphology of the DC was analysed 8 hours later. Preliminary experiments showed that this was the earliest time that the fluorescent channel of the confocal microscope could detect sufficient GFP

fluorescence to distinguish between uninfected and infected DC. Even at this stage of the infection, significant morphological changes could be seen. GFP+ve DC rapidly lost their dendritic morphology, but remained adherent to FN (figs. 4.5A+B). At higher DC confluence, evidence of clustering could be seen, raising the possibility that infected 'DC could influence uninfected DC functionally through direct cell-cell contact (fig. 4.5A).

As shown in the previous chapter, one of the ways in which DC respond to maturation stimuli (such as LPS) is to increase adherence to FN (fig. 3.2). Therefore, the effect of HSV-1 infection on the morphological response to LPS was assessed. After infection, FN-adherent DC were cultured in the presence or absence of 100 ng/ml LPS. As seen in fig. 4.5B, uninfected DC responded by increasing the number of adherent DC with dendritic morphology. However, LPS did not increase the number of infected, GFP+ve, DC with a dendritic shape. In sharp contrast, GFP-ve DC responded similarly to uninfected DC cultures, consistent with them being uninfected, and therefore free from the inhibitory effects of the virus (fig. 4.5B). These data argue against a scenario in which viral infection resulted in the secretion of soluble factors that prevented uninfected DC in the culture responding to maturation stimuli.

4.2.3.3 Changes in integrin expression after HSV-1 infection

DC interact with and adhere to FN via β 1 integrins (Le Varlet et al., 1992; D'amico et al., 1998), and thus the inability of HSV-1 infected DC to spread on FN could have been attributed to change in expression of these integrins. Expression of both β 1 and β 2 integrins was assessed by flow cytometry after HSV-1 infection. The β 1 integrins (CD29 and CD49d) were downregulated after infection (GFP+ve gate), but β 2 integrin (CD11a, CD11b, CD11c and CD18) expression was not altered (fig. 4.6). Hence, the loss in dendritic morphology on FN after HSV-1 infection could be attributed at least in part to the selective downregulation of β 1 integrins on the surface of the infected DC.

Section 4.2.3 has shown that HSV-1 infection results in the loss of dendritic morphology, and prevents DC responding to exogenous maturation stimuli by altering

their shape. Selective downregulation of $\beta 1$ integrin expression may have contributed to this. Functionally, this may impair DC migration out of the periphery and may also play a role in the loss of T cell stimulatory capacity of HSV-1 infected DC.

1





Α









Figure 4.4 Morphology of HSV-1 infected DC in suspension.

DC were infected with HSV-1 (MOI = 1) and (A) fixed with 2% gluteraldehyde and viewed on slides, or (B) viewed in tissue culture suspension by confocal microscopy, revealing small DC-DC clusters compared to those after LPS stimulation. (C) Analysis also revealed occasional syncytia formation (*Left panel* – confocal microscopy; *right panel* – EM). Scale bar represents $50\mu m$. Representative fields of at least three independent experiments.



Figure 4.5 Morphology of FN-adherent DC infected with HSV-1.

DC were allowed to adhere to a FN substratum for 2 hours and then infected with HSV-1 (MOI = 0.3). Cell morphology was assessed 8 hours after infection by light microscopy (A) and quantified from a total of 4 random fields from 2 independent experiments (B). Results expressed as mean \pm SEM. Scale bar represents 50µm.



Figure 4.6 Integrin expression of DC infected with HSV-1.

DC were infected with HSV-1 (MOI=1) and then cultured for 16 hours. Integrin expression was determined by flow cytometry. HSV-1 infected cells gated on GFP+ve. Representative of at least two independent experiments.

4.2.4 DC viability following HSV-1 infection

Both the rapid changes in morphology described in section 4.2.3 and the dramatic functional impairment in section 4.2.1, might be a result of rapid cell death after infection. A crude assessment by flow cytometry, analysing the percentage of cells detected within the customary live gate of DC, showed that, 40 hours after infection, there was a significant fall in the number of live DC (fig. 4.7). However, this assay is not a direct measure of cell viability, but rather an assessment of change in shape and granularity of cells after fixation with formaldehyde. Infection may render the cells more susceptible to fixation artefacts, and therefore shape changes detectable on the flow cytometer may not necessarily be indicative of cell death. Therefore, several assays were used to determine DC viability directly.

Previous studies using trypan blue exclusion revealed no difference in DC viability 16 hours following infection, with only a slight reduction after 40 hours (Mikloska et al., 2001; Samady et al., 2003). This was confirmed qualitatively during cell counting prior to setting up the functional studies described in section 4.2.1 (data not shown). However, the assessment of viability was also undertaken by measuring MTT reduction as a measure of overall cell metabolic activity (see section 2.16). DC demonstrated a non-significant trend in reduction in viability 16 hours after HSV-1 infection. The degree of cell death was significantly increased after a further day of culture, such that MTT reduction by infected cultures was >50% lower than that of uninfected DC (fig. 4.8).

The MTT reduction assay did not give any information regarding the type of cell death occurring. However, reduction in cell size, in conjunction with loss of cell viability, suggested apoptosis. EM of HSV-1 infected DC found several cells that fulfilled apoptotic criteria, namely cell rounding (fig. 4.9) and nuclear chromatin condensation, both with intact (fig. 4.9A) and lost (fig. 4.9B) nuclear envelope. There was also evidence of apoptotic DC that, in the absence of phagocytic removal, underwent secondary necrosis, as evidenced by cells with apoptotic morphology that had lost cell membrane integrity (fig. 4.9C). Occasionally, it was possible to observe HSV-1

capsids in the nuclei of these cells, confirming apoptosis in DC with clear evidence of HSV-1 infection (fig. 4.9B+C).

For quantitative assessment of apoptosis in these cultures, flow cytometry was used. A defining criterion of apoptotic cell death is the degradation of nuclear DNA. DNA fragments elute out of permeabilised cells and the remaining DNA stain less brightly with the DNA intercalating dye PI (Rad et al., 2003). Permeabilising DC with ethanol and staining them with PI will therefore allow assessment of the DNA content of the cell. DC are non-replicating cells, so all live cells were in the G0 phase of the cell cycle and the apoptotic hypodiploid population are the "sub-G0" population. A sub-G0 population was observed at 12 hours of infection, accounting for 6% of cells over the background rate of DC apoptosis in the cultures (fig. 4.10A). There was a progressive increase in the number of cells falling in the sub-G0 gate over time, but there was no distinct peak, as seen in other systems (Rad et al., 2003). This reflected the asynchronous induction of apoptosis in DC. Unfortunately, ethanol fixation impaired GFP detection, and as a result the data could not be gated on infected DC. However, the number of sub-G0 DC increased with MOI, strongly suggesting that it was infected DC that died by apoptosis (fig. 4.10B). It was noteworthy that even at 24 hours, only approximately 25% of DC infected at MOI of 1 (i.e. about half the number infected) were apoptotic, demonstrating that not all infected DC underwent apoptosis.

Section 4.2.4 shows that HSV-1 infection of DC does indeed result in cell death, which occurs partly by apoptosis. However, this is delayed and occurs asynchronously, suggesting that reduced T cell stimulatory capacity of DC cannot be attributed solely to virus induced cell death. Therefore, two facets of the DC-T cell interaction that have an important bearing on T cell activation, namely cytokine secretion and the maturation state of the DC, were studied in greater detail.

132



Figure 4.7 Change in DC size following HSV-1 infection.

DC were infected at MOI of 1 and cultured for 16, 40 and 64 hours. A typical FSC/SSC gate that includes the majority of DC is shown. Representative experiment shown (n = 3).



DC were infected at MOI of 1 and cultured for 16 or 40 hours. MTT reduction was assessed as described in Materials and Methods. * = p < 0.02 is the mean optical density of uninfected DC relative to infected cultures. Results expressed as mean optical density \pm SEM (n = 3). Statistical analysis by Student's *t* test.



Figure 4.9 Ultrastructural changes in DC morphology after HSV-1 infection. DC were infected with HSV-1 (MOI = 1) for 16 hours, fixed in 2% gluteraldehyde and prepared for analysis by EM (see materials & methods). Unaffected DC were observed (A), as well as apoptotic DC (A+B) and secondary necrotic DC (C). Some cells displayed evidence of HSV-1 infection (inserts in B+C). Representative cells shown.





4.2.5 IL-12 secretion following HSV-1 infection

One of the critical signals provided by DC to initiate T cell activation is cytokine secretion. One of these cytokines, IL-12, can synergise with APC costimulatory molecules to induce T cell proliferation and IFNγ secretion (Kubin et al., 1994; Murphy et al., 1994). Therefore, reduced stimulation of T cells by HSV-1 infected DC might have resulted from impaired secretion of this cytokine.

Fig. 4.11 shows that HSV-1 infection per se did not result in IL-12 secretion. In contrast, IL-12 p40 production in response to LPS stimulation was impaired in a significant and dose-dependent manner, with the maximum inhibition observed at the highest MOI. This inability to secrete IL-12 in response to appropriate stimuli may in turn reflect the inability of DC to respond to IL-12 inducing signals (e.g. CD40L (Luft et al., 2002b), LIGHT (Morel et al., 2003) or 4-1BBL (Futagawa et al., 2002; Wilcox et al., 2002)) during T cell contact in vivo, and thus hinder T cell priming.

4.2.6 HSV-1 induced changes in DC surface phenotype

4.2.6.1 HSV-1 effects on DC differentiation

Potent T cell stimulating capacity is a defining characteristic of DC relative to other APC. Another possible explanation for the dramatic reduction in T cell stimulatory capacity and IL-12 secretion was that HSV-1 infection differentiated DC into cells with more monocyte/macrophage features. This was assessed by analysing the surface expression of CD1a and CD14 following infection. The expression of both these molecules did not change following infection (fig. 4.12). Thus, DC did not dedifferentiate following infection.

4.2.6.2 HSV-1 effects on surface phenotype

The other major signal a T cell receives from a DC during contact are co-stimulatory signals which act in conjunction with TCR ligation by the MHC-peptide complex. Therefore, the MHC and co-stimulatory molecule expression of infected DC was assessed (fig. 4.12).

Immature DC were infected and the surface phenotype of these DC analysed after 16 hours of culture. Gating on GFP-ve and GFP+ve DC following infection distinguished between uninfected and infected DC populations respectively. DC were infected at a MOI of 0.3 to ensure the presence of a substantial number of uninfected DC. Fig. 4.12 shows that HSV-1 infection results in significant phenotypic maturation, as demonstrated most obviously by the increased expression of CD86, in both GFP-ve and GFP+ve DC. HLA-DQ and HLA-DR (fig. 6.11) expression on the surface of DC was also consistently higher in infected cultures. MHC class I levels were elevated only on GFP-ve DC. The data suggest that both infected and uninfected DC in these culture were induced to mature by HSV-1 infection.

When infected DC were cultured in the presence of 100 ng/ml of LPS (fig. 4.12), strikingly only GFP-ve DC were able to further upregulate CD86 above the level induced by HSV alone, whereas no increase occurred in GFP+ve DC. This difference was even more apparent for MHC class I expression. Interestingly, HLA-DQ (and HLA-DR) expression was not affected in this way. Virus infection did not affect the response to LPS of both GFP-ve and GFP+ve DC, permitting an increased expression of these molecules.

Therefore, section 4.2.6 demonstrated that HSV-1 infection did not differentiate DC into cells of a known lower T cell stimulatory capacity, such as macrophages. Concurrently, viral infection induced maturation of infected DC, which could have rendered these cells more potent APC. However, the refractory nature of the phenotype change to LPS suggests that infected DC cannot respond fully to maturation stimuli in their microenvironment. This could impair their ability to turn into potent mature APC, and thus explain the poor T cell stimulatory capacity of these cells (figs. 4.1 + 4.2).

138





DC were infected HSV-1 and cultured for 40 hours in the presence or absence of LPS (100ng/ml). IL-12 (p40) levels in the supernatant were measured by ELISA. * p<0.03 is uninfected DC relative to all other groups. Results expressed as mean \pm SEM (n = 4). Statistical analysis by Student's *t* test.



Figure 4.12 Phenotypic changes following HSV-1 infection of DC.

Immature DC were infected with HSV-1 (MOI = 0.3) and then cultured for a further 16 hours in the presence or absence of LPS (100 ng/ml). (A) Representative experiment (n = 3). (B) Mean of median fluorescence intensity (n = 3). Error bars represent SEM.

4.3 Discussion

This chapter has detailed the effects of HSV-1 infection on DC-mediated T cell stimulation, morphology, viability, IL-12 secretion and surface phenotype. The data paint a negative picture with regards to the effects of HSV-1 on DC physiology. However, based on changes in surface phenotype, both infected and uninfected DC also show evidence of activation in response to signals from the infection, and this will form the basis of further study (see chapters 5 and 6). The present discussion will aim to integrate the effects of the virus seen here with the effects of HSV-1 on other cell types, and to evaluate how these changes can explain the functional consequences. Furthermore, the data are placed in the context of the ability of the host to generate anti-HSV responses in vivo, and how they may relate to the control of viral infections in general.

4.3.1 HSV-1 effects on DC-mediated T cell proliferation

The data in figs. 4.1 and 4.2 clearly demonstrate that HSV-1 infection of DC inhibits their ability to stimulate T cell proliferation. Reduced activation of naïve T cells was determined by allogeneic MLR and ConA driven autologous MLR assays. These assays are non-physiological in nature in relation to an in vivo HSV response, and therefore, the ability to stimulate a memory T cell recall response was assessed by measuring proliferation in response to PPD exposure. Although this assay requires conventional antigen capture, processing and presentation, it did not directly assess the ability to elicit HSV-specific responses. Indeed, in a proportion of individuals, HSV-1 infected DC did elicit such memory responses, but these were difficult to assess quantitatively without a positive control.

4.3.1.1 HSV recall responses

The proliferative responses obtained from the coculture of infected DC and autologous T cells were strongly indicative of a memory response to HSV-1. The proportion of individuals responding was consistent with serological data suggesting that approximately 50% of the UK population has been previously exposed to HSV-1.
However, there was the possibility that the virus infection stimulated a non-specific response, perhaps to exogenous antigens such as proteins in the FCS in the medium. Therefore, the serostatus of the volunteers tested was analysed. Owing to administrative difficulties in obtaining a diagnostic laboratory reading, an antiviral assay was used. Although it was very sensitive and the data clear-cut, it is not possible to say whether the antibodies in the serum were specific for HSV-1 or were HSV-2 antibodies that cross-reacted for HSV-1 (Gorander et al., 2003). There is evidence that some T cell responses can be type specific (Eriksson et al., 2004), although the assays used in this thesis could not determine this. Nonetheless, the perfect correlation between the proliferative response and serostatus suggested that these were indeed memory responses to HSV, and that proliferation resulted from response to HSV-1 antigens in the DC-T cell coculture.

The mechanisms involved in initiating these recall responses were not clear, as fig. 4.2 indicated that DC directly infected with HSV-1 had very poor APC function. It was possible that CD4+ T cells controlled viral replication within the DC, as seen for HCMV, thereby permitting direct antigen presentation by infected DC (Le Roy et al., 2002). However, this was an unlikely scenario because the observed proliferation would have correlated directly with MOI of DC, such that the more DC infected, the greater would have been the responses. Furthermore, as the proliferation assays were set up 16 hours after DC infection, the viral effects on DC would have been exerted already. Therefore, it was unlikely that the APC driving T cell proliferation were the infected DC themselves.

In the absence of direct presentation, uninfected APC cross-presenting HSV antigens from dead/dying DC were most likely to be responsible for stimulating the T cells. It was possible that this was being mediated by contaminating APC in the T cell preparation. This was unlikely, not only from the purity of the T cell population (chapter 2), but also because the HSV recall assays were carried out in parallel with the PPD assays in fig. 4.2. In those experiments, T cells alone cultured in the presence of PPD did not proliferate, suggesting that this population of cells did not contain appreciable number of APC (data not shown). The only other population that could cross-present HSV antigens were the uninfected DC. This scenario would be consistent

with the inverse correlation seen between the number of DC infected and the degree of response. The proliferation was greatest at the lowest MOI, that is, in the presence of the greatest number of uninfected DC left in co-culture.

4.3.1.2 Cross-presentation in anti-viral responses

The role of cross-presentation in the stimulation of anti-viral T cell responses has been studied in detail recently. DC can take up exogenous viral proteins or apoptotic cells that have been infected with virus, present viral antigens on MHC class I molecules and stimulate CD8+ T cell responses (Bachmann et al., 1996; Albert et al., 1998b). This has also been shown recently for the herpesviruses HCMV and Epstein Barr Virus (EBV) (Arrode et al., 2000; Tabi et al., 2001; Bickham et al., 2003). Mechanistically, the accumulation of MHC class I processing machinery to phagosomes is believed to provide a suitable environment for the generation of MHC class I peptides for loading and presentation on the surface (Guermonprez et al., 2003; Ackerman et al., 2003). Although recent evidence suggests that other APC also possess this machinery to cross-present exogenous antigens (Ramirez and Sigal, 2002), the potency of DC as APC renders this cell type critical for generating T cell responses in this way (Jung et al., 2002).

From the evidence above, it is hypothesised that uninfected DC in HSV-1 infected cultures could have acquired HSV antigens from infected DC and stimulated the T cell responses seen in fig. 4.1. Although DC can acquire antigen from live cells (Harshyne et al., 2001), a significant source of antigen was also likely to have been apoptotic infected cells, as these cells could be targeted to DC for uptake (Albert et al., 1998a). Recent evidence also suggests that DC apoptosis in vivo enhances the immunogenicity of its antigens, and uptake of virally infected DC in the LN in this way may be important in enhancing cross-priming (Racanelli et al., 2004). This may also serve as a useful mechanism to transfer antigenic material in DC emigrated from the periphery to lymphoid organs (Huang et al., 2000; Belz et al., 2004).

The in vivo relevance of cross-presentation pathway in the generation of anti-HSV immunity has been recently demonstrated in mouse models, whereby stimulation of

CD8+ T cell responses is carried out by uninfected CD8 α + LN-resident DC (Smith et al., 2003b; Allan et al., 2003). The absence of viral DNA in these cells confirmed that infected DC were not responsible for the presentation. However, it is not yet clear whether the CD8 α + DC in LN are tissue resident DC or whether migrating DC can differentiate into these cells (Moron et al., 2002). The implication from the latter scenario is that the presenting DC are uninfected bystander DC at the infection site. In support of this notion, anti-HSV CD4+ T cells are stimulated by uninfected CD11b+ DC derived from the submucosa of infected sites (Zhao et al., 2003). Furthermore, excising the infection site reduces the length of time of HSV-1 antigen presentation and subsequent CTL activation (Stock et al., 2004).

4.3.1.3 HSV-1 mediated inhibition of T cell stimulation

The ability to stimulate T cell proliferation is a property reserved for a small number of cell types, professional APC. Studies carried out prior to and in parallel to the data presented here have shown that immature DC infected with HSV-1 have reduced capacity to stimulate allogeneic T cell proliferation (Salio et al., 1999). This inhibition is mirrored in HSV-1 infected macrophages (Hoves et al., 2001) and B cells (Barcy and Corey, 2001). Interestingly, mature DC infected with HSV-1 also demonstrated impaired ability to elicit T cell stimulation, despite HSV-1 having little effect on the surface phenotype of these cells (Kruse et al., 2000). Furthermore, the reduced ability of infected immature DC to stimulate T cells seems paradoxical in view of the more "mature" phenotype of infected DC (fig. 4.12) (Salio et al., 1999). This indicates that changes in surface phenotype are not the only determinants of infected DC ability to activate T cells. This issue is discussed in section 4.3.5.

In the general context of viral infections, disruption of DC function, and subsequently delaying the generation of a cellular immune response, would be a selective advantage to the virus' survival in the host. Indeed, many viral infections of DC result in an inhibition of the cell's ability to elicit T cell proliferation (see table 1.2). This concept will be discussed in more detail in chapter 7.

One level at which the virus could impair the DC's function is during the interaction with T cells. It is possible that the virus blocks an essential "back-signal" by T cells into the DC. There are many signals the DC receives from T cells, such as CD40L or 4-1BBL. Although the data in this chapter have shown that infected DC are refractory to further phenotype changes, IL-12 secretion and morphological changes in response to LPS, analysis of studies with other infections demonstrate that this conclusion cannot be extended to all ligands. Whereas, responses to LPS and CD40L are equally impaired following HCMV infection of DC (Moutaftsi et al., 2002), MV and parainfluenza virus (PIV) infected DC show impaired responses to CD40L but additive maturation with LPS or P(I:C) (Fugier-Vivier et al., 1997; Schnorr et al., 1997; Bartz et al., 2003). Therefore, the impaired response to LPS stimulation here suggests, but does not confirm, that DC may also be refractory to T cell-derived stimuli.

HSV-1 - infected

Theoretically viruses could also affect the DC ability to simulate T cells by exploiting the plastic differentiation pathway of myeloid cells to differentiate DC into cells with less potent APC potential, such as macrophages or back to their monocyte precursors (Chapuis et al., 1997; Palucka et al., 1998; Akashi et al., 2000). There is evidence that some viruses can affect the differentiation of monocytes into DC (Li et al., 2002; Gredmark and Soderberg-Naucler, 2003). However, the data presented here demonstrate that, at least within a day after HSV-1 infection, DC differentiation is maintained. Time points beyond this would be obscured either by cell death or the DC migration to lymph nodes and interaction with T cells, a process that can itself maintain DC viability (Bjorck et al., 1997).

4.3.1.4 HSV-1 effects on T cell function

As well as affecting the response of DC to the environment, the virus could have prevented T cell proliferation by having a direct inhibitory effect on the T cells themselves, either by inducing secretion of inhibitory cytokines, or through expression on the DC surface of immunosuppressive proteins that can subsequently interact with T cells. Subsequent to the completion of this work, HSV-1 gD has also been reported to have immunosuppressive actions on T cell proliferation, possibly by competing for the endogenous ligand of HVEM on T cells, LIGHT (La et al., 2002). This function

was only tested in the context of immobilised anti-CD3 mediated polyclonal T cell stimulation and it is not clear whether gD could mediate similar effects in a more conventional APC-T cell interaction. MV HA and MV F glycoprotein expression on DC is important for the inhibition of T cell proliferation (Klagge et al., 2000; Dubois et al., 2001). It has been hypothesised that the generalised immunosuppression observed after MV infection in vivo results from the non-specific inhibition of T cell responses by these glycoproteins. In contrast, the immunostimulatory properties of gD on other cell types (e.g. other DC - see chapter 6) may compensate for its immunomodulatory function. Alternatively, the more peripheral and localised nature of HSV infections in vivo restricts significantly its systemic access and functional consequence.

It is also possible that infected DC express endogenous inhibitory proteins. Following HCMV infection of DC, [umour necrosis factor-related apoptosis-inducing ligand (TRAIL) expression is upregulated, turning DC cytotoxic and killing neighbouring T cells (Raftery et al., 2001). Recent studies suggest that TRAIL expression is also upregulated on DC after HSV-1 infection, although this was only demonstrated by western blot (Muller et al., 2004). Therefore, an untested hypothesis is that TRAIL expression in T cell proliferation observed may have resulted from killing of T cells.

One other possibility was that virus replicating in DC (fig. 3.8) could have infected T cells directly. HSV-1 infection of activated T cells renders these cells susceptible to apoptosis (Raftery et al., 1999), possibly explaining the functional defect observed. In certain viral infections, especially HIV-1, DC-T cell transfer has formed the basis of models to explain the establishment of the infection in vivo. Virus picked up by DC in the periphery (e.g. in the genital mucosa) is "carried" by DC migration to the target CD4+ T cells (Geijtenbeek et al., 2000a). VZV infection and replication in DC permits trans infection of T cells (Abendroth et al., 2001), but this was not observed here with HSV-1. This may be because DC were washed prior to addition of T cells 16 hours after infection, and viral production after this time is minimal (fig. 3.8). Alternatively, T cells in co-culture with infected DC were insufficiently activated to render them susceptible to HSV-1 infection, as activating T cells by mAb cross linking CD3 does increase these cells' permissiveness (La et al., 2002). However, it is also possible that

HSV trans infection of T cells is not a property of this virus, as HSV is a more localised disease and does not require systemic spread of infection.

4.3.2 Morphological changes after infection

The physiological advantages of a dendritic morphology remain an area of debate. Hypothetically, it may be advantageous to both the main functions of DC. The large membrane surface area created by the processes allows a larger sampling volume for sensing danger in the surrounding environment. Equally, it also provides a larger stage for the interaction with T cells in secondary lymphoid organs. Surprisingly, there is little experimental data to support these theories, although recently, Rho GTPase modulation of the actin cytoskeleton in DC has been shown to affect DC migration, and contact and subsequent activation of T cells (Kobayashi et al., 2001; Swetman et al., 2002; Benvenuti et al., 2004a). From the virus' perspective, reducing both antigen sampling and presentation would be beneficial in order to impair the generation of a virus specific cellular response, and hence favour virus survival.

HSV-1 infection resulted in striking changes in the morphology of DC, most noticeably causing the loss of the characteristic membrane processes on non-adherent cells and dendrite formation on FN-adherent cells. These changes occurred in parallel with development of homotypic clusters between infected and uninfected DC and with occasional syncytia formation. Several explanations could account for these changes. In the light of reduced DC viability after infection, an important factor to take into account is cell death. However, it is also important to study the viral-host interplay in the regulation of changes in cell shape. Finally, changes in adhesion molecule expression can also affect cellular morphology on a substratum. These issues will be discussed in turn below.

4.3.2.1 Morphological changes vs. cell death effects

Apoptosis of any cell also incorporates a cell rounding stage and a loss of locomotive capacity on the underlying substratum. Given that HSV-1 infection of DC can ultimately result in apoptosis (figs. 4.9 and 4.10), this is an important issue. However,

morphological changes were apparent within 8 hours of infection, when cell viability was not yet affected. No significant changes in MTT reduction were seen until 16 hours after infection and hypodiploid apoptotic cells were only seen from 12 hours after infection. Furthermore, cell rounding induced by viruses can also occur in a caspase 'independent manner (Carthy et al., 1998). Therefore, it was likely that the morphological changes observed were secondary to specific viral induced changes rather than cell death alone.

It is worth noting that HSV-1 is a cytopathic virus, and one of the known steps in cytopathic effects is cell rounding. This raises interesting questions regarding the cell specificity of the effects observed here (i.e. whether HSV targets the cytoskeleton of DC specifically). Although it is unlikely that HSV has evolved to target DC morphology, the question could be addressed from another perspective: are DC particularly sensitive to the cytoskeleton modulating effects of HSV-1? Few cells have as elongated and delicate structures as DC, which are also more dynamic than they first appear. Changes in temperature, as well as short-term exposure (3 hours) to the protein synthesis inhibitor, cycloheximide, can result in rapid loss of these processes (data not shown). It is therefore likely that small changes in the cytoskeleton or upstream signalling in cells such as DC, would be observed at earlier times after HSV-1 infection. In support of this theory, HeLa cell morphology infection is remarkably unaffected 16 hours after infection (fig. 3.8). Furthermore, Vero cells infected with HSV-1 show an accumulation of cytoplasmic G-actin within 4 hours of infection, but do not show loss of integrity of actin microfilaments and cell morphology until 16 hours after infection. As there was no increase in the total actin concentration in these cells, the conversion of filamentous (F)-actin to globular (G)-actin in the early stages of infection was insufficient to result in gross morphological and cytoskeletal changes in those cells (Bedows et al., 1983). Therefore, it is possible that DC are more susceptible to equivalent changes in F-actin distribution/integrity, explaining the early changes in morphology seen.

4.3.2.2 Viral induced changes in DC cytoskeleton

The family of small Rho GTPases are important in regulating the turnover of the actin cytoskeleton, and therefore maintaining and inducing changes in cell morphology (Hall, 1998). Recently, their role in the regulation of the DC cytoskeleton has been investigated. DC endocytosis and macropinocytosis are dependent on the activity of Cdc42 and Rac (Garrett et al., 2000; West et al., 2000). These signalling molecules may regulate the formation of membrane ruffles and lamellipodia necessary to carry out these DC functions (Burns et al., 2001; Swetman et al., 2002). Viral interference with these signalling pathways could affect the formation of such structures, and therefore alter the morphology of the cell overall. In addition, cytoskeletal rearrangement is important for chemokine-mediated migration (Yanagawa and Onoe, 2002; Dong et al., 2003), and for antigen processing and presentation (Al Alwan et al., 2001; Al Alwan et al., 2003; Dong et al., 2003; Benvenuti et al., 2004a). Impairment of actin-mediated polarisation of surface molecules into the immunological synapse may also have contributed to the impaired T cell stimulation in these studies (Al Alwan et al., 2001; Al Alwan et al., 2003). Such altered cytoskeletal rearrangement was proposed to explain the equivalent functional inhibition by HSV-infected B lymphoblastoid cells (Barcy and Corey, 2001).

The viral effector molecules that mediate changes in the cytoskeleton are currently unclear. One candidate molecule is the protein product of the US3 gene. US3 from HSV-2 can prevent Rac- and Cdc42-mediated phosphorylation of one of the downstream regulators, PAK (Murata et al., 2000), which is implicated in regulating the actin cytoskeleton and cell morphological changes (Sells et al., 1997; Daniels et al., 1998). Both HSV-1 and HSV-2 US3 are highly homologous to PAK (Murata et al., 2000), and may prevent PAK phosphorylation by competing for interaction with guanine-nucleotide exchange factors. Inhibition of Cdc42- and Rac-regulated PAK activity would be consistent with the morphological changes seen after HSV-1 infection, as microinjection of dominant negative Rac or Cdc42 redistributes F-actin and induces a rapid retraction of dendrites in DC adhered to a FN substratum (Swetman et al., 2002). Recent microarray analysis has revealed that US3 is expressed as early as 2 hours post-infection (Stingley et al., 2000). The redistribution of G-actin within 4

hours of infection (Bedows et al., 1983) are consistent with an important role for US3 in early changes in the cytoskeleton of HSV-1 infected cells.

In vivo studies carried out with US3 deleted viruses result in severely attenuated footpad and vaginal infection (Nishiyama et al., 1992; Inagaki-Ohara et al., 2001). The reduced virulence did not stem from an impaired ability of the virus to replicate in cells, but rather was secondary to a more potent local inflammatory cell infiltrate, including DC and T cells, with increased secretion of IL-12 and IFNy. It is possible that US3 deleted viruses were not able to prevent apoptosis in infected epithelial cells (Leopardi et al., 1997) (see section 4.3.3.2), but that would not explain the normal replication kinetics in the infected tissues (Inagaki-Ohara et al., 2001). On the other hand, it is tempting to speculate that failure of the virus to disrupt the DC cytoskeleton allowed infected DC to present viral antigens rapidly. Coupled with the cross-presentation pathway, a quantitative increase in antigen presentation capacity may have elicited a more potent immune response that could clear the peripheral infection more effectively.

4.3.2.3 Integrin mediated change in morphology

Integrins are heterodimeric molecules composed of an α and β subunit. DC express both β 1 and β 2 integrins (fig. 4.6) (McCarthy et al., 1997; Woodhead et al., 1998; Ammon et al., 2000; Puig-Kroger et al., 2000). These molecules are important for DC migration to and from the blood and lymphatics, via peripheral tissues. Transendothelial migration requires adhesion to surface of endothelial cells by both β 1 and β 2 integrins (D'amico et al., 1998), whereas in the tissues, DC adhesion to and migration on the ECM occurs via β 1, but not β 2, integrins (Le Varlet et al., 1992; D'amico et al., 1998).

The only two known β 1 associating α chains expressed on MDDC are α 4 (CD49d) and α 5 (CD49e) (D'amico et al., 1998). These integrin heterodimerise with β 1 integrins to form VLA-4 and VLA-5 respectively. Both of these molecules are important in the adhesion of DC to FN, a major component of the ECM (Puig-Kroger et al., 2000). The β 2 integrin heterodimers that are important in DC migration across

endothelial barriers are $\alpha L\beta 2$ (CD11a/CD18), $\alpha M\beta 2$ (CD11b/CD18) and $\alpha X\beta 2$ (CD11c/CD18) (D'amico et al., 1998; Fiorini et al., 2002).

Following HSV-1 infection, the expression of the β 1 integrin (CD29) was downregulated on infected DC, whereas the β 2 integrin (CD18) levels were unaffected. The downregulation of CD29 has also been observed after HSV-1 infection of fibroblasts (Ito et al., 1995), and may therefore constitute a general HSV-1 modulatory mechanism. Interestingly, not all members of the herpesvirus family affect β 1 integrin expression in the same way. Whereas VZV also downregulates CD29 expression, HCMV results in upregulation of CD29 on fibroblasts (Ito et al., 1995).

Similarly to CD29, CD49d expression was also downregulated after HSV infection of DC. Unfortunately, no suitable CD49e antibodies were available for this study. As integrins exist in a dimeric form, changes in expression could be mediated by regulation of either subunit, although the antibodies used here only detected single integrin chains, giving no direct assessment of the quantity of functional heterodimer present on the surface. In that respect, it is noteworthy that CD29 expression is unaffected by DC maturation (Puig-Kroger et al., 2000), whereas CD49d is upregulated 48 hr after DC stimulation, in a p38 and NF-kB dependent manner (Aiba et al., 1993; Puig-Kroger et al., 2000). Reduced expression of both molecules suggests regulation by the virus, but the factors involved are unclear. CD49d is predominantly controlled at the transcriptional level in the steady state (Postigo et al., 1997; Puig-Kroger et al., 2000), and its levels could be controlled by changes in transcription rate during infection, as well as post-transcriptionally, by the viral proteins vhs and ICP27 (Kwong and Frenkel, 1987; Hardwicke and Sandri-Goldin, 1994). The observed differential regulation of β 1 and β 2 integrins is currently difficult to explain. Further study may elucidate differential regulatory mechanisms involved in the expression of these molecules.

As $\beta 1$ integrins on MDDC are FN receptors, viral downregulation could have played a role in the loss of dendritic morphology in infected DC. It is unclear what degree of $\beta 1$ integrin downregulation is required to have a significant functional effect on cell morphology. The rapid turnover of the cytoskeleton at the point of adhesion with the

ECM means that even a small reduction in turnover of adhesion molecules may result in the loss of cellular attachment and hence retraction of the dendrite. Furthermore, it should be noted that the data acquired in this chapter relate to whole cell expression of these molecules by flow cytometry. CD29 expression is critical to dendrite formation on FN by focal accumulation at the end of dendritic processes and attachment to the ECM (Swetman et al, submitted for publication). Therefore, the downregulation of expression may be more specific than has hitherto been suggested, and may thus have more dramatic effects on FN-adherent DC morphology.

The only other virus where DC morphology has been documented following infection is MV. The MV-infected cells were also rounded, but there was no data on the effect on the morphology of FN-adherent DC (Grosjean et al., 1997; Fugier-Vivier et al., 1997). Furthermore, as it is currently unknown whether any MV protein can interact with or affect the function of RhoGTPases, it is difficult to extrapolate relevant parallels with this virus to the effects seen here with HSV-1.

Another observed phenomenon that may be related to expression of adhesion molecules was the increased clustering of infected and uninfected DC. The absence of upregulation of $\beta 1$ or $\beta 2$ integrins might imply that the increased adhesion between DC was being mediated through the increased expression of alternative intercellular adhesion molecules, such as nectins (Takai and Nakanishi, 2003), but it is also possible that clusters were formed by the increased migration of uninfected DC towards infected ones, without quantitative or qualitative changes in adhesion molecules. This may explain the clusters seen in highly confluent FN-adherent DC, but cannot account for the many non-adherent clusters that were also seen. Furthermore, the functional relevance of such "heterotypic" clustering is unclear, because in this scenario it would be difficult to specify whether it is a virus induced effect or a host response to viral infection. The close proximity of infected and uninfected cells allows the intrinsic fusogenic ability of HSV-1 to form syncytia (Spear et al., 2000).

4.3.2.4 Effects of HSV-1 infection on DC migration

One other aspect of viral modulation of DC morphology is how it impinges on the ability of the cell to migrate. Emigration from the skin and entry into appropriate areas of lymphoid organs require locomotion in response to chemotactic gradients. One previous study raised this issue by determining the Ca2+ flux response to CCL19, the CCR7 ligand important in mediating mature DC migration to lymph nodes. The flux was abrogated in HSV-1 infected cells (Salio et al., 1999). However, the assay was carried out 24 hours after infection, by which time, under normal stimulatory circumstances, the majority of DC would have already migrated to a draining lymph node (de Vries et al., 2003). It would be interesting to study the migration response of DC very early after infection, as upregulation of chemokine receptors, especially CCR7, may occur in parallel to that of CD86 and MHC class II (fig. 4.12). This may be a critical step as CCR7 expression is inhibited following HCMV infection, corresponding to a decreased migration potential of these infected DC (Moutaftsi et al., 2004). However, chemokine receptor expression does not always necessarily correlate with migration potential due to uncoupling from downstream signalling pathways (Sozzani et al., 1997; D'amico et al., 2000; Penna et al., 2001). Furthermore, some viruses have evolved to secrete antagonistic chemokine analogues, which would disrupt chemokine-mediated migration even in the presence of functional chemokine receptors (Seet et al., 2003). Preliminary studies revealed that HSV-1 infected DC could not extend dendrites and migrate on FN like uninfected DC (data not shown) (Swetman et al., 2002)). Therefore, future studies should use chemokine gradients to test the migration potential of virally infected DC (Moutaftsi et al., 2004).

4.3.3 DC viability after HSV-1 infection

Viral induction of DC death represents a very direct strategy to annihilate the capacity to stimulate anti-viral immunity. However, both the two main forms of cell death, necrosis and apoptosis, create a dilemma for viruses.

4.3.3.1 HSV-1 and necrotic cell death

Necrotic cell death is uncontrolled, resulting in dispersal of a large amount of viral antigens in a proinflammatory environment. This can activate neighbouring DC, which alerts the immune system to the presence of infection, and may enhance DC cross-presentation of exogenous viral material to T cells (Sauter et al., 2000). It is interesting that HCMV replicates at low levels in DC, which has the effect of minimising necrotic cell death (Raftery et al., 2001). Similar findings are presented here for HSV-1, and low replication in lymphoid tissues may provide a selective advantage.

4.3.3.2 HSV-1 prevention and induction of apoptosis

Viral induction of apoptotic cell death, although self-contained and immunologically more silent (Sauter et al., 2000), does conflict with the evolutionary requirements of viruses to replicate. Therefore, many have developed sophisticated anti-apoptotic pathways to keep infected cells alive long enough to complete a full replicative cycle (Dobbelstein and Shenk, 1996; Lukac and Alwine, 1999; Aubert and Blaho, 2001). HSV-1 can prevent apoptosis in a number of cells it infects using several mechanisms designed to protect against a variety of death stimuli. These include glycoprotein J (gJ) (Jerome et al., 1999; Zhou et al., 2000), gD (Zhou et al., 2000; Medici et al., 2003), ICP4 (Leopardi and Roizman, 1996), ICP27 (Aubert and Blaho, 1999) and the US3 protein kinase (Leopardi et al., 1997; Hagglund et al., 2002). Loss of gJ or US3 renders cells susceptible to CTL induced apoptosis (Jerome et al., 2001), loss of ICP4 leads to mitochondria dysfunction and DNA fragmentation (Galvan et al., 1999), whereas loss of ICP27 results in the activation of caspase 3 and DNA damage (Aubert et al., 1999).

Anti-apoptotic mechanisms are also needed to protect infected cells from endogenous apoptosis inducing signals. For HSV, the precise identity of these components remains controversial. Recent studies have suggested that the apoptotic stimulus is dependent on IE gene transcription (Koyama and Adachi, 1997; Aubert et al., 1999; Sanfilippo et al., 2004). However, this does not explain apoptosis induced by gD deleted viruses, which cannot infect cells normally or express any viral genes (Zhou et al., 2000). In this scenario, only the endocytosed virion components could be the apoptotic ligands. It is also not yet clear whether these apoptotic stimuli render the cells more susceptible

to environmental apoptosis-inducing stimuli or whether they activate endogenous apoptotic pathways directly. Nevertheless, pro-apoptotic viral components are likely to have played a part in the evolution of some of the anti-apoptotic mechanisms described above.

4.3.3.3 HSV-1 induced apoptosis in DC

Despite the extensive list of anti-apoptotic proteins, it is intriguing that HSV-1 can prevent apoptosis following infection in some (Chou and Roizman, 1992; Leopardi et al., 1997; Jerome et al., 1999), but not other cell types (Galvan and Roizman, 1998). In cells of haematopoietic origin, apoptosis has been observed after HSV-1 infection of both PBMC (Tropea et al., 1995) and activated T cells (Ito et al., 1997), but not macrophages (Hoves et al., 2001). Other studies of HSV-1 infection of immature DC either did not address the issue of viability (Salio et al., 1999) or did not find a gross difference, as assessed by trypan blue exclusion assay (Mikloska et al., 2001; Samady et al., 2003). However this assay is not very sensitive, as it relies on the structural integrity of the host cell membrane. Therefore, DC viability was assessed by a MTT reduction assay, which demonstrated a substantial reduction in viability of HSV-1 infected DC 40 hours post infection. Although the changes in FSC/SSC affected the majority of the cells (fig. 4.7), this only translated into a 50% decrease in MTT reduction. One possible explanation could be that changes in cell size on the flow cytometer overestimate cell death due to increased susceptibility of HSV-1 infected cells to shrinkage after formaldehyde fixation.

The ultrastructural changes seen by EM (cell rounding, nuclear chromatin condensation) and the reduction in cell size detected by flow cytometry, indicate that apoptotic cell death was primarily responsible for the loss of cell viability. In contrast, necrosis would have caused an increase in cell size, followed by cell rupture and loss of structural integrity. This was not seen in the majority of DC infected with HSV-1. An assay commonly used for the quantitative assessment of apoptosis is the detection of phosphatidylserine (PS) leaflets on the extracellular portion of the apoptotic cell by annexin V binding. However, in HSV-1 infected cells, PS expression does not necessarily correlate with cell death, whereas DNA degradation does (Jerome et al.,

155

1998). Therefore, the loss of DNA integrity was detected by the presence of decreased PI staining by flow cytometry, a sub-G0 population. The detection of this population post-infection confirmed that HSV-1 infection did result in apoptosis of DC. However, entry into apoptosis was extremely slow and asynchronous, with only very slight changes seen one day after infection, and only 30% cells affected after 40 hours, even though transgene expression was detected within four hours of infection. EM studies confirmed that there was viral infection of some apoptotic cells. Unfortunately, the ethanol fixation used in this PI staining protocol did not prevent GFP from washing out of the cells, and therefore a direct 2-colour analysis of the viability of infected cells could not be carried out. However, the correlation between the number of infected DC and the degree of apoptosis strongly suggested that viral infection was the stimulus for apoptosis.

The number of DC in the sub-G0 region was consistently lower than the percentage of cells infected, suggesting that not all infected DC died by apoptosis. Some cell death could be attributed to rare necrotic cell death or to syncytia formation. Alternatively, some infected cells did not die. This may have occurred in a subpopulation of more mature DC in the DC cultures. This was observed occasionally by flow cytometry as a population (up to 10%) of cells expressing high levels of CD86 or MHC molecules, equivalent to those induced by LPS stimulation. This conclusion is consistent with the relative resistance of mature DC to HSV-1 induced cell death (Kruse et al., 2000).

4.3.3.4 Mechanisms responsible for HSV-1 induced apoptosis

The signalling pathways affected by virus infection that can induce apoptosis have not been elucidated. Recent studies have suggested that the level of JNK activation correlates with DC apoptosis following stimulation with H_2O_2 (Handley et al, submitted for publication). Activation of this pathway may also be important in DC apoptosis after HSV-1 infection, as the activation of JNK correlates with the degree of apoptosis in HSV-1 infected neurones in vivo (Perkins et al., 2003). Furthermore, HSV encodes a protein kinase, US3, which attenuates the activation of JNK and prevents apoptosis in infected cells (Murata et al., 2000), underlying the advantage to the virus to prevent cell apoptosis via this pathway. Other signalling pathways that may be important in inducing apoptosis include the NFAT, ERK and p38 pathways. Firstly, HSV-1 protein mediated inhibition of NFAT nuclear translocation (Scott et al., 2001) could promote apoptosis (Pyrzynska et al., 2001; Mosieniak et al., 1998), although in some cell types it may have the opposite effect (Srivastava et al., 1999). Secondly, HSV-1 expression of the protein kinase ICP10 activates ERK and subsequently prevents apoptosis in infected cells (Perkins et al., 2002). Finally, although no correlation has been found between the levels of p38 MAPK activation and apoptosis induction by HSV-1, p38 MAPK activation has been associated with the induction of apoptosis in some scenarios (Torcia et al., 2001; Tikhomirov and Carpenter, 2004). Indeed, DC pre-treated with the p38 MAPK inhibitor, SB203580, demonstrated an increase in FSC on the flow cytometer compared to infected cells not treated with the inhibitor (data not shown). Although not a formal assessment of cell viability, this preliminary observation is suggestive that the activation of p38 MAPK by HSV-1 is a pro-apoptotic event in DC, in contrast to no effect on H₂O₂ induced apoptosis (Handley et al, submitted for publication). Therefore, the cellular pathways involved in inducing cell death may be stimulus specific.

A possible host effector mechanism for the induction of apoptosis by HSV-1 in DC has recently been described. Early viral gene products rendered DC susceptible to autocrine/paracrine TRAIL and TNFα mediated apoptosis by downregulating the procaspase 8 inhibitory protein, c-FLIP, and up-regulating p53 (Muller et al., 2004; Muller et al., 2004). The downregulation of c-FLIP is likely to be an important effector pathway in HSV-1 mediated apoptosis in general. Macrophage viability is not affected by HSV-1 infection and the level of expression of c-FLIP in these cells is also unaffected (Hoves et al., 2001). Conversely HSV-1 infection of human umbilical vein endothelial cells (HUVEC) induces apoptosis at a similar rate as seen in immature DC and a similar reduction in c-FLIP expression was seen (Muller et al., 2004). These observations demonstrate the cell-specific nature of HSV-1 regulation of apoptosis. Furthermore, it suggests that HSV-1 is unlikely to have evolved specific mechanisms to induce apoptosis in immature DC and that DC are not more susceptible to HSV-1 induced apoptosis than other cell types. Of note, c-FLIP expression is elevated in

mature DC (Franchi et al., 2003; Lundqvist et al., 2002; Rescigno et al., 2000). Therefore, it is possible that HSV-1 cannot reduce c-FLIP expression in mature DC to levels that render the cells susceptible to apoptosis, explaining the preservation of viability by mature DC infected with HSV-1 (Kruse et al., 2000).

The discussion above does not explain the delayed kinetics of death or why the viral anti-apoptotic mechanisms do not rescue DC from HSV-1 induced apoptosis. One hypothesis revolves around the low levels of replication observed in DC. As viral replication in DC is lower than in a more permissive cell line, it suggests that viral gene expression would also be lower in DC. Consequently, it may take longer for sufficient quantities of the apoptotic IE gene transcripts to accumulate and trigger apoptosis in DC than in a more permissive cell (Sanfilippo et al., 2004). Concurrently, however, the lower gene expression in DC may also result in insufficient synthesis of viral anti-apoptotic proteins and the cell eventually undergoes apoptosis. Such a scenario demonstrates the delicate balance between pro- and anti-apoptotic signals that is also seen in other viral infections(Cella et al., 1999b). VV has apoptosis inducing components (Ramsey-Ewing and Moss, 1998) and induces apoptosis in DC (Engelmayer et al., 1999). Since the virus only abortively replicates in DC (Engelmayer et al., 1999), insufficient expression of anti-apoptotic proteins may allow apoptosis to ensue. Influenza virus components can also trigger apoptosis (Schultz-Cherry et al., 2001; Chen et al., 2001) and apoptotic cell death does occur after replication and infection in DC (Cella et al., 1999b).

Furthermore, VV, influenza virus and HSV-1 also demonstrate how the relationship between DC viability and T cell stimulation is not a straightforward one (Engelmayer et al., 1999; Cella et al., 1999b). Similar rates of apoptosis are seen for both infections, yet influenza virus infected DC stimulate T cell proliferation more efficiently than uninfected DC, whereas vaccinia virus DC are impaired, similar to the effects of HSV-1 infection. Influenza virus, but not VV, induces DC maturation, which may partly explain the difference in T cell stimulatory capacity, but HSV-1 also induces maturation without enhancing T cell stimulation. Although in vivo many other factors, such as migration to LN, would also need to be taken into consideration, the discriminating factor may relate to the timing of DC apoptosis in relation to T cell contact. Indeed, MV may have evolved to exploit that interaction specifically, as DC infected by this virus upregulate Fas expression, rendering them susceptible to T cell FasL-mediated apoptosis (Servet-Delprat et al., 2000a), despite the antiapoptotic CD40-CD40L interaction with activated T cells (Bjorck et al., 1997). This "induced" cell death is a very efficient way to target the death of DC at its most critical moment, during the DC-T cell interaction.

The discussion above highlights that cell death cannot be the sole component responsible for the inhibition of T cell stimulation by HSV-1 infected DC. Shorter functional assays of DC-T cell activation (ConA autologous assays – fig. 4.2C), which reduced the contribution of cell death during the assay, also showed impaired proliferative responses. Furthermore, despite little change in DC viability, a pronounced inhibition in T cell proliferation is seen following HSV-1 infection of mature DC (Salio et al., 1999; Kruse et al., 2000). The role of cell death could ultimately be addressed experimentally by artificially preventing apoptosis, perhaps using small pharmacological inhibitors of caspases, which may keep the DC alive throughout the length of the functional assays.

4.3.4 Impact on DC IL-12 secretion by HSV-1

Clearance of most viral infections is dependent on a Th1 skewed CD4+ and CD8+ T cell response. One of the key determinants is the cytokine context at the time of initial interaction between DC and T cells. IL-12 appears to be the initial signal for many Th1 responses (Trinchieri, 2003). In the context of HSV infection in vivo, mouse studies have implicated IL-12 in the control of the infection, acting predominantly upstream of the initiation of IFN γ -mediated protective T cell responses (Malmgaard and Paludan, 2003; Vollstedt et al., 2004). As a result, the protective effects of IL-12 have recently been harnessed therapeutically to design vaccination strategies for HSV in mice models (Sin et al., 1999; Lee et al., 2003a).

The cellular source for IL-12 secretion in vivo is critical to understanding how this cytokine regulates the ensuing immune response. Studies of peripheral lesions in mice have suggested that this cytokine is secreted by inflammatory cells and not by

parenchymal cells of the infected tissue (e.g. epithelial cells) (Kumaraguru and Rouse, 2002; Mikloska et al., 1998). Cells that can secrete IL-12 in vivo include neutrophils and macrophages, and they may play a role in these models (Trinchieri, 2003). However, DC can also secrete large amounts of IL-12 following appropriate stimulation via TLR ligands, as well as members of the TNF family, including CD40L, LIGHT and 4-1BBL on T cells (Schulz et al., 2000; Morel et al., 2003; Futagawa et al., 2002; Wilcox et al., 2002). HSV-1 DNA contains many hypomethylated CpG motifs (Honess et al., 1989) that are sufficient stimuli for IL-12 secretion in DC (Hemmi et al., 2000). However, HSV-1 infection alone was insufficient to initiate IL-12 secretion by DC in this system, consistent with previous findings in human myeloid DC (Ghanekar et al., 1996; Salio et al., 1999).

In vivo, the relative paucity of IL-12 inducing stimuli may be important to ensure that DC secrete this cytokine only under appropriate stimulation. However, this does not explain why viruses (including HSV) that possess known IL-12 inducing ligands (CpG DNA and dsRNA) do not always induce IL-12 secretion after infection (Moutaftsi et al., 2002; Plotnicky-Gilquin et al., 2001; Servet-Delprat et al., 2000b; Libraty et al., 2001). It is possible that a viral infection in vitro provides DC with insufficient quantities of stimulatory ligand, whereas in vivo DC respond to greater concentrations of extracellular ligand in the infection site. Alternatively, the expression of appropriate receptors may be an important variable. Human myeloid DC do not express TLR9 and this may explain the unresponsiveness to HSV DNA containing hypomethylated CpG motifs (Krug et al., 2001). In contrast, human PDC (Krug et al., 2001) and mouse macrophages (Hemmi et al., 2000) do express high levels of TLR9 and secrete IL-12 after HSV-1 infection (Krug et al., 2004; Malmgaard et al., 2000). However, the literature is not consistent with respect to the TLR9 dependence of IL-12 secretion. In the mouse, myeloid DC cultured from bone marrow, unlike their human counterparts, do express TLR9 (Boonstra et al., 2003). However, HSV-1 and HSV-2 infection of these cells does not induce IL-12 secretion (Jones et al., 2003), although there may be upregulation of IL-12 p40 mRNA synthesis (Kanangat et al., 1996). This suggests that hypomethylated CpG DNA may not be sufficient as an IL-12 inducing stimulus from HSV-1. Efficient replication in some cells may generate sufficient dsRNA intermediates to stimulate TLR3 induced IL-12 secretion (Alexopoulou et al., 2001),

whereas the low replication of HSV-1 in MDDC (fig. 3.8) may prevent sufficient quantities of stimulatory dsRNA to accumulate and activate TLR3 expressed in these cells (Krug et al., 2001). HSV-1 may also possess other IL-12 inducing ligands, independent of replication products that are detected by receptors not ubiquitously expressed. Consistent with this, UV-inactivation of HSV prior to infection of mouse macrophages abrogated IL-12 secretion (Malmgaard et al., 2000), whereas it remained intact following PDC infection (Krug et al., 2004). Therefore, the ability to secrete IL-12 may relate to both viral replication within cells, as well as receptor availability for appropriate stimulatory ligands.

Lack of IL-12 induction by viruses may also result from viral inhibitory mechanisms. IL-12 secretion in response to LPS by HSV-1 infected DC was impaired, in agreement with other studies (Salio et al., 1999; Jones et al., 2003). As most IL-12 is secreted 8-12 hours following DC stimulation (Langenkamp et al., 2000), the inhibition is unlikely to result simply from cell death. Similarly to HSV, DC infected with HCMV demonstrate reduced IL-12 secretion when simultaneously activated with an IL-12 inducing stimulus (e.g. LPS or CD40L) (Moutaftsi et al., 2002). The ability to secrete other cytokines (type I IFN) following infection, while IL-12 secretion is suppressed (Chapter 5) (Moutaftsi et al., 2002; Ho et al., 2001), supports a theory of targeted regulation of this cytokine, as does the specific inhibition of IL-12 secretion by post-transcriptional regulation by another herpesvirus, HHV-6 (Smith et al., 2003a).

The complexity of regulation of IL-12 secretion presents several levels where viral intervention may be important, including transcriptional, translational and post-translational (Trinchieri, 2003). The viral mechanisms affecting IL-12 secretion may differ dramatically between virus families. MV mediated inhibition has been attributed to the actions of MV nucleoprotein (MV NP) and the glycoproteins MV HA and MV F. Therefore, MV gene expression is not essential to reproduce the inhibition (Klagge et al., 2000; Marie et al., 2001; Dubois et al., 2001). In contrast, UV-inactivation of some viruses (e.g. HCMV and HSV-1) removes the inhibition of IL-12 secretion ((Moutaftsi et al., 2002), data not shown), reflecting regulation of this cytokine by a viral gene product. In addition, inhibition of IL-12 secretion may be specific to the stimulus. For example, MV infection of DC specifically inhibits CD40L induced IL-12

secretion, while enhancing it in response to LPS stimulation (Schnorr et al., 1997; Fugier-Vivier et al., 1997).

One transcription factor that has recently been implicated in the regulation of IL-12 p40 is nuclear factor of activated T cells (NFAT), which can bind the IL-12 p40 promoter and enhance its transcription (Zhu et al., 2003). In the resting state, NFAT resides in the cytoplasm and in response to an increase in intracellular [Ca2+], it is dephosphorylated by calcineurin, unmasking its nuclear localisation signal and allowing nuclear import. The ability of HSV to specifically prevent the nuclear translocation of NFAT may play a role in the inhibition of IL-12 secretion (Scott et al., 2001).

The reduced IL-12 secretion may have contributed to the impaired ability of HSV-1 infected DC to induce T cell proliferation. However, HSV-1 may also impair the secretion of other cytokines, such as IL-2, that are equally critical in the initiation of T cell proliferation, as seen in murine CMV (MCMV) infection of DC (Granucci et al., 2001; Andrews et al., 2001). It should also be noted that the data in fig. 4.11 only demonstrated inhibition in the secretion of IL-12 p40 by the entire DC population. The absolute dependence of the IL-12 p70 heterodimer, as well as the related cytokines, IL-23 and IL-27, for the p40 chain implies that the secretion of these functional cytokines would also be inhibited (Trinchieri, 2003). However, preliminary studies analysing the secretion of the functional heterodimer reveal that its secretion is only inhibited at much higher MOI than seen for IL-12 p40 (data not shown). This likely reflects a drawback of ELISA measurement of cytokines, as data in chapter 5 reveals that IL-12 p70 secretion by uninfected DC in those cultures could be enhanced greatly by paracrine type I IFN secreted and this may mask the inhibition in virus infected cells. Future studies should analyse the presence of IL-12 p70 intracellularly by flow cytometry, gating on both infected and uninfected cells, in order to definitively conclude the degree of inhibition seen for IL-12 secretion.

4.3.5 Changes in DC surface phenotype

The switch of DC functional state from immature to mature, represents a profound change in physiology. Therefore, not only is it tightly regulated, but it also presents itself as an attractive target for infecting viruses. Inhibiting the maturation of DC would render the antigen presenting cells incapable of initiating a potent antiviral immune response. Indeed antigen presentation to T cells by DC in an immature or semi-mature state, in the absence of sufficient costimulation or cytokine secretion, has been proposed as a mechanism for establishing tolerance or anergy in the T cell pool (Hawiger et al., 2001; Albert et al., 2001), a strategy that would clearly be advantageous for viral survival.

Two striking observations were made in this chapter regarding the change in expression of surface molecules in DC. Firstly, after HSV-1 infection, both infected and uninfected DC displayed more mature phenotypes relative to uninfected DC. This maturation response was indicative that virus infection resulted in activatory signals for both these cell populations. This will be discussed in more detail in chapters 5 and 6.

Secondly, despite the activated phenotype of infected cells, the expression of molecules such as MHC class I and CD86 was not as high as that seen following LPS stimulation alone, and infected DC were refractory to further maturation by LPS. In vivo, DC that acquire a maturation signal in the periphery (of any origin) would migrate to draining lymph nodes and are likely to receive further maturation stimuli from T cells and cytokines they encounter there(Wilcox et al., 2002). In the system used here, the second maturation signal was modelled with LPS, as this is a very well characterised stimulus. However, it is important to bear in mind that each stimulus signals in a receptor-specific manner, and thus changes seen in response to LPS may be specific to TLR4-induced signalling and difficult to extrapolate to other stimuli. Furthermore, the viral mechanisms responsible for the refractory response to LPS are still ill defined. There are several well-characterised immunomodulatory molecules in HSV-1 that could be responsible for the changes observed, but their relative expression and specific function in DC are currently unclear.

4.3.5.1 HSV-1 changes in MHC class I expression

The US12 gene of HSV-1 encodes for the ICP47 protein which interferes with peptide loading of MHC class I molecules by binding to the cytosolic side of the transporter associated with antigen processing (TAP) molecules on the ER. The result is lower expression of MHC class I molecules, including those presenting viral antigens on the surface of infected cells (York et al., 1994). However, there is no discernable change in the expression of MHC class I on the surface of DC after HSV-1 infection. The efficacy of ICP47 is cell-type dependent (York et al., 1994) and it is possible that ICP47 does not exert significant functional effects in DC. As potent APC, these cells may express larger amounts of TAP than in other cell types. Alternatively, maturation after infection may render the cells more resistant to the effects of ICP47. It is also possible that other viral proteins play a role in regulating the expression of MHC class I. Studies have shown that vhs prevents the recognition of infected targets by CD8+ T cells, suggesting an effect on MHC class I expression (Tigges et al., 1996). However, as vhs is a ribonuclease, its actions may be non-specific. The precise role of ICP47, vhs or any other candidate molecules, can only be determined by the use of viruses lacking functional genes and/or proteins.

4.3.5.2 HSV-1 changes in CD86 expression

The expression of CD86 in DC after infection follows a similar pattern to that of MHC class I, particularly the inhibition to further upregulation by LPS. Although the two proteins are transcribed from separate genes, the transcriptional control governing this process may be very similar, as both p38 MAPK and NF-kB are important (Yoshimura et al., 2001; Ardeshna et al., 2000). As well as affecting DC morphology, changes in Rac activation by HSV-1 US3 may also play a role in regulating expression of both MHC class I and CD86 (Murata et al., 2000; Jaksits et al., 2004). Therefore, changes in cell signalling induced by the virus may affect the expression of both these molecules, and explain the similar refractory response to LPS. Alternatively, virus infection induced the secretion of cytokines that inhibit the maturation of DC, such as TGF β (Geissmann et al., 1999), as seen after HCMV infection of fibroblasts (Arrode et al., 2002).

It is also possible that the virus expresses proteins that modulate the expression of CD86 (and possibly also other members of the B7 family) in immature DC specifically, in a manner analogous to the relationship between ICP47 and MHC class I. This is an unexplored hypothesis, but it is strengthened by recent studies of HSV-1 infection of mature DC. These cells are more resistant to HSV-1 induced changes than immature DC. The expression of MHC and co-stimulatory is unaffected after infection, whereas CD83 is selectively downregulated (Salio et al., 1999; Kruse et al., 2000), suggesting that HSV-1 may have evolved mechanisms to downregulate the expression of this molecule. It is interesting that HCMV and VV also specifically downregulate CD83 expression on mature DC (Hertel et al., 2003; Jenne et al., 2000). The mechanisms that regulate CD83 expression are unclear and it is not possible to exclude that CD83 may be targeted 'inadvertently' by viral mechanisms that modulate the surface expression of structurally unrelated host molecules (Tomazin et al., 1999).

4.3.5.3 HSV-1 changes in MHC class II expression

MHC class II molecule expression was affected differently to that of CD86 and MHC class I. It was elevated by infection, and in most experiments, this expression was increased further by LPS stimulation. There was no evidence of specific downregulation or refractoriness in the expression of MHC class II in HSV-1 infected DC. This observation may be best explained by the cellular regulation of MHC class II expression in DC, which differs from MHC class I and CD86. MHC class II molecules are 'stored' in endosomes in immature DC and, upon maturation, associate with the relevant peptides, and are shuttled to the surface (Inaba et al., 2000). As this process occurs simultaneously with transcriptional downregulation of MHC class II (Landmann et al., 2001), the regulation of these molecules could escape both transcriptional and post-transcriptional regulation by HSV-1 proteins (e.g. by vhs or ICP27). However, recent data has suggested that HSV-1 can disrupt MHC class II loading, by exporting HLA-DR molecules from the ER, thereby preventing their surface expression (Neumann et al., 2003). As HLA-DR expression increases to maximal levels in response to LPS within 5 hours of stimulation (data not shown), gB

may not affect the expression of this molecule on the surface of DC rapidly enough. This does not exclude the possibility that gB may affect the antigenic repertoire expressed by the DC after this time point. This protein may have played a role specifically in the reduced stimulation of PPD-specific memory CD4+ T cells (fig. 4.2A) (Pathan et al., 2001; Li et al., 1998). Similarly, protein antigen presentation is impaired in DC infected with HHV-6, whereas presentation of pre-processed peptides was intact, confirming that viruses may inhibit DC antigen presentation at the processing level (Kakimoto et al., 2002), possibly also mediated by disrupting the cell's cytoskeleton (Dong et al., 2003).

However, it is also a distinct possibility that the virus cannot synthesise sufficient quantities of gB to affect DC expression of MHC class II-peptide complexes, which are more stably expressed on the surface than on other cells (Cella et al., 1997). Therefore, although targeting of MHC class II pathway suggests a viral adaptation to target the function of APC specifically, it is also possible that the main evolutionary drive for this molecule is the inducible, but low, expression of MHC class II on infected epithelial cells (Albanesi et al., 1998), protecting these cells from CD4+ T cell recognition.

4.3.5.4 Responses of the uninfected GFP-ve population

Although GFP-ve DC upregulated both MHC class I and CD86 after LPS stimulation, it was noticeable that the expression of these molecules did not reach those of uninfected DC treated with LPS alone. Several reasons could explain this. Infected DC could express inhibitory molecules that prevent a complete response to maturation stimuli by neighbouring DC. Other viruses do indeed possess candidate inhibitory molecules (e.g. MV NP (Marie et al., 2001)). The only known HSV candidate molecule is gD, as this glycoprotein can also suppress T cell proliferation (La et al., 2002) and is expressed on the surface of DC (chapter 3). However, data in chapter 6 demonstrate that gD has stimulatory, not inhibitory activity on DC function. Other candidate molecules that can be excluded are other viral envelope glycoproteins, as their interaction with the DC surface do not prevent upregulation of CD86 to LPS stimulation (chapter 6). Nevertheless, despite these provisos, the behaviour of the

GFP-ve DC population was clearly different from that of infected cells and appeared, at least in part, free from some of the viral regulatory mechanisms seen in GFP+ve cells.

4.3.5.5 Relationship between phenotype changes and T cell stimulation

The changes in phenotype need to be analysed in relation to the functional inhibition seen in fig. 4.2. Alterations in the expression of MHC and co-stimulatory molecules would be predicted to have an impact on the ability to stimulate T cells. Preventing MHC molecule expression would hamper the activation of T cells both quantitatively and kinetically. Furthermore, the absence of co-stimulatory molecules may also induce tolerance or regulatory T cells (Jonuleit et al., 2000; Hawiger et al., 2001).

Studies of other viral infections of DC demonstrate that the relationship between surface phenotype, T cell responses and infection resolution is not so straightforward. Infections that prevent DC responses to maturation stimuli, such as LPS or CD40L, do indeed result in impaired stimulatory function (Salio et al., 1999; Moutaftsi et al., 2002; Engelmayer et al., 1999). It is interesting that it is only the large DNA viruses of the herpesvirus and poxvirus families that possess this function, suggesting that this is a highly evolved and complex viral strategy. However, it is equally noteworthy that DC maturation following infection per se does not generate more potent APC ubiquitously (Schnorr et al., 1997; Plotnicky-Gilquin et al., 2001). DC maturation following infection may contribute both to the disease pathogenesis (Raftery et al., 2002; Ho et al., 2001; Libraty et al., 2001), and also risk generating potent DC that present viral antigens. The spectre of fatal immunopathology to the host and/or rapid clearance of the virus may have selected for the evolution of viral mechanisms to curb the function of DC (Raftery et al., 2001; Moutaftsi et al., 2002; Marie et al., 2001).

4.4 Conclusions

HSV-1 infection of DC renders these cells poor stimulators of naïve and memory T cell proliferation. This chapter has attempted to elucidate some of the DC functions that are disrupted by the virus that may be responsible for the loss in antigen

presentation capacity seen. HSV-1 infection results in a rapid loss in dendritic morphology, and this loss in cytoskeletal plasticity may hamper their migration as well as their interaction with T cells. DC also undergo delayed apoptosis after infection, although its contribution to the functional impairment is difficult to discern. Noticeable, however, was the inability of infected DC to respond to further maturation stimuli after infection. This was demonstrated in the reduced ability to secrete IL-12, and the absence of this cytokine may be responsible for the inhibition in T cell proliferation. The inability to upregulate MHC class I and costimulatory molecules in response to further maturation stimuli may result in functional impairment of DC in vivo. This may include decreased or delayed migration to secondary lymphoid organs, as well as defective DC-T cell interaction.

The data presented in this chapter are consistent with the hypothesis that HSV-1 infection of DC impairs the ability of these cells to directly present and activate HSV-specific CD4 and CD8 T cells. This would subsequently delay the generation of the antiviral cellular and humoral response, perhaps long enough to allow the virus to establish itself in the host. The full implication of how this fits into a model of a peripheral herpetic lesion will be discussed in more detail in the final chapter of this thesis. However, this conclusion leaves unanswered two important questions: how are effective anti-HSV immune responses, present in all seropositive individuals, generated in vivo if DC are functionally impaired? Secondly, how can the knowledge gained from the viral-DC interaction be exploited therapeutically? The changes in surface phenotype after HSV infection may suggest answers for both these questions.

The maturation of uninfected DC in the culture suggests one potential mechanism to overcome the functional block on DC function after infection. The factors responsible for this bystander maturation will be the subject of investigation in the next chapter. On the other hand, the direct maturation in response to viral infection, albeit submaximal, implies a host response to the virus. The ability of critical cells of the innate immune system, such as DC, to recognise HSV-1 infection warrants further investigation. Identifying the immunostimulatory components of the virus may play an important role in vaccine design and will be the subject of the subsequent chapter.

Type I IFN: autocrine and paracrine priming factor

Chapter 5

Type I IFN: autocrine and paracrine priming factor

5.1 Introduction

The previous chapter detailed the functional consequences of HSV-1 infection of DC. Through inhibition of a variety of aspects of DC physiology, the virus prevented the ability of DC to stimulate T cell proliferation. However, HSV specific T cell memory responses could be initiated from individuals who had previously been exposed to the virus. The possible resolution of this paradox is that, in vivo, the onus for antigen presentation lies with bystander uninfected DC in the periphery of the infection. It is critical that these DC present HSV antigens to T cells with sufficient potency to stimulate effector cells, rather than induce tolerance or anergy in the T cell pool towards HSV antigens. One key determinant of this outcome is the maturation state of DC, such that presentation of antigen by immature DC can result in tolerance (Jonuleit et al., 2000; Hawiger et al., 2001). Important determinants of the infection outcome are the soluble factors secreted by infected DC, and how bystander DC respond to these. The rest of this chapter will attempt to characterise and detail these further.

The prime candidate mechanism is that of an immunostimulatory factor secreted after infection of DC, which can affect neighbouring uninfected DC. One major piece of evidence in support of this hypothesis was that uninfected GFP-ve DC in infected cultures displayed a more mature phenotype than the original immature population that was infected. It was possible that these activation signals originated from direct cell contact with infected DC. However, the activated phenotype was observed irrespective of the confluence in the tissue culture plate, suggesting that a soluble factor(s) secreted after infection was responsible for "bystander activation" of uninfected DC. This is an attractive model as, if extrapolated in vivo, it would allow DC neighbouring the infected cells and facing the risk of infection. Therefore, the objectives of this chapter will be to characterise the soluble factor(s) secreted after infection that is responsible for bystander DC activation in vitro, and determining how it may play a role in vivo in overcoming the block in DC function by HSV-1 infection. Other activities of this soluble factor on DC responses to TLR ligands will also be investigated and discussed.

5.2 Results

5.2.1 Bystander DC activation by HSV-1 infection

5.2.1.1 Phenotype changes

The changes in phenotype observed in the previous chapter (fig. 4.12) demonstrated that uninfected GFP-ve DC demonstrated elevated levels of CD86 and MHC class II relative to DC from cultures not exposed to virus. The possibility that this population represented infected DC not expressing GFP was excluded in chapter 3 (fig. 3.7). The addition of LPS to these cultures elevated the expression of CD86 and MHC class I in GFP-ve DC but not GFP+ve DC, further delineating the difference between the two populations (fig. 4.12).

To investigate the possibility that a soluble factor secreted after infection was responsible for the maturation of GFP-ve DC, supernatants were harvested from infected DC cultures and used to culture autologous uninfected DC (fig. 5.1). Although the production of HSV-1 in infected DC was low, it was important to remove any virus from these supernatants. This was carried out by ultracentrifuging the supernatant, and then filtering it through a $0.2\mu m$ filter. As a control, the supernatant from uninfected DC was treated in the same manner and was defined as supCON-DC. Supernatant from infected DC was defined as supHSV-DC. To confirm that supHSV-DC was devoid of infectious virus, it was titred on BHK cells in a viral infectivity assay. No infectious virus was detected (data not shown). Furthermore, culture of autologous DC with supHSV-DC yielded no GFP+ve DC (fig. 5.1A).

The most noticeable observation was that DC cultured in supHSV-DC upregulated both CD86 and HLA-DR (fig. 5.1B). Consistent with the upregulation of these molecules seen in GFP-ve DC, this was not as elevated as the effects seen with LPS (fig. 5.1B). Furthermore, this upregulation was also not refractory to further LPS stimulation (fig. 5.1B), excluding the presence of viral inhibitory molecules that rendered DC refractory to further stimulation as seen for infected DC (fig. 4.12). As a control, culturing DC with supCON-DC did not upregulate CD86 or HLA-DR expression, excluding the possibility that the maturation signal was ubiquitous in all supernatants from one-day old DC cultures.

Culturing DC in supHSV-DC had no effect on cell viability (data not shown). Therefore, the reduced viability observed in HSV-1 infected DC cultures (fig. 4.8) was likely to result from the direct effects of viral infection, rather than the release of soluble cytotoxic mediators.

5.2.1.2 Effects on IL-12 secretion

The secretion of IL-12 from DC cultured in supHSV-DC was measured next. Although supHSV-DC was not per se a stimulus for IL-12 secretion, it synergised with LPS and induced elevated levels of IL-12 p40 and p70 secretion relative to that seen with LPS stimulation of DC cultured in supCON-DC (fig. 5.2). Therefore, the soluble factors that mediated the changes in phenotype above also primed DC for elevated IL-12 secretion.



Figure 5.1 Effects of soluble factors released following infection on DC phenotype.

DC were infected with HSV-1 (MOI = 1) and supernatants were collected after 16 hours. The supernatants from both uninfected and HSV-1 infected DC were then treated as described in materials and methods (supCON-DC and supHSV-DC respectively), and added to fresh autologous DC. Flow cytometry detected the expression of GFP (A), and CD86 and HLA-DR (B). *(Left panel)* Representative experiment. *(Middle panel)* Mean of three independent experiments. *(Right panel)* LPS (100 ng/ml) was added to DC cultured in supCON-DC or supHSV-DC and the expression of CD86 and HLA-DR assessed. Mean of three independent experiments.

173









DC were infected with HSV-1 (MOI = 1) and supernatants were collected after 16 hours. The supernatants from both groups were then treated as described in materials and methods (supCON-DC and supHSV-DC) and added in the presence or absence of 100ng/ml LPS to fresh autologous DC. Secretion of (A) IL-12 p40 or (B) IL-12 p70 was measured by ELISA. Results expressed as mean \pm SEM (n = 3). * denotes p<0.01 relative to supCON-DC.

5.2.2 Characterisation of the soluble factor

5.2.2.1 Confirmation of type I IFN secretion by MDDC

Although a secreted viral protein could have been responsible for the above effects, cytokines secreted by infected DC were analysed, specifically TNF α and type I IFN (fig. 5.3A). TNF α was secreted, but at relatively low levels compared to those seen after LPS stimulation. The ability of TNF α to inhibit IL-12 secretion made it an unlikely candidate (Hodge-Dufour et al., 1998; Ma et al., 2000). On the other hand, the amount of type I IFN secreted (270 IU/ml) was consistent with previous studies of the effects of recombinant IFN α on upregulation co-stimulatory molecules on DC (Padovan et al., 2002) and on priming CD40-induced IL-12 p70 secretion (Luft et al., 2002b). Therefore, the presence of type I IFN in supHSV-DC was a likely candidate for the effects observed, and this hypothesis was tested further.

The source of type I IFN from DC cultures has been the subject of recent investigations, particularly since the identification of "natural interferon producing cells", or PDC, that can secrete large quantities of type I IFN when stimulated with many viruses, including HSV-1 (Siegal et al., 1999). Therefore, despite the purity of the myeloid DC population used in this study (fig. 3.3), it was important to demonstrate that the type I IFN released following HSV-1 infection did not derive from small numbers of contaminating PDC. This was confirmed by intracellular staining for IFN α (fig. 5.3B+C). The increase in intracellular IFN was mediated by a shift in fluorescence of the entire DC population, not by a subset of IFN-secreting DC (fig. 5.3B+C). As the type I IFN secretion system responds to a well described autocrine and paracrine positive feedback loop (Taniguchi and Takaoka, 2002), DC in these experiments were infected at a high MOI of 3. The resulting infection efficiency was near 90%, reducing to a minimum the number of uninfected DC, and ensuring that the stimulus for IFN secretion came directly from the infection process. Therefore, type I IFN detected in the supernatant of HSV-1 infected MDDC was derived directly from these cells.

5.2.2.2 Confirmation of antiviral activity of type I IFN

It has been known for many years that type I IFN can prevent HSV-1 replication by preventing the transcription of IE genes (Mittnacht et al., 1988; Oberman and Panet, 1988). In the context of a herpetic lesion, the release and paracrine activity of this family of cytokines would not only have immunostimulatory activity on bystander DC, but may also protect them from viral infection. Therefore, DC were cultured in the presence of 1000 IU/ml of IFNa for 24 hours prior to HSV-1 infection and then for a further 16 hours, also in the presence of IFN α . Consistent with antiviral activity, IFN α pretreatment prevented GFP expression (fig. 5.4A). Although this gene is controlled by the CMV IE promoter, IFNa has similar inhibitory effects on the transcription of this heterologous promoter as on HSV-1 IE genes (Nicholl and Preston, 1996). Therefore, although not directly assessed, the antiviral effect of type I IFN on GFP expression could confidently be extrapolated to HSV-1 genes. IFN α pretreated DC infected with HSV-1 also displayed a more activated phenotype than either untreated DC or IFN pretreated DC alone (fig. 5.4B), suggesting that infection induced activation of DC additional to that induced by IFN α alone (fig. 5.5A). Direct viral activation of DC, independent of type I IFN, is discussed in greater detail in chapter 6.

5.2.2.3 Effect of type I IFN on MDDC phenotype

To demonstrate that type I IFN in supHSV-DC could account for DC maturation, DC were cultured in recombinant IFN α , and the changes in surface phenotype assessed. IFN α induced the upregulation of CD86 and HLA-DR in a dose-dependent manner (Fig. 5.5A). Similarly to supHSV-DC, the upregulation of these molecules was less than the levels induced by LPS stimulation. When the two stimuli were added together, the upregulation of CD86 and HLA-DR was no higher than LPS alone (fig. 5.5B), analogous to the effects of supHSV-DC (fig. 5.1B).

One way to confirm that type I IFNs were indeed the active factors was to neutralise their activity in the supernatants. Type I IFNs are a family of cytokines, comprising many subtypes, but which all bind one receptor. Although neutralising antibodies are available that can bind many subtypes, a more straightforward approach was to use a blocking antibody against the receptor, which is comprised of two chains: IFNAR1 and IFNAR2. A neutralising monoclonal antibody that bound IFNAR2 was used. Initial experiments demonstrated that, consistent with previous findings (Padovan et al., 2002; Gauzzi et al., 2002), DC express this receptor, albeit at low levels (fig. 5.6A). To test its neutralizing ability, DC preincubated with the antibody, were cultured in recombinant 1000IU/ml IFN α , and upregulation of CD86 was assessed. This cytokine concentration was chosen to confirm that the antibody would neutralise amounts of type I IFN in excess of the concentration found in infected cultures (fig. 5.3A). The ability of IFN α to upregulate CD86 was totally abrogated by this antibody, confirming its ability to neutralise the biological activity of type I IFN in this system (figs. 5.6B (right panel), 5.9B+C and 5.10). Finally, addition of the antibody almost completely abrogated the ability of supHSV-DC to upregulate CD86, confirming that this cytokine was predominantly responsible for the activity in supHSV-DC (fig. 5.6B).

5.2.2.4 Confirmation of type I IFN effect on MDDC secretion of IL-12

Unfortunately, limitations in DC numbers precluded experiments to neutralise type I IFN activity and assessing its role in the IL-12 priming effect. However, in order to confirm that type I IFNs were able to enhance IL-12 secretion, DC were cultured in the presence of recombinant IFN α , as above, and simultaneously stimulated with LPS. Consistent with the effects of supHSV-DC, IFN α itself was not a stimulus for DC secretion of IL-12 p40 but could prime DC for increased LPS-mediated IL-12 p40 secretion (fig. 5.7A). This synergistic action was also observed for IL-12 p70 secretion (fig. 5.7B). Of note, the synergistic effect of IFN α on IL-12 p70 secretion was greater than that on IL-12 p40 secretion, inducing a 81% increase in IL-12 p70 over that of LPS alone, in comparison to a 61% increase for IL-12 p40. This was reflected in the increase in the IL-12 p70:p40 ratio, increasing the proportion of functional heterodimer secreted (fig. 5.7C).


Figure 5.3 Type I IFN secreted by MDDC infected with HSV-1.

(A) Supernatants from uninfected and HSV-1 infected DC, and DC stimulated with 100 ng/ml LPS were harvested 16 hours after infection and the secretion of TNF α and type I IFN secretion determined by ELISA and an antiviral bioassay respectively. The data are the mean of two and four independent experiments respectively. Error bars represent SEM. * denotes p<0.01 relative to uninfected DC. (B) MDDC were infected with HSV-1 at MOI of 3 and incubated for 12 hours in the presence of 2 μ M monensin. Production of IFN α was assessed by intracellular staining. Representative of three independent experiments. (C) As in (B) but displayed as frequency histogram. Representative of three independent experiments control) staining of three independent experiments. Error bars represent SEM.





DC were precultured in the presence or absence of 1000 IU/ml IFN α for 24 hours prior to infection with HSV-1. After 16 hours of culture in same conditions, the expression of (A) GFP, and (B) HLA DR and CD86 was assessed. Representative of two independent experiments.



Figure 5.5 Effect of IFN α on DC phenotype.

(A) (Left panel) MDDC were cultured for 16 hours in a range of concentrations of IFN α and changes in surface phenotype assessed. Data are mean MFI of three independent experiments. Error bars represent SEM. (Right panel) MDDC were cultured for 16 hours in 1000 IU/ml IFN α or 100 ng/ml LPS and changes in surface phenotype were assessed. Representative of three independent experiments. (B) As for (A) but in some groups DC were also co-cultured with both 1000 IU/ml IFN α and 100 ng/ml LPS. Mean of three independent experiments. Error bars represent SEM.



Figure 5.6 Role of type I IFN in supHSV-DC in inducing DC upregulation of CD86.

(A) Expression of IFNAR2 on DC. Representative of two independent experiments. (B) DC were cultured in supCON-DC, supHSV-DC or 1000 IU/ml IFN α for 16 hours in the presence of anti-IFNAR2 mAb or an isotype control and changes in CD86 expression were assessed. *(Left panel)* Representative experiment. *(Right panel)* Mean MFI of three independent experiments. Error bars represent SEM. # = S.E.M. = +/- 100 MFI.





DC were cultured in the presence or absence of 1000 IU/ml IFN α +/- 100 ng/ml LPS for 16 hours. Supernatants were assayed for IL-12 p40 (A) or IL-12 p70 (B) by ELISA. Results are mean of three independent experiments. Error bars represent SEM. * denotes p<0.01 relative to LPS only stimulated DC. (C) IL-12 p70/IL-12 p40 ratio calculated from (A) and (B) displayed as a bar chart.

5.2.3 Mechanism of type I IFN activity on DC

5.2.3.1 Signalling consequences of type I IFN

Despite inducing a moderate upregulation of HLA-DR and CD86, IFNa itself did not induce the secretion of IL-12, but was able to prime DC for elevated IL-12 secretion. It was likely that this effect was being mediated through the regulation of signalling pathways. The p38 MAPK pathway has been implicated in DC maturation and IL-12 secretion (Arrighi et al., 2001; Ardeshna et al., 2000; Puig-Kroger et al., 2001). Therefore, after short-term pulse of DC with IFN α , the activation of p38 was determined by assessing the phosphorylation state of this enzyme, which correlates with its kinase activity (Raingeaud et al., 1995). There was a significant degree of p38 phosphorylation, albeit less than that observed following LPS stimulation (fig 5.8A). Another pathway involved in maturation and IL-12 secretion is the NF-kB pathway (Yoshimura et al., 2001; Laderach et al., 2003). Its activation follows the degradation of IkB α , a cytosolic inhibitor of the NF-kB subunits. Loss of the IkB α protein permits NF-kB transcription factor subunits to translocate to the nucleus and induce transcription of NF-kB controlled genes (Karin and Ben Neriah, 2000; Yoshimura et al., 2001). IFN α induced some IkB α degradation, but this was again less than that induced by LPS stimulation alone. The degree of p38 phosphorylation and IkB α degradation induced by IFNa and LPS individually correlated with the resultant degree of upregulation of CD86 and MHC class II (fig. 5.5A), and may therefore have been a key determinant. Furthermore, culture with both IFNa and LPS resulted in little difference in the activation of p38, but a greater IkBa degradation relative to LPS stimulation alone (fig 5.8A). This additive increase in NF-kB activation could contribute to the elevated IL-12 secretion seen when DC were pulsed concurrently with the two stimuli (fig. 5.7).

5.2.3.2 Role of p38 MAPK activation in IFN α mediated increase in CD86 expression

The relationship between p38 activation and upregulation of CD86 by either IFN α or LPS, prompted the investigation of the role of this kinase in this change. DC were pre-

treated with the specific p38 kinase inhibitor, SB203580 (Tong et al., 1997) overnight prior to stimulation with IFN α or LPS and then for a further 16 hours. Upregulation of CD86 by IFN α occurred independently of p38 activity. In contrast, LPS induced upregulation was partially dependent on the activity of p38 MAPK (fig. 5.8B). Noticeably expression of CD86 under these conditions (i.e. in the absence of p38 activity) was similar to that induced by IFN α alone. Therefore, this section demonstrates that CD86 expression is regulated at two levels; lower expression is independent of p38 MAPK activity, whereas maximal expression is dependent on p38 MAPK.

5.2.4 Role of type I IFN in mediating TLR-dependent DC activation

5.2.4.1 Phenotype changes

Previous data has indicated that TLR ligation in myeloid DC can induce secretion of type I IFN (Cella et al., 1999b; Coccia et al., 2004). In agreement with this, 100 ng/ml of LPS induced DC to secrete approximately 100 IU/ml of type I IFN (fig. 5.3A). This concentration of IFN α could exert a functional effect on DC phenotype (fig. 5.5). Therefore, it was possible that a proportion of DC phenotype changes in response to LPS were mediated by the autocrine activity of secreted type I IFN (fig. 5.9A). This was tested by stimulating DC with 100 ng/ml LPS in the presence of neutralising anti-IFNAR2 antibody, as used in fig. 5.6. To confirm that the amount of antibody used in these assays was saturating, DC were cultured in parallel with 1000 IU/ml of IFN α (i.e. an excess relative to the amount secreted by DC stimulated with 100 ng/ml LPS) in the presence of the IFNAR2 antibody. This abrogated the upregulation of CD86 induced by IFN α , confirming that saturating quantities of antibody were used in these assays. However, upregulation of CD86 by LPS was unaffected by removing the effect of type I IFN. Thus, TLR4 induced maturation of DC phenotype was independent of the effects of type I IFN (fig. 5.9B).

To assess whether other TLRs induced maturation of DC which was dependent on type I IFN, the actions of the TLR3 ligand, double stranded RNA (dsRNA), were assessed. In vitro dsRNA can be mimicked by commercial preparation of Poly(I:C), which

ligates TLR3 and induces maturation of DC through the activation of NF-kB (Alexopoulou et al., 2001). Poly(I:C), like LPS, was used at the concentration that induced maximal upregulation of maturation markers ($25\mu g/ml$, data not shown). When $25\mu g/ml$ of Poly(I:C) were used to stimulate DC in the presence of neutralising anti-IFNAR2 antibodies, upregulation of CD86 was partially abrogated (fig. 5.9C). Although the DC appeared to be a uniform population (at least by CD1a+ expression), only a subset of DC was inhibited. Therefore, TLR3 mediated upregulation of CD86, in contrast to TLR4, was partially dependent on the autocrine / paracrine activity of type I IFN.

5.2.4.2 Cytokine secretion

Although TLR4-mediated upregulation of CD86 was independent of type I IFN activity, TLR4 ligation also induces secretion of proinflammatory cytokines. Two of these are IL-12 and TNF α (Banchereau and Steinman, 1998). The supernatants from the experiments in fig. 5.9B were analysed for the presence of TNF α and IL-12 p70 by ELISA. Surprisingly, while type I IFNs were not required for the increase in CD86 induced by LPS, they had a significant role in the secretion of both TNF α and IL-12 p70. Neutralising the activity of type I IFN, reduced the secretion of both these cytokines (fig. 5.10). It is difficult to determine precisely the relative contribution of IFN due to the variability in cytokine secretion between individuals. However, it was clear that the secretion of both these cytokines was not totally dependent on type I IFN, but that this factor played a significant role in ensuring maximal secretion.



Figure 5.8 Intracellular signals activated by IFN α in DC

(A) DC were cultured in the presence or absence of recombinant 1000 IU/ml IFN α +/- 100 ng/ml LPS for 60 minutes. Cell lysates were harvested and assayed for IkB α , phospho- and total-p38 MAPK by western blotting. Representative of two independent experiments. (B) DC were cultured in the presence or absence of the p38 MAPK inhibitor, SB203580, for 24 hours prior to stimulation with 1000 IU/ml IFN α or 100 ng/ml LPS and cultured for a further 16 hours +/- 3 μ M SB203580. (*Left panel*) Representative experiment. (*Right panel*) Mean of three independent experiments.



Figure 5.9 Role of autocrine/paracrine type I IFN on DC phenotype changes by LPS and Poly(I:C)

(A) Proposed model for autocrine/paracrine activation of DC by type I IFN after TLR ligation. (B+C) DC were pretreated with anti-IFNAR2 mAb or an isotype control, and then stimulated with 1000 IU/ml IFN α , 100 ng/ml LPS (B) or 25µg/ml Poly(I:C) (C). (Upper panel) Representative experiment. (Lower panel) Mean of three independent experiments. Error bars represent SEM.





Supernatants from the experiments in figure 5.9 (upper panel) were harvested and assayed for secretion of TNF α (middle panel) or IL-12 p70 (lower panel). Data are representative of at least three independent experiments.

5.3 Discussion

The antiviral immunity seen after HSV infections in vivo suggests the existence of mechanisms to bypass the loss in DC antigen presenting function resulting from HSV-1 infection. The previous chapter described that bystander uninfected DC had a more mature phenotype. Furthermore, HSV immune individuals were able to induce recall responses in the presence of HSV-1 infected DC, probably as a result of crosspresentation of viral antigens by uninfected bystander DC in the cultures. However, it is important to note that cross-presentation can result in two disparate scenarios: crosspriming or cross-tolerance and that the latter may result in T cell anergy or deletion, preventing the generation of an anti-viral response (Steinman et al., 2000). Therefore, it is important to understand the processes that skew responses towards cross-priming. One important determinant is the maturation state of the presenting DC, whereby cross-priming is favoured by mature DC (Hawiger et al., 2001). As well as the increased upregulation of MHC and co-stimulatory molecules, the secretion of cytokines, such as IL-12, is also important to ensure a potent T cell response (Albert et al., 2001). Recent data suggests that type I IFN plays an important role in skewing towards a cross-priming response in vivo (Dalod et al., 2003; Le Bon et al., 2003), possibly by acting on DC in secondary lymphoid organs and skewing towards Th1 responses (Foster et al., 2000). However, type I IFN stimulation of DC derived from an infectious site may be equally relevant in stimulating T cell responses. The remainder of this chapter will focus on two aspects of type I IFN activity: the response of bystander DC at the site of HSV infection and the role of type I IFNs in TLR-mediated activation of DC.

5.3.1 DC secretion of type I IFN

A prime candidate for the stimulatory factor secreted by HSV-1 infected DC was type I IFN, a family of cytokines made up of many subtypes (Taniguchi and Takaoka, 2002). The antiviral properties of type I IFN (fig. 5.4) have been appreciated for many years (Isaacs and Lindenmann, 1957), and are believed to play a major role in the control of HSV infections in vivo (Leib et al., 1999; Murphy et al., 2003). Most cell types can act as a source of type I IFN. However, a subset of DC, PDC, secrete high

levels of type I IFN when appropriately stimulated with certain viruses. It has been suggested that this discriminates PDC from myeloid DC (MDC) (Siegal et al., 1999; Foster et al., 2000; Cella et al., 1999b; Giacomini et al., 2001), although this notion has recently been challenged, since appropriate stimulation of myeloid DC can also induce secretion of large quantities of type I IFN, equivalent to that produced by PDC (Diebold et al., 2003).

Therefore, it was important to exclude the presence of PDC from the myeloid DC preparations used in the current study. The culture conditions used made the presence of PDC unlikely. IL-4, which is present in the DC differentiation cultures, has been shown to have an inhibitory effect on PDC survival (Rissoan et al., 1999). Furthermore, PDC express CD2 (Cella et al., 1999a), a marker used to deplete contaminating T and NK cells in the experimental protocol. The combination of myeloid markers expression in >99% of these DC and intracellular staining for IFN α confirmed that the source of type I IFN secreted into the supernatant of HSV infected cultures was undeniably MDDC (fig. 5.3). Thus, MDDC can secrete sufficient functionally active type I IFN following natural infection by the DNA virus HSV-1.

Other studies have attempted to quantify the secretion of type I IFN from MDDC after HSV-1 infection. Some have failed to detect any (Siegal et al., 1999; Mohty et al., 2003), whereas others have detected small amounts (Izaguirre et al., 2003). Independent of variation in experimental protocol, DC generation and HSV strain used, one common difference between this study and some others that have found no IFN, is the detection method for type I IFN. The bioassay used is more sensitive than ELISAs used by others. Furthermore, it is likely to detect a wider range of IFN subtypes than ELISAs, which are limited to the specificity of the antibodies used. This is an important distinction with regards to the mechanisms that regulate the secretion of type I IFN. Viral induction of IFN β transcription and secretion is followed by autocrine signalling through the IFNAR, and subsequent transcription of IFN α genes (see section 5.3.2). Consistent with the type I IFN detected in this thesis, IFN β transcription is increased in MDDC after HSV-1 infection (Izaguirre et al., 2003). However, the same study detected smaller increases in IFN α transcription and in a smaller number of IFN α subtypes than seen for PDC (Izaguirre et al., 2003). This

would explain the small shift in IFN α intracellular staining in fig. 5.3. Thus, the widespread use of IFN α ELISA for the detection of type I IFN is likely to generate negative data for MDDC secretion of type I IFN after HSV-1 infection.

The IFN secreted by MDDC is less than the amount secreted by PDC in response to HSV-1 (Siegal et al., 1999). However, as the amount secreted by MDDC is nonetheless sufficient to elicit functional changes in DC (figs. 5.1 and 5.2), this raises interesting questions about the in vivo advantage to the host of such potent IFN producing cells. This is discussed in greater detail in chapter 7.

The mechanisms responsible for type I IFN secretion by MDDC are also of interest. One viral component that may induce IFN secretion by MDDC is gD (fig. 6.17). This is discussed further in chapter 6. However, it is unclear what determines the differential quantity of IFN secreted by PDC compared to MDC. The recently described capacity of MDC to secrete large amount of IFN only when exposed to dsRNA in the cytoplasm (Diebold et al., 2003), suggests that PDC can detect components of HSV infection which MDC cannot, rather than an intrinsic inability of MDC to secrete type I IFN. This is supported by PDC secretion of IFN α in response to HSV independent of an IFN β autocrine feedback through the IFNAR (Barchet et al., 2002). A likely ligand that fits this model is HSV-1 DNA itself, which contains a high frequency of hypomethylated CpG motifs (Honess et al., 1989). The receptor for this ligand is TLR9, which in humans is expressed in PDC but not MDC (Krug et al., 2001), and exposure of PDC to HSV DNA promotes type I IFN production (Lund et al., 2003). The ability of PDC to secrete IFN α in response to stimulation with UVinactivated HSV is consistent with recognition of the viral genome (Siegal et al., 1999; Barchet et al., 2002). The secretion of IFN by MDDC is also independent of viral replication, but this is likely to involve recognition of different viral components, such as gD (chapter 6).

In support of the plasticity of viral recognition by MDC in the appropriate context, recent data has demonstrated that monocytes differentiated into DC in the presence of GM-CSF and IFN α (instead of IL-4), acquire the capacity to secrete IFN α in response to HSV-1 infection, although this is still inferior to the amount secreted by PDC

(Mohty et al., 2003). The response to HSV correlates with the expression of TLR7 in these cells. The ligand for this receptor is ssRNA, possibly with some specificity for viral sequence (Diebold et al., 2004; Heil et al., 2004). Thus, DC can acquire the capacity to detect products of viral replication in the correct environmental context (i.e. high concentration of IFN α), although HSV may also possess a novel TLR7 binding ligand. A model whereby monocytes are recruited to an infection site rich in IFN α that may prime these cells for recognition of HSV infection is an attractive one and further explored in chapter 7.

5.3.2 Role of type I IFN in DC physiology

The actions of type I IFN are mediated by interferon-stimulated gene (ISG) products. The transcription of these genes is regulated by interferon-stimulated regulatory elements (ISREs) within ISG promoters. Cell stimulation (by TLR or viral infection), results in the phosphorylation of interferon response factor-3 (IRF-3), and induces translocation of this cytoplasmic protein to the nucleus, where it binds to the promoters of some ISGs and in the IFN- β promoter. The subsequent secretion of IFN β results in autocrine stimulation of the IFNAR. In turn, this induces the formation of the IFN-stimulated gene factor-3 (ISGF3) or IFN α -associated factor (AAF). ISGF3 activates transcription of more ISGs and IRF-7, the latter promoting the transcription of IFN α genes, which can amplify the cell's ISG response further. AAF binds IFN γ -activated-site (GAS) elements which promote transcription of IRF-1, which in turn can bind to IRF-1 elements (IRFE) on target gene promoters (Taniguchi and Takaoka, 2002).

The most recognised actions of the IFN system are its antiviral properties (Isaacs and Lindenmann, 1957), but recent studies using mice lacking genes for IFNAR and various IRFs have suggested a role for type I IFNs in regulating several aspects of DC physiology. DC isolated from spleens of IFNAR KO mice possess a more immature phenotype, do not undergo further spontaneous maturation in vitro and are impaired in stimulating T cells (Montoya et al., 2002). These data imply that type I IFNs, acting in an autocrine fashion, are important in enhancing the APC capacity of DC. Furthermore, it is likely that type I IFNs also play a role in the differentiation of precursors into DC, as mice lacking IRF-2 or IRF-8 show selective deficiencies in

several, non-overlapping DC subsets. Furthermore, the absence of these transcription factors also impairs maturation of the remaining DC subsets, further supporting a role for type I IFN in DC responses to various stimuli (Tsujimura et al., 2003; Schiavoni et al., 2002; Aliberti et al., 2003; Ichikawa et al., 2004).

5.3.2.1 Type I IFN effects on immature DC phenotype

Previous studies have demonstrated that IFN α induces sub-maximal DC maturation (Lehner et al., 2001; Padovan et al., 2002; Luft et al., 2002b), and the data in this chapter extends this by demonstrating that the response is dose-dependent. It is important to note that the amount of type I IFN secreted following HSV-1 infection (about 250 'IU/ml) closely reproduced the sub-maximal upregulation of CD86 and HLA-DR observed in DC cultured in supHSV-DC. In vivo, the localised concentration of IFN- α achieved may be much higher, and drive full DC maturation, although this is likely to occur in conjunction with other stimuli in the environment.

As well as affecting DC maturation, the local release of type I IFN may impinge on differentiation of DC precursors. The ability of IFN α from SLE patient sera to induce spontaneous monocyte differentiation into DC suggests this is a relevant process in vivo (Blanco et al., 2001). Subsequent in vitro studies have investigated this phenomenon, in particular comparing with DC generated in GM-CSF and IL-4. Replacement of IL-4 with IFN α induces a proportion of monocytes to develop characteristics of more mature DC, as determined by surface phenotype and T cell stimulating capacity (Paquette et al., 1998; Wang et al., 1999a; Santini et al., 2000; Parlato et al., 2001). However, other studies have observed that similar treatment does not induce mature DC development (Dauer et al., 2003; Mohty et al., 2003), and the absence of IL-4 in the culture may also render the DC refractory to further stimulation (Dauer et al., 2003). The different times at which DC phenotype and function were analysed in these studies make direct comparison difficult.

The number of DC recovered in GM-CSF + IFN α cultures is lower than in the presence of GM-CSF and IL-4 (Paquette et al., 1998; Wang et al., 1999a; Dauer et al., 2003; Della et al., 2004). Furthermore, addition of IFN β to monocyte cultures

supplemented with GM-CSF, IL-4 and TNF α , conditions which normally generate mature DC, reduced the number of DC recovered (McRae et al., 2000a). It is possible that "over-accumulation" of activation signals induces DC death. In similar fashion, addition of IFN α and LPS to monocytes cultured in GM-CSF and IL-4 also induces rapid apoptosis (Lehner et al., 2001), but no such synergy in death occurs at equivalent LPS and IFN α concentrations in the immature DC cultures (data not shown and in (Lehner et al., 2001)). This may be due to different relative susceptibility to activation-induced cell death of differentiating and immature DC respectively. Alternatively, the length of time of stimulation may be important, as monocyte death is often analysed at least 3 days after culture in the various conditions, whereas DC phenotype changes are analysed within 24 hours of stimulation.

It is also important to remember that monocytes can differentiate into both macrophages and DC. The cytokine environment is important in influencing this outcome (Chapuis et al., 1997; Palucka et al., 1998). For example, both IL-10 and IFN γ can push towards macrophages (Allavena et al., 1998; Delneste et al., 2003), whereas IL-4 is important in enhancing differentiation into DC (Sallusto and Lanzavecchia, 1994). The absence of DC-specific markers makes it difficult to define cells as "non-DC", but the increased expression of CD14 in DC generated in the presence of IFN α may suggest cells tending more towards macrophage properties than those classically generated in the presence of GM-CSF and IL-4 (Mohty et al., 2003). Furthermore, the similarity in signalling between type I and type II IFN receptors, may be consistent with the ability of IFN γ to skew towards macrophages (Allavena et al., 1998; Delneste et al., 2003).

5.3.2.2 Type I IFN effects on immature DC IL-12 secretion

It is critical that the DC system possesses mechanisms to overcome the block in IL-12 secretion that results from HSV-1 infection (fig. 4.11). One mechanism proposed from this chapter is the secretion of type I IFN from HSV-1 infected DC that can prime bystander uninfected DC for IL-12 secretion, in the context of another inflammatory stimulus. However, further studies detailed that type I IFNs are also involved in

inducing maximal secretion of IL-12 by LPS alone (fig. 5.10), further underlying the critical priming role of this family of cytokines.

Although the synergistic effects observed with IFNa in this study are similar to those reported by others (Luft et al., 2002b; Heystek et al., 2003), the regulation of IL-12 secretion by type I IFN remains an area of controversy, as the effects reflect both cell type and species differences. For example, IFNa inhibits LPS induced IL-12 p40 secretion from monocytes (Hermann et al., 1998) and negative regulation of IL-12 secretion by type I IFN has been reported in mice (Cousens et al., 1997). The latter may be explained through type I IFN mediated PI3K activation (Yang et al., 2001), as PI3K can potentiate IL-12 secretion in humans (Re and Strominger, 2001) in contrast to its negative effects in the mouse (Fukao et al., 2002). Other contentious issues relate to the subtypes of IFN and the precise timing of IFN exposure. In contrast to IFN α (fig. 5.7) (Luft et al., 2002b), IFN- β does not have a priming effect on DC IL-12 secretion and may even inhibit it (McRae et al., 2000b; van Seventer et al., 2002). It seems paradoxical that two cytokines that signal through the same receptor may give rise to disparate functional consequences. However, recent data suggest that different type I IFNs bind to different sites on the IFNAR2 chain (Chill et al., 2003). The kinetics of binding to the receptor may be critical in determining the outcome of the cytokine interaction with the cell and the downstream signalling.

It has also been shown that DC generated in the presence of IFN α or IFN β during DC differentiation demonstrate impaired ability to secrete IL-12 (Paquette et al., 1998; Bartholome et al., 1999; Mohty et al., 2003). As the window for IL-12 secretion is narrow (between 8-12 hrs after stimulation), these cells may exhibit signs of "exhaustion" (Langenkamp et al., 2000). It is also possible that the absence of IL-4 in these cultures may deprive DC of the endogenous adjuvant activity of this cytokine for IL-12 secretion (Ebner et al., 2001). Indeed, in the mouse, DC do not secrete IL-12 p70 in response to a single stimulus (Schulz et al., 2000). The requirement to derive human DC from precursors in the presence of IL-4 may reduce the threshold for the secretion of IL-12. The species differences may also be intrinsic to the regulation of the transcription of the cytokine itself. For example, PI3K, may alter the threshold for secretion in humans compared to mice (Re and Strominger, 2001; Fukao et al., 2002).

In addition, the protein products of the IL-12 family may differ between the two species. Although both secrete excess IL-12 p40 compared to functional IL-12 p70 heterodimer, p40 from mouse cells can homodimerise and antagonise IL-12 p70 activity, whereas non-transfected human cells do not produce the homodimer, despite the presence of equivalently large excess quantities of IL-12 p40 secreted (Carra et al., 2000).

5.3.3 Role of type I IFN in TLR induced changes to DC

Given that previous studies have demonstrated that human DC stimulation by TLR ligands induces the secretion of type I IFN (Cella et al., 1999b), and that the secretion of type I IFN can affect the functional outcome of DC-pathogen interaction (Trottein et al., 2004), an immediate extension of the work was to assess the relative contribution of autocrine type I IFN activity on the changes in surface phenotype and cytokine secretion that occur during DC maturation by TLR ligands.

5.3.3.1 Type I IFN and phenotype changes to TLR3 and TLR4

All TLRs signal downstream via the Toll-interleukin 1 receptor-resistance (TIR) domain of the cytoplasmic tail of the receptor. This occurs through an adaptor protein, of which myeloid differentiation factor 88 (MyD88) is a prototype. However, LPS stimulation of DC deficient in MyD88 does not prevent phenotype changes (Kaisho et al., 2001). Another adaptor molecule that can associate with the TIR domain of TLR4 is TIR domain-containing adaptor protein (TIRAP), but this protein is also not essential in mediating LPS induced phenotypic maturation of DC (Yamamoto et al., 2002). However, absence of a third TLR adaptor molecule, TIR-domain-containing adaptor inducing interferon-beta (TRIF), does prevent this (Hoebe et al., 2003b). Noticeably, TRIF is important for LPS induced phosphorylation of IRF-3 (Fitzgerald et al., 2003), which in turn activates transcription of the IFN β gene (Sakaguchi et al., 2003). Therefore, this raises the possibility that autocrine type I IFN could mediate some of the maturation induced by LPS.

This question has been studied in the murine system using mice lacking the type I IFN receptor. The data show that autocrine type I IFN is required to allow maximal upregulation of CD40 and CD86 in DC in response to LPS (Hoshino et al., 2002; Honda et al., 2003). It is interesting that macrophages from IFNAR KO mice are completely unable to upregulate CD86 and MHC class II in response to LPS (Hoebe et al., 2003b), in contrast to more moderate effects on DC (Hoshino et al., 2002; Honda et al., 2003b), in contrast to more moderate effects on DC (Hoshino et al., 2002; Honda et al., 2003b). This cell type specific effects may be related to a variable contribution of the TRIF pathway to TLR4 induced gene expression in different leucocyte populations. A note of caution is required when comparing the surface phenotype changes between DC and macrophages because the primary output of macrophages in response to LPS may not be the upregulation of molecules involved in antigen presentation, but rather effector mechanisms involved in pathogen clearance, such as nitric oxide production (Gao et al., 1998).

There is little data available regarding the relative contribution of type I IFN in LPS mediated maturation of human DC. The data in this chapter suggests that type I IFN is not required in LPS induced upregulation of CD86 (fig. 5.9) and HLA-DR (data not shown), in agreement with another study (Gauzzi et al., 2002). The difference from mice may be related to the differential regulation of TLR4 between the two species. LPS stimulation of mouse macrophages downregulates TLR4 expression, whereas LPS stimulation of human macrophages results in upregulation (Matsuguchi et al., 2000; Poltorak et al., 1998a; Muzio et al., 2000). The reduced signalling through TLR4 may result in obligate amplification by a second stimulus in mouse cells to result in maximal upregulation of surface molecules. However, differences in DC models, and the culture conditions used might also contribute to the discrepancy, possibly priming human DC for stimulation at lower levels of LPS than mouse DC.

In contrast to LPS, dsRNA stimulation of DC is partly dependent on the autocrine activity of type I IFN. This is consistent with previous data in both DC and macrophages in mice (Honda et al., 2003; Hoebe et al., 2003b), although the contribution of IFN in mice is greater. The dependency on IFN is consistent with the recruitment of TRIF to TLR3 (Oshiumi et al., 2003), and the subsequent activation of IRF-3 and transcription of the IFN β gene (Sakaguchi et al., 2003). However, the

requirement for IFN in the TLR3 response may also relate to the role of anther adaptor molecule, TRAM. This molecule can interact with TLR4 but not TLR3, possibly increasing the relative strength of the direct stimulus via TLR4 relative to TLR3 (Fitzgerald et al., 2003; Yamamoto et al., 2003).

Mechanistically, the role of type I IFN in Poly(I:C) induced DC maturation was peculiar, in that removing the autocrine activity of IFN did not reduce CD86 upregulation uniformly. Rather, two discrete populations of cells were observed, one that expressed CD86 as high as untreated cells, and another expressing lower CD86 (fig. 5.9). These data are not directly comparable to that from mice where the original DC population is often not as uniform as that of human MDDC (Honda et al., 2003). Nevertheless, the continuous dose-response of DC phenotype to IFN (fig. 5.5) excluded the possibility that a critical concentration of type I IFN was required to trigger sufficient signalling to upregulate CD86 or HLA-DR. The positive regulation of TLR3 expression by type I IFN (Miettinen et al., 2001; Heinz et al., 2003; Tanabe et al., 2003), suggests that abolishing the activity of IFN may prevent sufficiently high expression of TLR3 to permit a maximal response to Poly(I:C). This regulation of receptor expression may be particularly relevant to TLR3 because, in the resting state, very low levels of this receptor are expressed on the DC surface (Matsumoto et al., 2003). This may be an important control mechanism to ensure that immune responses in the presence of dsRNA are generated only in the context of an ensuing antiviral response, where the local concentration of type I IFN and extracellular dsRNA may be high. Furthermore, in the presence of a high concentration of dsRNA but low type I IFN, as may occur towards the end of a viral infection when multiple CTL lysis events release large quantities of dsRNA, the absence of significant viral replication would reduce the IFN released. This model proposes that the DC is able to curb the cellular response in this scenario by not upregulating TLR3, reducing its sensitivity to the dsRNA.

5.3.3.2 Role of autocrine type I IFN in LPS induced cytokine secretion

The secretion of several proinflammatory cytokines occurs during DC maturation by TLR ligands. Murine macrophage and DC secretion or mRNA induction of $TNF\alpha$ and

198

IL-12 to LPS are not dependent on autocrine type I IFN activity (Hoebe et al., 2003b; Honda et al., 2003; Trottein et al., 2004). Similarly, the absence of IRF-3 does not affect the mRNA levels of TNFa in response to LPS (Sakaguchi et al., 2003). Interestingly, the secretion of both of these cytokines was dependent on TRIF, demonstrating that in DC maturation, TRIF is not required solely to induce the secretion of IFN β (Hoebe et al., 2003a). In contrast, the secretion of both TNF α and IL-12 by DC in this thesis was reduced when type I IFN activity was neutralised (fig. 5.10), despite the independence from autocrine type I IFN activity on changes in surface phenotype in response to LPS stimulation (figs. 5.9 and 5.10). The early downregulation of IFNAR after LPS stimulation suggests that the secretion of type I IFN occurred within 5 hours of LPS stimulation (Gauzzi et al., 2002). The one other study that has investigated this effect highlighted not only the complexity of signalling pathways downstream of TLR4, but also the cell specific regulation of these pathways. Transfection of human macrophages with adaptor molecule-expressing adenoviruses demonstrated that TIRAP is not required for the secretion of TNF α by macrophages in response to LPS. However, TIRAP was essential for LPS induced transcription of the IFN β gene, independently of the phosphorylation of IRF-3 (Andreakos et al., 2004). Not only does this imply that IFN β transcription does not result directly form IRF-3 activation, it also indicates that the secretion of $TNF\alpha$ by macrophages occurs independent of autocrine type I IFN activity, in contrast to the requirement in DC (fig. 5.10).

5.3.4 Signalling consequences of type I IFN on DC

5.3.4.1 Regulation of IL-12 and TNF α secretion

The priming effect of type I IFN for maximal IL-12 secretion may be explained by the signalling consequences of IFNAR ligation on DC. As IFN α alone is unable to induce IL-12 secretion, it is likely that the activation of p38 and NF-kB induced by this cytokine is insufficient to induce transcription of either subunit. Conversely, LPS activates these pathways sufficiently. Nevertheless, the synergistic secretion of this cytokine after co-stimulation with both IFN α and LPS suggests that IFNAR-derived signals play a role. IFNAR ligation induces the activation of NF-kB (fig. 5.8) (Yang et

al., 2000), as well as the generation of IRF-1, which can bind IRFE on gene promoters (see section 5.3.2). This is significant because both IL-12 subunit promoters contain NF-kB binding sites (Murphy et al., 1995; Grumont et al., 2001) and IRFEs, binding to which promotes gene transcription (Kollet et al., 2001; Maruyama et al., 2003; Liu et al., 2003).' The activation of p38 MAPK by IFN α (fig. 5.8) is unlikely to play a role during co-stimulation, as LPS alone activates p38 with the same kinetics as IFN α (data not shown) and to maximal levels, as judged by the lack of further activation when co-stimulated with IFN α (fig. 5.8).

However, autocrine type I IFN signalling after LPS stimulation alone may affect secretion of IL-12 differently. IFNa may suppress activation of the ERK pathway (Romerio et al., 2000), which may in turn promote IL-12 transcription(Puig-Kroger et al., 2001). Alternatively, the activation of p38 MAPK following LPS stimulation will have subsided, and thus IFN α may reactivate this pathway at the time of feedback (fig. 5.8). As type I IFN induced p38 MAPK activation can result in histone phosphorylation (Li et al., 2004), this process may mark the IL-12 promoters for increased IFNAR-derived NF-kB and IRF-1 recruitment (Saccani et al., 2002), because in both IL-12 subunit promoters, transcription factor access to at least one NF-kB binding site and/or one IRFE is regulated by chromatin remodelling via histone phosphorylation or acetylation (Goriely et al., 2003; Liu et al., 2003; Weinmann et al., 1999; Maruyama et al., 2003). However, it is also possible that LPS induced p38 activation causes a permanent change in histone structure that permits greater direct access to NF-kB and IRF-1 induced by autocrine type I IFN. Indeed, cells can demonstrate such temporal memory, as IFNy can prime macrophages for enhanced IL-12 secretion 16 hours prior to LPS exposure (Liu et al., 2003).

The signalling regulation of TNF α secretion by autocrine type I IFN may occur through similar mechanisms to those for IL-12, as TNF α transcription is positively regulated by p38 MAPK and NF-kB (Hoffmeyer et al., 1999; Zhu et al., 2000). One notable difference is that NF-kB recruitment to the TNF α promoter is not regulated by histone phosphorylation as for IL-12, although the study only analysed the role of one histone (Saccani et al., 2002). A model whereby type I IFN provides a secondary stimulus that promotes NF-kB recruitment to target promoter sites, either by inducing changes in nucleosome conformation or by initiating greater NF-kB and IRF-1 activation, is consistent with delayed targeting of NF-kB to some promoters. Whereas some genes are targeted within 30 minutes, for others this only occurs over 2 hours after LPS stimulation (Saccani et al., 2001). Murine BMDC treated with Poly(I:C) also demonstrate two waves of nuclear NF-kB recruitment. The second of these occurred 4-6 hours after initial stimulation, and was absent in mice lacking IFNAR, implicating type I IFN in this effect (Honda et al., 2003).

To validate this model, it will be important to demonstrate that LPS stimulation does indeed induce a second wave of NF-kB, IRF-1 and/or p38 activation. Furthermore, IFNAR neutralisation experiments should ablate this secondary wave of signalling, and thus reduce IL-12 and TNF α secretion. Equally, addition of p38 or NF-kB inhibitors after the first wave of activation of these pathways by LPS (i.e. after 2 hours) should also reduce the secretion of IL-12 and TNF α .

Intriguingly, type I IFN may also promote IL-12 and TNF α secretion through posttranscriptional regulation. The activation of p38 MAPK stabilises TNF α mRNA (Dean et al., 1999; Wang et al., 1999b; Brook et al., 2000; Neininger et al., 2002), and IL-12 secretion may also be regulated at the level of mRNA stability (Du and Sriram, 1998). Analysing the expression of cytokine mRNA after LPS stimulation in the presence of neutralising anti-IFNAR2 antibodies may determine whether this is a significant regulation mechanism in this system.

5.3.4.2 Differential regulation of CD86 and IL-12

The regulation of the expression of CD86 contrasted that of IL-12. IFN α was able to upregulate CD86 independent of the activity of p38 MAPK, suggesting that weaker signalling activation is required to induce transcription of this protein compared to IL-12. Furthermore, LPS induced upregulation was only partly dependent on p38 activity. The p38 independent signalling is likely to be mediated by NF-kB (Li et al., 1999;

Yoshimura et al., 2001). It is currently not known whether CD86 expression is predominantly regulated at the transcriptional level and whether chromatin remodelling is important. If so, it would suggest that the CD86 promoter is constitutively more accessible to NF-kB access than either of the IL-12 subunit promoters.

The lower requirement for signalling activation for CD86 would also explain why LPS stimulation alone is sufficient for maximal upregulation, independent of signals from autocrine type I IFN. The differential signalling requirement between CD86 and IL-12 spectra two thresholds of DC activation. Physiologically, this control may be important to ensure that cytokine secretion occurs only at the appropriate time and place, and under the correct stimulation (e.g. during contact with T cells).

5.4 Conclusions

This chapter aimed to investigate some of the mechanisms that DC use to bypass the HSV-1 induced inhibition of antigen presenting function described in chapter 4. Secretion and paracrine activity of type I IFN early after infection fits these criteria. Bystander maturation of neighbouring DC primes these cells for increased IL-12 secretion and induces sub-maximal activation. In this way, secretion of type I IFN in peripheral infection sites can serve two purposes. In a lesion where there is incapacity of DC in the core of the site to stimulate T cells, such as for HSV-1, type I IFN can skew the neighbouring DC that take up the infectious antigen to cross-prime instead of cross-tolerise. This is critical for the generation of an effective immune response that will resolve the infection.

Type I IFN also serves as an important extracellular second messenger to gauge the threat to the host. Although the DC system has evolved to recognise a large number of conserved pathogen molecules, it is important to tailor the amplitude of the immune response to the severity of the infection, in order to avoid undesirable immunopathology. Type I IFN secretion, also in response to non-viral stimuli, can serve such a function by amplifying the DC response where appropriate, but equally important, by not activating indiscriminately all neighbouring DC maximally. In the

event of the spread of infection, bystander cells are primed for stimulation by pathogen structures and can then undergo maximal maturation and secrete larger quantities of pro-inflammatory cytokines. This could occur both in the periphery and in the secondary lymphoid organs. The intracellular signals that regulate these properties may illustrate how DC have evolved to integrate and appropriately respond to endogenous and exogenous danger signals in their microenvironment. Mechanisms of HSV-1 activation and inhibition of DC

Chapter 6

1

Mechanisms of HSV-1 activation and inhibition of DC

6.1 Introduction

Chapter 4 demonstrated that DC infected with HSV-1 expressed higher levels of CD86 and HLA DR molecules compared to uninfected immature DC. Irrespective of the subsequent inhibition in function and response to maturation stimuli, this finding indicates that the viral infection is a maturation stimulus for DC. The viral infection could result in the secretion of factors that can activate infected cells in an autocrine manner. The obvious candidate is type I IFN detailed in chapter 5, as this was the only significant stimulatory ligand in the supernatant of infected DC cultures. MV induced maturation of DC is dependent on autocrine type I IFN activity (Dubois et al., 2001). Similarly, the autocrine action of TNF α after adenovirus (AdV) infection of DC is responsible for the maturation of the cells (Trevejo et al., 2001; Philpott et al., 2004). Alternatively, virus products can stimulate DC directly and induce maturation, independent of the autocrine activity of secreted cytokines (Lopez et al., 2003). The viral ligands can either take the form of virion components that activate cells upon infection (Fantuzzi et al., 1996), or alternatively they can be the products of the viral replication cycle (e.g. dsRNA intermediates of viral replication (Jacobs and Langland, 1996)). The two mechanisms may be linked, as exemplified by AdV penton base that is required to induce TNF α secretion and subsequent maturation of DC (Philpott et al., 2004). However, for the most part, viral factors responsible for inducing DC maturation remain ill defined.

Although intracellular delivery of virion components may trigger a DC response, an important interface for virus-cell interaction is the cell surface and receptors that can bind to viral ligands. The expression of entry receptors and their downstream signalling pathways may be important in such a scenario. Alternatively, viruses may possess conserved structures that are recognised by PRRs, as detailed in chapter 1, which may induce intracellular signals. The receptors could be expressed on the cell surface to recognise viral envelope structures (Bieback et al., 2002; Compton et al., 2003) or alternatively they may be expressed in endosomal structures to detect the viral genome during uncoating. Indeed, ssRNA, dsRNA and hypomethylated CpG DNA are all recognised by, TLR7 and TLR8, TLR3 and TLR9 respectively (Diebold et al., 2004; Heil et al., 2004; Alexopoulou et al., 2001; Hemmi et al., 2000), receptors that

205

are expressed in endosomes (Heil et al., 2003; Matsumoto et al., 2003; Ahmad-Nejad et al., 2002)

In the context of HSV-1, viral binding to cell surface can elicit signalling that will induce activation of pathways with a functional output (Paludan et al., 2001; Cheshenko et al., 2003). Whether these events are sufficient to induce the maturation of DC has not yet been investigated. Alternatively, HSV-1 induction of IFN responses may result in DC maturation directly (Lopez et al., 2003) or through the autocrine activity of secreted type I IFN (chapter 5 + (Dubois et al., 2001)). Furthermore, HSV-1 replication in DC (fig. 3.8) may also generate sufficient dsRNA molecules (Jacquemont and Roizman, 1975) that can ligate TLR3 or activate PKR, and induce DC maturation or type I IFN secretion (Alexopoulou et al., 2001; Diebold et al., 2003).

This chapter will aim to determine the HSV-1 ligands that elicit activation of DC and induce signalling changes, through the use of progressively inactivated or neutralised viruses. This approach will allow not only the investigation of the activatory components of the virus, but also help identify some of the components responsible for the inhibitory effects on DC function that were detailed in chapter 4. The emphasis will be on viral envelope components, specifically gD, and how they interact with receptors on the surface of DC. The mechanistic understanding of this host-pathogen interaction will be discussed in the context of other viral infections and how it may impinge on disease pathogenesis.

6.2 Results

6.2.1 Type I IFN independent HSV-1 maturation of DC

It was possible that the mature phenotype observed by infected DC (fig 4.12) was dependent on the autocrine activity of type I IFN secreted (fig 5.3). Therefore, DC were infected in the presence of anti-IFNAR2, and the upregulation of CD86 was assessed. This molecule was elevated by HSV-1 infection most consistently. IFN α stimulation of DC was used as a positive control to confirm the neutralisation of IFN by this antibody. In the presence of the same concentration of anti-IFNAR2 that neutralised IFN α and the IFN activity in supHSV-DC in fig. 5.6, HSV-1 induced upregulation of CD86 was not affected (fig. 6.1). These data demonstrate that HSV-1 stimulated activation of DC directly, independent of the autocrine activity of secreted type I IFN after infection.

6.2.2 Role of viral gene expression in DC activation and inhibition

To determine the role of viral gene products in the activation of DC, HSV-1 virions were inactivated by ultraviolet (UV) light (UV-HSV). UV light disrupts viral gene expression by generating dimers between adjacent pyrimidines (Tornaletti and Pfeifer, 1996). However, the protein structure of the virions is left intact, rendering UV-HSV theoretically infective but unable to express any viral genes. Both of these properties were assessed before functional experiments were attempted.

6.2.2.1 Efficiency of UV inactivation of HSV-1

Several factors, including the viral size, structure and genome type may determine the length of time required to inactivate different viruses by UV-light (Kwan A, personal communication). The efficiency of HSV-1 inactivation was assessed in order to find the optimum time of UV light exposure. UV-inactivation (i.e. loss of viral genome integrity) was assessed by analysing the expression of GFP after infection. The time course of UV-light exposure is shown in chapter 3 (fig. 3.6C). In agreement with the time of exposure used in other studies (Paludan and Mogensen, 2001), total viral

inactivation was achieved after 20 min of UV-light exposure. This was the time used for the rest of the studies.

6.2.2.2 Binding of UV-inactivated HSV-1

It was possible that the UV light exposure could damage the ability of UV-HSV to infect DC. Therefore, the binding capacity of UV-HSV to the DC surface was compared to non-treated HSV (WT-HSV). The DC surface was stained for gD 1 hour post infection to detect bound particles and envelope gD that had fused with the DC surface membrane. Fig. 6.2A shows that UV-HSV and WT-HSV bound DC with equivalent efficiency. It remained possible that any soluble gD in the viral preparation might also have bound to DC, and generated a false positive. However, as HSV-1 entry is mediated by gB/gC attachment to heparan sulphate (HS), followed by binding of gD to entry receptors (Spear et al., 2000), it was possible to prevent the attachment step by preincubaitng HSV virions with heparin prior to infection (Herold et al., 1994). As expected, this process reduced the infection efficiency dramatically (fig. 6.2B). Therefore, to exclude the presence of significant soluble gD being the source of the false positive result in fig. 6.2A, heparin-coated HSV-1 was added to DC, and the cells then stained for gD on the surface (fig. 6.2C). This experiment showed that in the presence of heparin, gD staining was abrogated. Thus, no significant soluble gD was present in the viral prep, and the gD staining in fig 6.2A represented the degree of virion-derived gD on DC.

6.2.2.3 Infectivity of UV-inactivated HSV

The experiments in section 6.2.2.2 only demonstrated that UV-HSV could bind to DC as efficiently as WT-HSV. It was also important to assess that UV treatment did not damage their ability to enter DC. VP16 is a tegument protein involved in maintaining the structure of virions and is also a viral gene transactivator (Wysocka and Herr, 2003). Therefore, intracellular staining of DC 2 hours post-infection for the presence of VP16 allowed an estimate of the ability of HSV virions to bind and enter DC (Aubert et al., 1999). WT-HSV infection revealed a small but consistent increase in VP16 expression in DC. The VP16 expression profile of UV-HSV infected DC 2 hours

post infection was near identical to that of WT-HSV, confirming that UV-HSV was able to infect DC with equal efficiency as WT-HSV (fig. 6.3A). VP16 staining could not be correlated with GFP expression at 2hr post-infection, as the expression of GFP was low or undetectable (fig. 3.6). However, UV-HSV infected DC did not express any GFP[']16 hours after infection, confirming the efficiency of inactivation by the UV-treatment protocol (fig. 6.3B).

The data in sections 6.2.2.2 and 6.2.2.3 confirm that UV treatment of HSV-1 prevented the expression of viral genes without impairing the ability of the virus to bind and enter DC. In this way, the effects on DC of viral binding and entry, but not gene expression, could be assessed.

6.2.3 Functional effects of UV-HSV infection of DC

6.2.3.1 T cell proliferation induced by UV-HSV infected DC

Fig. 6.4A demonstrates that UV-HSV infected DC can stimulate allogeneic T cell proliferation equally well as uninfected DC, in contrast to the strong inhibition seen with WT-HSV infected DC. Therefore, HSV-1 gene products are responsible for the majority of the inhibition in DC ability to stimulate T cell proliferation.

6.2.3.2 Cell viability of UV-HSV infected DC

Another striking effect of WT-HSV infection of DC was the loss of cell viability 40 hours after infection (fig 4.8). Using the MTT reduction assay again, fig. 6.4B demonstrates that UV-HSV infected DC maintained cell viability up to 40 hours after infection, in contrast to the effects of WT-HSV. Therefore, this suggests that viral gene products were responsible for the loss of DC viability after infection.

6.2.3.3 Morphology of UV-HSV infected DC

One of the earliest changes seen with WT-HSV infection of FN-adherent DC was the loss of dendritic morphology (fig. 4.5). This was not evident after UV-HSV infection, suggesting that viral gene products were responsible for the loss of dendritic shape

(fig. 6.4C). In contrast to treatment with LPS, UV-HSV infection did not increase the number of adherent dendritic shaped DC, and did not increase the length of the dendrites.

Another noticeable effect of WT-HSV infection was resistance to further DC stimulation by LPS. Again, this was most marked in effects on morphology. DC rounded up by WT-HSV were unable to regain a dendritic shape after LPS stimulation (fig 4.5). By contrast, UV-HSV infected DC, when further stimulated with LPS, increased DC adherence to FN and dendritic shape, in a manner indistinguishable from the effects of adding LPS alone to uninfected DC (fig. 6.4C). Therefore, viral gene products were responsible not only for the loss of dendritic shape, but also for the refractory morphological response to LPS.

6.2.3.4 Surface phenotype of UV-HSV infected DC

UV-HSV infection of DC resulted in similar changes in surface phenotype to those seen with WT-HSV, upregulating CD86 (fig. 6.5) and HLA-DR (data not shown) expression. However, in contrast to WT-HSV, LPS stimulation of UV-HSV infected DC resulted in further upregulation of CD86 and MHC class I (fig. 6.5).

In summary, section 6.2.3 showed that HSV-1 gene expression is responsible for the deleterious effects of HSV-1 infection on DC, rendering DC morphology and phenotype refractory to further stimulation by LPS, and also inducing cell death and preventing efficient T cell stimulation. However, the changes in phenotype also demonstrate that viral receptor binding and/or entry into DC is a maturation stimulus and this event will be dissected further below.



Figure 6.1 Role of autocrine type I IFN in upregulation of CD86 by HSV-1.

DC were infected with HSV-1 at MOI of 1 or treated with 1000 IU/ml IFN α in the presence of anti-IFNAR2 mAb or an isotype control antibody, and expression of CD86 assessed. WT-HSV data gated on GFP+ve DC. Representative of three independent experiments.



Figure 6.2 Assessment of gD on UV-HSV and WT-HSV binding to DC surface. DC were infected with UV-HSV or WT-HSV for 1 hour and analysed for expression of gD. (B) WT-HSV was incubated with increasing concentration of heparin and used to infect DC at MOI of 1. 16 hours after infection, the percentage of DC expressing GFP was assessed. (C) UV-HSV was incubated in the presence or absence of 12.5 μ g/ml of heparin before infecting DC at MOI of 1 for 1 hour, and the surface expression of gD analysed. All figures are representative of three independent experiments.



Figure 6.3 Assessment of infectivity and inactivation of UV-HSV. (A) DC were infected with UV-HSV or WT-HSV (MOI =1) for 2 hours and stained for intracellular HSV-1 VP16. (B) DC infected with UV-HSV or WT-HSV (MOI =1) were analysed for GFP expression 16 hours after infection. All figures are representative of at least three independent experiments.


Figure 6.4 Effects of viral gene expression on HSV-1 mediated changes in DC function, viability, and morphology.

(A) DC infected with UV-HSV or WT-HSV (MOI=1) were used to stimulate allogeneic T cell proliferation. Data are mean ³H-thymidine incorporation of three independent experiments. Error bars represent SEM. * denotes p<0.01 between WT-HSV and either UV-HSV or uninfected DC. (B) Viability of DC infected with UV-HSV or WT-HSV was analysed by MTT reduction assay 16 and 40 hours after infection. Data shown as mean optical density of four independent experiments. Error bars represent SEM. * denotes p<0.01. (C) DC adherent to fibronectin-coated glass coverslips were infected with UV-HSV or WT-HSV in the presence or absence of 100 ng/ml LPS. Morphology of DC was assessed 8 hours after infection. Phase contrast and GFP fluorescence displayed in separate panels for WT-HSV. Representative fields of at lease three independent experiments.



Figure 6.5 Effects of viral gene expression on HSV-1 mediated changes in DC phenotype.

DC infected with UV-HSV or WT-HSV for 16 hours in the presence or absence of 100 ng/ml LPS were analysed for expression of surface molecules. Numbers on dot plot represent MFI. Representative of three independent experiments.

6.2.4 Role of HSV binding to DC surface

The data in section 6.2.3 isolated the DC activation event as occurring prior to gene expression. This could have resulted either from virus binding to the DC surface, or from post entry events. To dissect these two possibilities, WT-HSV particles were modified further by formaldehyde fixation (FIX-HSV). This treatment cross-links the envelope glycoproteins, rendering them unable to fuse with the cell membrane and therefore non-infective. This was confirmed by the lack of GFP expression after overnight culture (fig. 6.6A). However, the virus' ability to bind to the DC surface remained intact, as determined by the presence of HSV-1 gD on the surface of DC 1 hour after infection (fig. 6.6B).

DC exposed to FIX-HSV upregulated the surface molecules, CD86 and HLA-DR similarly to WT-HSV (fig. 6.7A and 6.9+10+11). Furthermore, like UV-HSV, FIX-HSV infected DC were not refractory to further LPS stimulation, again in contrast to the effects of WT-HSV, as CD86 expression could be further elevated by simultaneous pulsing with LPS (fig. 6.7A).

Although the data in fig. 6.7A suggest that viral interaction with the DC surface is sufficient to induce DC maturation, the autocrine activity of type I IFN cannot be ruled out because the inhibitory effects of viral genes on type I IFN signalling are absent during FIX-HSV exposure (Yokota et al., 2001; Mossman et al., 2001; Yokota et al., 2004; Chee and Roizman, 2004). Therefore, DC 'infected' with FIX-HSV were treated with anti-IFNAR2 in the same way as in fig. 6.1, in order to determine the role of autocrine type I IFN in FIX-HSV induced DC maturation. Interestingly, the upregulation of CD86 by FIX-HSV was abrogated to a small degree by neutralising the activity of type I IFN (fig. 6.7B). These data suggest that type I IFN is secreted by DC stimulated with FIX-HSV, but that this cytokine is not wholly responsible for the mature phenotype of infected DC and that the cell receives a significant signal by virus binding to its surface.

In order to exclude that the changes in DC phenotype in fig. 6.7 occurred as a result of contaminants in the viral treatment protocol, DC were exposed to equivalent

concentrations of fixative and neutralising agent (formaldehyde and sodium bisulphite) in the absence of virus. This did not result in changes in CD86 expression (fig. 6.8 - upper row), in contrast to upregulation observed in the presence of WT-HSV (RPMI +/- sodium bisulphite groups) or FIX-HSV (formaldehyde + sodium bisulphite group) (fig. $6.8^{'}$ lower row).

6.2.5 Signalling consequences of HSV-1 binding to DC surface

6.2.5.1 Activation of signalling pathways by HSV-1

If HSV-1 binding to the cell surface alone was responsible for the increased expression of CD86 and HLA-DR, it was likely that this event was inducing the activation of signalling pathways inside the DC. As detailed in chapter 5, two effector pathways involved in the upregulation of CD86 are NF-kB and p38 MAPK. Although, HSV-1 infection can initiate NF-kB and p38 MAPK activation in other cell types (Patel et al., 1998; Zachos et al., 1999), this has not been demonstrated to result from early viral interaction with receptors on the cell surface.

Western blotting showed that WT-HSV infection of DC induced significant IkB α degradation within 30 minutes of infection (fig. 6.9A and data not shown). WT-HSV also induced significant phosphorylation of p38 after infection (fig. 6.9A) and this was quantified relative to total p38 by densitometry (fig. 6.9B). Interestingly, LPS mediated greater IkB α degradation and p38 activation than WT-HSV, correlating with the degree of maturation seen with LPS relative to the (sub-maximum) maturation observed following HSV-1 infection (fig. 4.12).

6.2.5.2 Role of p38 MAPK in HSV-1 induced DC activation

It was important to determine the significance of the activation of these proinflammatory pathways on DC activation by HSV-1. This can be carried out efficiently in vitro by preventing activation of the pathways described above and subsequently infecting DC with HSV-1. To study specifically the role of activation of these pathways emanating from the cell surface interaction with HSV-1, DC were

stimulated only with FIX-HSV in these assays. Other studies have shown that p38 and NF-kB can be activated at later times after infection with HSV-1 (Patel et al., 1998; Zachos et al., 1999), which would be difficult to distinguish from the effects of DC receptor ligation after infection.

Unfortunately, no inhibitor of NF-kB activity was available for these studies. On the other hand, the inhibitor SB203580, which prevents the kinase activity of phosphorylated p38, was used (Tong et al., 1997). DC treated with 100 ng/ml LPS were used as a positive control in these assays to ensure that the inhibitor prevented p38 MAPK activity sufficiently to observe effects on DC phenotype. Fig. 6.9C shows that 3µM SB203580 was able to partly prevent the upregulation of CD86 and HLA-DR following LPS stimulation (same data as in fig. 5.8). In parallel cultures, the upregulation of expression of these molecules on DC after FIX-HSV exposure was only prevented to a small degree. Therefore, these experiments demonstrate that p38 MAPK activation by viral interaction with DC surface receptors plays a minor role in the acquisition of a more activated phenotype seen with these DC. The implication and technical limitations of this experiment are considered in more detail in the discussion in section 6.3.3.3.







Figure 6.7 Surface phenotype changes on DC exposed to FIX-HSV. (A) DC infected with FIX-HSV or WT-HSV for 16 hours in the presence or absence of (A) 100 ng/ml LPS or (B) anti-IFNAR-2 antibody were analysed for expression of CD86. WT-HSV data gated on GFP+ve DC.



Figure 6.8 Effects of virus inactivation protocol on DC CD86 expression.

RPMI or HSV-1 were treated with various combination of RPMI or 3.8% formaldehyde, and HBSS or 17.5% sodium bisulphite prior to 'infection' of DC, and cultured for further 16 hours. Expression of CD86 (main panel) and GFP (insert) were assessed. All figures are representative of at least three independent experiments.

221



Figure 6.9 Signalling consequences of HSV-1 infection of DC.

(A) MDDC were infected with WT-HSV (MOI = 3) or stimulated with 100 ng/ml LPS for 60 minutes. Cell lysates were harvested and assayed by Western blot for IkBa, phospho-p38 and total p38. Representative of three independent experiments. (B) Data in (A) presented as ratio between band intensity of phospho- and total-p38 MAPK. Results are mean of three independent experiments. Error bars represent SEM. (C) DC were cultured in the presence or absence of 3µM SB203580 for 24 hours prior to stimulation with FIX-HSV or 100 ng/ml LPS. DC were cultured for a further 16 hours in the presence of SB203580 prior to assessment of CD86 and HLA-DR exposure. Results are mean of three independent experiments. Error bars represent SEM.

6.2.6 Receptors involved in HSV-1 induced activation of DC

Section 6.2.5 demonstrated that HSV-1 is able to ligate surface receptors on DC and induce the activation of p38 and NF-kB. An attempt to identify the receptors responsible was made using antibody neutralisation of candidate molecules. These included the known entry receptors, HVEM and nectin-1 (Spear et al., 2000) and TLR2, because HSV-1 may signal via this receptor (Kurt-Jones et al., 2004). HVEM activates NF-kB on some cells (Marsters et al., 1997), and TLR2 ligation can induce maturation of DC via NF-kB and p38 activation (Michelsen et al., 2001; Hertz et al., 2001; Re and Strominger, 2001). Fig. 3.4 (and data displayed again in figs. 6.11+12) documented that the two entry receptors are expressed on DC. The optimum amount of antibody used to neutralise the receptors was determined by utilising a saturating amount of antibody for surface staining. This approach was applied also for TLR2 and its expression is displayed in fig. 6.10A.

A caveat about such neutralisation experiments is that antibodies may not bind to ligand-binding sites, and thus may not prevent signal transduction. With regards to TLR2, this ambiguity was minimised by using a positive control TLR2 ligand, PGN, the activity of which can be partially neutralised in DC by the mAb used in these studies (Uehori et al., 2003). For the entry receptors, neutralisation of virus infection served as a useful indicator of prevention of viral ligand binding, the step that could be responsible for cell surface-derived signalling.

6.2.6.1 Role of TLR2 in DC activation by HSV-1

TLR2 is not known to be an entry receptor for HSV-1. This was confirmed in fig. 6.10B, which showed that this receptor is not essential for infection of DC, as determined by antibody blocking and subsequent GFP expression. When DCs were coated with anti-TLR2 antibody prior to HSV-1 infection, the upregulation of CD86 or HLA-DR by either FIX-HSV or WT-HSV was not prevented (fig 6.10C). This antibody also did not prevent maturation induced by the positive control PGN, which raises doubts about the ability of the mAb used for these experiments to prevent TLR2

ligand binding and signal transduction (fig. 6.10C). By extension, it is also difficult to conclude convincingly that HSV-1 does not activate DC via TLR2 ligation.

6.2.6.2 Role of nectin-1 in DC activation by HSV-1

There are many antibodies raised to nectin-1, but the clone used for these studies, CK41, can prevent infection of HSV-1 in some cell types (Krummenacher et al., 2000). However, saturating amounts of this antibody did not prevent HSV-1 infection of DC (fig. 6.11B). Furthermore, the antibody to nectin-1 did not prevent upregulation of CD86 or HLA-DR induced by either FIX-HSV or WT-HSV (fig. 6.11C). The saturating amounts of this antibody used in these studies should have ensured that HSV-1 could not interact with this receptor. However, the lack of neutralisation of infection questions the firm conclusion that all interactions with this receptor by the wirus was abrogated. Thus, while it appears unlikely that nectin-1 alone plays a significant role in DC maturation, this cannot be totally excluded by these data.

6.2.6.3 Role of HVEM in DC activation by HSV-1

There are few, if any, mAb to HVEM that have been demonstrated previously to neutralise infection. The "neutralising" one that was tested in this system, was not able to prevent infection (fig. 6.12A). Therefore, a neutralising rabbit polyclonal serum to HVEM was used to prevent HSV-1 infection by this receptor (Montgomery et al., 1996). The advantage of this approach was that the polyclonal nature made it more likely that all potential receptor ligation sites were covered, though the downside would be that any effect would make the binding site on the receptor difficult to map out directly or accurately.

Incubation of DC with anti-HVEM sera prior to HSV-1 exposure did reduce infectivity to a small degree (fig. 6.12B), consistent with previous reports (Salio et al., 1999). However, analogous to nectin-1, HVEM neutralisation did not prevent maturation by either FIX-HSV or WT-HSV (fig. 6.12C). As for nectin-1, the absence of a suitable positive control to check receptor neutralisation, and the inability to prevent infection significantly makes it difficult to exclude this receptor firmly from a role in DC

maturation by HSV-1. However, the partial neutralisation of infection suggested that saturating quantities of antibody were used, which could prevent virus-HVEM interaction and that the infected cells were likely to reflect redundancy in receptor usage (Krummenacher et al., 2004). Thus, the data in fig. 6.12 suggests that ligating HVEM is not essential for maturation of DC by HSV-1.

6.2.7 Receptor internalisation during infection process

The absent or incomplete neutralisation with TLR2, nectin-1 and HVEM antibodies seen in section 6.2.6 could have been due to the neutralised receptors undergoing rapid internalisation in the steady state, consequently exposing a large number of nonneutralised receptors for the virus to bind. The presence of "neutralising" antibodybound receptors on infected cells would suggest that those cells had been infected independently of that receptor and its downstream signalling. Therefore, DC were stained 16 hours after infection with anti-mouse IgG (for TLR2 and nectin-1) or antirabbit IgG (for HVEM), to assess the expression of the receptors on infected DC that had originally been coated with antibody. While the expression of TLR2 and nectin-1 was not detectable at this time (fig. 6.13A), rabbit IgG bound HVEM was still present, both in uninfected and infected DC (fig. 6.13B). This excluded preferential downregulation of this entry receptor after infection, as previously described for some other viruses (Crise et al., 1990; Welstead et al., 2004), but also confirmed that DC had been infected despite the expression of saturating quantities of antibody. As this was a polyclonal antibody with neutralising capacity, this suggested that virus binding to HVEM was not essential in DC maturation by HSV-1.





(A) Expression of TLR2 on DC. (B) DC pre-treated with anti-TLR2 antibody were infected with WT-HSV and GFP expression assessed after 16 hours. (C) DC were pre-treated in the presence or absence of anti-TLR2 antibody and then stimulated or infected with FIX-HSV, WT-HSV or PGN. DC were cultured for further 16 hours, and the expression of HLA-DR and CD86 assessed. All figures are representative of at least three independent experiments.



Figure 6.11 Effects of nectin-1 ligation on DC stimulation by HSV-1.

(A) Expression of nectin-1 on DC. (B) DC pre-treated with anti-nectin-1 antibody were infected with WT-HSV and GFP expression assessed after 16 hours. (C) DC were pre-treated in the presence or absence of anti-nectin-1 antibody and then stimulated or infected with FIX-HSV or WT-HSV. DC were cultured for further 16 hours, and the expression of HLA-DR and CD86 assessed. All figures are representative of at least three independent experiments.



Figure 6.12 Effects of HVEM ligation on DC stimulation by HSV-1.

(A+B) Left panel anti-HVEM mAb, right panel anti-HVEM antisera (A) Expression of HVEM on DC. (B) DC pre-treated with anti-HVEM antibody or antisera were infected with WT-HSV and GFP expression assessed after 16 hours. (C) DC were pre-treated in the presence or absence of anti-HVEM antisera, and then stimulated or infected with FIX-HSV or WT-HSV. DC were cultured for further 16 hours, and the expression of HLA-DR and CD86 assessed. All figures are representative of at least three independent experiments.



DC were pre-treated with anti-TLR2 mAb, anti-Nectin-1 mAb or anti-HVEM antisera, and cultured for 16 hours and then stained for the presence of antibody on the surface using PE-conjugated goat anti-mouse Ab (for TLR2 and Nectin-1) or PE-conjugated goat anti-rabbit Ab (for HVEM). All panels are representative of at least two independent experiments.

6.2.8 Viral ligands involved in DC activation

In light of the difficulty in defining the receptor(s) mediating DC activation, focus shifted towards defining the viral ligands, with the hope that candidate receptors would be easier to define in this way; and, that, perhaps addition of purified viral ligands to DC could provide a more robust system to identify the activatory receptor(s).

Having established that HSV-1 binding to the DC surface induces a degree of DC maturation and the activation of p38 and NF-kB pathways, envelope structures were investigated. HSV-1 entry into cells is mediated by gB and gC attachment to cell surface HS, followed by gD mediated entry via one of the entry receptors (Spear et al., 2000). The most likely ligand was gD, due to its critical role in binding surface receptors or entry into cells and its capacity to activate NF-kB in cells directly (Spear et al., 2000; Medici et al., 2003). Without ready access to viruses not expressing functional gD on their envelope, the role of gD in DC activation was determined by neutralisation of HSV-1 with a mAb to gD (clone LP2).

6.2.8.1 Role of gD in upregulation of CD86

The ability of LP2 to neutralise the infection of DC was assessed by measuring GFP expression after DC "infection" with neutralised virus. Fig. 6.14A shows that GFP expression was totally abrogated if HSV-1 was neutralised by gD. Treating UV-HSV, FIX-HSV and WT-HSV in this way abrogated the stimulus for the upregulation of CD86 (fig. 6.14B). The effect on FIX-HSV confirmed that extracellular binding of gD to a cell surface receptor was a sufficient activating signal to induce partial DC maturation seen following HSV-1 infection. To exclude the possibility that the presence of antibodies on the viral envelope could sterically hinder the interaction of viral ligands with the cell surface, HSV particles were also pre-coated with a non-neutralising anti-gD monoclonal antibody (AP7 clone) prior to addition to DC. As determined by GFP expression, this antibody did not prevent infection to a significant degree, in contrast to LP2 (fig. 6.15A). Furthermore, staining for the presence of mouse IgG on the surface of DC 16 hours after infection demonstrated that DC infected with AP7-coated HSV (i.e. the GFP+ve population), expressed mouse IgG on

the cell surface. This is likely to have originated from non-neutralised viral envelope that fused with the DC membrane.

DC exposed to FIX-HSV coated in AP7 demonstrated an equally mature phenotype as virus treated with an isotype control antibody, in contrast to the neutralising effect of LP2 (fig. 6.15B). Thus figs. 6.14 and 6.15 demonstrate that the ability of neutralising anti-gD antibodies to prevent DC maturation is likely to occur by preventing specific sites on gD from interacting with activatory surface receptors, rather than through non-specific steric effects or through interacting with inhibitory $Fc\gamma R$ on DC.

The neutralisation of gD did not exclude the role of other viral structures in the activation of DC totally. Two of these are gB and gC, which attach HSV-1 virions to HS on the DC surface (Spear et al., 2000). It was possible that gD neutralisation also prevented HS binding and that this interaction was an important activation signal in DC, as suggested for HCMV infection (Song et al., 2001). To assess whether gD-neutralised virions could attach to DC, the binding of these opsonised virions was assessed by flow cytometry, by staining DC 1 hour post infection with a PE-conjugated anti-mouse IgG antibody. As seen in fig. 6.16, significant signal could be detected when neutralised virions were added to DC. To determine the mechanism of their binding, opsonised virions were also incubated with 12.5 μ g/ml of heparin, prior to addition to DC and flow cytometry analysis. Fig. 6.16 clearly shows that the cell surface binding was sensitive to the heparin step, demonstrating that gD neutralisation of HSV-1 permitted virion attachment to HS on the cell surface and that attachment per se is not an activatory signal in DC.

6.2.8.2 Role of gD in type I IFN secretion by DC

Similar to the changes in surface phenotype, gD-neutralisation of WT-HSV with LP2 also abolished DC secretion of type I IFN (Fig 6.17). Equivalent IFN secretion to that seen with WT-HSV was also observed in 5 out of 7 individuals' DC infected with UV-HSV, and in 2 out of 3 individuals' DC exposed to FIX-HSV. Therefore, HSV-DC surface interaction is important for inducing IFN secretion, and viral envelope gD plays an important role in inducing this effect.

6.2.8.3 Role of gD in bystander DC activation

Chapter 5 confirmed that type I IFN was the only secreted factor that could mediate bystander activation of DC. However, the immunostimulatory property of gD described above, combined with the high level of expression of gD on the surface of infected DC (fig. 3.7), made it possible that direct contact between infected and uninfected DC could also have played a role in the maturation of GFP-ve DC (fig. 4.12). Therefore, DC were infected with HSV-1 and then cultured in the presence of anti-gD antibodies (LP2 clone), in order to neutralise the activity of the gD expressed on infected DC. It was important to determine that sufficient antibody was added to neutralise all gD activity, and this was carried out in two ways. Firstly, DCs were stained 16 hours after infection with an anti-mouse IgG antibody, in order to confirm that all cells expressing gD had indeed bound the antibody (fig. 6.18A). The expression of gD in this way was similar to that seen in fig. 3.7, suggesting that all gDexpressing cells were saturated with antibody. Secondly, virus production was determined by viral infectivity assay of the supernatant of these cells. This showed that HSV-1 in the supernatant was neutralised totally, confirming that sufficient unbound antibody in the medium was present to prevent secondary stimulation of GFP-ve DC by gD on HSV-1. In this experimental system, the expression of CD86 and HLA-DR on GFP-ve DC was not reduced when infected DC were cultured in the presence of anti-gD mAb, excluding that bystander uninfected DC were significantly activated by direct contact with gD expressing infected DC (fig. 6.18B).

6.2.9 Direct activation of DC by gD

Finally, to determine whether gD was solely responsible for activating DC, cells were exposed to a recombinant form of gD and the changes in surface phenotype assessed. The gD used in these studies was derived from a baculovirus system, which produces proteins analogous to mammalian-produced forms of gD, as determined by reactivity to various mAb that recognise discontinuous epitopes of gD (Chiang et al., 1994; Rux et al., 1998). The gD used in these studies was a truncated form, gD(285t), missing 21 amino acids from the C-terminal (Rux et al., 1998). Although this deletion does not

alter the structure drastically relative to full length gD, this gD variant has a higher affinity for HVEM and nectin-1 (Rux et al., 1998),(Krummenacher et al., 1998) and may have been more likely to elicit an effect on DC. The data demonstrated a dose dependent increase in the expression of CD86 (fig. 6.19). Although it is difficult to make a direct comparison between the concentration of dimeric gD added in solution and envelope associated glycoprotein, it was noticeable that even 50µg/ml of gD did not upregulate CD86 to the maximal levels observed with LPS stimulation, analogous to the effects of WT-HSV, or more relevantly, FIX-HSV. If the purity of the protein preparation can be confirmed, then these experiments are confirmation that gD alone can activate DC, although not excluding that other envelope glycoproteins may improve the efficiency of this activation or are themselves stimulatory ligands. These issues are discussed in more detail below.



Figure 6.14 Role of gD interaction with DC on CD86 expression.

WT-HSV was neutralised with anti-gD mAb (LP2 clone) and used to infect DC (MOI = 3). DC were cultured for 16 hours and GFP expression assessed. Representative of three independent experiments. (B) UV-HSV, FIX-HSV or WT-HSV were neutralised with anti-gD mAb (LP2 clone) and then used to infect DC for 16 hours. Expression of CD86 was analysed by flow cytometry. WT-HSV data gated on GFP+ve DC. Representative of three independent experiments.



Figure 6.15 Specificity of gD interaction with DC

(A) DC infected with WT-HSV exposed to neutralising (LP2 clone) or nonneutralising (AP7 clone) anti-gD mAb for 16 hours. Expression of mouse IgG and GFP was assessed by flow cytometry. Representative of three independent experiments. (B) DC infected with FIX-HSV exposed to LP2 or AP7 clones anti-gD mAb were infected for 16 hours and expression of CD86 was assessed by flow cytometry and presented as mean CD86 MFI of three independent experiments. Error bars represent SEM.



Figure 6.16 Attachment of HSV-1 to DC surface.

WT-HSV was neutralised with anti-gD mAb (lower panels) and then incubated in the presence or absence of 12.5 μ g/ml heparin prior to infection of DC for 1 hour. DC were stained for the presence of opsonised viral particles on the surface with PE-conjugated goat anti-mouse Ab. Representative of three independent experiments.



Figure 6.17 Role of gD in HSV-1 induced type I IFN secretion by DC. WT-HSV was neutralised with anti-gD mAb (LP2 clone) and then used to infect DC. Supernatants were harvested after 16 hours and type I IFN secretion determined by an antiviral bioassay. Results are mean of four independent experiments. Error bars represent SEM. * denotes p<0.01 relative to WT-HSV neutralised with an isotype control antibody.





DC were infected with WT-HSV (MOI =1) for 1 hour and cultured in the presence of anti-gD mAb (LP2 clone) for a further 16 hours. DC were stained with goat anti-mouse IgG (A), and HLA-DR and CD86 (B). WT-HSV data gated on GFP-ve DC. Representative of two independent experiments.





6.3 Discussion

This chapter sought to dissect the viral and host mechanisms responsible for both the activation and inhibition of DC following HSV-1 infection. Progressively inactivated and neutralised viruses demonstrated that the viral effects on DC morphology, viability and response to external stimuli were carried out predominantly by viral gene products. This suggests that these inhibitory mechanisms may have evolved to curb the activation of DC following early steps in the infection process. Dissection of the activation phenomena was complicated by inconclusive attempts to block DC surface receptors, but neutralisation of the activity of viral gD suggested that this glycoprotein may play a significant role in the direct activation of DC by HSV-1.

6.3.1 HSV-1 factors that inhibit DC functions

Infection of DC with MV impairs the function of the cells to a similar degree as HSV-1 does in chapter 4. However, a large amount of the inhibition has been attributed to the effects of immunosuppressive proteins interacting with DC and/or T cells (Klagge et al., 2000; Marie et al., 2001; Dubois et al., 2001). In contrast, de novo synthesis of viral gene products was predominantly responsible for the inhibitory effects seen following WT-HSV infection of immature DC. DC infected with UV-HSV remained viable, could stimulate T cell proliferation efficiently, had normal morphology and were able to respond to further maturation stimuli through changes in shape and through upregulation of CD86 and MHC class I to maximal levels, in sharp contrast to the effects seen with WT-HSV (chapter 4) (Salio et al., 1999) (figs. 6.4 and 6.5). These findings agreed with previous studies showing that UV-HSV infection of mature DC results in no loss of function (Kruse et al., 2000), but it is important to note that the phenotype of mature DC is more resistant to virus induced changes than immature DC (Salio et al., 1999), and therefore subtler effects of viral proteins might not have been observed. It is likely that DC maturation, in inducing a cellular state that enhances the APC function of the cell, also generates an antiviral state. The effector mechanisms are as yet unspecified, but are likely to inhibit viral gene expression at the transcriptional level (Bakri et al., 2001). The central role of the type I IFN signalling system both in DC maturation (chapter 5) and in antiviral activity is likely to be important in this

effect, either through the autocrine activity of type I IFN (chapter 5), or through activation of ISGs in the absence of IFN secretion, as demonstrated in the maturation of DC by some RNA viral infections (Lopez et al., 2003).

The specific viral components responsible for the changes observed have not been defined. A key role has been suggested for the tegument protein vhs, which inhibits protein synthesis by accelerating mRNA degradation (Kwong and Frenkel, 1987; Samady et al., 2003). However, the precise targets for vhs and other candidate viral immunomodulatory molecules remain to be elucidated. Recent studies have also suggested that vhs has specificity for mRNA with AU-rich elements (ARE) (Esclatine et al., 2004). In addition, a database of ARE containing mRNA has been set up, allowing candidate proteins for vhs-mediated degradation to be identified (http://rc.kfshrc.edu.sa/ared). These include IL-12 p35, IL-12 p40 and TNFa, possibly explaining the low secretion of these proteins by infected DC (figs. 4.11 and 5.3). The absence of IL-12 secretion may be particularly relevant because HSV-1 possess a ligand that could induce IL-12, gD. This is structurally related to LIGHT, a HVEM ligand that can induce IL-12 secretion from DC (Morel et al., 2003). Although HVEM is unlikely to play a major role in DC maturation by HSV-1 (fig. 6.12), it is also possible that ligation of this receptor may induce the secretion of IL-12 from DC primed for maturation by other receptor interactions, and that vhs prevents this response (fig. 4.11). Targeted IL-12 mRNA degradation by UV-inactivated HHV-6 raises the possibility that similar RNA degrading mechanisms are conserved among other herpesviruses (Smith et al., 2003a). Alternatively, if HSV-1 binds to DC via TLR2 (Kurt-Jones et al., 2004), then the absence of IL-12 secretion is consistent with the effects of ligating this receptor on DC (Weigt et al., 2003).

CD86 is an important negative in this discussion, as this molecule is upregulated by de novo protein synthesis (Li et al., 1999), but its mRNA does not contain ARE (<u>http://rc.kfshrc.edu.sa/ared/</u>). Therefore, its increased expression after HSV-1 infection may result from escape from vhs mediated degradation. However, the disparity between IL-12 secretion and CD86 expression is reminiscent of the effects of IFN α that were observed in chapter 5, and may similarly reflect an insufficient dose of

virus to ligate sufficient receptors to overcome p38 and NF-kB signalling thresholds controlling IL-12 secretion (see section 5.3.4.2).

6.3.2 Mechanism of HSV induced apoptosis

The absence of cell death after UV-HSV infection is consistent with the notion that viral IE gene expression is required to induce DC apoptosis after WT-HSV infection (Koyama and Adachi, 1997; Aubert et al., 1999; Sanfilippo et al., 2004). In the context of this chapter, it is also interesting to note that several studies have proposed that gD can prevent apoptosis. This may occur secondary to the accumulation of gD inside infected cells (Aubert et al., 2001), but gD binding to the cell surface, possibly also through autocrine activity of secreted gD (Murata et al., 2002), may be the critical antiapoptotic step. In that respect, addition of exogenous HSV-1 gD can prevent apoptosis in cells infected with gD- virus (Zhou et al., 2000; Zhou and Roizman, 2001) and gD can also prevent Fas-induced apoptosis in U937 (Medici et al., 2003). The anti-apoptotic action is believed to occur through interaction with the mannose receptor inside endosomes (Zhou and Roizman, 2002), resulting in NF-kB activation and upregulation of c-FLIP expression (Medici et al., 2003). Concurrently, the activation of these pathways may be responsible for the maturation of DC seen (Rescigno et al., 2000; Yoshimura et al., 2001; Lundqvist et al., 2002; Franchi et al., 2003).

6.3.3 HSV-1 induced DC maturation

6.3.3.1 HSV-1 activation of DC independent of type I IFN

Some viral infections of DC induce secretion of cytokines that can feedback in an autocrine/paracrine manner and induce cytokine-dependent maturation of DC (Dubois et al., 2001; Trevejo et al., 2001). Given that chapter 5 detailed how type I IFN secretion from HSV-1 infected DC could induce maturation of bystander DC, it was important to determine the relative role of this cytokine in the acquisition of a mature phenotype by infected DC. Figs. 6.1 and 6.7 demonstrate that this cytokine was largely not required for this effect. As the data in chapter 5 suggested that type I IFNs were the only major cytokines secreted that were responsible for the activation of DC (fig. 5.6), these data were suggestive that the stimulus for DC maturation following HSV-1

infection derived from the virus itself. The downregulation of IFNAR after LPS induced maturation of DC may explain the insensitivity to type I IFN (Gauzzi et al., 2002). However, LPS and Poly(I:C) induced type I IFN secretion rapidly enough to exert a significant autocrine effect (figs. 5.9 and 5.10). Nevertheless, it was possible that HSV-1 infection downregulated the expression of IFNAR more rapidly and efficiently than after TLR3/4 stimulation, possibly through the activity of a viral gene product (Chee and Roizman, 2004).

Partly through the activation of NF-kB, viruses can induce antiviral states in cells independent of the activity of secreted type I IFN (Lopez et al., 2003; Bose et al., 2003), emphasising the importance of direct pathways of viral activation of the cell. However, the data cannot exclude the role of cytokines other than type I IFN secreted after infection that do not activate bystander DC, but that could do so on infected DC, possibly in concert with a priming signal from the infection itself (Trevejo et al., 2001; Bose et al., 2003). Excluding other factors is further complicated by the partial dependence on type I IFN of FIX-HSV induced maturation relative to the absolute independence of WT-HSV. This could have been attributed to the ability of HSV-1 to disrupt type I IFN signalling in infected cells, and demonstrates how the relative role of different factors may be driven by the virus as much as by the cell itself (Mossman et al., 2001; Yokota et al., 2001; Yokota et al., 2004; Chee and Roizman, 2004; Duerst and Morrison, 2004).

It should also be remembered that the regulation of cytokine secretion and surface molecule expression in DC is closely related and rarely independent of each other (chapter 5). Therefore, although the studies in this thesis have aimed to define the relative importance of one stimulus (the virus) over another (type I IFN) in the maturation of DC, it is likely that in vivo the multi-stimulus environment in which the DC encounters the virus will affect the cellular response. The activity of other stimulatory factors in the microenvironment, as well virus interaction and any cytokine secretion, may summate and drive towards full DC maturation. Therefore, defining that virus interaction with DC plays a role, if not an exclusive one, in inducing some degree of maturation of these cells may still have a significant effect in vivo.

6.3.3.2 Viral interaction with the cell surface

Upregulation of CD86 and HLA-DR in response to UV-HSV and FIX-HSV suggested that virus binding to the cell surface was responsible for the activation of DC. This conclusion is consistent with observations in other cell types. One common sequelae of receptor ligation is intracellular calcium fluxes and subsequent phosphorylation of host proteins. Indeed, within 2 min of infection with HSV-1 infection, calcium fluxes are seen, and within 5 minutes significant levels of tyrosine phosphorylation of cytoplasmic and membrane proteins can be observed (Qie et al., 1999; Cheshenko et al., 2003). The downstream effects of these signals are still unclear. Protein phosphorylation may be important for the rearrangement of cytoskeleton for viral infection, by phosphorylating focal adhesion kinase (FAK), similar to the events after Shigella flexneri penetration of cells (Dehio et al., 1995). Alternatively, Ca2+ fluxes and protein phosphorylation may also be important in promoting envelope fusion (Dimitrov et al., 1993; Briand et al., 1997), as observed for the other herpesviruses EBV and HCMV (Dugas et al., 1988; Keay and Baldwin, 1996), and may also promote phosphorylation of proteins important in the activation of certain signal transduction pathways, such as NF-kB and p38 (Kelher et al., 2003). Indeed, HCMV enters cells in a similar manner to HSV-1 (Wang et al., 2003), and binding of this virus to the cell surface initiates Ca2+ signalling and receptor phosphorylation (Keay et al., 1995; Keay and Baldwin, 1996) that are important in downstream activation of the Sp1 and NF-kB pathways (Yurochko et al., 1997).

6.3.3.3 HSV-1 activation of NF-kB and p38 pathways

DC maturation by FIX-HSV suggested that virus binding to the cell surface initiated the activation of signalling pathways that induced some aspects of DC maturation. Both the p38 and NF-kB pathways mediate DC maturation (Ardeshna et al., 2000; Yoshimura et al., 2001), and both were activated by HSV-1 interaction with the DC surface. It is interesting that there was correlation between activation of p38 and NFkB, and the phenotype of HSV-1 infected DC compared to LPS, supporting the hypothesis that these pathways are important in the maturation of DC following HSV-1 infection. However, p38 inhibition had minimal effect on the upregulation of CD86 or HLA-DR by FIX-HSV. Similar to the effects of IFN α in fig. 5.8, this probably reflected a p38 independent upregulation of CD86 and HLA-DR. The reduced activation of both p38 and NF-kB compared to LPS would also explain why some molecules, such as MHC class I, are not upregulated to a great degree by FIX-HSV (data not shown). Nevertheless, the partial, although insufficient, activation of p38 MAPK raises the activation state of the cell nearer a threshold that would be required to elevate expression of several surface molecules (e.g. CD86) to maximal levels seen with LPS. Therefore, as suggested above, the partial activation of these signalling pathways may not be sufficient per se to induce drastic effects on DC phenotype, but in the absence of viral intervention, could summate with other stimuli the DC receive, and induce full maturation.

Transient activation of NF-kB has been reported after HSV-1 binding to the cell surface, but de novo gene expression is required to sustain this to higher levels (Patel et al., 1998; Amici et al., 2001). This suggests that viral gene expression and/or its products provide further stimuli for NF-kB activation. Alternatively, HSV proteins target inhibitory proteins of the NF-kB pathway (Amici et al., 2001), as also seen for HCMV (DeMeritt et al., 2004). The mechanisms employed to activate the NF-kB pathway in the earlier wave may involve both triggering of host pathways by viral entry and active viral modulation of these signals. The most likely scenario is that receptor triggering activates the IKK-IkB cascade that triggers NF-kB translocation to the nucleus (see section 6.3.5). However, vhs may also target IkBa mRNA, which contains ARE (http://rc.kfshrc.edu.sa/ared/). In contrast, the elevated levels of NF-kB activation at later stages of infection (Patel et al., 1998) may result from host recognition of products of viral replication. Accumulation of replication intermediate dsRNA may trigger TLR3 or activate PKR to induce NF-kB (Zamanian-Daryoush et al., 2000), as can an ER or ribotoxic stress response (Pahl and Baeuerle, 1997; Bunyard et al., 2003). However, HSV-1 expression of IE genes can also induce NF-kB activation within 2 hours of infection in macrophages through ROI-induced release of Ca2+ from mitochondria (Mogensen et al., 2003). It is unclear whether this mechanism of response to intracellular stress is specific to macrophages, cells that have a greater propensity to generate ROI, or whether ROI are more ubiquitous intracellular messengers in response to viral infection.

Irrespective of the mechanism, the early activation of the NF-kB pathway is likely to be important for the DC response to viruses. For example, HCMV induces early IkB α degradation and NF-kB activation within 30 minutes of infection of fibroblasts and monocytes (Yurochko et al., 1995; Yurochko and Huang, 1999; DeMeritt et al., 2004). In monocytes, this early phase of NF-kB is accompanied by activation of p38 MAPK. It can induce transcription and translation of immunological proteins, such IL-1 β , and can be mediated solely by glycoprotein interaction with surface receptors (Yurochko and Huang, 1999). As HCMV infection of fibroblasts does not result in the early wave of p38 MAPK seen in monocytes (Johnson et al., 2000), this demonstrates the importance of studying these early activation events in the appropriate cells. HSV-1 binding to DC may elicit similar signals to HCMV, but DC may possess lower requirements to activate particular signalling pathways in response to selected stimuli. The degree of NF-kB or p38 MAPK activation induced may also be critical. While prolonged NF-kB activation may be important for enhancing transcription of NF-kB responsive HSV genes (Rong et al., 1992) to enhance replication (Patel et al., 1998; Amici et al., 2001), lower levels of NF-kB activation induced by gD binding to the cell surface may be sufficient for the induction of a cellular antiviral response (Bose et al., 2003). In DC, this response is translated into maturation.

6.3.4 DC receptors involved in HSV-1 induced activation

Three candidate receptors responsible for initiating cell surface signalling were identified on the basis of their known properties either in response to or in the infection by HSV-1. TLR2 has been recently proposed as playing a role in cellular responses to HSV-1 (Kurt-Jones et al., 2004), whereas HVEM and nectin-1 are HSV-1 entry receptors (Montgomery et al., 1996; Geraghty et al., 1998). Of these three, TLR2 and HVEM have been shown to activate NF-kB in other cell types, consistent with being members of TLR and TNFR superfamilies respectively (Re and Strominger, 2001; Kurt-Jones et al., 2004; Marsters et al., 1997). TLR2 can also activate p38 (Agrawal et al., 2003), but recent studies suggest that HVEM may not be able to do so, as DC require TRAF6 for p38 activation (Mackey et al., 2003) and this factor does not associate with HVEM (Marsters et al., 1997). The central role of NF-kB and p38 activation in mediating DC activation suggested an important role for these receptors

(Clark et al., 1999; Neumann et al., 2000; Hofer et al., 2001; Yoshimura et al., 2001; Ardeshna et al., 2000). On the other hand, nectin-1 is a member of adhesion molecules and its downstream signalling is currently undefined.

Receptor neutralisation studies using monoclonal and polyclonal antibodies were inconclusive. Despite using saturating quantities of antibody, the maturation of DC by HSV-1 could not be attributed to one receptor. The efficacy of the TLR2 neutralising antibody could not be guaranteed, as the upregulation of CD86 and HLA-DR by PGN was not affected. However, as this antibody has only previously been shown to partially neutralise DC secretion of TNF α in response to PGN (Uehori et al., 2003), the expression of CD86 may not be as sensitive as TNF α secretion to incomplete reductions in signalling through TLR2 (see chapter 5). The same hypothesis may also explain the unaltered phenotype of DC exposed to HSV-1 in the presence of anti-TLR2 mAb. Future strategies will involve using another neutralising antibody to TLR2, clone 2392, which has been shown in side-by-side studies to have more potent neutralising capacity than a similar antibody clone to the one used in the experiments in this thesis (Iwaki et al., 2002).

For the entry receptors, the data is even more difficult to interpret in the absence of an absolute positive control and the redundancy in receptor usage by HSV-1 (Krummenacher et al., 2004). This is often bypassed by transfecting resistant cell lines with a single receptor. However, this is not a suitable control for DC. Therefore, the theoretical approach behind the experimental design used was that, despite the redundancy, DC coated with receptor antibody, but still infected, could be interpreted as having been infected independent of the neutralised receptor. This conclusion could be made with HVEM, as infected cells still expressed antibody-bound receptors 16 hours post-infection, but nectin-1 was internalised more rapidly. This may simply have reflected that anti-nectin-1 was a mAb, and therefore fewer antibodies were present initially on the DC surface. Nevertheless, the saturating amount of antibody added to the cells should have prevented interaction between gD and nectin-1 necessary for entry (Krummenacher et al., 2000). This suggested that nectin-1 was not involved in activating DC. The neutralisation of HVEM suggested a similar conclusion for this

receptor. Considered in conjunction with the polyclonal nature of the reagent used, this supported most strongly that HVEM was not solely responsible for the activation of DC.

The receptor neutralisation studies leave open two possibilities that might explain the inconclusive results. One of these is that HSV-1 induces DC maturation by binding to, and signalling via, receptors other than TLR2, HVEM or nectin-1. Indeed, gD interaction with chemokine receptors (Ankel et al., 1998) or the MR (Brunetti et al., 1994; Brunetti et al., 1995) can exert functional effects on cells (Ankel et al., 1998; Milone and Fitzgerald-Bocarsly, 1998a; Rong et al., 2003). The MR has also recently been shown to mature DC, though programming the cells to induce Th2 responses (Chieppa et al., 2003). Only neutralisation of these receptors could determine their role. An alternative scenario is that all (or two) of the receptors neutralised in these experiments contribute relatively similar weak activation signals to DC, and that HSV-1 ligates these receptors in a non-saturable manner. The titration of recombinant gD on DC supports this theory, as even at the highest concentration of 50 μ g/ml, the expression of CD86 did not plateau, suggesting that a non-saturating dose of glycoprotein had been added (fig. 6.19). This model predicts that neutralisation of one receptor would result in compensatory ligation of heterologous 'empty' receptors. In this way, the signals emanating from these receptors could compensate for the loss of signal from the neutralised receptors, resulting in no change in surface phenotype. This hypothesis could be tested by using combinations of neutralising antibodies, thereby preventing ligation of more than one receptor at any one time.

6.3.5 Viral ligands responsible for DC activation

The ability of FIX-HSV to induce maturation of DC suggested that HSV-1 envelope structures were able to ligate receptors on the surface of DC and transmit activation signals. The virus envelope is made up of at least 11 glycoproteins (Handler et al., 1996), but the most likely candidates were viral glycoproteins that interact with receptors during viral entry. These include gB and gC that interact with HS, and gD that binds to entry receptors (Spear et al., 2000). Fig. 6.16 excluded the role of gB and gC binding to HS as providing a significant stimulus for activation of the cell. It has

previously been suggested that HCMV, a virus that utilises similar mechanism to enter cells as HSV-1, can attach HS and transmit signals that induce upregulation of HLA-ABC in fibroblasts (Song et al., 2001). However, HS is a surface glycomatrix with no precise anchoring or defined intracellular signalling machinery, and the study did not exclude that inactivated virus was binding to an activatory surface receptor (Song et al., 2001).

It is important to note, that although the data in this chapter rules out the role of gB and gC binding to HS on DC as providing maturation signals, they do not exclude the role for these glycoproteins completely. Although, no other receptors have been described that bind gC, gB plays a role in inducing envelope-membrane fusion, along with gH-gL, and this event may play an important role in the activation of DC (see section 6.3.5.2).

6.3.5.1 Role of gD in DC activation

The biological activity of viral glycoproteins binding to and inducing^A_k response to infected cells has been previously described. EBV gp350 can induce TNF α alpha secretion in a NF-kB dependent mechanism (D'Addario et al., 2000). HIV-1 gp120 can also induce the secretion of a wide range of cytokines, depending on the cells stimulated and the microenvironment of the cells (Ameglio et al., 1994; Fantuzzi et al., 1996), and in DC, it may induce maturation (Fantuzzi et al., 2004).

Neutralisation of gD with LP2, rendered the virus particles non-infectious, but did not prevent virus attachment to the cell surface (figs. 6.14 and 6.16). From these findings, it was extrapolated that the neutralisation prevented the interaction of gD with receptors on DC. The loss of mature phenotype induced by both FIX-HSV and WT-HSV by LP2 neutralisation was suggestive of the role of gD in providing the initial maturation stimulus to DC. Previous studies analysing the ability of gD to bind to surface receptors back up a role for this glycoprotein in activating DC. Recombinant forms of gD alone are able to activate NF-kB in U937 monoblastoid cells (Medici et al., 2003), and induce secretion of TNF α from RAW 264.7 macrophage-like cell lines (Paludan and Mogensen, 2001) and type I IFN from PBMC (Ankel et al., 1998).
However, there are several caveats, both from the data presented here and from the literature, that currently precludes from concluding that gD is solely responsible for providing activation signals to DC.

It was possible that LP2-neutralised virus prevented DC maturation by interacting with inhibitory FcR on the DC surface. This is unlikely because when $Fc\gamma RII$, the only Fc γR expressed on human MDDC, are cross-linked, maturation of DC ensues, suggesting that this receptor possesses a cytoplasmic tail that can transmit positive rather than negative signals in DC (Banki et al., 2003). Furthermore, when HSV particles were coated with the non-neutralising anti-gD mAb, AP7, maturation was not prevented. On the assumption that AP7 coated virions had an equal possibility of interacting with Fc γ RII as those neutralised with LP2, then the interaction with Fc γ RII was unlikely to play a significant role in this system.

An alternative explanation for the effect of LP2 neutralisation on DC phenotype was that the antibody sterically hindered the binding of other viral components that were responsible for the activation of DC. AP7 binding did not prevent maturation, suggesting that the LP2 binding domain of gD was responsible for activation of DC. However, as AP7 is a non-neutralising antibody, it is likely to bind a different epitope of gD to LP2, one that is unlikely to prevent receptor interaction. Therefore, AP7 may not be the best control antibody to exclude steric hindrance. An antibody to a component closer to the receptor binding complex, such as gH, may be a more appropriate control.

The neutralisation and resultant loss in DC maturation by LP2 may assist in determining the receptor used for virus activation. Co-sedimentation of antibodybound HSV-1 with truncated soluble forms of receptor demonstrated that LP2 binding to gD prevented viral interaction with nectin-1 but not HVEM (Nicola et al., 1998; Krummenacher et al., 1998). These data would appear to exclude the role of HVEM in DC maturation by HSV-1. However LP2 can prevent HSV infection in CHO cells (which are naturally resistant to HSV) transfected and expressing only HVEM (Nicola et al., 1998), suggesting that even though LP2 allows gD-HVEM binding, it may still affect receptor-induced conformational change of gD (Carfi et al., 2001) or fusion of the viral envelope with the ceil membrane. As both steps may be important in virus induced signalling (see section 6.3.3), then the role of HVEM in virus induced signalling in DC cannot be excluded definitively from the LP2 neutralising experiments.

In order to determine the direct role of gD in DC activation, recombinant gD was added directly to DC. This glycoprotein induced the upregulation of CD86 in a dosedependent manner, suggesting that alone it could induce activation signals in DC directly (fig. 6.19). However, it is important to confirm that the effects observed were gD-specific. The glycoprotein was prepared from a baculovirus transfection system, and therefore should not have contained any bacterial contaminants. Furthermore, proteins purified from insect cells added to DC do not induce maturation (Chan et al., 2003). Nevertheless, it will be important to exclude any contaminants in the protein preparation. One approach may be to precipitate the protein out of the culture medium (prior to addition to DC) using anti-gD antibodies. Alternative approaches include attempting to neutralise the receptor that the protein may be ligating, which has the added advantage of suggesting which receptor HSV-1 binds to activate DC.

6.3.5.2 Role of non-gD viral components in DC activation

Evidence that viral components other than gD alone bind to receptors to elicit signalling from the cell surface has emanated from the use of deletion mutants. Ca2+ fluxes elicited by viral infection do not occur with gL- viruses, implying that gD receptor interaction alone is insufficient to induce tyrosine phosphorylation, and that post-binding events are required for activation (Cheshenko et al., 2003). These viruses are also less potent at eliciting TNF α production from macrophages (Paludan and Mogensen, 2001).

These findings appear to contradict the data presented in this thesis, specifically the activation of DC by FIX-HSV and recombinant gD alone. One possible explanation is that the gD binding to cells in the glycoprotein deletion mutants is aberrant. However, this is unlikely because the removal of individual glycoproteins from the viral envelope does not grossly affect the quantity and conformation of the remaining ones

(Rodger et al., 2001). Another possibility is that the activation of DC by recombinant gD was an artefact independent of any contaminants in the protein preparation. The gD used in fig. 6.19 was gD(285t), a truncated form of the full length glycoprotein that is missing the 21 amino acids from the C-terminal end (Rux et al., 1998). This form of gD binds to HVEM and nectin-1 with a 100-time greater affinity than the full length protein (Rux et al., 1998; Krummenacher et al., 1998). In this thesis, the truncated protein was used in order to determine whether an effect could be observed that would suggest that gD could activate DC directly. Use of the full-length protein may determine whether the difference in protein affinity impinges on the downstream signalling, as this has not been investigated previously. No effect on DC phenotype by the full length gD would also suggest that the gD(285)t preparation was free of contaminants, as both preparations are purified using the same gD column (Rux et al., 1998). In keeping with this hypothesis, fixation of the viral envelope may have crosslinked the closely juxtapositioned envelope glycoproteins into a conformation that can bind DC receptors with greater affinity than the wild type glycoproteins (Handler et al., 1996; Rodger et al., 2001). Thus, fixation may have generated a ligand that was artificially over-efficient at stimulating DC, perhaps in a similar manner to gD(285)t. This may also explain the common observation that FIX-HSV could elicit more potent maturation of DC than UV-HSV or WT-HSV (data not shown).

However, the limitations of the FIX-HSV model do not detract from the conclusion that virus-surface binding activated DC in a gD-dependent manner. It is likely, though, that gD can activate cells in concert with other viral ligands, possibly through their role in inducing a conformational change in gD and/or the bound receptors to induce signalling downstream (Carfi et al., 2001; Krummenacher et al., 2002). Consistent with this, HSV-1 virions that are unable to fuse with the cell membrane cannot initiate Ca2+ fluxes or tyrosine phosphorylation (Cheshenko et al., 2003; Qie et al., 1999). However, only alpha-herpesviruses possess gD-like molecules that are bone fide ligands for receptors, whereas beta- and gamma-herpesvirus entry glycoproteins also possess fusogenic capacity. Indeed, the fusogenic capacity of HCMV entry glycoprotein, gB, is required to activate an interferon response following surface binding (Netterwald et al., 2004).

One approach to determine the relative roles of these HSV-1 accessory glycoproteins would be to use neutralising antibodies to prevent their binding to the cell surface. Antibodies to gB may affect viral attachment to the HS on DC (Laquerre et al., 1998), but antibodies to gH or gL would prevent infection by preventing envelope-membrane fusion. Theoretically, this approach should permit gD ligation to surface receptors, excluding the role of non-gD proteins. It has also not been excluded that gH-L binds an unknown surface receptor, and the addition of recombinant gH-L may exclude this possibility and/or further address the issue of protein purity. The use of deletion viruses (e.g. gD- vs. gH- or gL- HSV-1) to "infect" DC would also aid in dissecting this scenario.

6.3.6 HSV-1 induced cytokine secretion

6.3.6.1 Secretion of type I IFN

The secretion of type I IFN induced by UV-HSV and FIX-HSV infection in the majority of individuals tested is consistent with replication-independent trigger for IFN secretion. Recent data shows that PDC can also respond to HSV-1 infection in this manner (Hochrein et al., 2004), although PDC may uniquely recognise ligands that induce greater type I IFN secretion (see section 5.3.1). The data presented in this chapter also indicate that HSV-1 gD plays an important role in type I IFN secretion by MDDC (fig. 6.17). The potential receptors responsible include CXCR4 and CCR3, neither of which are implicated in the HSV-1 infection process (Ankel et al., 1998). Alternatively, MR can bind the heavily glycosylated gD (Brunetti et al., 1994; Brunetti et al., 1995), and has been implicated in HSV-1 induced secretion of type I IFN (Milone and Fitzgerald-Bocarsly, 1998a; Rong et al., 2003). Irrespective of the receptor, the early role of PKC and Ca2+ fluxes in the induction of type I IFN secretion, supports the triggering of surface receptors by the virus to elicit this response (Li et al., 1996).

For the same technical limitations described in section 6.3.5, the loss of IFN secretion by LP2 anti-gD mAb does not prove that this glycoprotein is solely responsible for inducing type I IFN secretion. As for the effects on phenotype, it is likely that gD may induce type I IFN secretion with the assistance of the gB, gH-L fusion complex. Indeed, the induction of IRF-3 by HSV-1 only occurs in the presence of both binding and fusing with the cell surface (Preston et al., 2001). Similarly, HCMV can triggering of IRF-3 activation and ISG transcription occurs through both binding of gB to the cell surface and subsequent post-attachment events (i.e. envelope fusion) (Preston et al., 2001; Boyle et al., 1999; Netterwald et al., 2004). Therefore, it appears that herpesviruses can trigger an IFN response early after infection, prior to viral gene expression that may modulate it. However, it is unclear whether this pathway relates to a host recognition of viral entry, analogous to recognition of products of viral replication (e.g. dsRNA), and whether it is a significant selective pressure for the evolution of HSV-1 mechanisms to inhibit signalling of the type I IFN system signalling (Nicholl et al., 2000; Mossman et al., 2001; Lin et al., 2004; Collins et al., 2004). This concept is discussed further in section 6.3.7.

6.3.6.2 Secretion of other cytokines

In a mouse macrophage cell line, HSV-1 replication induces NF-kB activation and IL-12 secretion (Malmgaard et al., 2000). In the same cells, HSV-2 also triggers TNF α secretion (Paludan et al., 2001), via binding of gD to the cell surface, viral DNA delivery to the nucleus and viral replication (Paludan and Mogensen, 2001). Similar multi-layered mechanisms have also been implicated in macrophage production of CCL5 in response to HSV infection. Viral replication is required to trigger production of this chemokine, through the activation of NF-kB, PKR and IRF-3 (Melchjorsen et al., 2002; Melchjorsen and Paludan, 2003). NF-kB activation in DC may have been sufficient to induce low degree of TNF α secretion, but insufficient for IL-12, further highlighting the cell-dependent outcomes of signalling activation.

6.3.7 Functional outcome of HSV-1 triggering of host signalling pathways

It is important to relate the mechanistic discussion in earlier sections to the outcome of infection. The phenotype changes induced by HSV-1 on DC are mild, and may not result per se in a significant increase in antigen presenting function of the DC, but may

condition the cell's Th skewing capacity. Equally, the viral induced signalling changes may also favour the survival and replication of the virus. The survival advantages of the host-pathogen interaction are discussed below.

6.3.7.1 Pathogen recognition: advantage to host survival

Recent studies have focused on the ability of the innate immune system, of which DC are a central component, to recognise PAMPs and swiftly initiate anti-pathogen responses (chapter 1). Considerable interest has been generated by the role of the evolutionarily conserved TLR family in the regulation of viral infections. Known viral ligands are detailed in table 1.1. The direct activation of DC by HSV-1 would appear to be consistent with the concept of host-pathogen recognition and indeed, some studies have shown that the binding of viral structures to TLR favours host survival. TLR9 and TLR3 are important in the control of murine cytomegalovirus infection (Tabeta et al., 2004). The hypomethylated CpG rich genome of HCMV, along with the dsRNA intermediate, are likely to be the PAMPs respectively. Recent studies have also suggested that TLR7 can recognise ssRNA (Heil et al., 2003; Diebold et al., 2004) and that this receptor is important for the clearance of the ssRNA virus, vesicular stomatitis virus (VSV) (Lund et al., 2004). Similar protection studies have been extended to the human pathogen, RSV. The resolution of bronchiolitis caused by this virus is TLR dependent, as the RSV F protein is recognised by TLR4 (Kurt-Jones et al., 2000; Haynes et al., 2001; Haeberle et al., 2002). Although it has also been suggested that the IL-12 receptor axis is more critical than TLR4 in some mice models (Ehl et al., 2004), polymorphisms in the human population have confirmed that TLR4 does play an important role in RSV infection in vivo (Tal et al., 2004). The evolutionary advantage to host immunity of TLR recognition in some infections is further highlighted by the expression of proteins by pox viruses that inhibit TLR signalling (Harte et al., 2003).

6.3.7.2 Receptor triggering: advantage to viral survival

The presence or expression of viral ligands that bind TLR may not necessarily correspond to a protective advantage to the host. TLR3 is not required for clearance of the dsRNA virus LCMV, and TLR9 expression does not impact on the replication or

clearance of HSV-1 after epidermal infection (Krug et al., 2004), despite the DNA genome of this virus being rich in CpG motifs that can ligate TLR9 (Honess et al., 1989; Lund et al., 2003). Furthermore, in some mouse models, the expression of TLRs, promotes the establishment of wild type viruses over more attenuated forms (Jude et al., 2003).

There is also now evidence to demonstrate that viruses have evolved to exploit TLR binding and signalling. Murine mammary tumour virus (MMTV) can bind TLR4 and upregulate expression of its own receptor (Rassa et al., 2002). Interestingly viral-TLR4 binding induces the maturation of DC, demonstrating how increased viral infectivity may be of sufficient advantage to trade off against activation of these cells. Similarly, MV HA binding to TLR2 also upregulates the MV receptor, CD150 (Murabayashi et al., 2002; Bieback et al., 2002). Interestingly, the attenuated vaccine strain of MV uses CD46 as an entry receptor, and does not bind TLR2 (Bieback et al., 2002).

Virus proteins may also bind non-TLR receptors to favour their own survival. MV NP can bind $Fc\gamma R$ on DC, and suppress their IL-12 secretion (Marie et al., 2001). The ability of many viruses to bind to c-type lectins (see section 1.8.1.2) may also be exploited as evidenced by the Th2 conditioning initiated by MR cross-linking (Chieppa et al., 2003), and inhibition of DC maturation by *Mycobacterium tuberculosis* binding to DC-SIGN (Geijtenbeek et al., 2003).

Closer analysis of the biology of HSV-1 replication reveals how binding to stimulatory receptors may be advantageous. Calcium fluxes induced by HSV-1 binding may increase the efficiency of viral infection by aiding transport of VP16 to the nucleus, possibly through actin remodelling (Cheshenko et al., 2003). Activation of the NF-kB, and p38 MAPK pathways can prevent apoptosis (Goodkin et al., 2003) and enhance the efficiency of viral replication (Patel et al., 1998; Zachos et al., 1999), as many viral gene promoters are positively regulated by NF-kB (Rong et al., 1992). Consequently, it is conceivable that viral activation of NF-kB and p38 are desired events, analogous to the regulation of the ERK and JNK pathways, where the virus expresses proteins that specifically induce or prevent the activation of these pathways respectively, in order to prevent cellular apoptosis (Perkins et al., 2002; Murata et al., 2000).

Although a genetically stable virus, HSV-1 can evolve under selective pressure, as seen by the acquisition of acyclovir resistance (Danve-Szatanek et al., 2004). Therefore, this discussion suggests that HSV-1 has conserved stimulatory structures to activate the host cell signalling machinery to favour its own replication. As the activation of both p38 MAPK and NF-kB in DC primes these cells' maturation and subsequent initiation of antiviral responses, it is hypothesised that this activation is the driving force for the evolution of viral mechanisms to dampen the function of DC that are activated early by the initial infection process (chapter 4 +section 6.3.1). The interesting implication from this model is that the ability of DC to 'recognise' and mature in response to HSV-1 infection may result primarily from the dependence of the virus to activate certain signalling pathways for its own replicative advantage. Identification of the receptors involved would also aid in determining how these virusinduced signals affect the T cell stimulatory capacity. In that respect, the ability of HSV-1 to bind MR or TLR2 (Milone and Fitzgerald-Bocarsly, 1998a; Kurt-Jones et al., 2004) may skew towards Th2 responses (Chieppa et al., 2003; Agrawal et al., 2003).

6.4 Conclusions

This chapter has sought to detail the mechanisms involved in both the activation and inhibition of DC after infection with HSV-1. Although a precise study using mutant viruses is required to explore the latter issue fully, the data demonstrated that viral genes expressed, rather than virion components, were responsible for the suppression in DC function observed in chapter 4. In contrast, interaction of the viral envelope with the surface of DC induced the activation of pro-inflammatory signalling pathways and the upregulation of MHC class II and co-stimulatory molecules. A major player in this activation was gD, although its precise role and the receptors with which it interacts remains elusive. The virally induced activation of DC needs to be detailed further both qualitatively and quantitatively, as it may alter the accepted dogma of pathogen recognition by host receptors. Furthermore, the consequence of HSV-1 entry triggering changes in DC physiology has important implication both in a model of HSV-1

-

infection and for therapeutic uses of HSV-1 as a gene delivery vector. These issues will be discussed in the final chapter of this thesis

•

-

General discussion

-

Chapter 7

i

General discussion

7.1 Introduction

The work in this thesis has addressed the interaction between HSV-1 and DC. Disruption of DC function would appear to be a candidate addition to the list of strategies this virus possesses to evade or delay the immune response. Furthermore, the pressures to escape the generation of immunity by DC that are activated by early events in the infection process, may have led to selection of a virus which meets the stringent requirement to interfere with as many DC functions as possible. However, this in vitro data must be reconciled with in vivo observations that potent and well-characterised cellular responses occur following HSV-1 infection. The interplay between virus and the host in a model of HSV-1 infection will be discussed in this chapter. The application of this model to a more clinical setting will also be assessed. Finally, the central role of DC in the pathogenesis of other viral infections will be discussed.

7.2 Model for establishment and resolution of peripheral HSV-1 lesion

7.2.1 Role of DC impairment

HSV-1 infects the epidermis, where the predominant cell type it encounters is keratinocytes. These cells allow efficient replication (Mikloska et al., 1996). DC are represented in the epidermis by LC, which may be infected at the earliest stages of infection. LC share many characteristics with the myeloid non-Langerhans DC described in this thesis: potent ability to stimulate naïve T cells, dendritic morphology and a similar expression profile of surface molecules (e.g. MHC class II, CD1a) (Caux et al., 1996; Ito et al., 1999; Canque et al., 2000). Therefore, LC may well be affected negatively by HSV-1 infection in a similar manner to these DCs. Infected LC are unlikely to be able to migrate out of the skin into draining LNs, and will not stimulate T cells effectively if they get there. The absence of detectable virus in draining lymph nodes of infected peripheral sites supports this model (Mueller et al., 2002; Zhao et al., 2003).

Impaired LC function inhibits the stimulatory capacity of the first APC encountered, which delays the generation of the adaptive immune response. The subsequent delay in infiltrating CD4+ and CD8+ T cells at the infection site allows the virus sufficient time to replicate and infect a second round of keratinocytes (and LC). This would not only increase the size of the lesion, but also increase the local virus titre, favouring successful transmission of the infection to the next individual. The absence of CD8+T cells early in infection would also allow efficient infection of nerve endings (van Lint et al., 2004). Furthermore, LC infection may specifically enhance the efficiency of nervous system infection and subsequent establishment of latency. It is noticeable that there is a close anatomical association between LC and c-type nerve endings in the epidermis, such that 60% of these nerves are associated with LC (Hosoi et al., 1993). Although chapter 3 demonstrated that viral production by infected DC is low, the local concentration of produced virus may be sufficiently high to infect the nerve endings, allowing retrograde travel to the DRG and the establishment of latency (Jones, 2003). The relationship between LC and nerve endings also suggests that LC would be one of the first cells infected after reactivation from latency, favouring the establishment of secondary skin lesions.

The involvement of other skin DC subsets in this model is less well defined, as there is little data analysing the spread of lesions in vivo. Specifically, it is unclear whether a lytic lesion breaks through the epidermal-dermal barrier. Recent data analysing HSV-2 infection of the genital mucosa suggests that this represents an anatomical barrier to the infection (Zhao et al., 2003), but it is unclear whether this also occurs in the skin. As a result, it is unclear whether dermal DC would also be infected and affected by the virus. An important factor may be determined by the depth of the skin abrasion that allowed the initial viral breach of the epidermal barrier, and the dose of viral inoculum of the infection, which determines the number of cells infected initially and the subsequent rate of spread of the lytic infection.

7.2.2 Role of bystander DC

Anti-HSV cellular responses in vivo demonstrate that the host can bypass the initial block in antigen presentation and harness the potency of DC. The bystander activity of

type I IFN in this process was suggested in chapter 5 and is discussed in section 7.2.3. This model proposes that the epidermis surrounding the lesion, as well as the underlying dermal/submucosal tissue, will contain uninfected DC, and that while the function of the infected DC in the epidermis at the centre of the lesion may be impaired, uninfected "bystander" dermal/submucosal DC at the margin may take up dead or dying virally infected cells, and then cross-present the viral antigens in the draining LN (see section 4.3.1.2). The correlation in fig. 4.1 between uninfected DC in infected cultures and HSV recall proliferative response supports this hypothesis.

Data from murine models supports the role of uninfected bystander DC in initiating the antiviral response. Neither epidermal nor vaginal infection resulted in the presence of detectable HSV DNA in the draining LN (Mueller et al., 2002; Zhao et al., 2003), excluding both direct presentation by infected DC and direct viral drainage into the LN via lymphatics, as seen after footpad infection (Mueller et al., 2002). However, there are differences between the DC subsets involved in antigen presentation. Following epidermal HSV-1 infection, virus specific CD8+ T cells were shown to be activated exclusively by LN resident CD8 α + DC (Allan et al., 2003; Smith et al., 2003b). In contrast, after vaginal infection with HSV-2, virus specific CD4+ T cells were activated by CD11b+ submucosal DC (Zhao et al., 2003). It remains unclear whether this disparity relates to differences between CD4+ and CD8+ T cells (Pooley et al., 2001) or between HSV-1 and HSV-2 infection. Furthermore, the direct ontogeny of DC subsets is still unresolved (Shortman and Liu, 2002), and migrating myeloid CD11b+ DC may differentiate into CD8 α + DC in the LN (Moron et al., 2002). Nevertheless, the preferential role of mouse $CD8\alpha + DC$ in cross-presentation to CD8+T cells is consistent with the model of indirect presentation (Iyoda et al., 2002; Schulz and Reis e Sousa, 2002).

Early murine studies had demonstrated an inverse correlation between LC numbers in the skin of mice and the severity of the lesion after epidermal HSV-1 infection (Sprecher and Becker, 1989). However, both of the models above demonstrated that LC were not responsible for CD4+ or CD8+ T cell stimulation (Zhao et al., 2003; Allan et al., 2003). Therefore, in the context of an epidermal infection, the primary role of LC may be antigen transfer to draining LN rather than direct T cell stimulation. Such an 'antigenic vector' role of DC from the periphery to LN has been proposed previously (Huang et al., 2000; Belz et al., 2004). The decline in LC numbers migrating to the LN after infection (Zhao et al., 2003) may reflect the speed of the spread of infection in the epidermis and/or the synchronous early emigration of uninfected LC from that compartment.

The antigen presenting role of PDC in this model also remains a mystery. PDC express TLR9, which can ligate HSV DNA that is rich in hypomethylated CpG motifs (Lund et al., 2003). However, abrogating this pathway does not hamper the control of viral replication or resolution of peripheral HSV-1 lesions (Krug et al., 2004). This may relate to the redundant role of this pathway in type I IFN secretion in response to HSV-1 (Hochrein et al., 2004). Furthermore, despite PDC accumulation in LNs after peripheral infection (Smith et al., 2003b; Yoneyama et al., 2004), PDC are not involved in direct antigen presentation to responding T cells after peripheral infection (Smith et al., 2003; Allan et al., 2003). This may relate to their relatively poor ability to stimulate naïve T cell proliferation (Krug et al., 2003) or their inability to cross-present exogenous antigen from DC draining peripheral infection sites (Salio et al., 2004).

Uninfected bystander DC at the site of infection may migrate towards the lesion. Murine studies of genital mucosa infection show a rapid recruitment of DC within 24 hours of HSV infection. The driving signals are likely to be chemokines, such as CCL3, secreted at the site of infection by keratinocytes (Mikloska et al., 1998) that may recruit MDC (Yoneyama et al., 2004). These DC could have originated either from the migration of tissue resident dermal DC towards the lesion, or the recruitment of monocyte precursors from blood that have differentiated into immature DC (Randolph et al., 1999). Type I IFN secreted at the infection site is likely to promote this differentiation, as discussed in chapter 5 and below.

7.2.3 Role of type I IFNs in resolution of peripheral HSV-1 infection

Type I interferons may be key mediators in terms of the cross-presentation model outlined above, both at the site of infection and in secondary lymphoid organs during the generation of the antiviral response.

7.2.3.1 Type I IFN in the periphery

In the early stages of infection, type I IFN can be secreted by non-DC (Lebre et al., 2003), as well as from infected myeloid DC in tissue (fig. 5.3). In addition to their antiviral activity, type I IFNs may also be an important danger signal detected by bystander uninfected DC, inducing maturation and migration of DC cross-presenting antigens from dying infected cells (chapter 5) (Padovan et al., 2002; Luft et al., 2002b). This initial IFN release may also be important in aiding the differentiation of monocyte precursors of DC (Randolph et al., 1998; Blanco et al., 2001), generating MDC that secrete greater quantities of IFN α following exposure to HSV-1 (Mohty et al., 2003). Concurrent recruitment of PDC to the site of infection (Wollenberg et al., 2002) may amplify the local type I IFN production by responding to HSV DNA (Lund et al., 2003; Lundberg et al., 2003a; Krug et al., 2004).

7.2.3.2 Type I IFN in secondary lymphoid organs

DC may also interact with HSV in peripheral blood, as mucocutaneous HSV infection results in systemic viraemia in a large proportion of individuals (Youssef et al., 2002), and this encounter provides an alternative pathway for the recruitment of both DC subsets to LNs after local infection (Cella et al., 1999a; Yoneyama et al., 2004). Although local type I IFN secretion by MDC may be sufficient in the DC-T cell microenvironment, the large IFN secretion by recruited PDC may be critical to ensure that DC subsets (cross-)presenting viral antigens result in priming of the immune response, as opposed to inducing tolerance (Hawiger et al., 2001; Dalod et al., 2003; Le Bon et al., 2003; Bjorck, 2004). The mechanisms by which this occurs may revolve around increasing IL-12 secretion by the presenting DC (Albert et al., 2001) or increasing the time of contact and interaction between DC and T cells (Benvenuti et al., 2004b).

7.3 HSV-DC interactions: implications from laboratory to clinic

The knowledge gained from studying the relationship between HSV-1 and DC can be applied to a more clinical setting. Not only can HSV-1 infection affect the pathogenesis of other viral infections, but it may also be exploited for the efficient delivery of therapeutic genes into DC, such as in the field of tumour immunotherapy. Furthermore, the central role of DC in the generation of immunity to HSV suggests that detailing the functional consequences of the interaction between virus and the cell may be beneficial in developing rational designs for vaccines to HSV.

7.3.1 Impingement on pathogenesis of HIV-1 infection

The original cohort of HIV infected individuals was identified on the basis of more severe HSV disease. This has subsequently been explained by the immunodeficiency induced by HIV-1 which results in more frequent reactivations from latency and lack of immunological control of peripheral infections (Siegal et al., 1981; Schacker et al., 1998b). However, more recent data has suggested that HSV infection may also accelerate HIV disease progression towards AIDS. HSV reactivation is a significant predictor of onset of AIDS (Hennessey et al., 2000), and plasma HIV RNA levels can also increase during clinical HSV reactivation (Mole et al., 1997), possibly explaining data showing a moderate positive effects on survival of HIV infected individuals treated with acyclovir (Ioannidis et al., 1998).

Active HSV lesions may provide portals for HIV entry and exit (Schacker et al., 1998a), but HSV-1 may induce more direct effects that result in increased HIV replication. One of these is pseudotyping of HIV-1 particles in dually infected cells, subsequently increasing the tropism of the HIV particles (Heng et al., 1994; Calistri et al., 1999). Another possibility is that cytokines secreted after HSV infection induce bystander HIV infected cells to enhance viral replication (Duh et al., 1989). Alternatively, HSV-1 infection per se is sufficient to enhance HIV replication, as supported by evidence of increased HIV replication in cells infected with both HIV-1 and HSV-1 (Heng et al., 1994). The increased replication may result from enhanced

transcription of the HIV LTR in cells that harbour latent HIV, through increased NFkB activation of the LTR (Mosca et al., 1987; Margolis et al., 1992). Transfection experiments and the use of HSV deletion mutants have proposed important roles for HSV ICP0, ICP4 and ICP27, although the status of HIV-1 infection and the cell type infected may determine their relative contribution (Albrecht et al., 1989; Golden et al., 1992; Margolis et al., 1992; Schafer et al., 1996). It is also possible that HSV promotes post-transcriptional HIV replication through the chaperone functions of US11 (Diaz et al., 1996).

Although the precise details of this dual viral interaction are still to be determined, the implications of this molecular interaction are twofold. Firstly, HSV activation of DC may not only induce the migration of HIV-laden DC to the LN, but also increase viral replication and thus trans infection of CD4+ T cells (Geijtenbeek et al., 2000a). Secondly, it implies that successful immunological control of HSV infection prior to the acquisition of HIV infection, or during the clinical latency stage of infection may reduce the number of HIV latent reservoirs that may be 'restimulated' by HSV infection, and limit or delay the progression to AIDS.

7.3.2 Development of a vaccine to HSV-1

The role of the cellular adaptive immune response in preventing neuronal infection, resolving peripheral infections and controlling the reactivation from latency, justifies the hope that a vaccine can limit the severity, and hence morbidity, of secondary infection (Liu et al., 2000; van Lint et al., 2004). Furthermore, less frequent reactivation may also reduce the number of episodes of asymptomatic viral shedding and therefore, the horizontal and vertical transmission rate in the population. Recent efforts have included the use of a recombinant gB/gD combined subunit vaccine. However, a large clinical trial showed that this offered no protection, despite eliciting strong neutralising antibody responses (Corey et al., 1999). Another recombinant gD vaccine, given with a more Th1-inducing adjuvant, demonstrated protection from HSV-2 disease only in women who were seronegative for HSV-1 and -2 at the time of vaccination (Stanberry et al., 2002). Interestingly, discovery of a recent locus of resistance to HSV in mice was also sex linked, affording protection in females

(Lundberg et al., 2003b). Animal studies demonstrated that the potency of the vaccine was dependent on the use of a Th1 skewing adjuvant (Bourne et al., 2003). Chapter 1 detailed how DC are major determinants of the Th1/Th2 balance in vivo, underlying that modulation of DC function may be important in improving vaccine design.

One approach to target DC in vivo is through DNA vaccination (Larregina et al., 2001). An inherent advantage of DNA over recombinant proteins is that the bacterial DNA from which plasmids are derived possesses intrinsic Th1 inducing adjuvanticity (Hemmi et al., 2000), as they are rich in hypomethylated CpG motifs that activate innate immunity via TLR9 ligation (Harandi et al., 2003). In addition, viral antigens may be administered together with genes coding for molecules which further enhance the function of DC as a "natural adjuvant". The combination of viral and immunological genes may ensure that DC transmit full activation signals to T cells, minimising the induction of regulatory T cells, which could limit the potency of the anti-HSV immune response both qualitatively and quantitatively (Suvas et al., 2003). Vaccination with DNA encoding IL-12 and IL-18 can result in DC transfection and expression of these cytokines at the time of T cell contact, skewing towards a Th1 response (Lee et al., 2003a). Enhanced migration of DC to secondary lymphoid organs has also been achieved, by immunising with DNA encoding the chemokine genes, CCL19 and CCL21 (Lee et al., 2003b), presumably by enhancing the source of these chemokines in the LN (Gunn et al., 1998; Ngo et al., 1998), recruiting further DC and T cells (Forster et al., 1999; Sallusto et al., 1999).

The potency of immunity is further enhanced by combining DNA vaccines with attenuated recombinant poxviruses in heterologous prime boost strategies (Toka et al., 2003). The mechanisms responsible for this enhanced efficacy are unclear but may relate to the interaction of DC with poxviruses. Wild type VV does not induce DC maturation (Engelmayer et al., 1999), but DC infected with the attenuated modified vaccinia virus Ankara (MVA) strain or with canarypox are induced to mature (Drillien et al., 2004; Ignatius et al., 2000). This may also explain the protection offered against lethal HSV-1 infection by the adoptive transfer of DC from MVA-treated mice (Franchini et al., 2004).

In addition, the data presented in chapter 6 illustrates that gD itself may possess adjuvant properties. It would be ideal to conceive gD as both adjuvant and antigen, but the data from vaccine trials using recombinant gD suggest that this does not occur to a significant degree in vivo. One possibility is that insufficient gD is delivered to DC in adjuvant form to stimulate them. In this respect, the role of other envelope glycoproteins and their mechanism of interaction with DC receptors that induces maturation will need to be evaluated, as they may synergise in transmitting positive signals to DC (see section 6.3.5).

7.3.3 Use of HSV-1 as a gene delivery vector

HSV-1 has been proposed to be a suitable vector for the delivery of antigens to DC. Its ability to infect DC efficiently and express engineered transgenes to a high degree (as determined by GFP expression in this thesis) are attractive properties as an ex vivo gene delivery vector (Coffin et al., 1998). The relative advantages and disadvantages of HSV-1 as a gene delivery vector relative to other viruses and non-viruses is beyond the scope of this thesis, and has been reviewed elsewhere recently (Lachmann, 2004). Nevertheless, data presented here demonstrate that the virus' ability to disrupt the function of DC requires these pathogenic viral factors to be identified and removed, in order to achieve a successful therapeutic application. Initial attempts have suggested that it is feasible to produce efficient, yet attenuated, HSV-1 vectors for gene delivery to DC (Samady et al., 2003). The deletion of IE genes to attenuate the pathogenicity of HSV-1 is consistent with the data using UV-HSV in chapter 6.

One way in which it has been proposed to exploit HSV infection of DC is to transfect the cells ex vivo, and then reintroduce them subcutaneously. As a result, a critical step in the success of this approach is subsequent emigration of DC from the skin to draining LNs. Recent evidence has shown that this process is highly inefficient in immature DC (de Vries et al., 2003). Maturation of DC induced by HSV-1 (chapter 6) suggests that the initial virus activation may prime DC for increased migration potential, enhancing the number of transfected DC reaching LNs, and stimulating appropriate antiviral, and therefore anti-transgene, immune responses.

7.4 Central role of DC in the pathogenesis of viral infections

The model described above (section 7.2) of a human peripheral herpetic lesion is partly based on data acquired in mice studies. The advantages of mouse models include the evaluation of disease outcomes in the presence of an intact immune system. However, many human viruses are restricted to infect human cells only, requiring the use of other members of the same virus family to infect primates (e.g. SIV infection of macaques), or transgenic mouse models (e.g. CD46 transgenic mice susceptible to MV infection (Marie et al., 2001)). These scenarios are imperfect, as they require extrapolation to humans or interpretation of virus infections in a host the virus has not evolved to replicate in. Where mouse models are available, the virus' life cycle may be different to that in humans. For example, HSV-1 ICP47 binds human TAP to prevent MHC class I loading and expression, but binds and inhibits murine TAP much less efficiently (Tomazin et al., 1998). The murine equivalent of HCMV, murine CMV (MCMV), uses different immunomodulatory genes and mechanisms to HCMV (Tortorella et al., 2000).

The analysis of ex-vivo derived human DC has also been be carried out but its usefulness is limited only to certain scenarios, particularly in systemic infections, such as HCV or HIV, where DC in circulation are infected by the virus (Navas et al., 2002; Goutagny et al., 2003; Donaghy et al., 2003). Changes in the numbers of DC in peripheral blood of these patients can also be used to infer the role of these cells in the control of the viral infection (Pichyangkul et al., 2003). However, this analysis must be interpreted cautiously, as changes in DC numbers could occur equally from suppression of haematopoiesis, or extravasation of DC to infected lesions in the periphery or into secondary lymphoid organs.

In light of the difficulties in interpreting animal and ex-vivo derived DC data, extrapolation of in vitro interaction of DC with viruses can also aid understanding the functional consequence in vivo. Analysis of disease outcomes may allow the result of in vitro infection to predict the role of DC infection in vivo. For example, DV and hantavirus, despite being of distinct viral phylogeny, both result in haemorrhagic fevers in man. Upon DC infection, both viruses induce maturation and elevate the resultant capacity to stimulate T cells (Ho et al., 2001; Raftery et al., 2002), suggesting

that DC infected with these viruses may accelerate the generation of the immunopathology responsible for the infection morbidity. Conversely, it could be that certain viruses have evolved to enhance or dampen the function of DC specifically in order to, increase their own survival and transmission. This would be most clearly evident by considering the topic from within specific virus families, and therefore this approach is used to explore the question below.

7.4.1 Herpesviruses

HSV-1, HSV-2, VZV, HCMV or HHV-6 infection of DC results in an impaired ability to stimulate T cell proliferation (Chapter 4) (Jones et al., 2003; Morrow et al., 2003; Moutaftsi et al., 2002; Kakimoto et al., 2002). Not only would this delay the generation of the antiviral response, but the infection of DC may also be central to the survival and transmission of these viruses in vivo. This is supported by analysing viral effects on various levels of the DC lifecycle.

It has been suggested that trans-infection of T cells from VZV infected DC is a critical initial step required for T cell mediated dissemination of the virus to the skin, resulting in the characteristic skin vesicular rash that may be important for the transmission of the virus (Abendroth et al., 2001; Ku et al., 2004). Although the infection of T cells by other herpesviruses has not yet been documented, the ability of HCMV to bind DC-SIGN suggests that this receptor may aid HCMV infection of T cells in trans, analogously to that seen with HIV-1 (Halary et al., 2002; Geijtenbeek et al., 2000a). Infection of T cells is not seen following co-culture with HSV-1 infected DC in vitro (chapter 4), but infected T cells could have been deleted by fratricide prior to detection (Raftery et al., 1999).

Some herpesviruses also prevent the differentiation of DC from their precursors. Both EBV and HCMV can infect monocytes and prevent their differentiation into immature DC (Li et al., 2002; Gredmark and Soderberg-Naucler, 2003; Guerreiro-Cacais et al., 2004). A decrease in DC numbers may allow sufficient time for EBV and HCMV to establish latency. As TLR ligands can also prevent the differentiation of monocytes into DC (Xie et al., 2003), it is not yet possible to determine whether this effect is

virally driven or secondary to the viral induced activation of monocytes (D'Addario et al., 2000; Simmen et al., 2001).

Herpesviruses may also have modified the expression of molecules important in the DC-T cell interaction. HSV-1 and HCMV infected DC are refractory to increases in co-stimulatory molecule expression after DC maturation (Chapter 4) (Salio et al., 1999; Moutaftsi et al., 2002). It is interesting to note that both HSV-1 and VZV specifically downregulate CD83 upon infection of mature DC (Kruse et al., 2000; Morrow et al., 2003). The functional significance of this event is not yet established but its conservation among these two α -herpesviruses suggests that it may favour their survival. In addition, HCMV infected DC become cytotoxic and kill surrounding T cells by FasL and TRAIL dependent mechanisms. Though not all T cells die in this manner, the ones that escape are anergic, suggesting that regulatory cytokines may also be secreted from HCMV infected DC (Raftery et al., 2001). Recent data demonstrating that HSV-1 also induces upregulation of TRAIL and Fas in infected DC suggests that this virus may utilise similar mechanisms to disrupt the proliferation of T cells (Muller et al., 2004), and that herpesviruses may have evolved to target the DC-T cell interaction specifically.

DC may also be central to herpesvirus establishment of latency, a characteristic of all herpesviruses. While EBV is latent in B cells, one of the HCMV reservoirs are myeloid DC progenitor cells, thus potentially undergoing latency in the precursors of DC themselves (Hahn et al., 1998). The central role of DC (specifically LC) in the establishment of HSV-1 latency was discussed in section 7.2.1.

7.4.2 Paramyxoviruses

MV, parainfluenza virus (PIV) and RSV all impair infected DC ability to activate T cells (Grosjean et al., 1997; Fugier-Vivier et al., 1997; Plotnicky-Gilquin et al., 2001; Bartz et al., 2003). Although the viral strategies used by these viruses are similar to those used by herpesviruses, such as TRAIL mediated induction of T cell apoptosis by MV infected DC (Vidalain et al., 2000a), distinct differences are also apparent. Despite inducing a more mature phenotype than uninfected DC (Servet-Delprat et al., 2000b),

the overall reduced ability of MV to stimulate T cell proliferation can be attributed at least partly to a direct interaction with MV NP, MV HA and MV F on DC or T cells (Klagge et al., 2000; Marie et al., 2001; Dubois et al., 2001). This mechanism can disrupt IL-12 secretion (Fugier-Vivier et al., 1997; Marie et al., 2001), and may also dampen T cell responses to irrelevant antigens non-specifically, possibly explaining post-MV infection immunosuppression (Marie et al., 2001; Klagge et al., 2000). A similar paradox between DC maturation and impaired T cell stimulation is seen following PIV infection of DC. UV-inactivation of PIV does not prevent the functional inhibition, suggesting that envelope glycoprotein-mediated immunosuppression, similar to MV, may also occur with this virus (Plotnicky-Gilquin et al., 2001).

The CD40 signalling pathway may be targeted by paramyxoviruses. CD40 ligation enhances MV replication (Fugier-Vivier et al., 1997) and CD40L stimulated IL-12 secretion by MV-infected DC is inhibited, in contrast to the enhanced secretion with LPS stimulation (Fugier-Vivier et al., 1997; Schnorr et al., 1997). Following RSV infection, DC are able to enhance secretion of IL-12p70 and T cell stimulation in response to poly (I:C) stimulation, but not to CD40L (Bartz et al., 2003). For both MV and RSV, a gene product appears to play a role, as UV-inactivation of the viruses negated the effect by RSV completely and also partially inhibited that of MV. Coupling these observations with data demonstrating enhanced MV replication in DC after CD40 ligation, it suggests that homologous mechanisms may exist within this virus family to target the CD40 signalling pathway in DC, both to suppress the function of the cell and to enhance the virus' own replication.

7.4.3 Retroviruses

HIV-1 results in immunodeficiency predominantly as a result of the long-term depletion of CD4+ T cells. A hypothetical role for DC in the initiation of this immunosuppression has been proposed for many years. The functional consequences of DC exposure to HIV-1 are still controversial, with some studies describing no inhibition in T cell stimulatory capacity (Sapp et al., 1999; Chougnet et al., 1999), and others showing reduced DC function (Donaghy et al., 2003; Granelli-Piperno et al., 2004). HIV-1 replication in DC alone is low compared to that observed in T cells, but

this is increased following ligation with activated T cells (Granelli-Piperno et al., 1999; MacDougall et al., 2002). This increased inoculum of virus could infect neighbouring CD4+ T cells and demonstrates a similar adaptation as seen for MV (Fugier-Vivier et al., 1997). The ability of HIV-1 to bind DC-SIGN and infect T cells in trans has led to a suggestion that HIV-1 has evolved to exploit DC as a vector to LNs (Geijtenbeek et al., 2000a). Recent evidence has suggested that the HIV-1 protein Nef actively promotes the upregulation and polarisation of DC-SIGN towards T cells in a heterotypic cluster (Sol-Foulon et al., 2002). Along with DC-SIGN mediated signals impinging on DC IL-12 secretion (Geijtenbeek et al., 2003), these observations support the importance of DC-SIGN in the pathogenesis of HIV-1 infection, and may explain the protection from HIV-1 infection offered by DC-SIGN variants (Liu et al., 2004).

DC may also be central to the dissemination of another retrovirus, HTLV-1. DC infection in mucosae may transport the virus to lymphoid organs, where DC scanning of a large number of T cells results in the establishment of numerous cell-to-cell contacts that the virus could exploit to infect many T cells (Igakura et al., 2003). This virus has been linked to the development of an aggressive leukaemia, ATL, in 0.05-0.1% of infected individuals (Tajima, 1990). Non-specific viral impairment of DC function may play a role in its pathogenesis, as asymptomatic HTLV-1 infected individuals have reduced T cell responses to EBV and PPD antigens (Katsuki et al., 1987; Tachibana et al., 1988). In addition, CD4+ T cells from ATL patients are deficient in CD40L expression (Makino et al., 2001), a molecule that is induced by interaction with DC (Johnson-Leger et al., 1998; Peng et al., 1998), possibly explaining the opportunistic infections that occur in ATL patients (Shimoyama, 1991). In this respect, it is notable that DC derived from monocytes infected with HTLV-1 in vitro, or derived from monocytes of HTLV-1 patients, are poor stimulators of HTLV-1 specific T cell responses (Makino et al., 2000). In some individuals, a consequential impaired DC stimulation of CD8+ CTL could reduce the ability to control escape mutants that may give rise to ATL (Furukawa et al., 2001).

7.4.4 Other viruses

The filoviruses, Ebola and Marburg viruses, result in fatal haemorrhagic fevers that are a consequence of endothelial cell mediated coagulopathy induced by cytokines of the innate immune system (Geisbert et al., 2003b). However, strong cellular adaptive immune responses correlate with a positive outcome (Jahrling et al., 1996; Baize et al., 1999). In vitro infection of DC with Ebola and Marburg viruses results in impaired stimulation of T cells (Bosio et al., 2003). Recent studies of nonhuman primate models of infection have identified DC as early targets for Ebola virus infection (Geisbert et al., 2003a). Furthermore, Ebola virus infection of DC via DC-SIGN (Alvarez et al., 2002) may facilitate trans infection of T cells, possibly explaining the high level of lymphocyte apoptosis observed in lymphoid organs (Geisbert et al., 2003a). Therefore, specific filovirus targeting of DC in vivo may delay the generation of a rapid and potent antiviral response necessary to clear the infection.

Poxviruses possess a large genome and subsequently a vast array of immunomodulatory molecules, some of which may affect the function of DC specifically. These can modulate MHC class I presentation, and cytokine, chemokine and complement networks (Seet et al., 2003). However, although several aspects of DC physiology could be targeted, VV infection of DC reveals a selective downregulation of CD80 and CD83, with minor effect on MHC and CD86 expression (Jenne et al., 2000). The change in expression of certain molecules expressed by professional antigen presenting cells may be advantageous to the virus' survival and suggests selective viral targeting of DC molecules.

7.5 Future work

This project has generated many questions that would be of interest to follow up through further experiments. Many of the finer details of future directions and of lack of investigation in the literature have been addressed in the relevant chapters. Below are outlined a few broad themes that may be of particular interest.

One important question that remains unanswered are the early effects in vivo on the morphology and migration of DC exposed to HSV-1. Studying chemokine-driven

migration of DC shortly after infection in vitro may be a useful approach, and human skin explants would enable the study of this phenomenon in a more relevant setting. Both these approaches may shed light on whether the morphological changes induced by the virus are critical to prevent migration out of the skin or in the antigen presentation setting in LNs. The added advantage of the skin model would be to demonstrate human DC infected with HSV-1 for the first time, possibly aided by the use of a GFP expressing virus (Larregina et al., 2001).

Simpler multicellular in vitro systems can also be set up to study the cross talk between cytokines/chemokines secreted by infected keratinocytes and DC. It would be interesting to study how the combination of signals emanating from these cells affects the Th skew of the ensuing antiviral T cell response initiated by DC. In a similar manner, the autocrine/paracrine role of type I IFNs in the amplification of immune responses to ligands of the innate immune system has not been documented in detail in humans and deserves further attention.

The interaction between virus and DC should also be examined further. For the use of DC as a vector, work understanding the inhibitory viral genes is ongoing in a collaborating laboratory and may both identify homologous mechanisms used within the herpesvirus family and also highlight novel regulatory components of DC functions. Ultimately, microarray studies of mutant viruses may help identify cellular pathways that are suppressed by the virus. Parallel viral gene expression profiling would both determine the HSV-1 genes that are expressed in DC, and also suggest candidate molecules for the changes in the cellular transcriptome identified.

Future work should also identify the roles of viral envelope glycoproteins and DC receptors responsible for the activation of DC. In particular, this study should be extended to a more detailed analysis of the functional consequence of this interaction. The cytokines secreted and intracellular signalling pathways activated would aid in shaping future vaccine designs. For example, glycoproteins can be designed with greater affinity for one receptor over another (Whitbeck et al., 1997; Krummenacher et al., 1999), shaping their adjuvant properties.

Finally, establishing a role in humans for DC in the regulation of HSV-1 infection needs to be determined. In particular, the role of DC in the establishment of latency after primary infection, and in the recurrence of symptomatic infection after viral reactivation, needs to be investigated. One could envisage that a dual chamber co-culture system of human skin explants and peripheral nerves may provide a useful approach (Mikloska and Cunningham, 2001). However, a central role of DC in the pathogenesis of HSV-1 infection may also be derived from genetic epidemiological approaches involving individuals infected with HSV-1 with varying disease severity and frequency of symptomatic recurrence. A relationship between susceptibility and variation in molecules central to the function of DC would support the hypothesis that this cell type does indeed play a critical role in the control and outcome of this infection in man.

References

References

-

.

Abe,K. and Saito,H. (1998). Amyloid [beta] protein inhibits cellular MTT reduction not by suppression of mitochondrial succinate dehydrogenase but by acceleration of MTT formazan exocytosis in cultured rat cortical astrocytes. Neuroscience Research 31, 295-305.

Abendroth, A., Morrow, G., Cunningham, A.L., and Slobedman, B. (2001). Varicella-zoster virus infection of human dendritic cells and transmission to T cells: implications for virus dissemination in the host. J Virol. 75, 6183-6192.

Ackerman, A.L., Kyritsis, C., Tampe, R., and Cresswell, P. (2003). Early phagosomes in dendritic cells form a cellular compartment sufficient for cross presentation of exogenous antigens. Proc. Natl. Acad. Sci U. S. A. *100*, 12889-12894.

Agnese, D.M., Calvano, J.E., Hahm, S.J., Coyle, S.M., Corbett, S.A., Calvano, S.E., and Lowry, S.F. (2002). Human toll-like receptor 4 mutations but not CD14 polymorphisms are associated with an increased risk of gram-negative infections. J Infect. Dis. *186*, 1522-1525.

Agrawal,S., Agrawal,A., Doughty,B., Gerwitz,A., Blenis,J., Van Dyke,T., and Pulendran,B. (2003). Cutting edge: different Toll-like receptor agonists instruct dendritic cells to induce distinct Th responses via differential modulation of extracellular signal-regulated kinase-mitogen-activated protein kinase and c-Fos. J Immunol *171*, 4984-4989.

Ahmad-Nejad, P., Hacker, H., Rutz, M., Bauer, S., Vabulas, R.M., and Wagner, H. (2002). Bacterial CpG-DNA and lipopolysaccharides activate Toll-like receptors at distinct cellular compartments. Eur. J Immunol *32*, 1958-1968.

Ahmed, M., Lock, M., Miller, C.G., and Fraser, N.W. (2002). Regions of the herpes simplex virus type 1 latency-associated transcript that protect cells from apoptosis in vitro and protect neuronal cells in vivo. J Virol 76, 717-729.

Ahn,K., Meyer,T.H., Uebel,S., Sempe,P., Djaballah,H., Yang,Y., Peterson,P.A., Fruh,K., and Tampe,R. (1996). Molecular mechanism and species specificity of TAP inhibition by herpes simplex virus ICP47. EMBO J. 15, 3247-3255.

Aiba,S., Nakagawa,S., Ozawa,H., Miyake,K., Yagita,H., and Tagami,H. (1993). Up-regulation of alpha 4 integrin on activated Langerhans cells: analysis of adhesion molecules on Langerhans cells relating to their migration from skin to draining lymph nodes. J. Invest Dermatol. *100*, 143-147.

Aicher, A., Shu, G.L., Magaletti, D., Mulvania, T., Pezzutto, A., Craxton, A., and Clark, E.A. (1999). Differential role for p38 mitogen-activated protein kinase in regulating CD40-induced gene expression in dendritic cells and B cells. J. Immunol. *163*, 5786-5795.

Akashi,K., Traver,D., Miyamoto,T., and Weissman,I.L. (2000). A clonogenic common myeloid progenitor that gives rise to all myeloid lineages. Nature 404, 193-197.

Aksoy, E., Amraoui, Z., Goriely, S., Goldman, M., and Willems, F. (2002). Critical role of protein kinase C epsilon for lipopolysaccharide-induced IL-12 synthesis in monocyte-derived dendritic cells. Eur. J Immunol *32*, 3040-3049.

Al Alwan, M.M., Liwski, R.S., Haeryfar, S.M., Baldridge, W.H., Hoskin, D.W., Rowden, G., and West, K.A. (2003). Cutting edge: dendritic cell actin cytoskeletal polarization during immunological synapse formation is highly antigen-dependent. J Immunol 171, 4479-4483.

Al Alwan, M.M., Rowden, G., Lee, T.D., and West, K.A. (2001). The dendritic cell cytoskeleton is critical for the formation of the immunological synapse. J Immunol *166*, 1452-1456.

Albanesi, C., Cavani, A., and Girolomoni, G. (1998). Interferon-gamma-stimulated human keratinocytes express the genes necessary for the production of peptide-loaded MHC class II molecules. J Invest. Dermatol. *110*, 138-142.

Albert, M.L., Jegathesan, M., and Darnell, R.B. (2001). Dendritic cell maturation is required for the cross-tolerization of CD8+ T cells. Nat. Immunol 2, 1010-1017.

Albert,M.L., Pearce,S.F., Francisco,L.M., Sauter,B., Roy,P., Silverstein,R.L., and Bhardwaj,N. (1998a). Immature dendritic cells phagocytose apoptotic cells via alphavbeta5 and CD36, and cross-present antigens to cytotoxic T lymphocytes. J Exp. Med *188*, 1359-1368.

Albert, M.L., Sauter, B., and Bhardwaj, N. (1998b). Dendritic cells acquire antigen from apoptotic cells and induce class I-restricted CTLs. Nature *392*, 86-89.

Albrecht, M.A., DeLuca, N.A., Byrn, R.A., Schaffer, P.A., and Hammer, S.M. (1989). The herpes simplex virus immediate-early protein, ICP4, is required to potentiate replication of human immunodeficiency virus in CD4+ lymphocytes. J Virol *63*, 1861-1868.

Alderman, C.J., Bunyard, P.R., Chain, B.M., Foreman, J.C., Leake, D.S., and Katz, D.R. (2002). Effects of oxidised low density lipoprotein on dendritic cells: a possible immunoregulatory component of the atherogenic micro-environment? Cardiovasc. Res. *55*, 806-819.

Alexopoulou,L., Holt,A.C., Medzhitov,R., and Flavell,R.A. (2001). Recognition of double-stranded RNA and activation of NF-kappaB by Toll-like receptor 3. Nature *413*, 732-738.

Aliberti, J., Schulz, O., Pennington, D.J., Tsujimura, H., Sousa, C.R., Ozato, K., and Sher, A. (2003). Essential role for ICSBP in the in vivo development of murine CD8alpha + dendritic cells. Blood 101, 305-310.

Allan,R.S., Smith,C.M., Belz,G.T., van Lint,A.L., Wakim,L.M., Heath,W.R., and Carbone,F.R. (2003). Epidermal viral immunity induced by CD8alpha+ dendritic cells but not by Langerhans cells. Science 301, 1925-1928.

Allavena, P., Piemonti, L., Longoni, D.; Bernasconi, S., Stoppacciaro, A., Ruco, L., and Mantovani, A. (1998). IL-10 prevents the differentiation of monocytes to dendritic cells but promotes their maturation to macrophages. Eur. J Immunol 28, 359-369.

Alvarez, C.P., Lasala, F., Carrillo, J., Muniz, O., Corbi, A.L., and Delgado, R. (2002). C-type lectins DC-SIGN and L-SIGN mediate cellular entry by Ebola virus in cis and in trans. J Virol. *76*, 6841-6844.

Ameglio, F., Capobianchi, M.R., Castilletti, C., Cordiali, F.P., Fais, S., Trento, E., and Dianzani, F. (1994). Recombinant gp120 induces IL-10 in resting peripheral blood mononuclear cells; correlation with the induction of other cytokines. Clin. Exp. Immunol *95*, 455-458.

Amici, C., Belardo, G., Rossi, A., and Santoro, M.G. (2001). Activation of I kappa b kinase by herpes simplex virus type 1. A novel target for anti-herpetic therapy. J Biol. Chem. 276, 28759-28766.

Ammon, C., Meyer, S.P., Schwarzfischer, L., Krause, S.W., Andreesen, R., and Kreutz, M. (2000). Comparative analysis of integrin expression on monocyte-derived macrophages and monocyte-derived dendritic cells. Immunology *100*, 364-369.

An,H., Yu,Y., Zhang,M., Xu,H., Qi,R., Yan,X., Liu,S., Wang,W., Guo,Z., Guo,J., Qin,Z., and Cao,X. (2002). Involvement of ERK, p38 and NF-kappaB signal transduction in regulation of TLR2, TLR4 and TLR9 gene expression induced by lipopolysaccharide in mouse dendritic cells. Immunology *106*, 38-45.

Andreakos, E., Sacre, S.M., Smith, C., Lundberg, A., Kiriakidis, S., Stonehouse, T., Monaco, C., Feldmann, M., and Foxwell, B.M. (2004). Distinct pathways of LPS-induced NF-{kappa}B activation and cytokine production in human myeloid and nonmyeloid cells defined by selective utilization of MyD88 and Mal/TIRAP. Blood *103*, 2229-2237.

Andreakos, E., Smith, C., Monaco, C., Brennan, F.M., Foxwell, B.M., and Feldmann, M. (2003). Ikappa B kinase 2 but not NF-kappa B-inducing kinase is essential for effective DC antigen presentation in the allogeneic mixed lymphocyte reaction. Blood *101*, 983-991.

Andrews, D.M., Andoniou, C.E., Granucci, F., Ricciardi-Castagnoli, P., and Degli-Esposti, M.A. (2001). Infection of dendritic cells by murine cytomegalovirus induces functional paralysis. Nat. Immunol 2, 1077-1084.

Ankel,H., Westra,D.F., Welling-Wester,S., and Lebon,P. (1998). Induction of interferon-alpha by glycoprotein D of herpes simplex virus: a possible role of chemokine receptors. Virology 251, 317-326.

Ardeshna,K.M., Pizzey,A.R., Devereux,S., and Khwaja,A. (2000). The PI3 kinase, p38 SAP kinase, and NF-kappaB signal transduction pathways are involved in the survival and maturation of lipopolysaccharide-stimulated human monocyte-derived dendritic cells. Blood *96*, 1039-1046.

Arrighi,J.F., Rebsamen,M., Rousset,F., Kindler,V., and Hauser,C. (2001). A critical role for p38 mitogen-activated protein kinase in the maturation of human blood-derived dendritic cells induced by lipopolysaccharide, TNF-alpha, and contact sensitizers. J. Immunol. *166*, 3837-3845.

Arrode, G., Boccaccio, C., Abastado, J.P., and Davrinche, C. (2002). Cross-presentation of human cytomegalovirus pp65 (UL83) to CD8+ T cells is regulated by virus-induced, soluble-mediator-dependent maturation of dendritic cells. J Virol. *76*, 142-150.

Arrode, G., Boccaccio, C., Lule, J., Allart, S., Moinard, N., Abastado, J.P., Alam, A., and Davrinche, C. (2000). Incoming Human Cytomegalovirus pp65 (UL83) Contained in Apoptotic Infected Fibroblasts Is Cross-Presented to CD8+ T Cells by Dendritic Cells. J. Virol. 74, 10018-10024.

Asanuma,H., Sharp,M., Maecker,H.T., Maino,V.C., and Arvin,A.M. (2000). Frequencies of memory T cells specific for varicella-zoster virus, herpes simplex virus, and cytomegalovirus by intracellular detection of cytokine expression. J Infect. Dis. *181*, 859-866.

Ashley, R., Benedetti, J., and Corey, L. (1985). Humoral immune response to HSV-1 and HSV-2 viral proteins in patients with primary genital herpes. J Med. Virol *17*, 153-166.

Ashley, R.L., Corey, L., Dalessio, J., Wilson, P., Remington, M., Barnum, G., and Trethewey, P. (1994). Protein-specific cervical antibody responses to primary genital herpes simplex virus type 2 infections. J Infect. Dis. 170, 20-26.

Asselin-Paturel, C., Boonstra, A., Dalod, M., Durand, I., Yessaad, N., Dezutter-Dambuyant, C., Vicari, A., O'Garra, A., Biron, C., Briere, F., and Trinchieri, G. (2001). Mouse type I IFN-producing cells are immature APCs with plasmacytoid morphology. Nat. Immunol *2*, 1144-1150.

Athie-Morales, V., Smits, H.H., Cantrell, D.A., and Hilkens, C.M. (2004). Sustained IL-12 signaling is required for Th1 development. J Immunol 172, 61-69.

Aubert, M. and Blaho, J.A. (1999). The herpes simplex virus type 1 regulatory protein ICP27 is required for the prevention of apoptosis in infected human cells. J. Virol. 73, 2803-2813.

Aubert, M. and Blaho, J.A. (2001). Modulation of apoptosis during herpes simplex virus infection in human cells. Microbes. Infect. *3*, 859-866.

Aubert, M., O'Toole, J., and Blaho, J.A. (1999). Induction and prevention of apoptosis in human HEp-2 cells by herpes simplex virus type 1. J. Virol. 73, 10359-10370.

Aubert, M., Rice, S.A., and Blaho, J.A. (2001). Accumulation of herpes simplex virus type 1 early and leaky-late proteins correlates with apoptosis prevention in infected human HEp-2 cells. J Virol 75, 1013-1030.

Bachmann,M.F., Lutz,M.B., Layton,G.T., Harris,S.J., Fehr,T., Rescigno,M., and Ricciardi-Castagnoli,P. (1996). Dendritic cells process exogenous viral proteins and virus-like particles for class I presentation to CD8+ cytotoxic T lymphocytes. Eur. J Immunol *26*, 2595-2600.

Baize,S., Leroy,E.M., Georges-Courbot,M.C., Capron,M., Lansoud-Soukate,J., Debre,P., Fisher-Hoch,S.P., McCormick,J.B., and Georges,A.J. (1999). Defective humoral responses and extensive intravascular apoptosis are associated with fatal outcome in Ebola virus-infected patients. Nat. Med. *5*, 423-426.

Bakri,Y., Schiffer,C., Zennou,V., Charneau,P., Kahn,E., Benjouad,A., Gluckman,J.C., and Canque,B. (2001). The maturation of dendritic cells results in postintegration inhibition of HIV-1 replication. J Immunol *166*, 3780-3788.

Banchereau, J. and Steinman, R.M. (1998). Dendritic cells and the control of immunity. Nature 392, 245-252.

Bangert, C., Friedl, J., Stary, G., Stingl, G., and Kopp, T. (2003). Immunopathologic features of allergic contact dermatitis in humans: participation of plasmacytoid dendritic cells in the pathogenesis of the disease? J Invest. Dermatol. *121*, 1409-1418.

Banki,Z., Kacani,L., Mullauer,B., Wilflingseder,D., Obermoser,G., Niederegger,H., Schennach,H., Sprinzl,G.M., Sepp,N., Erdei,A., Dierich,M.P., and Stoiber,H. (2003). Cross-linking of CD32 induces maturation of human monocyte-derived dendritic cells via NF-kappa B signaling pathway. J Immunol *170*, 3963-3970.

Barchet, W., Cella, M., Odermatt, B., Asselin-Paturel, C., Colonna, M., and Kalinke, U. (2002). Virusinduced Interferon {alpha} Production by a Dendritic Cell Subset in the Absence of Feedback Signaling In Vivo. J. Exp. Med. 195, 507-516.

Barcy,S. and Corey,L. (2001). Herpes simplex inhibits the capacity of lymphoblastoid B cell lines to stimulate CD4+ T cells. J Immunol *166*, 6242-6249.

Barry,S.C., Seppen,J., Ramesh,N., Foster,J.L., Seyama,K., Ochs,H.D., Garcia,J.V., and Osborne,W.R. (2000). Lentiviral and murine retroviral transduction of T cells for expression of human CD40 ligand. Hum. Gene Ther. 11, 323-332.

Bartholome,E.J., Willems,F., Crusiaux,A., Thielemans,K., Schandene,L., and Goldman,M. (1999). IFNbeta interferes with the differentiation of dendritic cells from peripheral blood mononuclear cells: selective inhibition of CD40-dependent interleukin-12 secretion. J Interferon Cytokine Res. *19*, 471-478. Bartz,H., Turkel,O., Hoffjan,S., Rothoeft,T., Gonschorek,A., and Schauer,U. (2003). Respiratory syncytial virus decreases the capacity of myeloid dendritic cells to induce interferon-gamma in naive T cells. Immunology *109*, 49-57.

Bauer, M., Redecke, V., Ellwart, J.W., Scherer, B., Kremer, J.P., Wagner, H., and Lipford, G.B. (2001). Bacterial CpG-DNA triggers activation and maturation of human CD11c-, CD123+ dendritic cells. J. Immunol. *166*, 5000-5007.

Beck,K., Meyer-Konig,U., Weidmann,M., Nern,C., and Hufert,F.T. (2003). Human cytomegalovirus impairs dendritic cell function: a novel mechanism of human cytomegalovirus immune escape. Eur. J. Immunol. *33*, 1528-1538.

Bedows, E., Rao, K.M., and Welsh, M.J. (1983). Fate of microfilaments in vero cells infected with measles virus and herpes simplex virus type 1. Mol. Cell. Biol. *3*, 712-719.

Belz,G.T., Smith,C.M., Kleinert,L., Reading,P., Brooks,A., Shortman,K., Carbone,F.R., and Heath,W.R. (2004). Distinct migrating and nonmigrating dendritic cell populations are involved in MHC class I-restricted antigen presentation after lung infection with virus. Proc. Natl. Acad. Sci U. S. A. 101, 8670-8675.

Benvenuti, F., Hugues, S., Walmsley, M., Ruf, S., Fetler, L., Popoff, M., Tybulewicz, V.L., and Amigorena, S. (2004a). Requirement of Rac1 and Rac2 expression by mature dendritic cells for T cell priming. Science *305*, 1150-1153.

Benvenuti,F., Lagaudriere-Gesbert,C., Grandjean,I., Jancic,C., Hivroz,C., Trautmann,A., Lantz,O., and Amigorena,S. (2004b). Dendritic cell maturation controls adhesion, synapse formation, and the duration of the interactions with naive T lymphocytes. J Immunol *172*, 292-301.

Bickham,K., Goodman,K., Paludan,C., Nikiforow,S., Tsang,M.L., Steinman,R.M., and Munz,C. (2003). Dendritic cells initiate immune control of epstein-barr virus transformation of B lymphocytes in vitro. J. Exp. Med. 198, 1653-1663.

Bieback,K., Lien,E., Klagge,I.M., Avota,E., Schneider-Schaulies,J., Duprex,W.P., Wagner,H., Kirschning,C.J., ter,M., V, and Schneider-Schaulies,S. (2002). Hemagglutinin protein of wild-type measles virus activates toll-like receptor 2 signaling. J. Virol. *76*, 8729-8736.

Biron, C.A., Byron, K.S., and Sullivan, J.L. (1989). Severe herpesvirus infections in an adolescent without natural killer cells. N Engl J Med *320*, 1731-1735.

Bjorck, P. (2004). Dendritic cells exposed to herpes simplex virus in vivo do not produce IFN-alpha after rechallenge with virus in vitro and exhibit decreased T cell alloreactivity. J Immunol *172*, 5396-5404.

Bjorck, P., Banchereau, J., and Flores-Romo, L. (1997). CD40 ligation counteracts Fas-induced apoptosis of human dendritic cells. Int. Immunol *9*, 365-372.

Blanco, P., Palucka, A.K., Gill, M., Pascual, V., and Banchereau, J. (2001). Induction of dendritic cell differentiation by IFN-alpha in systemic lupus erythematosus. Science 294, 1540-1543.

Blander, J.M. and Medzhitov, R. (2004). Regulation of phagosome maturation by signals from toll-like receptors. Science 304, 1014-1018.

Blom,B., Ligthart,S.J.W.C., Schotte,R., and Spits,H. (2002). Developmental origin of pre-DC2. Human Immunology 63, 1072-1080.

Bonifaz,L., Bonnyay,D., Mahnke,K., Rivera,M., Nussenzweig,M.C., and Steinman,R.M. (2002). Efficient targeting of protein antigen to the dendritic cell receptor DEC-205 in the steady state leads to antigen presentation on major histocompatibility complex class I products and peripheral CD8+ T cell tolerance. J Exp. Med. *196*, 1627-1638.

Boonstra,A., Asselin-Paturel,C., Gilliet,M., Crain,C., Trinchieri,G., Liu,Y.J., and O'Garra,A. (2003). Flexibility of mouse classical and plasmacytoid-derived dendritic cells in directing T helper type 1 and 2 cell development: dependency on antigen dose and differential toll-like receptor ligation. J. Exp. Med. *197*, 101-109.

Bose,S., Kar,N., Maitra,R., DiDonato,J.A., and Banerjee,A.K. (2003). Temporal activation of NF-{kappa}B regulates an interferon-independent innate antiviral response against cytoplasmic RNA viruses. PNAS 100, 10890-10895.

Bosio, C.M., Aman, M.J., Grogan, C., Hogan, R., Ruthel, G., Negley, D., Mohamadzadeh, M., Bavari, S., and Schmaljohn, A. (2003). Ebola and Marburg Viruses Replicate in Monocyte-Derived Dendritic Cells without Inducing the Production of Cytokines and Full Maturation. J Infect. Dis. *188*, 1630-1638.

Bourne,N., Bravo,F.J., Francotte,M., Bernstein,D.I., Myers,M.G., Slaoui,M., and Stanberry,L.R. (2003). Herpes simplex virus (HSV) type 2 glycoprotein D subunit vaccines and protection against genital HSV-1 or HSV-2 disease in guinea pigs. J. Infect. Dis. *187*, 542-549.

Bousso, P. and Robey, E. (2003). Dynamics of CD8+ T cell priming by dendritic cells in intact lymph nodes. Nat. Immunol 4, 579-585.

Boyle,K.A., Pietropaolo,R.L., and Compton,T. (1999). Engagement of the Cellular Receptor for Glycoprotein B of Human Cytomegalovirus Activates the Interferon-Responsive Pathway. Mol. Cell. Biol. 19, 3607-3613.

Brand, U., Bellinghausen, I., Enk, A.H., Jonuleit, H., Becker, D., Knop, J., and Saloga, J. (1998). Influence of extracellular matrix proteins on the development of cultured human dendritic cells. Eur. J. Immunol. 28, 1673-1680.

Briand,G., Barbeau,B., and Tremblay,M. (1997). Binding of HIV-1 to its receptor induces tyrosine phosphorylation of several CD4-associated proteins, including the phosphatidylinositol 3-kinase. Virology 228, 171-179.

Brinkmann, V., Geiger, T., Alkan, S., and Heusser, C.H. (1993). Interferon alpha increases the frequency of interferon gamma-producing human CD4+ T cells. J. Exp. Med. *178*, 1655-1663.

Brook, M., Sully, G., Clark, A.R., and Saklatvala, J. (2000). Regulation of tumour necrosis factor alpha mRNA stability by the mitogen-activated protein kinase p38 signalling cascade. FEBS. Lett. *483*, 57-61.

Brown,Z.A., Selke,S., Zeh,J., Kopelman,J., Maslow,A., Ashley,R.L., Watts,D.H., Berry,S., Herd,M., and Corey,L. (1997). The Acquisition of Herpes Simplex Virus during Pregnancy. N Engl J Med 337, 509-516.

Brunetti,C.R., Burke,R.L., Hoflack,B., Ludwig,T., Dingwell,K.S., and Johnson,D.C. (1995). Role of mannose-6-phosphate receptors in herpes simplex virus entry into cells and cell-to-cell transmission. J. Virol. *69*, 3517-3528.

Brunetti,C.R., Burke,R.L., Kornfeld,S., Gregory,W., Masiarz,F.R., Dingwell,K.S., and Johnson,D.C. (1994). Herpes simplex virus glycoprotein D acquires mannose 6-phosphate residues and binds to mannose 6-phosphate receptors. J. Biol. Chem. 269, 17067-17074.

Bruun, T., Kristoffersen, A.K., Rollag, H., and Degre, M. (1998). Interaction of herpes simplex virus with mononuclear phagocytes is dependent on the differentiation stage of the cells. APMIS. *106*, 305-314.

Bunyard,P., Handley,M., Pollara,G., Rutault,K., Wood,I., Chaudry,M., Alderman,C., Foreman,J., Katz,D.R., and Chain,B.M. (2003). Ribotoxic stress activates p38 and JNK kinases and modulates the antigen-presenting activity of dendritic cells. Mol. Immunol. *39*, 815-827.

Burns,S., Thrasher,A.J., Blundell,M.P., Machesky,L., and Jones,G.E. (2001). Configuration of human dendritic cell cytoskeleton by Rho GTPases, the WAS protein, and differentiation. Blood. *98*, 1142-1149.

Calistri, A., Parolin, C., Pizzato, M., Calvi, P., Giaretta, I., and Palu, G. (1999). Herpes simplex virus chronically infected human T lymphocytes are susceptible to HIV-1 superinfection and support HIV-1 pseudotyping. J Acquir. Immune. Defic. Syndr. 21, 90-98.
Canque,B., Camus,S., Dalloul,A., Kahn,E., Yagello,M., Dezutter-Dambuyant,C., Schmitt,D., Schmitt,C., and Gluckman,J.C. (2000). Characterization of dendritic cell differentiation pathways from cord blood CD34(+)CD7(+)CD45RA(+) hematopoietic progenitor cells. Blood *96*, 3748-3756.

Cantin, E., Tanamachi, B., and Openshaw, H. (1999). Role for gamma interferon in control of herpes simplex virus type 1 reactivation. J Virol 73, 3418-3423.

Cantin,E.M., Hinton,D.R., Chen,J., and Openshaw,H. (1995). Gamma interferon expression during acute and latent nervous system infection by herpes simplex virus type 1. J. Virol. *69*, 4898-4905.

Carfi,A., Willis,S.H., Whitbeck,J.C., Krummenacher,C., Cohen,G.H., Eisenberg,R.J., and Wiley,D.C. (2001). Herpes simplex virus glycoprotein D bound to the human receptor HveA. Mol. Cell *8*, 169-179.

Carra,G., Gerosa,F., and Trinchieri,G. (2000). Biosynthesis and posttranslational regulation of human IL-12. J Immunol 164, 4752-4761.

Carter,K.L., Ward,P.L., and Roizman,B. (1996). Characterization of the products of the U(L)43 gene of herpes simplex virus 1: potential implications for regulation of gene expression by antisense transcription. J Virol. 70, 7663-7668.

Carthy,C.M., Granville,D.J., Watson,K.A., Anderson,D.R., Wilson,J.E., Yang,D., Hunt,D.W., and McManus,B.M. (1998). Caspase activation and specific cleavage of substrates after coxsackievirus B3-induced cytopathic effect in HeLa cells. J Virol. 72, 7669-7675.

Caux,C., Vanbervliet,B., Massacrier,C., Dezutter-Dambuyant,C., Saint-Vis,B., Jacquet,C., Yoneda,K., Imamura,S., Schmitt,D., and Banchereau,J. (1996). CD34+ hematopoietic progenitors from human cord blood differentiate along two independent dendritic cell pathways in response to GM-CSF+TNF alpha. J. Exp. Med. *184*, 695-706.

Cella, M., Engering, A., Pinet, V., Pieters, J., and Lanzavecchia, A. (1997). Inflammatory stimuli induce accumulation of MHC class II complexes on dendritic cells. Nature *388*, 782-787.

Cella, M., Facchetti, F., Lanzavecchia, A., and Colonna, M. (2000). Plasmacytoid dendritic cells activated by influenza virus and CD40L drive a potent TH1 polarization. Nat. Immunol *1*, 305-310.

Cella, M., Jarrossay, D., Facchetti, F., Alebardi, O., Nakajima, H., Lanzavecchia, A., and Colonna, M. (1999a). Plasmacytoid monocytes migrate to inflamed lymph nodes and produce large amounts of type I interferon. Nat. Med. 5, 919-923.

Cella, M., Salio, M., Sakakibara, Y., Langen, H., Julkunen, I., and Lanzavecchia, A. (1999b). Maturation, activation, and protection of dendritic cells induced by double-stranded RNA. J Exp. Med. 189, 821-829.

Chan, C.W.Y., Kay, L.S., Khadaroo, R.G., Chan, M.W.C., Lakatoo, S., Young, K.J., Zhang, L., Gorczynski, R.M., Cattral, M., Rotstein, O., and Levy, G.A. (2003). Soluble Fibrinogen-Like Protein 2/Fibroleukin Exhibits Immunosuppressive Properties: Suppressing T Cell Proliferation and Inhibiting Maturation of Bone Marrow-Derived Dendritic Cells. J Immunol *170*, 4036-4044.

Chapuis, F., Rosenzwajg, M., Yagello, M., Ekman, M., Biberfeld, P., and Gluckman, J.C. (1997). Differentiation of human dendritic cells from monocytes in vitro. Eur. J. Immunol. 27, 431-441.

Chee, A.V. and Roizman, B. (2004). Herpes simplex virus 1 gene products occlude the interferon signaling pathway at multiple sites. J Virol 78, 4185-4196.

Chen,S.H., Kramer,M.F., Schaffer,P.A., and Coen,D.M. (1997). A viral function represses accumulation of transcripts from productive-cycle genes in mouse ganglia latently infected with herpes simplex virus. J Virol 71, 5878-5884.

Chen, S.H., Lee, L.Y., Garber, D.A., Schaffer, P.A., Knipe, D.M., and Coen, D.M. (2002). Neither LAT nor open reading frame P mutations increase expression of spliced or intron-containing ICP0 transcripts in mouse ganglia latently infected with herpes simplex virus. J Virol *76*, 4764-4772.

Chen, W., Calvo, P.A., Malide, D., Gibbs, J., Schubert, U., Bacik, I., Basta, S., O'Neill, R., Schickli, J., Palese, P., Henklein, P., Bennink, J.R., and Yewdell, J.W. (2001). A novel influenza A virus mitochondrial protein that induces cell death. Nat. Med. 7, 1306-1312.

Cheshenko, N., Del Rosario, B., Woda, C., Marcellino, D., Satlin, L.M., and Herold, B.C. (2003). Herpes simplex virus triggers activation of calcium-signaling pathways. J Cell. Biol. *163*, 283-293.

Chiang,H.Y., Cohen,G.H., and Eisenberg,R.J. (1994). Identification of functional regions of herpes simplex virus glycoprotein gD by using linker-insertion mutagenesis. J Virol *68*, 2529-2543.

Chieppa, M., Bianchi, G., Doni, A., Del Prete, A., Sironi, M., Laskarin, G., Monti, P., Piemonti, L., Biondi, A., Mantovani, A., Introna, M., and Allavena, P. (2003). Cross-linking of the mannose receptor on monocyte-derived dendritic cells activates an anti-inflammatory immunosuppressive program. J Immunol *171*, 4552-4560.

Chill,J.H., Quadt,S.R., Levy,R., Schreiber,G., and Anglister,J. (2003). The Human Type I Interferon Receptor: NMR Structure Reveals the Molecular Basis of Ligand Binding. Structure 11, 791-802.

Cho,S.S., Bacon,C.M., Sudarshan,C., Rees,R.C., Finbloom,D., Pine,R., and O'Shea,J.J. (1996). Activation of STAT4 by IL-12 and IFN-alpha: evidence for the involvement of ligand-induced tyrosine and serine phosphorylation. J Immunol *157*, 4781-4789.

Chou, J., Chen, J.J., Gross, M., and Roizman, B. (1995). Association of a M(r) 90,000 phosphoprotein with protein kinase PKR in cells exhibiting enhanced phosphorylation of translation initiation factor eIF-2

alpha and premature shutoff of protein synthesis after infection with gamma 134.5- mutants of herpes simplex virus 1. Proc. Natl. Acad. Sci. U. S. A. 92, 10516-10520.

Chou, J. and Roizman, B. (1992). The gamma 1(34.5) gene of herpes simplex virus 1 precludes neuroblastoma cells from triggering total shutoff of protein synthesis characteristic of programed cell death in neuronal cells. Proc. Natl. Acad. Sci. U. S. A 89, 3266-3270.

Chougnet, C., Cohen, S.S., Kawamura, T., Landay, A.L., Kessler, H.A., Thomas, E., Blauvelt, A., and Shearer, G.M. (1999). Normal immune function of monocyte-derived dendritic cells from HIV-infected individuals: implications for immunotherapy. J. Immunol. *163*, 1666-1673.

Clark,G.J., Gunningham,S., Troy,A., Vuckovic,S., and Hart,D.N. (1999). Expression of the RelB transcription factor correlates with the activation of human dendritic cells. Immunology *98*, 189-196.

Clemens, M.J. and Elia, A. (1997). The double-stranded RNA-dependent protein kinase PKR: structure and function. J. Interferon. Cytokine. Res. 17, 503-524.

Cocchi,F., Lopez,M., Menotti,L., Aoubala,M., Dubreuil,P., and Campadelli-Fiume,G. (1998a). The V domain of herpesvirus Ig-like receptor (HIgR) contains a major functional region in herpes simplex virus-1 entry into cells and interacts physically with the viral glycoprotein D. Proc. Natl. Acad. Sci U. S. A. 95, 15700-15705.

Cocchi,F., Menotti,L., Mirandola,P., Lopez,M., and Campadelli-Fiume,G. (1998b). The ectodomain of a novel member of the immunoglobulin subfamily related to the poliovirus receptor has the attributes of a bona fide receptor for herpes simplex virus types 1 and 2 in human cells. J Virol *72*, 9992-10002.

Coccia,E.M., Severa,M., Giacomini,E., Monneron,D., Remoli,M.E., Julkunen,I., Cella,M., Lande,R., and Uze,G. (2004). Viral infection and Toll-like receptor agonists induce a differential expression of type I and lambda interferons in human plasmacytoid and monocyte-derived dendritic cells. Eur. J Immunol *34*, 796-805.

Coffin,R.S., MacLean,A.R., Latchman,D.S., and Brown,S.M. (1996). Gene delivery to the central and peripheral nervous systems of mice using HSV1 ICP34.5 deletion mutant vectors. Gene Ther. *3*, 886-891.

Coffin,R.S., Thomas,S.K., Thomas,N.S., Lilley,C.E., Pizzey,A.R., Griffiths,C.H., Gibb,B.J., Wagstaff,M.J., Inges,S.J., Binks,M.H., Chain,B.M., Thrasher,A.J., Rutault,K., and Latchman,D.S. (1998). Pure populations of transduced primary human cells can be produced using GFP expressing herpes virus vectors and flow cytometry. Gene Ther. *5*, 718-722.

Collins,S.E., Noyce,R.S., and Mossman,K.L. (2004). Innate Cellular Response to Virus Particle Entry Requires IRF3 but Not Virus Replication. J. Virol. 78, 1706-1717.

Compton, T., Kurt-Jones, E.A., Boehrae, K.W., Belko, J., Latz, E., Golenbock, D.T., and Finberg, R.W. (2003). Human cytomegalovirus activates inflammatory cytokine responses via CD14 and Toll-like receptor 2. J. Virol. 77, 4588-4596.

Compton, T., Nowlin, D.M., and Cooper, N.R. (1993). Initiation of human cytomegalovirus infection requires initial interaction with cell surface heparan sulfate. Virology. *193*, 834-841.

Connolly,S.A., Whitbeck,J.J., Rux,A.H., Krummenacher,C., van Drunen Littel-van den Hurk, Cohen,G.H., and Eisenberg,R.J. (2001). Glycoprotein D homologs in herpes simplex virus type 1, pseudorabies virus, and bovine herpes virus type 1 bind directly to human HveC(nectin-1) with different affinities. Virology *280*, 7-18.

Corey,L., Langenberg,A.G., Ashley,R., Sekulovich,R.E., Izu,A.E., Douglas,J.M., Jr., Handsfield,H.H., Warren,T., Marr,L., Tyring,S., DiCarlo,R., Adimora,A.A., Leone,P., Dekker,C.L., Burke,R.L., Leong,W.P., and Straus,S.E. (1999). Recombinant glycoprotein vaccine for the prevention of genital HSV-2 infection: two randomized controlled trials. Chiron HSV Vaccine Study Group. JAMA 282, 331-340.

Cousens, L.P., Orange, J.S., Su, H.C., and Biron, C.A. (1997). Interferon-alpha/beta inhibition of interleukin 12 and interferon-gamma production in vitro and endogenously during viral infection. Proc. Natl. Acad. Sci. U. S. A 94, 634-639.

Crise,B., Buonocore,L., and Rose,J.K. (1990). CD4 is retained in the endoplasmic reticulum by the human immunodeficiency virus type 1 glycoprotein precursor. J Virol *64*, 5585-5593.

Cunningham, A.L. and Merigan, T.C. (1984). Leu-3+ T cells produce gamma-interferon in patients with recurrent herpes labialis. J Immunol *132*, 197-202.

Cunningham, A.L., Nelson, P.A., Fathman, C.G., and Merigan, T.C. (1985a). Interferon gamma production by herpes simplex virus antigen-specific T cell clones from patients with recurrent herpes labialis. J. Gen. Virol. 66 (*Pt 2*), 249-258.

Cunningham, A.L., Turner, R.R., Miller, A.C., Para, M.F., and Merigan, T.C. (1985b). Evolution of recurrent herpes simplex lesions. An immunohistologic study. J Clin. Invest 75, 226-233.

D'Addario, M., Ahmad, A., Morgan, A., and Menezes, J. (2000). Binding of the Epstein-Barr virus major envelope glycoprotein gp350 results in the upregulation of the TNF-alpha gene expression in monocytic cells via NF-kappaB involving PKC, PI3-K and tyrosine kinases. J Mol. Biol. *298*, 765-778.

D'amico,G., Bianchi,G., Bernasconi,S., Bersani,L., Piemonti,L., Sozzani,S., Mantovani,A., and Allavena,P. (1998). Adhesion, transendothelial migration, and reverse transmigration of in vitro cultured dendritic cells. Blood *92*, 207-214.

D'amico,G., Frascaroli,G., Bianchi,G., Transidico,P., Doni,A., Vecchi,A., Sozzani,S., Allavena,P., and Mantovani,A. (2000). Uncoupling of inflammatory chemokine receptors by IL-10: generation of functional decoys. Nat. Immunol *1*, 387-391.

Dalloul,A., Oksenhendler,E., Chosidow,O., Ribaud,P., Carcelain,G., Louvet,S., Massip,P., Lebon,P., and Autran,B. (2004). Severe herpes virus (HSV-2) infection in two patients with myelodysplasia and undetectable NK cells and plasmacytoid dendritic cells in the blood. Journal of Clinical Virology *30*, 329-336.

Dalod, M., Hamilton, T., Salomon, R., Salazar-Mather, T.P., Henry, S.C., Hamilton, J.D., and Biron, C.A. (2003). Dendritic Cell Responses to Early Murine Cytomegalovirus Infection: Subset Functional Specialization and Differential Regulation by Interferon {alpha}/{beta}. J Exp. Med. 197, 885-898.

Daniels, R.H., Hall, P.S., and Bokoch, G.M. (1998). 'Membrane targeting of p21-activated kinase 1 (PAK1) induces neurite outgrowth from PC12 cells. EMBO. J. 17, 754-764.

Danve-Szatanek, C., Aymard, M., Thouvenot, D., Morfin, F., Agius, G., Bertin, I., Billaudel, S., Chanzy, B., Coste-Burel, M., Finkielsztejn, L., Fleury, H., Hadou, T., Henquell, C., Lafeuille, H., Lafon, M.E., Le Faou, A., Legrand, M.C., Maille, L., Mengelle, C., Morand, P., Morinet, F., Nicand, E., Omar, S., Picard, B., Pozzetto, B., Puel, J., Raoult, D., Scieux, C., Segondy, M., Seigneurin, J.M., Teyssou, R., and Zandotti, C. (2004). Surveillance Network for Herpes Simplex Virus Resistance to Antiviral Drugs: 3-Year Follow-Up. J. Clin. Microbiol. *42*, 242-249.

Dauer, M., Pohl, K., Obermaier, B., Meskendahl, T., Robe, J., Schnurr, M., Endres, S., and Eigler, A. (2003). Interferon-alpha disables dendritic cell precursors: dendritic cells derived from interferon-alpha-treated monocytes are defective in maturation and T-cell stimulation. Immunology *110*, 38-47.

de Jong,E.C., Vieira,P.L., Kalinski,P., Schuitemaker,J.H.N., Tanaka,Y., Wierenga,E.A., Yazdanbakhsh,M., and Kapsenberg,M.L. (2002). Microbial Compounds Selectively Induce Th1 Cell-Promoting or Th2 Cell-Promoting Dendritic Cells In Vitro with Diverse Th Cell-Polarizing Signals. J Immunol *168*, 1704-1709.

de Vries,I.J., Krooshoop,D.J.E.B., Scharenborg,N.M., Lesterhuis,W.J., Diepstra,J.H., van Muijen,G.N.P., Strijk,S.P., Ruers,T.J., Boerman,O.C., Oyen,W.J.G., Adema,G.J., Punt,C.J.A., and Figdor,C.G. (2003). Effective Migration of Antigen-pulsed Dendritic Cells to Lymph Nodes in Melanoma Patients Is Determined by Their Maturation State. Cancer Res *63*, 12-17.

Dean, J.L., Brook, M., Clark, A.R., and Saklatvala, J. (1999). p38 Mitogen-activated Protein Kinase Regulates Cyclooxygenase-2 mRNA Stability and Transcription in Lipopolysaccharide-treated Human Monocytes. J. Biol. Chem. 274, 264-269. Dehio, C., Prevost, M.C., and Sansonctti, P.J. (1995). Invasion of epithelial cells by Shigella flexneri induces tyrosine phosphorylation of cortactin by a pp60c-src-mediated signalling pathway. EMBO J 14, 2471-2482.

Della,B.S., Nicola,S., Riva,A., Biasin,M., Clerici,M., and Villa,M.L. (2004). Functional repertoire of dendritic célls generated in granulocyte macrophage-colony stimulating factor and interferon-alpha. J Leukoc. Biol. 75, 106-116.

Delneste, Y., Charbonnier, P., Herbault, N., Magistrelli, G., Caron, G., Bonnefoy, J.Y., and Jeannin, P. (2003). Interferon-gamma switches monocyte differentiation from dendritic cells to macrophages. Blood *101*, 143-150.

DeMeritt,I.B., Milford,L.E., and Yurochko,A.D. (2004). Activation of the NF-{kappa}B Pathway in Human Cytomegalovirus-Infected Cells Is Necessary for Efficient Transactivation of the Major Immediate-Early Promoter. J. Virol. 78, 4498-4507.

Deshpande, S.P., Kumaraguru, U., and Rouse, B.T. (2000). Dual Role of B Cells in Mediating Innate and Acquired Immunity to Herpes Simplex Virus Infections. Cellular Immunology 202, 79-87.

Diaz, J.J., Dodon, M.D., Schaerer-Uthurralt, N., Simonin, D., Kindbeiter, K., Gazzolo, L., and Madjar, J.J. (1996). Post-transcriptional transactivation of human retroviral envelope glycoprotein expression by herpes simplex virus Us11 protein. Nature *379*, 273-277.

Didierlaurent, A., Ferrero, I., Otten, L.A., Dubois, B., Reinhardt, M., Carlsen, H., Blomhoff, R., Akira, S., Kraehenbuhl, J.P., and Sirard, J.C. (2004). Flagellin Promotes Myeloid Differentiation Factor 88-Dependent Development of Th2-Type Response. J Immunol *172*, 6922-6930.

Diebold,S.S., Kaisho,T., Hemmi,H., Akira,S., and Reis e Sousa (2004). Innate antiviral responses by means of TLR7-mediated recognition of single-stranded RNA. Science. *303*, 1529-1531.

Diebold,S.S., Montoya,M., Unger,H., Alexopoulou,L., Roy,P., Haswell,L.E., Al Shamkhani,A., Flavell,R., Borrow,P., and Reis e Sousa (2003). Viral infection switches non-plasmacytoid dendritic cells into high interferon producers. Nature 424, 324-328.

Dieu,M.C., Vanbervliet,B., Vicari,A., Bridon,J.M., Oldham,E., Ait-Yahia,S., Briere,F., Zlotnik,A., Lebecque,S., and Caux,C. (1998). Selective recruitment of immature and mature dendritic cells by distinct chemokines expressed in different anatomic sites. J Exp. Med. *188*, 373-386.

Dillon,S., Agrawal,A., Van Dyke,T., Landreth,G., McCauley,L., Koh,A., Maliszewski,C., Akira,S., and Pulendran,B. (2004). A Toll-like receptor 2 ligand stimulates Th2 responses in vivo, via induction of extracellular signal-regulated kinase mitogen-activated protein kinase and c-Fos in dendritic cells. J Immunol *172*, 4733-4743.

Dimitrov,D.S., Broder,C.C., Berger,E.A., and Blumenthal,R. (1993). Calcium ions are required for cell fusion mediated by the CD4-human immunodeficiency virus type 1 envelope glycoprotein interaction. J Virol *67*, 1647-1652.

Dobbelstein, M. and Shenk, T. (1996). Protection against apoptosis by the vaccinia virus SPI-2 (B13R) gene product. J. Virol. 70, 6479-6485.

Dobrowolski,S., Harter,M., and Stacey,D.W. (1994). Cellular ras activity is required for passage through multiple points of the G0/G1 phase in BALB/c 3T3 cells. Mol. Cell Biol. 14, 5441-5449.

Dohner,K., Wolfstein,A., Prank,U., Echeverri,C., Dujardin,D., Vallee,R., and Sodeik,B. (2002). Function of dynein and dynactin in herpes simplex virus capsid transport. Mol. Biol. Cell 13, 2795-2809.

Donaghy,H., Gazzard,B., Gotch,F., and Patterson,S. (2003). Dysfunction and infection of freshly isolated blood myeloid and plasmacytoid dendritic cells in patients infected with HIV-1. Blood 101, 4505-4511.

Dong, R., Cwynarski, K., Entwistle, A., Marelli-Berg, F., Dazzi, F., Simpson, E., Goldman, J.M., Melo, J.V., Lechler, R.I., Bellantuono, I., Ridley, A., and Lombardi, G. (2003). Dendritic cells from CML patients have altered actin organization, reduced antigen processing, and impaired migration. Blood. *101*, 3560-3567.

Doni,A., Peri,G., Chieppa,M., Allavena,P., Pasqualini,F., Vago,L., Romani,L., Garlanda,C., and Mantovani,A. (2003). Production of the soluble pattern recognition receptor PTX3 by myeloid, but not plasmacytoid, dendritic cells. Eur. J-Immunol *33*, 2886-2893.

Doyle,S.E., O'Connell,R.M., Miranda,G.A., Vaidya,S.A., Chow,E.K., Liu,P.T., Suzuki,S., Suzuki,N., Modlin,R.L., Yeh,W.C., Lane,T.F., and Cheng,G. (2004). Toll-like receptors induce a phagocytic gene program through p38. J Exp. Med *199*, 81-90.

Drillien, R., Spehner, D., and Hanau, D. (2004). Modified vaccinia virus Ankara induces moderate activation of human dendritic cells. J Gen Virol *85*, 2167-2175.

Du,C. and Sriram,S. (1998). Mechanism of inhibition of LPS-induced IL-12p40 production by IL-10 and TGF-beta in ANA-1 cells. J Leukoc. Biol. 64, 92-97.

Dubin,G., Socolof,E., Frank,I., and Friedman,H.M. (1991). Herpes simplex virus type 1 Fc receptor protects infected cells from antibody-dependent cellular cytotoxicity. J Virol 65, 7046-7050.

Dubois, B., Lamy, P.J., Chemin, K., Lachaux, A., and Kaiserlian, D. (2001). Measles virus exploits dendritic cells to suppress CD4+ T-cell proliferation via expression of surface viral glycoproteins independently of T-cell trans-infection. Cell Immunol. *214*, 173-183.

Duerst, R.J. and Morrison, L.A. (2004). Herpes simplex virus 2 virion host shutoff protein interferes with type I interferon production and responsiveness. Virology *322*, 158-167.

Dugas, B., Delfraissy, J.F., Calenda, A., Peuchmaur, M., Wallon, C., Rannou, M.T., and Galanaud, P. (1988). Activation and infection of B cells by Epstein-Barr virus. Role of calcium mobilization and of protein kinase C translocation. J Immunol 141, 4344-4351.

Duh,E.J., Maury,W.J., Folks,T.M., Fauci,A.S., and Rabson,A.B. (1989). Tumor necrosis factor alpha activates human immunodeficiency virus type 1 through induction of nuclear factor binding to the NF-kappa B sites in the long terminal repeat. Proc. Natl. Acad. Sci U. S. A. *86*, 5974-5978.

Dzionek,A., Sohma,Y., Nagafune,J., Cella,M., Colonna,M., Facchetti,F., Gunther,G., Johnston,I., Lanzavecchia,A., Nagasaka,T., Okada,T., Vermi,W., Winkels,G., Yamamoto,T., Zysk,M., Yamaguchi,Y., and Schmitz,J. (2001). BDCA-2, a novel plasmacytoid dendritic cell-specific type II C-type lectin, mediates antigen capture and is a potent inhibitor of interferon alpha/beta induction. J Exp. Med *194*, 1823-1834.

Eberle, F., Dubreuil, P., Mattei, M.G., Devilard, E., and Lopez, M. (1995). The human PRR2 gene, related to the human poliovirus receptor gene (PVR), is the true homolog of the murine MPH gene. Gene *159*, 267-272.

Ebner, S., Ehammer, Z., Holzmann, S., Schwingshackl, P., Forstner, M., Stoitzner, P., Huemer, G.M., Fritsch, P., and Romani, N. (2004). Expression of C-type lectin receptors by subsets of dendritic cells in human skin. Int. Immunol *16*, 877-887.

Ebner,S., Ratzinger,G., Krosbacher,B., Schmuth,M., Weiss,A., Reider,D., Kroczek,R.A., Herold,M., Heufler,C., Fritsch,P., and Romani,N. (2001). Production of IL-12 by Human Monocyte-Derived Dendritic Cells Is Optimal When the Stimulus Is Given at the Onset of Maturation, and Is Further Enhanced by IL-4. J Immunol *166*, 633-641.

Ehl,S., Bischoff,R., Ostler,T., Vallbracht,S., Schulte-Monting,J., Poltorak,A., and Freudenberg,M. (2004). The role of Toll-like receptor 4 versus interleukin-12 in immunity to respiratory syncytial virus. Eur. J Immunol *34*, 1146-1153.

Eisenbarth,S.C., Piggott,D.A., Huleatt,J.W., Visintin,I., Herrick,C.A., and Bottomly,K. (2002). Lipopolysaccharide-enhanced, Toll-like Receptor 4-dependent T Helper Cell Type 2 Responses to Inhaled Antigen. J. Exp. Med. 196, 1645-1651.

Engelmayer, J., Larsson, M., Subklewe, M., Chahroudi, A., Cox, W.I., Steinman, R.M., and Bhardwaj, N. (1999). Vaccinia virus inhibits the maturation of human dendritic cells: a novel mechanism of immune evasion. J Immunol *163*, 6762-6768.

Eriksson,K., Bellner,L., Gorander,S., Lowhagen,G.B., Tunback,P., Rydberg,K., and Liljeqvist,J.A. (2004). CD4(+) T-cell responses to herpes simplex virus type 2 (HSV-2) glycoprotein G are type specific and differ in symptomatic and asymptomatic HSV-2-infected individuals. J Gen Virol *85*, 2139-2147.

Esclatine, A., Taddeo, B., Evans, L., and Roizman, B. (2004). The herpes simplex virus 1 UL41 genedependent destabilization of cellular RNAs is selective and may be sequence-specific. Proc. Natl. Acad. Sci. U. S. A. 101, 3603-3608.

Fantuzzi,L., Gessani,S., Borghi,P., Varano,B., Conti,L., Puddu,P., and Belardelli,F. (1996). Induction of interleukin-12 (IL-12) by recombinant glycoprotein gp120 of human immunodeficiency virus type 1 in human monocytes/macrophages: requirement of gamma interferon for IL-12 secretion. J Virol 70, 4121-4124.

Fantuzzi, L., Purificato, C., Donato, K., Belardelli, F., and Gessani, S. (2004). Human Immunodeficiency Virus Type 1 gp120 Induces Abnormal Maturation and Functional Alterations of Dendritic Cells: a Novel Mechanism for AIDS Pathogenesis. J Virol 78, 9763-9772.

Farassati,F., Yang,A.D., and Lee,P.W. (2001). Oncogenes in Ras signalling pathway dictate host-cell permissiveness to herpes simplex virus 1. Nat. Cell. Biol. *3*, 745-750.

Farkas, L., Beiske, K., Lund-Johansen, F., Brandtzaeg, P., and Jahnsen, F.L. (2001). Plasmacytoid dendritic cells (natural interferon- alpha/beta-producing cells) accumulate in cutaneous lupus erythematosus lesions. Am. J. Pathol. *159*, 237-243.

Feldman,L.T., Ellison,A.R., Voytek,C.C., Yang,L., Krause,P., and Margolis,T.P. (2002). Spontaneous molecular reactivation of herpes simplex virus type 1 latency in mice. Proc. Natl. Acad. Sci. U. S. A. 99, 978-983.

Fiorini, M., Vermi, W., Facchetti, F., Moratto, D., Alessandri, G., Notarangelo, L., Caruso, A., Grigolato, P., Ugazio, A.G., Notarangelo, L.D., and Badolato, R. (2002). Defective migration of monocyte-derived dendritic cells in LAD-1 immunodeficiency. J. Leukoc. Biol. *72*, 650-656.

Fitzgerald,K.A., Rowe,D.C., Barnes,B.J., Caffrey,D.R., Visintin,A., Latz,E., Monks,B., Pitha,P.M., and Golenbock,D.T. (2003). LPS-TLR4 signaling to IRF-3/7 and NF-kappaB involves the toll adapters TRAM and TRIF. J Exp. Med. *198*, 1043-1055.

Fohrer, H., Audit, I.M., Sainz, A., Schmitt, C., Dezutter-Dambuyant, C., and Dalloul, A.H. (2004). Analysis of transcription factors in thymic and CD34+ progenitor-derived plasmacytoid and myeloid dendritic cells: evidence for distinct expression profiles. Experimental Hematology *32*, 104-112.

Fonteneau, J.F., Gilliet, M., Larsson, M., Dasilva, I., Munz, C., Liu, Y.J., and Bhardwaj, N. (2003). Activation of influenza virus-specific CD4+ and CD8+ T cells: a new role for plasmacytoid dendritic cells in adaptive immunity. Blood *101*, 3520-3526.

Forster, R., Schubel, A., Breitfeld, D., Kremmer, E., Renner-Muller, I., Wolf, E., and Lipp, M. (1999). CCR7 coordinates the primary immune response by establishing functional microenvironments in secondary lymphoid organs. Cell *99*, 23-33.

Foster,G.R., Germain,C., Jones,M., Lechler,R.I., and Lombardi,G. (2000). Human T cells elicit IFNalpha secretion from dendritic cells following cell to cell interactions. Eur. J Immunol *30*, 3228-3235.

Franchi,L., Condo,I., Tomassini,B., Nicolo,C., and Testi,R. (2003). A caspaselike activity is triggered by LPS and is required for survival of human dendritic cells. Blood. *102*, 2910-2915.

Franchini,M., Hefti,H., Vollstedt,S., Glanzmann,B., Riesen,M., Ackermann,M., Chaplin,P., Shortman,K., and Suter,M. (2004). Dendritic cells from mice neonatally vaccinated with modified vaccinia virus Ankara transfer resistance against herpes simplex virus type I to naive one-week-old mice. J Immunol *172*, 6304-6312.

Frank, I. and Friedman, H.M. (1989). A novel function of the herpes simplex virus type 1 Fc receptor: participation in bipolar bridging of antiviral immunoglobulin G. J Virol *63*, 4479-4488.

Friedman,H.M., Cohen,G.H., Eisenberg,R.J., Seidel,C.A., and Cines,D.B. (1984). Glycoprotein C of herpes simplex virus 1 acts as a receptor for the C3b complement component on infected cells. Nature. *309*, 633-635.

Friedman,H.M., Wang,L., Fishman,N.O., Lambris,J.D., Eisenberg,R.J., Cohen,G.H., and Lubinski,J. (1996). Immune evasion properties of herpes simplex virus type 1 glycoprotein gC. J Virol 70, 4253-4260.

Fries, L.F., Friedman, H.M., Cohen, G.H., Eisenberg, R.J., Hammer, C.H., and Frank, M.M. (1986). Glycoprotein C of herpes simplex virus 1 is an inhibitor of the complement cascade. J Immunol 137, 1636-1641.

Fugier-Vivier, I., Servet-Delprat, C., Rivailler, P., Rissoan, M.C., Liu, Y.J., and Rabourdin-Combe, C. (1997). Measles virus suppresses cell-mediated immunity by interfering with the survival and functions of dendritic and T cells. J Exp. Med. *186*, 813-823.

Fukao, T., Tanabe, M., Terauchi, Y., Ota, T., Matsuda, S., Asano, T., Kadowaki, T., Takeuchi, T., and Koyasu, S. (2002). PI3K-mediated negative feedback regulation of IL-12 production in DCs. Nat. Immunol. *3*, 875-881.

Furukawa,Y., Kubota,R., Tara,M., Izumo,S., and Osame,M. (2001). Existence of escape mutant in HTLV-I tax during the development of adult T-cell leukemia. Blood *97*, 987-993.

Futagawa,T., Akiba,H., Kodama,T., Takeda,K., Hosoda,Y., Yagita,H., and Okumura,K. (2002). Expression and function of 4-1BB and 4-1BB ligand on murine dendritic cells. Int. Immunol *14*, 275-286.

Galvan, V., Brandimarti, R., and Roizman, B. (1999). Herpes simplex virus 1 blocks caspase-3independent and caspase-dependent pathways to cell death. J. Virol. 73, 3219-3226.

Galvan, V. and Roizman, B. (1998). Herpes simplex virus 1 induces and blocks apoptosis at multiple steps during infection and protects cells from exogenous inducers in a cell-type-dependent manner. Proc. Natl. Acad. Sci. U. S. A 95, 3931-3936.

Gantner, B.N., Simmons, R.M., Canavera, S.J., Akira, S., and Underhill, D.M. (2003). Collaborative Induction of Inflammatory Responses by Dectin-1 and Toll-like Receptor 2. J. Exp. Med. 197, 1107-1117.

Gao, J.J., Filla, M.B., Fultz, M.J., Vogel, S.N., Russell, S.W., and Murphy, W.J. (1998). Autocrine/Paracrine IFN-{alpha} {beta} Mediates the Lipopolysaccharide-Induced Activation of Transcription Factor Stat1 {alpha} in Mouse Macrophages: Pivotal Role of Stat1 {alpha} in Induction of the Inducible Nitric Oxide Synthase Gene. J Immunol *161*, 4803-4810.

Garlanda, C., Hirsch, E., Bozza, S., Salustri, A., De Acetis, M., Nota, R., Maccagno, A., Riva, F., Bottazzi, B., Peri, G., Doni, A., Vago, L., Botto, M., De Santis, R., Carminati, P., Siracusa, G., Altruda, F., Vecchi, A., Romani, L., and Mantovani, A. (2002). Non-redundant role of the long pentraxin PTX3 in anti-fungal innate immune response. Nature. *420*, 182-186.

Garrett, W.S., Chen, L.M., Kroschewski, R., Ebersold, M., Turley, S., Trombetta, S., Galan, J.E., and Mellman, I. (2000). Developmental control of endocytosis in dendritic cells by Cdc42. Cell *102*, 325-334.

Gauzzi, M.C., Canini, I., Eid, P., Belardelli, F., and Gessani, S. (2002). Loss of type I IFN receptors and impaired IFN responsiveness during terminal maturation of monocyte-derived human dendritic cells. J Immunol *169*, 3038-3045.

Gebhardt, B.M. and Hill, J.M. (1990). Cellular neuroimmunologic responses to ocular herpes simplex virus infection. J Neuroimmunol. 28, 227-236.

Geijtenbeek, T.B., Kwon, D.S., Torensma, R., Van Vliet, S.J., van Duijnhoven, G.C., Middel, J., Cornelissen, J.L., Nottet, H.S., KewalRamani, V.N., Littman, D.R., Figdor, C.G., and Van Kooyk, Y. (2000a). DC-SIGN, a dendritic cell-specific HIV-1-binding protein that enhances trans-infection of T cells. Cell *100*, 587-597.

Geijtenbeek, T.B., Torensma, R., Van Vliet, S.J., van Duijnhoven, G.C., Adema, G.J., Van Kooyk, Y., and Figdor, C.G. (2000b). Identification of DC-SIGN, a novel dendritic cell-specific ICAM-3 receptor that supports primary immune responses. Cell *100*, 575-585.

Geijtenbeek, T.B., Van Vliet, S.J., Engering, A., 't Hart, B.A., and Van Kooyk, Y. (2004). Self- and nonself-recognition by C-type lectins on dendritic cells. Annu. Rev. Immunol 22, 33-54.

Geijtenbeek, T.B., Van Vliet, S.J., Koppel, E.A., Sanchez-Hernandez, M., Vandenbroucke-Grauls, C.M., Appelmelk, B., and Van Kooyk, Y. (2003). Mycobacteria target DC-SIGN to suppress dendritic cell function. J. Exp. Med. *197*, 7-17.

Geisbert,T.W., Hensley,L.E., Larsen,T., Young,H.A., Reed,D.S., Geisbert,J.B., Scott,D.P., Kagan,E., Jahrling,P.B., and Davis,K.J. (2003a). Pathogenesis of Ebola Hemorrhagic Fever in Cynomolgus Macaques: Evidence that Dendritic Cells Are Early and Sustained Targets of Infection. Am. J Pathol. *163*, 2347-2370.

Geisbert, T.W., Young, H.A., Jahrling, P.B., Davis, K.J., Larsen, T., Kagan, E., and Hensley, L.E. (2003b). Pathogenesis of Ebola Hemorrhagic Fever in Primate Models: Evidence that Hemorrhage Is Not a Direct Effect of Virus-Induced Cytolysis of Endothelial Cells. Am. J Pathol. *163*, 2371-2382.

Geissmann, F., Dieu-Nosjean, M.C., Dezutter, C., Valladeau, J., Kayal, S., Leborgne, M., Brousse, N., Saeland, S., and Davoust, J. (2002). Accumulation of immature Langerhans cells in human lymph nodes draining chronically inflamed skin. J Exp. Med *196*, 417-430.

Geissmann,F., Revy,P., Regnault,A., Lepelletier,Y., Dy,M., Brousse,N., Amigorena,S., Hermine,O., and Durandy,A. (1999). TGF-beta 1 prevents the noncognate maturation of human dendritic Langerhans cells. J Immunol *162*, 4567-4575.

Geraghty,R.J., Krummenacher,C., Cohen,G.H., Eisenberg,R.J., and Spear,P.G. (1998). Entry of alphaherpesviruses mediated by poliovirus receptor-related protein 1 and poliovirus receptor. Science 280, 1618-1620.

Ghanekar, S., Zheng, L., Logar, A., Navratil, J., Borowski, L., Gupta, P., and Rinaldo, C. (1996). Cytokine expression by human peripheral blood dendritic cells stimulated in vitro with HIV-1 and herpes simplex virus. J. Immunol. *157*, 4028-4036.

Giacomini, E., Iona, E., Ferroni, L., Miettinen, M., Fattorini, L., Orefici, G., Julkunen, I., and Coccia, E.M. (2001). Infection of human macrophages and dendritic cells with Mycobacterium tuberculosis induces a differential cytokine gene expression that modulates T cell response. J Immunol *166*, 7033-7041.

Girardin, S.E., Boneca, I.G., Carneiro, L.A.M., Antignac, A., Jehanno, M., Viala, J., Tedin, K., Taha, M.K., Labigne, A., Zathringer, U., Coyle, A.J., DiStefano, P.S., Bertin, J., Sansonetti, P.J., and Philpott, D.J.

(2003a). Nod1 Detects a Unique Muropeptide from Gram-Negative Bacterial Peptidoglycan. Science 300, 1584-1587.

Girardin,S.E., Boneca,I.G., Viala,J., Chamaillard,M., Labigne,A., Thomas,G., Philpott,D.J., and Sansonetti,P.J. (2003b). Nod2 Is a General Sensor of Peptidoglycan through Muramyl Dipeptide (MDP) Detection. J. Biol. Chem. 278, 8869-8872.

Golden, M.P., Kim, S., Hammer, S.M., Ladd, E.A., Schaffer, P.A., DeLuca, N., and Albrecht, M.A. (1992). Activation of human immunodeficiency virus by herpes simplex virus. J Infect. Dis. *166*, 494-499.

Goldsmith,K., Chen,W., Johnson,D.C., and Hendricks,R.L. (1998). Infected cell protein (ICP)47 enhances herpes simplex virus neurovirulence by blocking the CD8+ T cell response. J Exp. Med 187, 341-348.

Goodkin, M.L., Ting, A.T., and Blaho, J.A. (2003). NF-kappaB is required for apoptosis prevention during herpes simplex virus type 1 infection. J Virol 77, 7261-7280.

Gorander, S., Svennerholm, B., and Liljeqvist, J.A. (2003). Secreted Portion of Glycoprotein G of Herpes Simplex Virus Type 2 Is a Novel Antigen for Type-Discriminating Serology. J. Clin. Microbiol. 41, 3681-3686.

Goriely,S., Demonte,D., Nizet,S., De Wit,D., Willems,F., Goldman,M., and Van Lint,C. (2003). Human IL-12(p35) gene activation involves selective remodeling of a single nucleosome within a region of the promoter containing critical Sp1-binding sites. Blood *101*, 4894-4902.

Gosset, P., Bureau, F., Angeli, V., Pichavant, M., Faveeuw, C., Tonnel, A.B., and Trottein, F. (2003). Prostaglandin D2 affects the maturation of human monocyte-derived dendritic cells: consequence on the polarization of naive Th cells. J Immunol *170*, 4943-4952.

Goutagny, N., Fatmi, A., Ledinghen, V.D., Penin, F., Couzigou, P., Inchauspe, G., and Bain, C. (2003). Evidence of Viral Replication in Circulating Dendritic Cells during Hepatitis C Virus Infection. J Infect. Dis. *187*, 1951-1958.

Goutagny, N., Vieux, C., Decullier, E., Ligeoix, B., Epstein, A., Trepo, C., Couzigou, P., Inchauspe, G., and Bain, C. (2004). Quantification and functional analysis of plasmacytoid dendritic cells in patients with chronic hepatitis C virus infection. J Infect. Dis. *189*, 1646-1655.

Granelli-Piperno, A., Finkel, V., Delgado, E., and Steinman, R.M. (1999). Virus replication begins in dendritic cells during the transmission of HIV-1 from mature dendritic cells to T cells. Curr. Biol. 9, 21-29.

Granelli-Piperno, A., Golebiowska, A., Trumpfheller, C., Siegal, F.P., and Steinman, R.M. (2004). HIV-1infected monocyte-derived dendritic cells do not undergo maturation but can elicit IL-10 production and T cell regulation. Proc. Natl. Acad. Sci U. S. A. 101, 7669-7674.

Granucci, F., Petralia, F., Urbano, M., Citterio, S., Di Tota, F., Santambrogio, L., and Ricciardi-Castagnoli, P. (2003). The scavenger receptor MARCO mediates cytoskeleton rearrangements in dendritic cells and microglia. Blood *102*, 2940-2947.

Granucci,F., Vizzardelli,C., Pavelka,N., Feau,S., Persico,M., Virzi,E., Rescigno,M., Moro,G., and Ricciardi-Castagnoli,P. (2001). Inducible IL-2 production by dendritic cells revealed by global gene expression analysis. Nat. Immunol *2*, 882-888.

Gredmark,S. and Soderberg-Naucler,C. (2003). Human cytomegalovirus inhibits differentiation of monocytes into dendritic cells with the consequence of depressed immunological functions. J Virol. 77, 10943-10956.

Grosjean, I., Caux, C., Bella, C., Berger, I., Wild, F., Banchereau, J., and Kaiserlian, D. (1997). Measles virus infects human dendritic cells and blocks their allostimulatory properties for CD4+ T cells. J Exp. Med. 186, 801-812.

Grouard,G., Rissoan,M.C., Filgueira,L., Durand,I., Banchereau,J., and Liu,Y.J. (1997). The Enigmatic Plasmacytoid T Cells Develop into Dendritic Cells with Interleukin (IL)-3 and CD40-Ligand. J. Exp. Med. 185, 1101-1112.

Grubor-Bauk, B., Simmons, A., Mayrhofer, G., and Speck, P.G. (2003). Impaired clearance of herpes simplex virus type 1 from mice lacking CD1d or NKT cells expressing the semivariant V alpha 14-J alpha 281 TCR. J. Immunol. *170*, 1430-1434.

Gruenheid, S., Gatzke, L., Meadows, H., and Tufaro, F. (1993). Herpes simplex virus infection and propagation in a mouse L cell mutant lacking heparan sulfate proteoglycans. J Virol. 67, 93-100.

Grumont, R., Hochrein, H., O'Keeffe, M., Gugasyan, R., White, C., Caminschi, I., Cook, W., and Gerondakis, S. (2001). c-Rel regulates interleukin 12 p70 expression in CD8(+) dendritic cells by specifically inducing p35 gene transcription. J. Exp. Med. 194, 1021-1032.

Gu,L., Tseng,S., Horner,R.M., Tam,C., Loda,M., and Rollins,B.J. (2000). Control of TH2 polarization by the chemokine monocyte chemoattractant protein-1. Nature. 404, 407-411.

Guermonprez, P., Saveanu, L., Kleijmeer, M., Davoust, J., van Endert, P., and Amigorena, S. (2003). ERphagosome fusion defines an MHC class I cross-presentation compartment in dendritic cells. Nature 425, 397-402. Guerreiro-Cacais,A.O., Li,L., Donati,D., Bejarano,M.T., Morgan,A., Masucci,M.G., Hutt-Fletcher,L., and Levitsky,V. (2004). Capacity of Epstein-Barr virus to infect monocytes and inhibit their development into dendritic cells is affected by the cell type supporting virus replication. J Gen Virol *85*, 2767-2778.

Guler, M.L., Jacobson, N.G., Gubler, U., and Murphy, K.M. (1997). T cell genetic background determines maintenance of IL-12 signaling: effects on BALB/c and B10.D2 T helper cell type 1 phenotype development. J Immunol *159*, 1767-1774.

Gunn,M.D., Tangemann,K., Tam,C., Cyster,J.G., Rosen,S.D., and Williams,L.T. (1998). A chemokine expressed in lymphoid high endothelial venules promotes the adhesion and chemotaxis of naive T lymphocytes. Proc. Natl. Acad. Sci U. S. A. 95, 258-263.

Guo,Z., Zhang,M., An,H., Chen,W., Liu,S., Guo,J., Yu,Y., and Cao,X. (2003). Fas ligation induces ILlbeta-dependent maturation and IL-lbeta-independent survival of dendritic cells: different roles of ERK and NF-kappaB signaling pathways. Blood *102*, 4441-4447.

Haeberle,H.A., Takizawa,R., Casola,A., Brasier,A.R., Dieterich,H.J., Van Rooijen,N., Gatalica,Z., and Garofalo,R.P. (2002). Respiratory syncytial virus-induced activation of nuclear factor-kappaB in the lung involves alveolar macrophages and toll-like receptor 4-dependent pathways. J. Infect. Dis. *186*, 1199-1206.

Hagglund,R., Munger,J., Poon,A.P., and Roizman,B. (2002). U(S)3 protein kinase of herpes simplex virus 1 blocks caspase 3 activation induced by the products of U(S)1.5 and U(L)13 genes and modulates expression of transduced U(S)1.5 open reading frame in a cell type-specific manner. J Virol. *76*, 743-754.

Hagmann, M., Georgiev, O., Schaffner, W., and Douville, P. (1995a). Transcription factors interacting with herpes simplex virus alpha gene promoters in sensory neurons. Nucleic. Acids. Res. 23, 4978-4985.

Hagmann, M., Georgiev, O., Schaffner, W., and Douville, P. (1995b). Transcription factors interacting with herpes simplex virus alpha gene promoters in sensory neurons. Nucl. Acids. Res. 23, 4978-4985.

Hahn,G., Jores,R., and Mocarski,E.S. (1998). Cytomegalovirus remains latent in a common precursor of dendritic and myeloid cells. Proc. Natl. Acad. Sci. U. S. A 95, 3937-3942.

Halary, F., Amara, A., Lortat-Jacob, H., Messerle, M., Delaunay, T., Houles, C., Fieschi, F., Arenzana-Seisdedos, F., Moreau, J.F., and Dechanet-Merville, J. (2002). Human cytomegalovirus binding to DC-SIGN is required for dendritic cell infection and target cell trans-infection. Immunity. *17*, 653-664.

Halford, W.P., Gebhardt, B.M., and Carr, D.J. (1996). Persistent cytokine expression in trigeminal ganglion latently infected with herpes simplex virus type 1. J. Immunol. *157*, 3542-3549.

Halford,W.P., Gebhardt,B.M., and Carr,D.J. (1997). Acyclovir blocks cytokine gene expression in trigeminal ganglia latently infected with herpes simplex virus type 1. Virology 238, 53-63.

Hall, A. (1998). Rho GTPases and the actin cytoskeleton. Science. 279, 509-514.

Handler, C.G., Eisenberg, R.J., and Cohen, G.H. (1996). Oligomeric structure of glycoproteins in herpes simplex virus type 1. J Virol 70, 6067-6070.

Harandi, A.M., Eriksson, K., and Holmgren, J. (2003). A protective role of locally administered immunostimulatory CpG oligodeoxynucleotide in a mouse model of genital herpes infection. J. Virol. 77, 953-962.

Hardwicke, M.A. and Sandri-Goldin, R.M. (1994). The herpes simplex virus regulatory protein ICP27 contributes to the decrease in cellular mRNA levels during infection. J. Virol. *68*, 4797-4810.

Harshyne, L.A., Watkins, S.C., Gambotto, A., and Barratt-Boyes, S.M. (2001). Dendritic cells acquire antigens from live cells for cross-presentation to CTL. J. Immunol. *166*, 3717-3723.

Harte, M.T., Haga, I.R., Maloney, G., Gray, P., Reading, P.C., Bartlett, N.W., Smith, G.L., Bowie, A., and O'Neill, L.A. (2003). The poxvirus protein A52R targets Toll-like receptor signaling complexes to suppress host defense. J Exp. Med *197*, 343-351.

Hawiger, D., Inaba, K., Dorsett, Y., Guo, M., Mahnke, K., Rivera, M., Ravetch, J.V., Steinman, R.M., and Nussenzweig, M.C. (2001). Dendritic cells induce peripheral T cell unresponsiveness under steady state conditions in vivo. J. Exp. Med. *194*, 769-779.

Hawn,T.R., Verbon,A., Lettinga,K.D., Zhao,L.P., Li,S.S., Laws,R.J., Skerrett,S.J., Beutler,B., Schroeder,L., Nachman,A., Ozinsky,A., Smith,K.D., and Aderem,A. (2003). A common dominant TLR5 stop codon polymorphism abolishes flagellin signaling and is associated with susceptibility to legionnaires' disease. J Exp. Med *198*, 1563-1572.

Hayashi,F., Smith,K.D., Ozinsky,A., Hawn,T.R., Yi,E.C., Goodlett,D.R., Eng,J.K., Akira,S., Underhill,D.M., and Aderem,A. (2001). The innate immune response to bacterial flagellin is mediated by Toll-like receptor 5. Nature. *410*, 1099-1103.

Haynes,L.M., Moore,D.D., Kurt-Jones,E.A., Finberg,R.W., Anderson,L.J., and Tripp,R.A. (2001). Involvement of toll-like receptor 4 in innate immunity to respiratory syncytial virus. J. Virol. 75, 10730-10737.

Heeg, U., Dienes, H.P., Muller, S., and Falke, D. (1986). Involvement of actin-containing microfilaments in HSV-induced cytopathology and the influence of inhibitors of glycosylation. Arch. Virol. *91*, 257-270.

Heil,F., Ahmad-Nejad,P., Hemmi,H., Hochrein,H., Ampenberger,F., Gellert,T., Dietrich,H., Lipford,G., Takeda,K., Akira,S., Wagner,H., and Bauer,S. (2003). The Toll-like receptor 7 (TLR7)-specific stimulus loxoribine uncovers a strong relationship within the TLR7, 8 and 9 subfamily. Eur. J Immunol *33*, 2987-2997.

Heil,F., Hemmi,H., Hochrein,H., Ampenberger,F., Kirschning,C., Akira,S., Lipford,G., Wagner,H., and Bauer,S. (2004). Species-specific recognition of single-stranded RNA via toll-like receptor 7 and 8. Science. *303*, 1526-1529.

Heinz, S., Haehnel, V., Karaghiosoff, M., Schwarzfischer, L., Muller, M., Krause, S.W., and Rehli, M. (2003). Species-specific regulation of Toll-like receptor 3 genes in men and mice. J Biol. Chem. 278, 21502-21509.

Hemmi,H., Takeuchi,O., Kawai,T., Kaisho,T., Sato,S., Sanjo,H., Matsumoto,M., Hoshino,K., Wagner,H., Takeda,K., and Akira,S. (2000). A Toll-like receptor recognizes bacterial DNA. Nature 408, 740-745.

Heng,M.C., Heng,S.Y., and Allen,S.G. (1994). Co-infection and synergy of human immunodeficiency virus-1 and herpes simplex virus-1. Lancet. 343, 255-258.

Hennessey, K.A., Giorgi, J.V., Kaplan, A.H., Visscher, B.R., Gange, S., Margolick, J.B., Riddler, S., Phair, J., and Detels, R. (2000). AIDS onset at high CD4+ cell levels is associated with high HIV load. AIDS. Res. Hum. Retroviruses. 16, 103-107.

Hermann, P., Rubio, M., Nakajima, T., Delespesse, G., and Sarfati, M. (1998). IFN-{alpha} Priming of Human Monocytes Differentially Regulates Gram-Positive and Gram-Negative Bacteria-Induced IL-10 Release and Selectively Enhances IL-12p70, CD80, and MHC Class I Expression. J Immunol *161*, 2011-2018.

Herold,B.C., Gerber,S.I., Polonsky,T., Belval,B.J., Shaklee,P.N., and Holme,K. (1995). Identification of structural features of heparin required for inhibition of herpes simplex virus type 1 binding. Virology 206, 1108-1116.

Herold,B.C., Visalli,R.J., Susmarski,N., Brandt,C.R., and Spear,P.G. (1994). Glycoprotein Cindependent binding of herpes simplex virus to cells requires cell surface heparan sulphate and glycoprotein B. J Gen. Virol. 75 (*Pt 6*), 1211-1222.

Herold,B.C., WuDunn,D., Soltys,N., and Spear,P.G. (1991). Glycoprotein C of herpes simplex virus type 1 plays a principal role in the adsorption of virus to cells and in infectivity. J Virol. 65, 1090-1098.

Hertel,L., Lacaille,V.G., Strobl,H., Mellins,E.D., and Mocarski,E.S. (2003). Susceptibility of Immature and Mature Langerhans Cell-Type Dendritic Cells to Infection and Immunomodulation by Human Cytomegalovirus. J. Virol. 77, 7563-7574.

Hertz,C.J., Kiertscher,S.M., Godowski,P.J., Bouis,D.A., Norgard,M.V., Roth,M.D., and Modlin,R.L. (2001). Microbial lipopeptides stimulate dendritic cell maturation via Toll-like receptor 2. J Immunol *166*, 2444-2450.

Heystek,H.C., den Drijver,B., Kapsenberg,M.L., van Lier,R.A., and De Jong,E.C. (2003). Type I IFNs differentially modulate IL-12p70 production by human dendritic cells depending on the maturation status of the cells and counteract IFN-gamma-mediated signaling. Clin. Immunol *107*, 170-177.

Hishizawa, M., Imada, K., Kitawaki, T., Ueda, M., Kadowaki, N., and Uchiyama, T. (2004). Depletion and impaired interferon-alpha-producing capacity of blood plasmacytoid dendritic cells in human T-cell leukaemia virus type I-infected individuals. Br. J Haematol. *125*, 568-575.

Ho,L.J., Wang,J.J., Shaio,M.F., Kao,C.L., Chang,D.M., Han,S.W., and Lai,J.H. (2001). Infection of human dendritic-cells by dengue virus causes cell maturation and cytokine production. J Immunol 166, 1499-1506.

Hochrein,H., Schlatter,B., O'Keeffe,M., Wagner,C., Schmitz,F., Schiemann,M., Bauer,S., Suter,M., and Wagner,H. (2004). Herpes simplex virus type-1 induces IFN-alpha production via Toll-like receptor 9-dependent and -independent pathways. Proc. Natl. Acad. Sci U. S. A. *101*, 11416-11421.

Hodge-Dufour, J., Marino, M.W., Horton, M.R., Jungbluth, A., Burdick, M.D., Strieter, R.M., Noble, P.W., Hunter, C.A., and Pure, E. (1998). Inhibition of interferon gamma áinduced interleukin 12áproduction: A potential mechanism for the anti-inflammatory activities of tumor necrosis factor. PNAS *95*, 13806-13811.

Hoebe,K., Du,X., Georgel,P., Janssen,E., Tabeta,K., Kim,S.O., Goode,J., Lin,P., Mann,N., Mudd,S., Crozat,K., Sovath,S., Han,J., and Beutler,B. (2003a). Identification of Lps2 as a key transducer of MyD88-independent TIR signalling. Nature. 424, 743-748.

Hoebe,K., Janssen,E.M., Kim,S.O., Alexopoulou,L., Flavell,R.A., Han,J., and Beutler,B. (2003b). Upregulation of costimulatory molecules induced by lipopolysaccharide and double-stranded RNA occurs by Trif-dependent and Trif-independent pathways. Nat. Immunol *4*, 1223-1229.

Hofer, S., Rescigno, M., Granucci, F., Citterio, S., Francolini, M., and Ricciardi-Castagnoli, P. (2001). Differential activation of NF-kappa B subunits in dendritic cells in response to Gram-negative bacteria and to lipopolysaccharide. Microbes. Infect. *3*, 259-265.

Hoffmeyer, A., Grosse-Wilde, A., Flory, E., Neufeld, B., Kunz, M., Rapp, U.R., and Ludwig, S. (1999). Different mitogen-activated protein kinase signaling pathways cooperate to regulate tumor necrosis factor alpha gene expression in T lymphocytes. J Biol. Chem. 274, 4319-4327.

Honda, K., Sakaguchi, S., Nakajima, C., Watanabe, A., Yanai, H., Matsumoto, M., Ohteki, T., Kaisho, T., Takaoka, A., Akira, S., Seya, T., and Taniguchi, T. (2003). Selective contribution of IFN-alpha/beta

signaling to the maturation of dendritic cells induced by double-stranded RNA or viral infection. Proc. Natl. Acad. Sci. U. S. A *100*, 10872-10877.

Honess, R.W., Gompels, U.A., Barrell, B.G., Craxton, M., Cameron, K.R., Staden, R., Chang, Y.N., and Hayward, G.S. (1989). Deviations from expected frequencies of CpG dinucleotides in herpesvirus DNAs may be diagnostic of differences in the states of their latent genomes. J Gen. Virol. 70 (Pt 4), 837-855.

Hoshino, K., Kaisho, T., Iwabe, T., Takeuchi, O., and Akira, S. (2002). Differential involvement of IFNbeta in Toll-like receptor-stimulated dendritic cell activation. Int. Immunol. *14*, 1225-1231.

Hosoi, J., Murphy, G.F., Egan, C.L., Lerner, E.A., Grabbe, S., Asahina, A., and Granstein, R.D. (1993). Regulation of Langerhans cell function by nerves containing calcitonin gene-related peptide. Nature *363*, 159-163.

Hoves, S., Niller, H.H., Krause, S.W., Straub, R., Gluck, T., Mountz, J.D., Scholmerich, J., and Fleck, M. (2001). Decreased T cell stimulatory capacity of monocyte-derived human macrophages following herpes simplex virus type 1 infection. Scand. J Immunol *54*, 93-99.

Huang,F.P., Platt,N., Wykes,M., Major,J.R., Powell,T.J., Jenkins,C.D., and MacPherson,G.G. (2000). A discrete subpopulation of dendritic cells transports apoptotic intestinal epithelial cells to T cell areas of mesenteric lymph nodes. J Exp. Med *191*, 435-444.

Ibrahim, M.A., Chain, B.M., and Katz, D.R. (1995). The injured cell: the role of the dendritic cell system as a sentinel receptor pathway. Immunol. Today 16, 181-186.

Ichikawa, E., Hida, S., Omatsu, Y., Shimoyama, S., Takahara, K., Miyagawa, S., Inaba, K., and Taki, S. (2004). Defective development of splenic and epidermal CD4+ dendritic cells in mice deficient for IFN regulatory factor-2. PNAS *101*, 3909-3914.

Igakura, T., Stinchcombe, J.C., Goon, P.K., Taylor, G.P., Weber, J.N., Griffiths, G.M., Tanaka, Y., Osame, M., and Bangham, C.R. (2003). Spread of HTLV-I between lymphocytes by virus-induced polarization of the cytoskeleton. Science 299, 1713-1716.

Ignatius, R., Marovich, M., Mehlhop, E., Villamide, L., Mahnke, K., Cox, W.I., Isdell, F., Frankel, S.S., Mascola, J.R., Steinman, R.M., and Pope, M. (2000). Canarypox virus-induced maturation of dendritic cells is mediated by apoptotic cell death and tumor necrosis factor alpha secretion. J Virol. 74, 11329-11338.

Inaba,K., Turley,S., Iyoda,T., Yamaide,F., Shimoyama,S., Reis e Sousa, Germain,R.N., Mellman,I., and Steinman,R.M. (2000). The formation of immunogenic major histocompatibility complex class II-peptide ligands in lysosomal compartments of dendritic cells is regulated by inflammatory stimuli. J Exp. Med. *191*, 927-936.

Inagaki-Ohara,K., Iwasaki,T., Watanabe,D., Kurata,T., and Nishiyama,Y. (2001). Effect of the deletion of US2 and US3 from herpes simplex virus type 2 on immune responses in the murine vagina following intravaginal infection. Vaccine. *20*, 98-104.

Inman,M., Perng,G.C., Henderson,G., Ghiasi,H., Nesburn,A.B., Wechsler,S.L., and Jones,C. (2001). Region of herpes simplex virus type 1 latency-associated transcript sufficient for wild-type spontaneous reactivation promotes cell survival in tissue culture. J Virol 75, 3636-3646.

Ioannidis, J.P., Collier, A.C., Cooper, D.A., Corey, L., Fiddian, A.P., Gazzard, B.G., Griffiths, P.D., Contopoulos-Ioannidis, D.G., Lau, J., Pavia, A.T., Saag, M.S., Spruance, S.L., and Youle, M.S. (1998). Clinical efficacy of high-dose acyclovir in patients with human immunodeficiency virus infection: a meta-analysis of randomized individual patient data. J Infect. Dis. *178*, 349-359.

Isaacs,A.A. and Lindenmann,J. (1957). Virus interference. I. The interferon. Proc. R. Soc. Lond B Biol. Sci. 147, 258-267.

Ito,M., Watanabe,M., Ihara,T., Kamiya,H., and Sakurai,M. (1995). Increased expression of adhesion molecules (CD54, CD29 and CD44) on fibroblasts infected with cytomegalovirus. Microbiol. Immunol. *39*, 129-133.

Ito,M., Watanabe,M., Kamiya,H., and Sakurai,M. (1997). Herpes simplex virus type 1 induces apoptosis in peripheral blood T lymphocytes. J Infect. Dis. 175, 1220-1224.

Ito,T., Amakawa,R., Kaisho,T., Hemmi,H., Tajima,K., Uehira,K., Ozaki,Y., Tomizawa,H., Akira,S., and Fukuhara,S. (2002). Interferon-alpha and interleukin-12 are induced differentially by Toll-like receptor 7 ligands in human blood dendritic cell subsets. J Exp. Med *195*, 1507-1512.

Ito,T., Inaba,M., Inaba,K., Toki,J., Sogo,S., Iguchi,T., Adachi,Y., Yamaguchi,K., Amakawa,R., Valladeau,J., Saeland,S., Fukuhara,S., and Ikehara,S. (1999). A CD1a+/CD11c+ subset of human blood dendritic cells is a direct precursor of Langerhans cells. J Immunol *163*, 1409-1419.

Iwaki, D., Mitsuzawa, H., Murakami, S., Sano, H., Konishi, M., Akino, T., and Kuroki, Y. (2002). The Extracellular Toll-like Receptor 2 Domain Directly Binds Peptidoglycan Derived from Staphylococcus aureus. J. Biol. Chem. 277, 24315-24320.

Iyoda, T., Shimoyama, S., Liu, K., Omatsu, Y., Akiyama, Y., Maeda, Y., Takahara, K., Steinman, R.M., and Inaba, K. (2002). The CD8+ dendritic cell subset selectively endocytoses dying cells in culture and in vivo. J. Exp. Med. *195*, 1289-1302.

Izaguirre, A., Barnes, B.J., Amrute, S., Yeow, W.S., Megjugorac, N., Dai, J., Feng, D., Chung, E., Pitha, P.M., and Fitzgerald-Bocarsly, P. (2003). Comparative analysis of IRF and IFN-alpha expression in human plasmacytoid and monocyte-derived dendritic cells. J Leukoc. Biol.

Jacobs,B.L. and Langland,J.O. (1996). When two strands are better than one: the mediators and modulators of the cellular responses to double-stranded RNA. Virology 219, 339-349.

~

Jacquemont, B. and Roizman, B. (1975). RNA synthesis in cells infected with herpes simplex virus. X. Properties of viral symmetric transcripts and of double-stranded RNA prepared from them. J. Virol. 15, 707-713.

Jahnsen, F.L., Lund-Johansen, F., Dunne, J.F., Farkas, L., Haye, R., and Brandtzaeg, P. (2000). Experimentally induced recruitment of plasmacytoid (CD123high) dendritic cells in human nasal allergy. J. Immunol. 165, 4062-4068.

Jahrling, P.B., Geisbert, J., Swearengen, J.R., Jaax, G.P., Lewis, T., Huggins, J.W., Schmidt, J.J., LeDuc, J.W., and Peters, C.J. (1996). Passive immunization of Ebola virus-infected cynomolgus monkeys with immunoglobulin from hyperimmune horses. Arch. Virol. Suppl 11, 135-140.

Jaksits,S., Bauer,W., Kriehuber,E., Zeyda,M., Stulnig,T.M., Stingl,G., Fiebiger,E., and Maurer,D. (2004). Lipid raft-associated GTPase signaling controls morphology and CD8+ T cell stimulatory capacity of human dendritic cells. J Immunol *173*, 1628-1639.

Janeway, C.A., Jr. (1992). The immune system evolved to discriminate infectious nonself from noninfectious self. Immunol Today. 13, 11-16.

Jenne, L., Hauser, C., Arrighi, J.F., Saurat, J.H., and Hugin, A.W. (2000). Poxvirus as a vector to transduce human dendritic cells for immunotherapy: abortive infection but reduced APC function. Gene Ther. 7, 1575-1583.

Jerome,K.R., Chen,Z., Lang,R., Torres,M.R., Hofmeister,J., Smith,S., Fox,R., Froelich,C.J., and Corey,L. (2001). HSV and glycoprotein J inhibit caspase activation and apoptosis induced by granzyme B or Fas. J. Immunol. *167*, 3928-3935.

Jerome,K.R., Fox,R., Chen,Z., Sears,A.E., Lee,H., and Corey,L. (1999). Herpes simplex virus inhibits apoptosis through the action of two genes, Us5 and Us3. J Virol. 73, 8950-8957.

Jerome, K.R., Tait, J.F., Koelle, D.M., and Corey, L. (1998). Herpes simplex virus type 1 renders infected cells resistant to cytotoxic T-lymphocyte-induced apoptosis. J. Virol. 72, 436-441.

Johansson, P.J., Myhre, E.B., and Blomberg, J. (1985). Specificity of Fc receptors induced by herpes simplex virus type 1: comparison of immunoglobulin G from different animal species. J Virol 56, 489-494.

Johnson-Leger, C., Christensen, J., and Klaus, G.G. (1998). CD28 co-stimulation stabilizes the expression of the CD40 ligand on T cells. Int. Immunol *10*, 1083-1091.

Johnson, D.C., Frame, M.C., Ligas, M.W., Cross, A.M., and Stow, N.D. (1988). Herpes simplex virus immunoglobulin G Fc receptor activity depends on a complex of two viral glycoproteins, gE and gI. J Virol *62*, 1347-1354.

Johnson, R:A., Huong, S.M., and Huang, E.S. (2000). Activation of the mitogen-activated protein kinase p38 by human cytomegalovirus infection through two distinct pathways: a novel mechanism for activation of p38. J. Virol. 74, 1158-1167.

Jones, C.A., Fernandez, M., Herc, K., Bosnjak, L., Miranda-Saksena, M., Boadle, R.A., and Cunningham, A. (2003). Herpes simplex virus type 2 induces rapid cell death and functional impairment of murine dendritic cells in vitro. J. Virol. 77, 11139-11149.

Jones, C. (2003). Herpes Simplex Virus Type 1 and Bovine Herpesvirus 1 Latency. Clin. Microbiol. Rev. 16, 79-95.

Jonuleit,H., Schmitt,E., Schuler,G., Knop,J., and Enk,A.H. (2000). Induction of interleukin 10producing, nonproliferating CD4(+) T cells with regulatory properties by repetitive stimulation with allogeneic immature human dendritic cells. J Exp. Med *192*, 1213-1222.

Jude, B.A., Pobezinskaya, Y., Bishop, J., Parke, S., Medzhitov, R.M., Chervonsky, A.V., and Golovkina, T.V. (2003). Subversion of the innate immune system by a retrovirus. Nat. Immunol 4, 573-578.

Jung,S., Unutmaz,D., Wong,P., Sano,G., De los,S.K., Sparwasser,T., Wu,S., Vuthoori,S., Ko,K., Zavala,F., Pamer,E.G., Littman,D.R., and Lang,R.A. (2002). In vivo depletion of CD11c(+) dendritic cells abrogates priming of CD8(+) T cells by exogenous cell-associated antigens. Immunity. *17*, 211-220.

Jurk, M., Heil, F., Vollmer, J., Schetter, C., Krieg, A.M., Wagner, H., Lipford, G., and Bauer, S. (2002). Human TLR7 or TLR8 independently confer responsiveness to the antiviral compound R-848. Nat. Immunol 3, 499.

Kadowaki, N., Antonenko, S., Lau, J.Y.-N., and Liu, Y.J. (2000). Natural Interferon {alpha}/{beta}producing Cells Link Innate and Adaptive Immunity. J. Exp. Med. 192, 219-226.

Kaisho, T., Takeuchi, O., Kawai, T., Hoshino, K., and Akira, S. (2001). Endotoxin-induced maturation of MyD88-deficient dendritic cells. J Immunol 166, 5688-5694.

Kakimoto, M., Hasegawa, A., Fujita, S., and Yasukawa, M. (2002). Phenotypic and functional alterations of dendritic cells induced by human herpesvirus 6 infection. J. Virol. 76, 10338-10345.

Kalinski, P., Schuitemaker, J.H., Hilkens, C.M., and Kapsenberg, M.L. (1998). Prostaglandin E2 induces the final maturation of IL-12-deficient CD1a+CD83+ dendritic cells: the levels of IL-12 are determined

during the final dendritic cell maturation and are resistant to further modulation. J Immunol 161, 2804-2809.

-

Kanangat,S., Thomas,J., Gangappa,S., Babu,J.S., and Rouse,B.T. (1996). Herpes simplex virus type 1mediated up-regulation of IL-12 (p40) mRNA expression. Implications in immunopathogenesis and protection. J Immunol 156, 1110-1116.

Kanto, T., Hayashi, N., Takehara, T., Tatsumi, T., Kuzushita, N., Ito, A., Sasaki, Y., Kasahara, A., and Hori, M. (1999). Impaired allostimulatory capacity of peripheral blood dendritic cells recovered from hepatitis C virus-infected individuals. J. Immunol. *162*, 5584-5591.

Kaplan, M.H., Sun, Y.L., Hoey, T., and Grusby, M.J. (1996). Impaired IL-12 responses and enhanced development of Th2 cells in Stat4-deficient mice. Nature. *382*, 174-177.

Kariko,K., Bhuyan,P., Capodici,J., and Weissman,D. (2004a). Small interfering RNAs mediate sequence-independent gene suppression and induce immune activation by signaling through toll-like receptor 3. J Immunol *172*, 6545-6549.

Kariko,K., Ni,H., Capodici,J., Lamphier,M., and Weissman,D. (2004b). mRNA is an endogenous ligand for Toll-like receptor 3. J Biol. Chem. 279, 12542-12550.

Karin, M. and Ben Neriah, Y. (2000). Phosphorylation meets ubiquitination: the control of NF-[kappa]B activity. Annu. Rev. Immunol 18, 621-663.

Katsuki, T., Katsuki, K., Imai, J., and Hinuma, Y. (1987). Immune suppression in healthy carriers of adult T-cell leukemia retrovirus (HTLV-I): impairment of T-cell control of Epstein-Barr virus-infected B-cells. Jpn. J Cancer Res 78, 639-642.

Keay,S. and Baldwin,B.R. (1996). Evidence for the role of cell protein phosphorylation in human cytomegalovirus/host cell fusion. J Gen Virol 77 (Pt 10), 2597-2604.

Keay,S., Baldwin,B.R., Smith,M.W., Wasserman,S.S., and Goldman,W.F. (1995). Increases in [Ca2+]i mediated by the 92.5-kDa putative cell membrane receptor for HCMV gp86. Am. J Physiol. 269, C11-C21.

Kelher, M.R., Ambruso, D.R., Elzi, D.J., Anderson, S.M., Paterson, A.J., Thurman, G.W., and Silliman, C.C. (2003). Formyl-Met-Leu-Phe induces calcium-dependent tyrosine phosphorylation of Rel-1 in neutrophils. Cell Calcium *34*, 445-455.

Keller, R., Peitchel, R., Goldman, J.N., and Goldman, M. (1976). An IgG-Fc receptor induced in cytomegalovirus-infected human fibroblasts. J Immunol 116, 772-777.

Kemp,L.M., Estridge,J.K., Brennan,A., Katz,D.R., and Latchman,D.S. (1990). Mononuclear phagocytes and HSV-1 infection: increased permissivity in differentiated U937 cells is mediated by post-transcriptional regulation of viral immediate-early gene expression. J Leukoc. Biol. 47, 483-489.

Khanna,K:M., Bonneau,R.H., Kinchington,P.R., and Hendricks,R.L. (2003). Herpes simplex virusspecific memory CD8+ T cells are selectively activated and retained in latently infected sensory ganglia. Immunity. *18*, 593-603.

Kjellen, L. and Lindahl, U. (1991). Proteoglycans: structures and interactions. Annu. Rev. Biochem. 60, 443-475.

Klagge,I.M., ter,M., V, and Schneider-Schaulies,S. (2000). Measles virus-induced promotion of dendritic cell maturation by soluble mediators does not overcome the immunosuppressive activity of viral glycoproteins on the cell surface. Eur. J Immunol *30*, 2741-2750.

Kobayashi, M., Azuma, E., Ido, M., Hirayama, M., Jiang, Q., Iwamoto, S., Kumamoto, T., Yamamoto, H., Sakurai, M., and Komada, Y. (2001). A pivotal role of Rho GTPase in the regulation of morphology and function of dendritic cells. J. Immunol. *167*, 3585-3591.

Koelle,D.M., Frank,J.M., Johnson,M.L., and Kwok,W.W. (1998a). Recognition of herpes simplex virus type 2 tegument proteins by CD4 T cells infiltrating human genital herpes lesions. J Virol 72, 7476-7483.

Koelle,D.M., Liu,Z., McClurkan,C.L., Cevallos,R.C., Vieira,J., Hosken,N.A., Meseda,C.A., Snow,D.C., Wald,A., and Corey,L. (2003). Immunodominance among herpes simplex virus-specific CD8 T cells expressing a tissue-specific homing receptor. Proc. Natl. Acad. Sci. U. S. A *100*, 12899-12904.

Koelle, D.M., Posavad, C.M., Barnum, G.R., Johnson, M.L., Frank, J.M., and Corey, L. (1998b). Clearance of HSV-2 from recurrent genital lesions correlates with infiltration of HSV-specific cytotoxic T lymphocytes. J Clin. Invest *101*, 1500-1508.

Kollet, J., Witek, C., Gentry, J.D., Liu, X., Schwartzbach, S.D., and Petro, T.M. (2001). Deletional analysis of the murine IL-12 p35 promoter comparing IFN-gamma and lipopolysaccharide stimulation. J. Immunol. *167*, 5653-5663.

Koyama, A.H. and Adachi, A. (1997). Induction of apoptosis by herpes simplex virus type 1. J. Gen. Virol. 78 (Pt 11), 2909-2912.

Koyama, A.H. and Miwa, Y. (1997). Suppression of apoptotic DNA fragmentation in herpes simplex virus type 1-infected cells. J. Virol. 71, 2567-2571.

Kramer, M.F. and Coen, D.M. (1995). Quantification of transcripts from the ICP4 and thymidine kinase genes in mouse ganglia latently infected with herpes simplex virus. J Virol 69, 1389-1399.

Kristie, T.M., Vogel, J.L., and Sears, A.E. (1999). Nuclear localization of the C1 factor (host cell factor) in sensory neurons correlates with reactivation of herpes simplex virus from latency. Proc. Natl. Acad. Sci. U. S. A. *96*, 1229-1233.

Krug,A., Towarowski,A., Britsch,S., Rothenfusser,S., Hornung,V., Bals,R., Giese,T., Engelmann,H., Endres,S., Krieg,A.M., and Hartmann,G. (2001). Toll-like receptor expression reveals CpG DNA as a unique microbial stimulus for plasmacytoid dendritic cells which synergizes with CD40 ligand to induce high amounts of IL-12. Eur. J. Immunol. *31*, 3026-3037.

Krug,A., Veeraswamy,R., Pekosz,A., Kanagawa,O., Unanue,E.R., Colonna,M., and Cella,M. (2003). Interferon-producing cells fail to induce proliferation of naive T cells but can promote expansion and T helper 1 differentiation of antigen-experienced unpolarized T cells. J. Exp. Med. *197*, 899-906.

Krug, A., Luker, G.D., Barchet, W., Leib, D.A., Akira, S., and Colonna, M. (2004). Herpes simplex virus type 1 activates murine natural interferon-producing cells through toll-like receptor 9. Blood *103*, 1433-1437.

Krummenacher, C., Baribaud, F., Ponce, D.L., Baribaud, I., Whitbeck, J.C., Xu, R., Cohen, G.H., and Eisenberg, R.J. (2004). Comparative usage of herpesvirus entry mediator A and nectin-1 by laboratory strains and clinical isolates of herpes simplex virus. Virology. *322*, 286-299.

Krummenacher, C., Baribaud, I., Ponce, D.L., Whitbeck, J.C., Lou, H., Cohen, G.H., and Eisenberg, R.J. (2000). Localization of a binding site for herpes simplex virus glycoprotein D on herpesvirus entry mediator C by using antireceptor monoclonal antibodies. J. Virol. *74*, 10863-10872.

Krummenacher, C., Baribaud, I., Sanzo, J.F., Cohen, G.H., and Eisenberg, R.J. (2002). Effects of herpes simplex virus on structure and function of nectin-1/HveC. J Virol 76, 2424-2433.

Krummenacher, C., Nicola, A.V., Whitbeck, J.C., Lou, H., Hou, W., Lambris, J.D., Geraghty, R.J., Spear, P.G., Cohen, G.H., and Eisenberg, R.J. (1998). Herpes simplex virus glycoprotein D can bind to poliovirus receptor-related protein 1 or herpesvirus entry mediator, two structurally unrelated mediators of virus entry. J. Virol. 72, 7064-7074.

Krummenacher, C., Rux, A.H., Whitbeck, J.C., Ponce-de-Leon, M., Lou, H., Baribaud, I., Hou, W., Zou, C., Geraghty, R.J., Spear, P.G., Eisenberg, R.J., and Cohen, G.H. (1999). The first immunoglobulin-like domain of HveC is sufficient to bind herpes simplex virus gD with full affinity, while the third domain is involved in oligomerization of HveC. J Virol *73*, 8127-8137.

Kruse, M., Rosorius, O., Kratzer, F., Stelz, G., Kuhnt, C., Schuler, G., Hauber, J., and Steinkasserer, A. (2000). Mature dendritic cells infected with herpes simplex virus type 1 exhibit inhibited T-cell stimulatory capacity. J Virol. 74, 7127-7136.

Ku,C.C., Zerboni,L., Ito,H., Graham,B.S., Wallace,M., and Arvin,A.M. (2004). Varicella-Zoster Virus Transfer to Skin by T Cells and Modulation of Viral Replication by Epidermal Cell Interferon-{alpha}. J. Exp. Med. 200, 917-925.

Kubin, M., 'Kamoun, M., and Trinchieri, G. (1994). Interleukin 12 synergizes with B7/CD28 interaction in inducing efficient proliferation and cytokine production of human T cells. J Exp. Med. *180*, 211-222.

Kuklin, N.A., Daheshia, M., Chun, S., and Rouse, B.T. (1998). Role of Mucosal Immunity in Herpes Simplex Virus Infection. J Immunol *160*, 5998-6003.

Kumaraguru,U. and Rouse,B.T. (2002). The IL-12 response to herpes simplex virus is mainly a paracrine response of reactive inflammatory cells. J. Leukoc. Biol. 72, 564-570.

Kurt-Jones, E.A., Chan, M., Zhou, S., Wang, J., Reed, G., Bronson, R., Arnold, M.M., Knipe, D.M., and Finberg, R.W. (2004). Herpes simplex virus 1 interaction with Toll-like receptor 2 contributes to lethal encephalitis. Proc. Natl. Acad. Sci. U. S. A.

Kurt-Jones, E.A., Popova, L., Kwinn, L., Haynes, L.M., Jones, L.P., Tripp, R.A., Walsh, E.E., Freeman, M.W., Golenbock, D.T., Anderson, L.J., and Finberg, R.W. (2000). Pattern recognition receptors TLR4 and CD14 mediate response to respiratory syncytial virus. Nat. Immunol. *1*, 398-401.

Kwon,B.S., Tan,K.B., Ni,J., Oh,K.O., Lee,Z.H., Kim,K.K., Kim,Y.J., Wang,S., Gentz,R., Yu,G.L., Harrop,J., Lyn,S.D., Silverman,C., Porter,T.G., Truneh,A., and Young,P.R. (1997). A newly identified member of the tumor necrosis factor receptor superfamily with a wide tissue distribution and involvement in lymphocyte activation. J. Biol. Chem. 272, 14272-14276.

Kwong, A.D. and Frenkel, N. (1987). Herpes simplex virus-infected cells contain a function(s) that destabilizes both host and viral mRNAs. Proc. Natl. Acad. Sci. U. S. A 84, 1926-1930.

La,S., Kim,J., Kwon,B.S., and Kwon,B. (2002). Herpes simplex virus type 1 glycoprotein D inhibits T-cell proliferation. Mol. Cells 14, 398-403.

Lachmann, R. (2004). Herpes simplex virus-based vectors. Int. J Exp. Pathol. 85, 177-190.

Laderach, D., Compagno, D., Danos, O., Vainchenker, W., and Galy, A. (2003). RNA interference shows critical requirement for NF-kappa B p50 in the production of IL-12 by human dendritic cells. J Immunol *171*, 1750-1757.

Landmann,S., Muhlethaler-Mottet,A., Bernasconi,L., Suter,T., Waldburger,J.M., Masternak,K., Arrighi,J.F., Hauser,C., Fontana,A., and Reith,W. (2001). Maturation of dendritic cells is accompanied by rapid transcriptional silencing of class II transactivator (CIITA) expression. J Exp. Med. *194*, 379-391.

Langenkamp, A., Messi, M., Lanzavecchia, A., and Sallusto, F. (2000). Kinetics of dendritic cell activation: impact on priming of TH1, TH2 and nonpolarized T cells. Nat. Immunol. 1, 311-316.

Laquerre, S., Argnani, R., Anderson, D.B., Zucchini, S., Manservigi, R., and Glorioso, J.C. (1998). Heparan sulfate proteoglycan binding by herpes simplex virus type 1 glycoproteins B and C, which differ in their contributions to virus attachment, penetration, and cell-to-cell spread. J Virol. *72*, 6119-6130.

Larregina, A.T., Watkins, S.C., Erdos, G., Spencer, L.A., Storkus, W.J., Beer, S.D., and Falo, L.D., Jr. (2001). Direct transfection and activation of human cutaneous dendritic cells. Gene Ther. *8*, 608-617.

Le Bon,A., Etchart,N., Rossmann,C., Ashton,M., Hou,S., Gewert,D., Borrow,P., and Tough,D.F. (2003). Cross-priming of CD8+ T cells stimulated by virus-induced type I interferon. Nat. Immunol 4, 1009-1015.

Le Roy, E., Baron, M., Faigle, W., Clement, D., Lewinsohn, D.M., Streblow, D.N., Nelson, J.A., Amigorena, S., and Davignon, J.L. (2002). Infection of APC by Human Cytomegalovirus Controlled Through Recognition of Endogenous Nuclear Immediate Early Protein 1 by Specific CD4+ T Lymphocytes. J Immunol *169*, 1293-1301.

Le Varlet, B., Staquet, M.J., Dezutter-Dambuyant, C., Delorme, P., and Schmitt, D. (1992). In vitro adhesion of human epidermal Langerhans cells to laminin and fibronectin occurs through beta 1 integrin receptors. J. Leukoc. Biol. 51, 415-420.

Leadbetter, E.A., Rifkin, I.R., Hohlbaum, A.M., Beaudette, B.C., Shlomchik, M.J., and Marshak-Rothstein, A. (2002). Chromatin-IgG complexes activate B cells by dual engagement of IgM and Toll-like receptors. Nature. *416*, 603-607.

Lebre, M.C., Antons, J.C., Kalinski, P., Schuitemaker, J.H., van Capel, T.M., Kapsenberg, M.L., and De Jong, E.C. (2003). Double-stranded RNA-exposed human keratinocytes promote Th1 responses by inducing a Type-1 polarized phenotype in dendritic cells: role of keratinocyte-derived tumor necrosis factor alpha, type I interferons, and interleukin-18. J Invest Dermatol *120*, 990-997.

Lee,S., Gierynska,M., Eo,S.K., Kuklin,N., and Rouse,B.T. (2003a). Influence of DNA encoding cytokines on systemic and mucosal immunity following genetic vaccination against herpes simplex virus. Microbes. Infect. 5, 571-578.

Lee, Y., Eo, S.K., Rouse, R.J., and Rouse, B.T. (2003b). Influence of CCR7 ligand DNA preexposure on the magnitude and duration of immunity. Virology 312, 169-180.

Lehner, M., Felzmann, T., Clodi, K., and Holter, W. (2001). Type I interferons in combination with bacterial stimuli induce apoptosis of monocyte-derived dendritic cells. Blood *98*, 736-742.

Leib, D.A., Harrison, T.E., Laslo, K.M., Machalek, M.A., Moorman, N.J., and Virgin, H.W. (1999). Interferons regulate the phenotype of wild-type and mutant herpes simplex viruses in vivo. J Exp. Med *189*, 663-672.

Lens, J.W., Drexhage, H.A., Benson, W., and Balfour, B.M. (1983). A study of cells present in lymph draining from a contact allergic reaction in pigs sensitized to DNFB. Immunology. 49, 415-422.

Leon, B., Martinez, d.H., Parrillas, V., Vargas, H.H., Sanchez-Mateos, P., Longo, N., Lopez-Bravo, M., and Ardavin, C. (2004). Dendritic cell differentiation potential of mouse monocytes: monocytes represent immediate precursors of CD8- and CD8+ splenic dendritic cells. Blood *103*, 2668-2676.

Leopardi, R. and Roizman, B. (1996). The herpes simplex virus major regulatory protein ICP4 blocks apoptosis induced by the virus or by hyperthermia. Proc. Natl. Acad. Sci. U. S. A 93, 9583-9587.

Leopardi, R., Van Sant, C., and Roizman, B. (1997). The herpes simplex virus 1 protein kinase US3 is required for protection from apoptosis induced by the virus. Proc. Natl. Acad. Sci. U. S. A 94, 7891-7896.

Li,J., Liu,Z., Jiang,S., Cortesini,R., Lederman,S., and Suciu-Foca,N. (1999). T suppressor lymphocytes inhibit NF-kappa B-mediated transcription of CD86 gene in APC. J Immunol *163*, 6386-6392.

Li,L., Liu,D., Hutt-Fletcher,L., Morgan,A., Masucci,M.G., and Levitsky,V. (2002). Epstein-Barr virus inhibits the development of dendritic cells by promoting apoptosis of their monocyte precursors in the presence of granulocyte macrophage-colony-stimulating factor and interleukin-4. Blood *99*, 3725-3734.

Li,P.G., Oppezzi,L., Seri,M., Westby,M., Caroli,F., Fenoglio,D., Lancia,F., Ferraris,A., Bottone,L., Valle,M.T., Kunkl,A., Romeo,G., Dalgleish,A.G., and Manca,F. (1998). Repertoire breadth of human CD4+ T cells specific for HIV gp120 and p66 (primary antigens) or for PPD and tetanus toxoid (secondary antigens). Hum. Immunol 59, 137-148.

Li,Q., Feldman,M., Harmon,C., and Fitzgerald-Bocarsly,P. (1996). Role of tyrosine kinases, protein kinase C, and protein kinase A in the regulation of interferon-alpha production induced by herpes simplex virus type 1. J Interferon Cytokine Res. *16*, 109-118.

Li,Y., Sassano,A., Majchrzak,B., Deb,D.K., Levy,D.E., Gaestel,M., Nebreda,A.R., Fish,E.N., and Platanias,L.C. (2004). Role of p38alpha Map kinase in Type I interferon signaling. J. Biol. Chem. 279, 970-979.

Libraty, D.H., Pichyangkul, S., Ajariyakhajorn, C., Endy, T.P., and Ennis, F.A. (2001). Human dendritic cells are activated by dengue virus infection: enhancement by gamma interferon and implications for disease pathogenesis. J Virol. *75*, 3501-3508.

Lillycrop,K.A., Estridge,J.K., and Latchman,D.S. (1993). The octamer binding protein Oct-2 inhibits transactivation of the herpes simplex virus immediate-early genes by the virion protein Vmw65. Virology *196*, 888-891.

Lin,G., Simmons,G., Pohlmann,S., Baribaud,F., Ni,H., Leslie,G.J., Haggarty,B.S., Bates,P., Weissman,D., Hoxie,J.A., and Doms,R.W. (2003). Differential N-linked glycosylation of human immunodeficiency virus and Ebola virus envelope glycoproteins modulates interactions with DC-SIGN and DC-SIGNR. J Virol 77, 1337-1346.

Lin,R., Noyce,R.S., Collins,S.E., Everett,R.D., and Mossman,K.L. (2004). The Herpes Simplex Virus ICP0 RING Finger Domain Inhibits IRF3- and IRF7-Mediated Activation of Interferon-Stimulated Genes. J. Virol. 78, 1675-1684.

Linehan, M.M., Richman, S., Krummenacher, C., Eisenberg, R.J., Cohen, G.H., and Iwasaki, A. (2004). In vivo role of nectin-1 in entry of herpes simplex virus type 1 (HSV-1) and HSV-2 through the vaginal mucosa. J. Virol. 78, 2530-2536.

Litwin, V., Jackson, W., and Grose, C. (1992). Receptor properties of two varicella-zoster virus glycoproteins, gpI and gpIV, homologous to herpes simplex virus gE and gI. J Virol *66*, 3643-3651.

Liu,H., Hwangbo,Y., Holte,S., Lee,J., Wang,C., Kaupp,N., Zhu,H., Celum,C., Corey,L., McElrath,M.J., and Zhu,T. (2004). Analysis of genetic polymorphisms in CCR5, CCR2, stromal cell-derived factor-1, RANTES, and dendritic cell-specific intercellular adhesion molecule-3-grabbing nonintegrin in seronegative individuals repeatedly exposed to HIV-1. J Infect. Dis. *190*, 1055-1058.

Liu, J., Cao, S., Herman, L.M., and Ma, X. (2003). Differential Regulation of Interleukin (IL)-12 p35 and p40 Gene Expression and Interferon (IFN)-{gamma}-primed IL-12 Production by IFN Regulatory Factor 1. J. Exp. Med. 198, 1265-1276.

Liu,T., Khanna,K.M., Chen,X., Fink,D.J., and Hendricks,R.L. (2000). CD8(+) T cells can block herpes simplex virus type 1 (HSV-1) reactivation from latency in sensory neurons. J. Exp. Med. 191, 1459-1466.

Liu, T., Tang, Q., and Hendricks, R.L. (1996). Inflammatory infiltration of the trigeminal ganglion after herpes simplex virus type 1 corneal infection. J. Virol. 70, 264-271.

Liu, Y., Peterson, D.A., Kimura, H., and Schubert, D. (1997). Mechanism of cellular 3-(4,5dimethylthiazol-2-yl)-2,5-diphenyltetrazolium bromide (MTT) reduction. J Neurochem. 69, 581-593.

Liu, Y. and Schubert, D. (1997). Cytotoxic amyloid peptides inhibit cellular 3-(4,5-dimethylthiazol-2-yl)-2,5-diphenyltetrazolium bromide (MTT) reduction by enhancing MTT formazan exocytosis. J Neurochem. 69, 2285-2293. Llodra, J., Angeli, V., Liu, J., Trogan, E., Fisher, E.A., and Randolph, G.J. (2004). From the Cover: Emigration of monocyte-derived cells from atherosclerotic lesions characterizes regressive, but not progressive, plaques. Proc. Natl. Acad. Sci U. S. A. 101, 11779-11784.

Longman, R.S., Talal, A.H., Jacobson, I.M., Albert, M.L., and Rice, C.M. (2004). Presence of functional dendritic cells in patients chronically infected with hepatitis C virus. Blood *103*, 1026-1029.

Lopez,C.B., Garcia-Sastre,A., Williams,B.R., and Moran,T.M. (2003). Type I interferon induction pathway, but not released interferon, participates in the maturation of dendritic cells induced by negative-strand RNA viruses. J Infect. Dis. *187*, 1126-1136.

Lore,K., Betts,M.R., Brenchley,J.M., Kuruppu,J., Khojasteh,S., Perfetto,S., Roederer,M., Seder,R.A., and Koup,R.A. (2003). Toll-like receptor ligands modulate dendritic cells to augment cytomegalovirusand HIV-1-specific T cell responses. J Immunol *171*, 4320-4328.

Lozach, P.Y., Lortat-Jacob, H., De Lacroix, D.L., Staropoli, I., Foung, S., Amara, A., Houles, C., Fieschi, F., Schwartz 0, Virelizier, J.L., Arenzana-Seisdedos, F., and Altmeyer, R. (2003). DC-SIGN and L-SIGN are high-affinity binding receptors for hepatitis C Virus glycoprotein E2. J. Biol. Chem.

Lu,H.T., Yang,D.D., Wysk,M., Gatti,E., Mellman,I., Davis,R.J., and Flavell,R.A. (1999). Defective IL-12 production in mitogen-activated protein (MAP) kinase kinase 3 (Mkk3)-deficient mice. EMBO J. 18, 1845-1857.

Lubinski, J., Wang, L., Mastellos, D., Sahu, A., Lambris, J.D., and Friedman, H.M. (1999). In vivo role of complement-interacting domains of herpes simplex virus type 1 glycoprotein gC. J Exp. Med. 190, 1637-1646.

Lubinski, J.M., Wang, L., Soulika, A.M., Burger, R., Wetsel, R.A., Colten, H., Cohen, G.H., Eisenberg, R.J., Lambris, J.D., and Friedman, H.M. (1998). Herpes simplex virus type 1 glycoprotein gC mediates immune evasion in vivo. J Virol *72*, 8257-8263.

Luft,T., Jefford,M., Luetjens,P., Hochrein,H., Masterman,K.A., Maliszewski,C., Shortman,K., Cebon,J., and Maraskovsky,E. (2002a). IL-1{beta} Enhances CD40 Ligand-Mediated Cytokine Secretion by Human Dendritic Cells (DC): A Mechanism for T Cell-Independent DC Activation. J Immunol *168*, 713-722.

Luft,T., Luetjens,P., Hochrein,H., Toy,T., Masterman,K.A., Rizkalla,M., Maliszewski,C., Shortman,K., Cebon,J., and Maraskovsky,E. (2002b). IFN-{alpha} enhances CD40 ligand-mediated activation of immature monocyte-derived dendritic cells. Int. Immunol. *14*, 367-380.

Lukac, D.M. and Alwine, J.C. (1999). Effects of human cytomegalovirus major immediate-early proteins in controlling the cell cycle and inhibiting apoptosis: studies with ts13 cells. J. Virol. *73*, 2825-2831.

Lund, J., Sato, A., Akira, S., Medzhitov, R., and Iwasaki, A. (2003). Toll-like receptor 9-mediated recognition of Herpes simplex virus-2 by plasmacytoid dendritic cells. J. Exp. Med. 198, 513-520.

-

Lund, J.M., Alexopoulou, L., Sato, A., Karow, M., Adams, N.C., Gale, N.W., Iwasaki, A., and Flavell, R.A. (2004). Recognition of single-stranded RNA viruses by Toll-like receptor 7. Proc. Natl. Acad. Sci U. S. A. 101, 5598-5603.

Lundberg, P., Welander, P., Han, X., and Cantin, E. (2003a). Herpes simplex virus type 1 DNA is immunostimulatory in vitro and in vivo. J. Virol. 77, 11158-11169.

Lundberg,P., Welander,P., Openshaw,H., Nalbandian,C., Edwards,C., Moldawer,L., and Cantin,E. (2003b). A locus on mouse chromosome 6 that determines resistance to herpes simplex virus also influences reactivation, while an unlinked locus augments resistance of female mice. J. Virol. 77, 11661-11673.

Lundqvist, A., Nagata, T., Kiessling, R., and Pisa, P. (2002). Mature dendritic cells are protected from Fas/CD95-mediated apoptosis by upregulation of Bcl-X(L). Cancer. Immunol. Immunother. 51, 139-144.

Ma,X., Sun,J., Papasavvas,E., Riemann,H., Robertson,S., Marshall,J., Bailer,R.T., Moore,A., Donnelly,R.P., Trinchieri,G., and Montaner,L.J. (2000). Inhibition of IL-12 Production in Human Monocyte-Derived Macrophages by TNF. J Immunol *164*, 1722-1729.

Mabit,H., Nakano,M.Y., Prank,U., Saam,B., Dohner,K., Sodeik,B., and Greber,U.F. (2002). Intact microtubules support adenovirus and herpes simplex virus infections. J. Virol. 76, 9962-9971.

MacDougall,T.H., Shattock,R.J., Madsen,C., Chain,B.M., and Katz,D.R. (2002). Regulation of primary HIV-1 isolate replication in dendritic cells. Clin. Exp. Immunol. *127*, 66-71.

Mackey, M.F., Wang, Z., Eichelberg, K., and Germain, R.N. (2003). Distinct contributions of different CD40 TRAF binding sites to CD154-induced dendritic cell maturation and IL-12 secretion. Eur. J. Immunol. *33*, 779-789.

MacLean,C.A., Efstathiou,S., Elliott,M.L., Jamieson,F.E., and McGeoch,D.J. (1991). Investigation of herpes simplex virus type 1 genes encoding multiply inserted membrane proteins. J. Gen. Virol. 72 (*Pt* 4), 897-906.

Mahanty,S., Hutchinson,K., Agarwal,S., McRae,M., Rollin,P.E., and Pulendran,B. (2003). Cutting edge: impairment of dendritic cells and adaptive immunity by Ebola and Lassa viruses. J Immunol 170, 2797-2801.

Makino, M., Utsunomiya, A., Maeda, Y., Shimokubo, S., Izumo, S., and Baba, M. (2001). Association of CD40 ligand expression on HTLV-I-infected T cells and maturation of dendritic cells. Scand. J Immunol 54, 574-581.

Makino, M., Wakamatsu, S., Shimokubo, S., Arima, N., and Baba, M. (2000). Production of functionally deficient dendritic cells from HTLV-I-infected monocytes: implications for the dendritic cell defect in adult T cell leukemia. Virology 274, 140-148.

Malkin, J.E. (2004). Epidemiology of genital herpes simplex virus infection in developed countries. Herpes. 11 Suppl 1, 2A-23A.

Malmgaard,L. and Paludan,S.R. (2003). Interferon (IFN)-alpha/beta, interleukin (IL)-12 and IL-18 coordinately induce production of IFN-gamma during infection with herpes simplex virus type 2. J. Gen. Virol. 84, 2497-2500.

Malmgaard,L., Paludan,S.R., Mogensen,S.C., and Ellermann-Eriksen,S. (2000). Herpes simplex virus type 2 induces secretion of IL-12 by macrophages through a mechanism involving NF-kappaB. J Gen. Virol. 81 Pt 12, 3011-3020.

Margolis, D.M., Rabson, A.B., Straus, S.E., and Ostrove, J.M. (1992). Transactivation of the HIV-1 LTR by HSV-1 immediate-early genes. Virology *186*, 788-791.

Marie, J.C., Kehren, J., Trescol-Biemont, M.C., Evlashev, A., Valentin, H., Walzer, T., Tedone, R., Loveland, B., Nicolas, J.F., Rabourdin-Combe, C., and Horvat, B. (2001). Mechanism of measles virus-induced suppression of inflammatory immune responses. Immunity. *14*, 69-79.

Marozin, S., Prank, U., and Sodeik, B. (2004). Herpes simplex virus type 1 infection of polarized epithelial cells requires microtubules and access to receptors present at cell-cell contact sites. J. Gen. Virol. *85*, 775-786.

Marsters,S.A., Ayres,T.M., Skubatch,M., Gray,C.L., Rothe,M., and Ashkenazi,A. (1997). Herpesvirus entry mediator, a member of the tumor necrosis factor receptor (TNFR) family, interacts with members of the TNFR-associated factor family and activates the transcription factors NF-kappaB and AP-1. J Biol. Chem. *272*, 14029-14032.

Maruyama,S., Sumita,K., Shen,H., Kanoh,M., Xu,X., Sato,M., Matsumoto,M., Shinomiya,H., and Asano,Y. (2003). Identification of IFN regulatory factor-1 binding site in IL-12 p40 gene promoter. J. Immunol. *170*, 997-1001.

Mason,N., Aliberti,J., Caamano,J.C., Liou,H.C., and Hunter,C.A. (2002). Cutting edge: identification of c-Rel-dependent and -independent pathways of IL-12 production during infectious and inflammatory stimuli. J Immunol *168*, 2590-2594.

Mathur,R.K., Awasthi,A., Wadhone,P., Ramanamurthy,B., and Saha,B. (2004). Reciprocal CD40 signals through p38MAPK and ERK-1/2 induce counteracting immune responses. Nat. Med 10, 540-544.

Matsuguchi, T., Musikacharoen, T., Ogawa, T., and Yoshikai, Y. (2000). Gene expressions of Toll-like receptor 2, but not Toll-like receptor 4, is induced by LPS and inflammatory cytokines in mouse macrophages. J Immunol *165*, 5767-5772.

Matsumoto, M., Funami, K., Tanabe, M., Oshiumi, H., Shingai, M., Seto, Y., Yamamoto, A., and Seya, T. (2003). Subcellular Localization of Toll-Like Receptor 3 in Human Dendritic Cells. J Immunol 171, 3154-3162.

Matzinger, P. (1998). An innate sense of danger. Semin. Immunol 10, 399-415.

McCaffrey,R.L., Fawcett,P., O'Riordan,M., Lee,K.D., Havell,E.A., Brown,P.O., and Portnoy,D.A. (2004). From the Cover: A specific gene expression program triggered by Gram-positive bacteria in the cytosol. PNAS *101*, 11386-11391.

McCarthy,D.A., Macey,M.G., Bedford,P.A., Knight,S.C., Dumonde,D.C., and Brown,K.A. (1997). Adhesion molecules are upregulated on dendritic cells isolated from human blood. Immunology 92, 244-251.

McRae,B.L., Nagai,T., Semnani,R.T., van Seventer,J.M., and van Seventer,G.A. (2000a). Interferonalpha and -beta inhibit the in vitro differentiation of immunocompetent human dendritic cells from CD14(+) precursors. Blood *96*, 210-217.

McRae, B.L., Beilfuss, B.A., and Seventer, G.A. (2000b). IFN-{beta} Differentially Regulates CD40-Induced Cytokine Secretion by Human Dendritic Cells. J Immunol *164*, 23-28.

Medici,M.A., Sciortino,M.T., Perri,D., Amici,C., Avitabile,E., Ciotti,M., Balestrieri,E., De Smaele,E., Franzoso,G., and Mastino,A. (2003). Protection by herpes simplex virus glycoprotein D against Fasmediated apoptosis: role of nuclear factor kappaB. J. Biol. Chem. 278, 36059-36067.

Medzhitov, R., Preston-Hurlburt, P., and Janeway, C.A., Jr. (1997). A human homologue of the Drosophila Toll protein signals activation of adaptive immunity. Nature 388, 394-397.

Melchjorsen, J. and Paludan, S.R. (2003). Induction of RANTES/CCL5 by herpes simplex virus is regulated by nuclear factor kappa B and interferon regulatory factor 3. J. Gen. Virol. 84, 2491-2495.

Melchjorsen, J., Pedersen, F.S., Mogensen, S.C., and Paludan, S.R. (2002). Herpes simplex virus selectively induces expression of the CC chemokine RANTES/CCL5 in macrophages through a mechanism dependent on PKR and ICP0. J Virol *76*, 2780-2788.

Michelsen,K.S., Aicher,A., Mohaupt,M., Hartung,T., Dimmeler,S., Kirschning,C.J., and Schumann,R.R. (2001). The role of toll-like receptors (TLRs) in bacteria-induced maturation of murine dendritic cells (DCS). Peptidoglycan and lipoteichoic acid are inducers of DC maturation and require TLR2. J Biol. Chem. 276, 25680-25686.

Miettinen, M., Sareneva, T., Julkunen, I., and Matikainen, S. (2001). IFNs activate toll-like receptor gene expression in viral infections. Genes. Immun. 2, 349-355.

Mikloska,Z., Bosnjak,L., and Cunningham,A.L. (2001). Immature monocyte-derived dendritic cells are productively infected with herpes simplex virus type 1. J Virol. 75, 5958-5964.

Mikloska,Z. and Cunningham,A.L. (2001). Alpha and gamma interferons inhibit herpes simplex virus type 1 infection and spread in epidermal cells after axonal transmission. J Virol. 75, 11821-11826.

Mikloska,Z., Danis,V.A., Adams,S., Lloyd,A.R., Adrian,D.L., and Cunningham,A.L. (1998). In vivo production of cytokines and beta (C-C) chemokines in human recurrent herpes simplex lesions--do herpes simplex virus-infected keratinocytes contribute to their production? J Infect. Dis. 177, 827-838.

Mikloska,Z., Kesson,A.M., Penfold,M.E., and Cunningham,A.L. (1996). Herpes simplex virus protein targets for CD4 and CD8 lymphocyte cytotoxicity in cultured epidermal keratinocytes treated with interferon-gamma. J. Infect. Dis. 173, 7-17.

Miller, M.J., Wei, S.H., Parker, I., and Cahalan, M.D. (2002). Two-photon imaging of lymphocyte motility and antigen response in intact lymph node. Science. 296, 1869-1873.

Miller, M.J., Hejazi, A.S., Wei, S.H., Cahalan, M.D., and Parker, I. (2004). T cell repertoire scanning is promoted by dynamic dendritic cell behavior and random T cell motility in the lymph node. PNAS *101*, 998-1003.

Milligan,G.N. and Bernstein,D.I. (1997). Interferon-[gamma] Enhances Resolution of Herpes Simplex Virus Type 2 Infection of the Murine Genital Tract. Virology 229, 259-268.

Milone, M.C. and Fitzgerald-Bocarsly, P. (1998b). The mannose receptor mediates induction of IFNalpha in peripheral blood dendritic cells by enveloped RNA and DNA viruses. J Immunol 161, 2391-2399.

Milone, M.C. and Fitzgerald-Bocarsly, P. (1998a). The mannose receptor mediates induction of IFNalpha in peripheral blood dendritic cells by enveloped RNA and DNA viruses. J Immunol *161*, 2391-2399.

Minami, M., Kita, M., Yan, X.Q., Yamamoto, T., Iida, T., Sekikawa, K., Iwakura, Y., and Imanishi, J. (2002). Role of IFN-gamma and tumor necrosis factor-alpha in herpes simplex virus type 1 infection. J Interferon. Cytokine. Res. 22, 671-676.

Mittnacht, S., Straub, P., Kirchner, H., and Jacobsen, H. (1988). Interferon treatment inhibits onset of herpes simplex virus immediate-early transcription. Virology *164*, 201-210.

Mogensen, T.H., Melchjorsen, J., Hollsberg, P., and Paludan, S.R. (2003). Activation of NF-kappa B in virus-infected macrophages is dependent on mitochondrial oxidative stress and intracellular calcium: downstream involvement of the kinases TGF-beta-activated kinase 1, mitogen-activated kinase/extracellular signal-regulated kinase kinase 1, and I kappa B kinase. J Immunol *170*, 6224-6233.

Mohty, M., Vialle-Castellano, A., Nunes, J.A., Isnardon, D., Olive, D., and Gaugler, B. (2003). IFN-alpha skews monocyte differentiation into Toll-like receptor 7-expressing dendritic cells with potent functional activities. J Immunol *171*, 3385-3393.

Mole,L., Ripich,S., Margolis,D., and Holodniy,M. (1997). The impact of active herpes simplex virus infection on human immunodeficiency virus load. J Infect. Dis. 176, 766-770.

Montgomery, R.I., Warner, M.S., Lum, B.J., and Spear, P.G. (1996). Herpes simplex virus-1 entry into cells mediated by a novel member of the TNF/NGF receptor family. Cell 87, 427-436.

Montoya, M., Schiavoni, G., Mattei, F., Gresser, I., Belardelli, F., Borrow, P., and Tough, D.F. (2002). Type I interferons produced by dendritic cells promote their phenotypic and functional activation. Blood *99*, 3263-3271.

Morel,Y., Truneh,A., Costello,R.T., and Olive,D. (2003). LIGHT, a new TNF superfamily member, is essential for memory T helper cell-mediated activation of dendritic cells. Eur. J Immunol *33*, 3213-3219.

Moron,G., Rueda,P., Casal,I., and Leclerc,C. (2002). CD8alpha- CD11b+ dendritic cells present exogenous virus-like particles to CD8+ T cells and subsequently express CD8alpha and CD205 molecules. J. Exp. Med. 195, 1233-1245.

Morrow, G., Slobedman, B., Cunningham, A.L., and Abendroth, A. (2003). Varicella-zoster virus productively infects mature dendritic cells and alters their immune function. J Virol. 77, 4950-4959.

Mosca,J.D., Bednarik,D.P., Raj,N.B., Rosen,C.A., Sodroski,J.G., Haseltine,W.A., Hayward,G.S., and Pitha,P.M. (1987). Activation of human immunodeficiency virus by herpesvirus infection: identification of a region within the long terminal repeat that responds to a trans-acting factor encoded by herpes simplex virus 1. Proc. Natl. Acad. Sci U. S. A. 84, 7408-7412.

Mosieniak, G., Pyrzynska, B., and Kaminska, B. (1998). Nuclear factor of activated T cells (NFAT) as a new component of the signal transduction pathway in glioma cells. J Neurochem. 71, 134-141.

Mosmann, T. (1983). Rapid colorimetric assay for cellular growth and survival: application to proliferation and cytotoxicity assays. J Immunol Methods. 65, 55-63.

Mossman,K.L., Macgregor,P.F., Rozmus,J.J., Goryachev,A.B., Edwards,A.M., and Smiley,J.R. (2001). Herpes simplex virus triggers and then disarms a host antiviral response. J Virol. *75*, 750-758.

Moutaftsi, M., Mehl, A.M., Borysiewicz, L.K., and Tabi, Z. (2002). Human cytomegalovirus inhibits maturation and impairs function of monocyte-derived dendritic cells. Blood *99*, 2913-2921.

Moutaftsi,M., Brennan,P., Spector,S.A., and Tabi,Z. (2004). Impaired Lymphoid Chemokine-Mediated Migration due to a Block on the Chemokine Receptor Switch in Human Cytomegalovirus-Infected Dendritic Cells. J. Virol. 78, 3046-3054.

Mueller,S.N., Jones,C.M., Smith,C.M., Heath,W.R., and Carbone,F.R. (2002). Rapid cytotoxic T lymphocyte activation occurs in the draining lymph nodes after cutaneous herpes simplex virus infection as a result of early antigen presentation and not the presence of virus. J. Exp. Med. *195*, 651-656.

Muller, D.B., Raftery, M.J., Kather, A., Giese, T., and Schonrich, G. (2004). Frontline: Induction of apoptosis and modulation of c-FLIPL and p53 in immature dendritic cells infected with herpes simplex virus. Eur. J Immunol *34*, 941-951.

Murabayashi,N., Kurita-Taniguchi,M., Ayata,M., Matsumoto,M., Ogura,H., and Seya,T. (2002). Susceptibility of human dendritic cells (DCs) to measles virus (MV) depends on their activation stages in conjunction with the level of CDw150: role of Toll stimulators in DC maturation and MV amplification. Microbes. Infect. *4*, 785-794.

Murata, T., Goshima, F., Daikoku, T., Takakuwa, H., and Nishiyama, Y. (2000). Expression of herpes simplex virus type 2 US3 affects the Cdc42/Rac pathway and attenuates c-Jun N-terminal kinase activation. Genes Cells 5, 1017-1027.

Murata, T., Goshima, F., Takakuwa, H., and Nishiyama, Y. (2002). Excretion of herpes simplex virus type 2 glycoprotein D into the culture medium. J Gen Virol *83*, 2791-2795.

Murphy,E.E., Terres,G., Macatonia,S.E., Hsieh,C.S., Mattson,J., Lanier,L., Wysocka,M., Trinchieri,G., Murphy,K., and O'Garra,A. (1994). B7 and interleukin 12 cooperate for proliferation and interferon gamma production by mouse T helper clones that are unresponsive to B7 costimulation. J Exp. Med. *180*, 223-231.

Murphy, J.A., Duerst, R.J., Smith, T.J., and Morrison, L.A. (2003). Herpes simplex virus type 2 virion host shutoff protein regulates alpha/beta interferon but not adaptive immune responses during primary infection in vivo. J. Virol. 77, 9337-9345.

Murphy, T.L., Cleveland, M.G., Kulesza, P., Magram, J., and Murphy, K.M. (1995). Regulation of interleukin 12 p40 expression through an NF-kappa B half-site. Mol. Cell Biol. 15, 5258-5267.
Muzio, M., Bosisio, D., Polentarutti, N., D'amico, G., Stoppacciaro, A., Mancinelli, R., van't Veer, C., Penton-Rol, G., Ruco, L.P., Allavena, P., and Mantovani, A. (2000). Differential expression and regulation of toll-like receptors (TLR) in human leukocytes: selective expression of TLR3 in dendritic cells. J Immunol *164*, 5998-6004.

Nagashunmugam, T., Lubinski, J., Wang, L., Goldstein, L.T., Weeks, B.S., Sundaresan, P., Kang, E.H., Dubin, G., and Friedman, H.M. (1998). In vivo immune evasion mediated by the herpes simplex virus type 1 immunoglobulin G Fc receptor. J Virol. 72, 5351-5359.

Navas, M.C., Fuchs, A., Schvoerer, E., Bohbot, A., Aubertin, A.M., and Stoll-Keller, F. (2002). Dendritic cell susceptibility to hepatitis C virus genotype 1 infection. J Med Virol *67*, 152-161.

Neininger, A., Kontoyiannis, D., Kotlyarov, A., Winzen, R., Eckert, R., Volk, H.D., Holtmann, H., Kollias, G., and Gaestel, M. (2002). MK2 targets AU-rich elements and regulates biosynthesis of tumor necrosis factor and interleukin-6 independently at different post-transcriptional levels. J Biol. Chem. 277, 3065-3068.

Nersting, J., Svenson, M., Andersen, V., and Bendtzen, K. (2003). Maturation of human dendritic cells by monocyte-conditioned medium is dependent upon trace amounts of lipopolysaccharide inducing tumour necrosis factor [alpha]. Immunology Letters *89*, 59-65.

Netterwald, J.R., Jones, T.R., Britt, W.J., Yang, S.J., McCrone, I.P., and Zhu, H. (2004). Postattachment Events Associated with Viral Entry Are Necessary for Induction of Interferon-Stimulated Genes by Human Cytomegalovirus. J. Virol. 78, 6688-6691.

Neumann, J., Eis-Hubinger, A.M., and Koch, N. (2003). Herpes simplex virus type 1 targets the MHC class II processing pathway for immune evasion. J. Immunol. *171*, 3075-3083.

Neumann,M., Fries,H., Scheicher,C., Keikavoussi,P., Kolb-Maurer,A., Brocker,E., Serfling,E., and Kampgen,E. (2000). Differential expression of Rel/NF-kappaB and octamer factors is a hallmark of the generation and maturation of dendritic cells. Blood *95*, 277-285.

Ngo,V.N., Tang,H.L., and Cyster,J.G. (1998). Epstein-Barr virus-induced molecule 1 ligand chemokine is expressed by dendritic cells in lymphoid tissues and strongly attracts naive T cells and activated B cells. J Exp. Med *188*, 181-191.

Nicholl,M.J. and Preston,C.M. (1996). Inhibition of herpes simplex virus type 1 immediate-early gene expression by alpha interferon is not VP16 specific. J. Virol. 70, 6336-6339.

Nicholl,M.J., Robinson,L.H., and Preston,C.M. (2000). Activation of cellular interferon-responsive genes after infection of human cells with herpes simplex virus type 1. J Gen. Virol. *81*, 2215-2218.

Nicola,A.V., McEvoy,A.M., and Straus,S.E. (2003). Roles for endocytosis and low pH in herpes simplex virus entry into HeLa and Chinese hamster ovary cells. J Virol 77, 5324-5332.

Nicola,A.V., Ponce,D.L., Xu,R., Hou,W., Whitbeck,J.C., Krummenacher,C., Montgomery,R.I., Spear,P.G., Eisenberg,R.J., and Cohen,G.H. (1998). Monoclonal antibodies to distinct sites on herpes simplex virus (HSV) glycoprotein D block HSV binding to HVEM. J. Virol. *72*, 3595-3601.

Nigou, J., Zelle-Rieser, C., Gilleron, M., Thurnher, M., and Puzo, G. (2001). Mannosylated lipoarabinomannans inhibit IL-12 production by human dendritic cells: evidence for a negative signal delivered through the mannose receptor. J Immunol *166*, 7477-7485.

Nilsen,N., Nonstad,U., Khan,N., Knetter,C.F., Akira,S., Sundan,A., Espevik,T., and Lien,E. (2004). Lipopolysaccharide and double-stranded RNA up-regulate toll-like receptor 2 independently of myeloid differentiation factor 88. J Biol. Chem. 279, 39727-39735.

Nishiyama,Y., Yamada,Y., Kurachi,R., and Daikoku,T. (1992). Construction of a US3 lacZ insertion mutant of herpes simplex virus type 2 and characterization of its phenotype in vitro and in vivo. Virology. 190, 256-268.

Oberman, F. and Panet, A. (1988). Inhibition of transcription of herpes simplex virus immediate early genes in interferon-treated human cells. J Gen Virol 69, 1167-1177.

Ohshima,Y., Yang,L.P., Uchiyama,T., Tanaka,Y., Baum,P., Sergerie,M., Hermann,P., and Delespesse,G. (1998). OX40 Costimulation Enhances Interleukin-4 (IL-4) Expression at Priming and Promotes the Differentiation of Naive Human CD4+ T Cells Into High IL-4-Producing Effectors. Blood *92*, 3338-3345.

Oppmann,B., Lesley,R., Blom,B., Timans,J.C., Xu,Y., Hunte,B., Vega,F., Yu,N., Wang,J., Singh,K., Zonin,F., Vaisberg,E., Churakova,T., Liu,M., Gorman,D., Wagner,J., Zurawski,S., Liu,Y., Abrams,J.S., Moore,K.W., Rennick,D., Waal-Malefyt,R., Hannum,C., Bazan,J.F., and Kastelein,R.A. (2000). Novel p19 protein engages IL-12p40 to form a cytokine, IL-23, with biological activities similar as well as distinct from IL-12. Immunity. *13*, 715-725.

Oshiumi,H., Matsumoto,M., Funami,K., Akazawa,T., and Seya,T. (2003). TICAM-1, an adaptor molecule that participates in Toll-like receptor 3-mediated interferon-beta induction. Nat. Immunol 4, 161-167.

Ouaaz,F., Arron,J., Zheng,Y., Choi,Y., and Beg,A.A. (2002). Dendritic cell development and survival require distinct NF-kappaB subunits. Immunity. *16*, 257-270.

Ozinsky, A., Underhill, D.M., Fontenot, J.D., Hajjar, A.M., Smith, K.D., Wilson, C.B., Schroeder, L., and Aderem, A. (2000). The repertoire for pattern recognition of pathogens by the innate immune system is defined by cooperation between Toll-like receptors. PNAS *97*, 13766-13771.

Padovan, E., Spagnoli, G.C., Ferrantini, M., and Heberer, M. (2002). IFN-alpha2a induces IP-10/CXCL10 and MIG/CXCL9 production in monocyte-derived dendritic cells and enhances their capacity to attract and stimulate CD8+ effector T cells. J Leukoc. Biol. *71*, 669-676.

Pahl,H.L. and Baeuerle,P.A. (1997). The ER-overload response: activation of NF-[kappa]B. Trends in Biochemical Sciences 22, 63-67.

Palliser, D., Huang, Q., Hacohen, N., Lamontagne, S.P., Guillen, E., Young, R.A., and Eisen, H.N. (2004). A role for Toll-like receptor 4 in dendritic cell activation and cytolytic CD8+ T cell differentiation in response to a recombinant heat shock fusion protein. J Immunol *172*, 2885-2893.

Palucka,K.A., Taquet,N., Sanchez-Chapuis,F., and Gluckman,J.C. (1998). Dendritic cells as the terminal stage of monocyte differentiation. J Immunol *160*, 4587-4595.

Paludan,S.R., Ellermann-Eriksen,S., Kruys,V., and Mogensen,S.C. (2001). Expression of TNF-alpha by herpes simplex virus-infected macrophages is regulated by a dual mechanism: transcriptional regulation by NF-kappa B and activating transcription factor 2/Jun and translational regulation through the AU-rich region of the 3' untranslated region. J. Immunol. *167*, 2202-2208.

Paludan,S.R. and Mogensen,S.C. (2001). Virus-cell interactions regulating induction of tumor necrosis factor alpha production in macrophages infected with herpes simplex virus. J Virol. 75, 10170-10178.

Panther,E., Corinti,S., Idzko,M., Herouy,Y., Napp,M., la Sala,A., Girolomoni,G., and Norgauer,J. (2003). Adenosine affects expression of membrane molecules, cytokine and chemokine release, and the T-cell stimulatory capacity of human dendritic cells. Blood *101*, 3985-3990.

Paquette, R.L., Hsu, N.C., Kiertscher, S.M., Park, A.N., Tran, L., Roth, M.D., and Glaspy, J.A. (1998). Interferon-alpha and granulocyte-macrophage colony-stimulating factor differentiate peripheral blood monocytes into potent antigen-presenting cells. J Leukoc. Biol. *64*, 358-367.

Para,M.F., Goldstein,L., and Spear,P.G. (1982). Similarities and differences in the Fc-binding glycoprotein (gE) of herpes simplex virus types 1 and 2 and tentative mapping of the viral gene for this glycoprotein. J Virol 41, 137-144.

Parlato,S., Santini,S.M., Lapenta,C., Di Pucchio,T., Logozzi,M., Spada,M., Giammarioli,A.M., Malorni,W., Fais,S., and Belardelli,F. (2001). Expression of CCR-7, MIP-3beta, and Th-1 chemokines in type I IFN-induced monocyte-derived dendritic cells: importance for the rapid acquisition of potent migratory and functional activities. Blood *98*, 3022-3029.

Parr,E.L. and Parr,M.B. (1997). Immunoglobulin G is the main protective antibody in mouse vaginal secretions after vaginal immunization with attenuated herpes simplex virus type 2. J. Virol. 71, 8109-8115.

Parr,M.B. and Parr,E.L. (2000). Immunity to vaginal herpes simplex virus-2 infection in B-cell knockout mice. Immunology *101*, 126-131.

Partida-Sanchez, S., Goodrich, S., Kusser, K., Oppenheimer, N., Randall, T.D., and Lund, F.E. (2004). Regulation of Dendritic Cell Trafficking by the ADP-Ribosyl Cyclase CD38: Impact on the Development of Humoral Immunity. Immunity 20, 279-291.

Patel,A., Hanson,J., McLean,T.I., Olgiate,J., Hilton,M., Miller,W.E., and Bachenheimer,S.L. (1998). Herpes simplex type 1 induction of persistent NF-kappa B nuclear translocation increases the efficiency of virus replication. Virology 247, 212-222.

Pathan,A.A., Wilkinson,K.A., Klenerman,P., McShane,H., Davidson,R.N., Pasvol,G., Hill,A.V., and Lalvani,A. (2001). Direct ex vivo analysis of antigen-specific IFN-gamma-secreting CD4 T cells in Yuberculosis-infected individuals: associations with clinical disease state and effect of treatment. J Immunol *167*, 5217-5225.

Peng,X., Remacle,J.E., Kasran,A., Huylebroeck,D., and Ceuppens,J.L. (1998). IL-12 up-regulates CD40 ligand (CD154) expression on human T cells. J Immunol *160*, 1166-1172.

Penna,G., Sozzani,S., and Adorini,L. (2001). Cutting Edge: Selective Usage of Chemokine Receptors by Plasmacytoid Dendritic Cells. J Immunol *167*, 1862-1866.

Pereira, R.A., Scalzo, A., and Simmons, A. (2001). Cutting edge: a NK complex-linked locus governs acute versus latent herpes simplex virus infection of neurons. J. Immunol. *166*, 5869-5873.

Perkins, D., Gyure, K.A., Pereira, E.F., and Aurelian, L. (2003). Herpes simplex virus type 1-induced encephalitis has an apoptotic component associated with activation of c-Jun N-terminal kinase. J Neurovirol. 9, 101-111.

Perkins, D., Pereira, E.F., Gober, M., Yarowsky, P.J., and Aurelian, L. (2002). The herpes simplex virus type 2 R1 protein kinase (ICP10 PK) blocks apoptosis in hippocampal neurons, involving activation of the MEK/MAPK survival pathway. J Virol 76, 1435-1449.

Perng,G.C., Jones,C., Ciacci-Zanella,J., Stone,M., Henderson,G., Yukht,A., Slanina,S.M., Hofman,F.M., Ghiasi,H., Nesburn,A.B., and Wechsler,S.L. (2000). Virus-induced neuronal apoptosis blocked by the herpes simplex virus latency-associated transcript. Science. 287, 1500-1503.

Pflanz,S., Timans,J.C., Cheung,J., Rosales,R., Kanzler,H., Gilbert,J., Hibbert,L., Churakova,T., Travis,M., Vaisberg,E., Blumenschein,W.M., Mattson,J.D., Wagner,J.L., To,W., Zurawski,S., McClanahan,T.K., Gorman,D.M., Bazan,J.F., de Waal,M.R., Rennick,D., and Kastelein,R.A. (2002). IL-27, a heterodimeric cytokine composed of EBI3 and p28 protein, induces proliferation of naive CD4(+) T cells. Immunity. *16*, 779-790.

Philpott,N.J., Nociari,M., Elkon,K.B., and Falck-Pedersen,E. (2004). Adenovirus-induced maturation of dendritic cells through a PI3 kinase-mediated TNF-alpha induction pathway. Proc. Natl. Acad. Sci. U. S. A. 101, 6200-6205.

Pichyangkul, S., Endy, T.P., Kalayanarooj, S., Nisalak, A., Yongvanitchit, K., Green, S., Rothman, A.L., Ennis, F.A., and Libraty, D.H. (2003). A blunted blood plasmacytoid dendritic cell response to an acute systemic viral infection is associated with increased disease severity. J Immunol *171*, 5571-5578.

Pickl,W.F., Majdic,O., Kohl,P., Stockl,J., Riedl,E., Scheinecker,C., Bello-Fernandez,C., and Knapp,W. (1996). Molecular and functional characteristics of dendritic cells generated from highly purified CD14+ peripheral blood monocytes. J. Immunol. *157*, 3850-3859.

Plotnicky-Gilquin,H., Cyblat,D., Aubry,J.P., Delneste,Y., Blaecke,A., Bonnefoy,J.Y., Corvaia,N., and Jeannin,P. (2001). Differential effects of parainfluenza virus type 3 on human monocytes and dendritic cells. Virology 285, 82-90.

Poltorak, A., He, X., Smirnova, I., Liu, M.Y., Van Huffel, C., Du, X., Birdwell, D., Alejos, E., Silva, M., Galanos, C., Freudenberg, M., Ricciardi-Castagnoli, P., Layton, B., and Beutler, B. (1998a). Defective LPS signaling in C3H/HeJ and C57BL/10ScCr mice: mutations in Tlr4 gene. Science. *282*, 2085-2088.

Poltorak, A., He, X., Smirnova, I., Liu, M.Y., Huffel, C.V., Du, X., Birdwell, D., Alejos, E., Silva, M., Galanos, C., Freudenberg, M., Ricciardi-Castagnoli, P., Layton, B., and Beutler, B. (1998b). Defective LPS Signaling in C3H/HeJ and C57BL/10ScCr Mice: Mutations in Tlr4 Gene. Science *282*, 2085-2088.

Pooley,J.L., Heath,W.R., and Shortman,K. (2001). Cutting edge: intravenous soluble antigen is presented to CD4 T cells by CD8- dendritic cells, but cross-presented to CD8 T cells by CD8+ dendritic cells. J. Immunol. *166*, 5327-5330.

Porcelli,S.A., Segelke,B.W., Sugita,M., Wilson,I.A., and Brenner,M.B. (1998). The CD1 family of lipid antigen-presenting molecules. Immunol. Today. 19, 362-368.

Posavad,C.M., Koelle,D.M., Shaughnessy,M.F., and Corey,L. (1997). Severe genital herpes infections in HIV-infected individuals with impaired herpes simplex virus-specific CD8+ cytotoxic T lymphocyte responses. Proc. Natl. Acad. Sci. U. S. A *94*, 10289-10294.

Posavad,C.M., Wald,A., Hosken,N., Huang,M.L., Koelle,D.M., Ashley,R.L., and Corey,L. (2003). T cell immunity to herpes simplex viruses in seronegative subjects: silent infection or acquired immunity? J Immunol *170*, 4380-4388.

Postigo, A.A., Sheppard, A.M., Mucenski, M.L., and Dean, D.C. (1997). c-Myb and Ets proteins synergize to overcome transcriptional repression by ZEB. EMBO J. *16*, 3924-3934.

Preston, C.M., Harman, A.N., and Nicholl, M.J. (2001). Activation of interferon response factor-3 in human cells infected with herpes simplex virus type 1 or human cytomegalovirus. J Virol. 75, 8909-8916.

Puig-Kroger, A., Relloso, M., Fernandez-Capetillo, O., Zubiaga, A., Silva, A., Bernabeu, C., and Corbi, A.L. (2001). Extracellular signal-regulated protein kinase signaling pathway negatively regulates the phenotypic and functional maturation of monocyte-derived human dendritic cells. Blood *98*, 2175-2182.

Puig-Kroger, A., Sanz-Rodriguez, F., Longo, N., Sanchez-Mateos, P., Botella, L., Teixido, J., Bernabeu, C., and Corbi, A.L. (2000). Maturation-dependent expression and function of the CD49d integrin on monocyte-derived human dendritic cells. J. Immunol. *165*, 4338-4345.

Pyrzynska,B., Lis,A., Mosieniak,G., and Kaminska,B. (2001). Cyclosporin A-sensitive signaling pathway involving calcineurin regulates survival of reactive astrocytes. Neurochem. Int. 38, 409-415.

Qi,H., Denning,T.L., and Soong,L. (2003). Differential Induction of Interleukin-10 and Interleukin-12 in Dendritic Cells by Microbial Toll-Like Receptor Activators and Skewing of T-Cell Cytokine Profiles. Infect. Immun. 71, 3337-3342.

Qie,L., Marcellino,D., and Herold,B.C. (1999). Herpes simplex virus entry is associated with tyrosine phosphorylation of cellular proteins. Virology 256, 220-227.

Racanelli,V., Behrens,S.E., Aliberti,J., and Rehermann,B. (2004). Dendritic cells transfected with cytopathic self-replicating RNA induce crosspriming of CD8+ T cells and antiviral immunity. Immunity. 20, 47-58.

Rad,A.N., Pollara,G., Sohaib,S.M., Chiang,C., Chain,B.M., and Katz,D.R. (2003). The differential influence of allogeneic tumor cell death via DNA damage on dendritic cell maturation and antigen presentation. Cancer Res. *63*, 5143-5150.

Raftery,M.J., Behrens,C.K., Muller,A., Krammer,P.H., Walczak,H., and Schonrich,G. (1999). Herpes simplex virus type 1 infection of activated cytotoxic T cells: Induction of fratricide as a mechanism of viral immune evasion. J Exp. Med. *190*, 1103-1114.

Raftery, M.J., Kraus, A.A., Ulrich, R., Kruger, D.H., and Schonrich, G. (2002). Hantavirus infection of dendritic cells. J Virol. 76, 10724-10733.

Raftery, M.J., Schwab, M., Eibert, S.M., Samstag, Y., Walczak, H., and Schonrich, G. (2001). Targeting the function of mature dendritic cells by human cytomegalovirus: a multilayered viral defense strategy. Immunity. *15*, 997-1009.

Raingeaud, J., Gupta, S., Dickens, M., and Han, J. (1995). Pro-inflammatory Cytokines and Environmental Stress Cause p38 Mitogen-activated Protein Kinase Activation by Dual Phosphorylation on Tyrosine and Threonine. J. Biol. Chem. 270, 7420-7426.

Ramirez, M.C. and Sigal, L.J. (2002). Macrophages and dendritic cells use the cytosolic pathway to rapidly cross-present antigen from live, vaccinia-infected cells. J. Immunol. *169*, 6733-6742.

Ramsey-Ewing, A. and Moss, B. (1998). Apoptosis Induced by a Postbinding Step of Vaccinia Virus Entry into Chinese Hamster Ovary Cells*1. Virology 242, 138-149.

Randolph,G.J., Beaulieu,S., Lebecque,S., Steinman,R.M., and Muller,W.A. (1998). Differentiation of monocytes into dendritic cells in a model of transendothelial trafficking. Science 282, 480-483.

Randolph,G.J., Inaba,K., Robbiani,D.F., Steinman,R.M., and Muller,W.A. (1999). Differentiation of phagocytic monocytes into lymph node dendritic cells in vivo. Immunity 11, 753-761.

Rassa, J.C., Meyers, J.L., Zhang, Y., Kudaravalli, R., and Ross, S.R. (2002). Murine retroviruses activate B cells via interaction with toll-like receptor 4. PNAS *99*, 2281-2286.

Re,F. and Strominger,J.L. (2001). Toll-like receptor 2 (TLR2) and TLR4 differentially activate human dendritic cells. J. Biol. Chem. 276, 37692-37699.

Reading, P.C., Miller, J.L., and Anders, E.M. (2000). Involvement of the mannose receptor in infection of macrophages by influenza virus. J Virol. 74, 5190-5197.

Rescigno, M., Martino, M., Sutherland, C.L., Gold, M.R., and Ricciardi-Castagnoli, P. (1998). Dendritic cell survival and maturation are regulated by different signaling pathways. J. Exp. Med. *188*, 2175-2180.

Rescigno, M., Piguet, V., Valzasina, B., Lens, S., Zubler, R., French, L., Kindler, V., Tschopp, J., and Ricciardi-Castagnoli, P. (2000). Fas engagement induces the maturation of dendritic cells (DCs), the release of interleukin (IL)-1beta, and the production of interferon gamma in the absence of IL-12 during DC-T cell cognate interaction: a new role for Fas ligand in inflammatory responses. J. Exp. Med. *192*, 1661-1668.

Riegler, S., Hebart, H., Einsele, H., Brossart, P., Jahn, G., and Sinzger, C. (2000). Monocyte-derived dendritic cells are permissive to the complete replicative cycle of human cytomegalovirus. J Gen. Virol. 81, 393-399.

Rissoan, M.C., Soumelis, V., Kadowaki, N., Grouard, G., Briere, F., de Waal, M.R., and Liu, Y.J. (1999). Reciprocal control of T helper cell and dendritic cell differentiation. Science 283, 1183-1186. Robbiani, D.F., Finch, R.A., Jager, D., Muller, W.A., Sartorelli, A.C., and Randolph, G.J. (2000). The Leukotriene C4 Transporter MRP1 Regulates CCL19 (MIP-3[beta], ELC)-Dependent Mobilization of Dendritic Cells to Lymph Nodes. Cell *103*, 757-768.

Rodger, G., Boname, J., Bell, S., and Minson, T. (2001). Assembly and organization of glycoproteins B, C, D, and H in herpes simplex virus type 1 particles lacking individual glycoproteins: No evidence for the formation of a complex of these molecules. J Virol *75*, 710-716.

Rogge,L., Barberis-Maino,L., Biffi,M., Passini,N., Presky,D.H., Gubler,U., and Sinigaglia,F. (1997). Selective Expression of an Interleukin-12 Receptor Component by Human T Helper 1 Cells. J. Exp. Med. *185*, 825-832.

Romerio, F., Riva, A., and Zella, D. (2000). Interferon-alpha2b reduces phosphorylation and activity of MEK and ERK through a Ras/Raf-independent mechanism. Br. J. Cancer 83, 532-538.

Rong,B.L., Libermann,T.A., Kogawa,K., Ghosh,S., Cao,L.X., Pavan-Langston,D., and Dunkel,E.C. (1992). HSV-1-inducible proteins bind to NF-kappa B-like sites in the HSV-1 genome. Virology *189*, 750-756.

Rong,Q., Alexander,T.S., Koski,G.K., and Rosenthal,K.S. (2003). Multiple mechanisms for HSV-1 induction of interferon alpha production by peripheral blood mononuclear cells. Arch. Virol. *148*, 329-344.

Rotta,G., Edwards,E.W., Sangaletti,S., Bennett,C., Ronzoni,S., Colombo,M.P., Steinman,R.M., Randolph,G.J., and Rescigno,M. (2003). Lipopolysaccharide or Whole Bacteria Block the Conversion of Inflammatory Monocytes into Dendritic Cells In Vivo. J. Exp. Med. *198*, 1253-1263.

Rux,A.H., Willis,S.H., Nicola,A.V., Hou,W., Peng,C., Lou,H., Cohen,G.H., and Eisenberg,R.J. (1998). Functional region IV of glycoprotein D from herpes simplex virus modulates glycoprotein binding to the herpesvirus entry mediator. J Virol *72*, 7091-7098.

Saccani,S., Pantano,S., and Natoli,G. (2001). Two waves of nuclear factor kappaB recruitment to target promoters. J Exp. Med. *193*, 1351-1359.

Saccani,S., Pantano,S., and Natoli,G. (2002). p38-Dependent marking of inflammatory genes for increased NF-kappa B recruitment. Nat. Immunol. *3*, 69-75.

Sakaguchi,S., Negishi,H., Asagiri,M., Nakajima,C., Mizutani,T., Takaoka,A., Honda,K., and Taniguchi,T. (2003). Essential role of IRF-3 in lipopolysaccharide-induced interferon-beta gene expression and endotoxin shock. Biochem. Biophys. Res. Commun. *306*, 860-866.

Salio, M., Cella, M., Suter, M., and Lanzavecchia, A. (1999). Inhibition of dendritic cell maturation by herpes simplex virus. Eur. J Immunol 29, 3245-3253.

Salio, M., Palmowski, M.J., Atzberger, A., Hermans, I.F., and Cerundolo, V. (2004). CpG-matured Murine Plasmacytoid Dendritic Cells Are Capable of In Vivo Priming of Functional CD8 T Cell Responses to Endogenous but Not Exogenous Antigens. J. Exp. Med. *199*, 567-579.

Sallusto, F., Cella, M., 'Danieli, C., and Lanzavecchia, A. (1995). Dendritic cells use macropinocytosis and the mannose receptor to concentrate macromolecules in the major histocompatibility complex class II compartment: downregulation by cytokines and bacterial products. J. Exp. Med. *182*, 389-400.

Sallusto, F. and Lanzavecchia, A. (1994). Efficient presentation of soluble antigen by cultured human dendritic cells is maintained by granulocyte/macrophage colony-stimulating factor plus interleukin 4 and downregulated by tumor necrosis factor alpha. J. Exp. Med. *179*, 1109-1118.

Sallusto, F., Lenig, D., Forster, R., Lipp, M., and Lanzavecchia, A. (1999). Two subsets of memory T lymphocytes with distinct homing potentials and effector functions. Nature *401*, 708-712.

Salomon, B. and Bluestone, J.A. (1998). Cutting Edge: LFA-1 Interaction with ICAM-1 and ICAM-2 Regulates Th2 Cytokine Production. J Immunol *161*, 5138-5142.

Samady,L., Costigliola,E., MacCormac,L., McGrath,Y., Cleverley,S., Lilley,C.E., Smith,J., Latchman,D.S., Chain,B., and Coffin,R.S. (2003). Deletion of the virion host shutoff protein (vhs) from herpes simplex virus (HSV) relieves the viral block to dendritic cell activation: potential of vhs- HSV vectors for dendritic cell-mediated immunotherapy. J. Virol. *77*, 3768-3776.

Sambucetti,L.C., Cherrington,J.M., Wilkinson,G.W., and Mocarski,E.S. (1989). NF-kappa B activation of the cytomegalovirus enhancer is mediated by a viral transactivator and by T cell stimulation. EMBO J 8, 4251-4258.

Sanchez-Sanchez,N., Riol-Blanco,L., de la Rosa,G., Puig-Kroger,A., Garcia-Bordas,J., Martin,D., Longo,N., Cuadrado,A., Cabanas,C., Corbi,A.L., Sanchez-Mateos,P., and Rodriguez-Fernandez,J.L. (2004). Chemokine receptor CCR7 induces intracellular signaling that inhibits apoptosis of mature dendritic cells. Blood *104*, 619-625.

Sandberg,K., Gobl,A.E., Funa,K., and Alm,G.V. (1989). Characterization of the blood mononuclear leucocytes producing alpha interferon after stimulation with herpes simplex virus in vitro, by means of combined immunohistochemical staining and in situ RNA-RNA hybridization. Scand. J Immunol 29, 651-658.

Sanfilippo,C.M., Chirimuuta,F.N., and Blaho,J.A. (2004). Herpes simplex virus type 1 immediate-early gene expression is required for the induction of apoptosis in human epithelial HEp-2 cells. J. Virol. 78, 224-239.

Santini,S.M., Lapenta,C., Logozzi,M., Parlato,S., Spada,M., Di Pucchio,T., and Belardelli,F. (2000). Type I interferon as a powerful adjuvant for monocyte-derived dendritic cell development and activity in vitro and in Hu-PBL-SCID mice. J Exp. Med. *191*, 1777-1788.

Sapp,M., Engelmayér,J., Larsson,M., Granelli-Piperno,A., Steinman,R., and Bhardwaj,N. (1999). Dendritic cells generated from blood monocytes of HIV-1 patients are not infected and act as competent antigen presenting cells eliciting potent T-cell responses. Immunol. Lett. *66*, 121-128.

Sato,K., Kawasaki,H., Nagayama,H., Enomoto,M., Morimoto,C., Tadokoro,K., Juji,T., and Takahashi,T.A. (2001). Signaling events following chemokine receptor ligation in human dendritic cells at different developmental stages. Int. Immunol. *13*, 167-179.

Sauter, B., Albert, M.L., Francisco, L., Larsson, M., Somersan, S., and Bhardwaj, N. (2000). Consequences of cell death: exposure to necrotic tumor cells, but not primary tissue cells or apoptotic cells, induces the maturation of immunostimulatory dendritic cells. J Exp. Med. *191*, 423-434.

Sawtell,N.M. (2003). Quantitative Analysis of Herpes Simplex Virus Reactivation In Vivo Demonstrates that Reactivation in the Nervous System Is Not Inhibited at Early Times Postinoculation. J. Virol. 77, 4127-4138.

Scandella, E., Men, Y., Gillessen, S., Forster, R., and Groettrup, M. (2002). Prostaglandin E2 is a key factor for CCR7 surface expression and migration of monocyte-derived dendritic cells. Blood *100*, 1354-1361.

Schacker, T., Ryncarz, A.J., Goddard, J., Diem, K., Shaughnessy, M., and Corey, L. (1998a). Frequent recovery of HIV-1 from genital herpes simplex virus lesions in HIV-1-infected men. JAMA. 280, 61-66.

Schacker, T., Zeh, J., Hu, H.L., Hill, E., and Corey, L. (1998b). Frequency of symptomatic and asymptomatic herpes simplex virus type 2 reactivations among human immunodeficiency virus-infected men. J. Infect. Dis. *178*, 1616-1622.

Schafer,S.L., Vlach,J., and Pitha,P.M. (1996). Cooperation between herpes simplex virus type 1encoded ICP0 and Tat to support transcription of human immunodeficiency virus type 1 long terminal repeat in vivo can occur in the absence of the TAR binding site. J Virol 70, 6937-6946.

Schelhaas, M., Jansen, M., Haase, I., and Knebel-Morsdorf, D. (2003). Herpes simplex virus type 1 exhibits a tropism for basal entry in polarized epithelial cells. J. Gen. Virol. *84*, 2473-2484.

Schiavoni,G., Mattei,F., Sestili,P., Borghi,P., Venditti,M., Morse,H.C., III, Belardelli,F., and Gabriele,L. (2002). ICSBP Is Essential for the Development of Mouse Type I Interferon-producing Cells and for the Generation and Activation of CD8{alpha}+ Dendritic Cells. J. Exp. Med. *196*, 1415-1425.

Schnorr, J.J., Xanthakos, S., Keikavoussi, P., Kampgen, E., ter, M., V, and Schneider-Schaulies, S. (1997). Induction of maturation of human blood dendritic cell precursors by measles virus is associated with immunosuppression. Proc. Natl. Acad. Sci. U. S. A *94*, 5326-5331.

Schnurr,M., Toy,T., [']Shin,A., Hartmann,G., Rothenfusser,S., Soellner,J., Davis,I.D., Cebon,J., and Maraskovsky,E. (2004). Role of adenosine receptors in regulating chemotaxis and cytokine production of plasmacytoid dendritic cells. Blood *103*, 1391-1397.

Schon, E., Harandi, A.M., Nordstrom, I., Holmgren, J., and Eriksson, K. (2001). Dendritic cell vaccination protects mice against lethality caused by genital herpes simplex virus type 2 infection. J Reprod. Immunol 50, 87-104.

Schultz-Cherry,S., Dybdahl-Sissoko,N., Neumann,G., Kawaoka,Y., and Hinshaw,V.S. (2001). Influenza virus ns1 protein induces apoptosis in cultured cells. J Virol. *75*, 7875-7881.

Schulz,O., Edwards,A.D., Schito,M., Aliberti,J., Manickasingham,S., Sher,A., and Reis e Sousa (2000). CD40 triggering of heterodimeric IL-12 p70 production by dendritic cells in vivo requires a microbial priming signal. Immunity. *13*, 453-462.

Schulz,O. and Reis e Sousa,C. (2002). Cross-presentation of cell-associated antigens by CD8alpha+ dendritic cells is attributable to their ability to internalize dead cells. Immunology *107*, 183-189.

Schwandner, R., Dziarski, R., Wesche, H., Rothe, M., and Kirschning, C.J. (1999). Peptidoglycan- and lipoteichoic acid-induced cell activation is mediated by toll-like receptor 2. J Biol. Chem. 274, 17406-17409.

Scott,E.S., Malcomber,S., and O'Hare,P. (2001). Nuclear translocation and activation of the transcription factor NFAT is blocked by herpes simplex virus infection. J Virol *75*, 9955-9965.

Seet, B.T., Johnston, J.B., Brunetti, C.R., Barrett, J.W., Everett, H., Cameron, C., Sypula, J., Nazarian, S.H., Lucas, A., and McFadden, G. (2003). Poxviruses and immune evasion. Annu. Rev. Immunol 21, 377-423.

Sells,M.A., Knaus,U.G., Bagrodia,S., Ambrose,D.M., Bokoch,G.M., and Chernoff,J. (1997). Human p21-activated kinase (Pak1) regulates actin organization in mammalian cells. Curr. Biol. 7, 202-210.

Servet-Delprat, C., Vidalain, P.O., Azocar, O., Le Deist, F., Fischer, A., and Rabourdin-Combe, C. (2000a). Consequences of Fas-mediated human dendritic cell apoptosis induced by measles virus. J Virol. 74, 4387-4393.

Servet-Delprat, C., Vidalain, P.O., Bausinger, H., Manie, S., Le Deist, F., Azocar, O., Hanau, D., Fischer, A., and Rabourdin-Combe, C. (2000b). Measles virus induces abnormal differentiation of CD40 ligand-activated human dendritic cells. J Immunol *164*, 1753-1760.

Shi,Y., Evans,J.E., and Rock,K.L. (2003). Molecular identification of a danger signal that alerts the immune system to dying cells. Nature. 425, 516-521.

Shieh, M.T., WuDunn, D., Montgomery, R.I., Esko, J.D., and Spear, P.G. (1992). Cell surface receptors for herpes simplex virus are heparan sulfate proteoglycans. J Cell Biol. *116*, 1273-1281.

Shigematsu, H., Reizis, B., Iwasaki, H., Mizuno, S., Hu, D., Traver, D., Leder, P., Sakaguchi, N., and Akashi, K. (2004). Plasmacytoid dendritic cells activate lymphoid-specific genetic programs irrespective of their cellular origin. Immunity 21, 43-53.

Shimoyama, M. (1991). Diagnostic criteria and classification of clinical subtypes of adult T-cell leukaemia-lymphoma. A report from the Lymphoma Study Group (1984-87). Br. J Haematol. 79, 428-437.

Shortman,K. and Liu,Y.J. (2002). Mouse and human dendritic cell subtypes. Nat. Rev. Immunol. 2, 151-161.

Shukla,D., Liu,J., Blaiklock,P., Shworak,N.W., Bai,X., Esko,J.D., Cohen,G.H., Eisenberg,R.J., Rosenberg,R.D., and Spear,P.G. (1999). A novel role for 3-O-sulfated heparan sulfate in herpes simplex virus 1 entry. Cell *99*, 13-22.

Shutt,D.C., Daniels,K.J., Carolan,E.J., Hill,A.C., and Soll,D.R. (2000). Changes in the motility, morphology, and F-actin architecture of human dendritic cells in an in vitro model of dendritic cell development. Cell. Motil. Cytoskeleton. *46*, 200-221.

Siegal,F.P., Kadowaki,N., Shodell,M., Fitzgerald-Bocarsly,P.A., Shah,K., Ho,S., Antonenko,S., and Liu,Y.J. (1999). The nature of the principal type 1 interferon-producing cells in human blood. Science 284, 1835-1837.

Siegal,F.P., Lopez,C., Hammer,G.S., Brown,A.E., Kornfeld,S.J., Gold,J., Hassett,J., Hirschman,S.Z., Cunningham-Rundles,C., Adelsberg,B.R., and . (1981). Severe acquired immunodeficiency in male homosexuals, manifested by chronic perianal ulcerative herpes simplex lesions. N. Engl. J Med. *305*, 1439-1444.

Simmen,K.A., Singh,J., Luukkonen,B.G.M., Lopper,M., Bittner,A., Miller,N.E., Jackson,M.R., Compton,T., and Fruh,K. (2001). Global modulation of cellular transcription by human cytomegalovirus is initiated by viral glycoprotein B. PNAS *98*, 7140-7145.

Sin,J.I., Kim,J.J., Arnold,R.L., Shroff,K.E., McCallus,D., Pachuk,C., McElhiney,S.P., Wolf,M.W., Pompa-de Bruin,S.J., Higgins,T.J., Ciccarelli,R.B., and Weiner,D.B. (1999). IL-12 gene as a DNA vaccine adjuvant in a herpes mouse model: IL-12 enhances Th1-type CD4+ T cell-mediated protective immunity against herpes simplex virus-2 challenge. J. Immunol. *162*, 2912-2921.

Sing,A., Rost,D., Tvardovskaia,N., Roggenkamp,A., Wiedemann,A., Kirschning,C.J., Aepfelbacher,M., and Heesemann,J. (2002). Yersinia V-antigen exploits toll-like receptor 2 and CD14 for interleukin 10-mediated immunosuppression. J Exp. Med *196*, 1017-1024.

Sloan,D.D., Zahariadis,G., Posavad,C.M., Pate,N.T., Kussick,S.J., and Jerome,K.R. (2003). CTL are inactivated by herpes simplex virus-infected cells expressing a viral protein kinase. J Immunol 171, 6733-6741.

Smith,A., Santoro,F., Di Lullo,G., Dagna,L., Verani,A., and Lusso,P. (2003a). Selective suppression of IL-12 production by human herpesvirus 6. Blood *102*, 2877-2884.

Smith,C.M., Belz,G.T., Wilson,N.S., Villadangos,J.A., Shortman,K., Carbone,F.R., and Heath,W.R. (2003b). Cutting Edge: Conventional CD8{alpha}+ Dendritic Cells Are Preferentially Involved in CTL Priming After Footpad Infection with Herpes Simplex Virus-1. J Immunol *170*, 4437-4440.

Snijders, A., Kalinski, P., Hilkens, C.M., and Kapsenberg, M.L. (1998). High-level IL-12 production by human dendritic cells requires two signals. Int. Immunol. 10, 1593-1598.

Sodeik, B., Ebersold, M.W., and Helenius, A. (1997). Microtubule-mediated transport of incoming herpes simplex virus 1 capsids to the nucleus. J. Cell Biol. *136*, 1007-1021.

Sol-Foulon, N., Moris, A., Nobile, C., Boccaccio, C., Engering, A., Abastado, J.P., Heard, J.M., Van Kooyk, Y., and Schwartz, O. (2002). HIV-1 Nef-induced upregulation of DC-SIGN in dendritic cells promotes lymphocyte clustering and viral spread. Immunity. *16*, 145-155.

Song,B.H., Lee,G.C., Moon,M.S., Cho,Y.H., and Lee,C.H. (2001). Human cytomegalovirus binding to heparan sulfate proteoglycans on the cell surface and/or entry stimulates the expression of human leukocyte antigen class I. J. Gen. Virol. *82*, 2405-2413.

Sozzani,S., Luini,W., Borsatti,A., Polentarutti,N., Zhou,D., Piemonti,L., D'amico,G., Power,C.A., Wells,T.N., Gobbi,M., Allavena,P., and Mantovani,A. (1997). Receptor expression and responsiveness of human dendritic cells to a defined set of CC and CXC chemokines. J Immunol *159*, 1993-2000.

Spear, P.G. (2002). Viral interactions with receptors in cell junctions and effects on junctional stability. Dev. Cell *3*, 462-464.

Spear, P.G., Eisenberg, R.J., and Cohen, G.H. (2000). Three classes of cell surface receptors for alphaherpesvirus entry. Virology 275, 1-8.

Sprecher, E. and Becker, Y. (1986). Skin Langerhans cells play an essential role in the defense against HSV-1 infection. Arch. Virol. *91*, 341-349.

Sprecher, E. and Becker, Y. (1989). Langerhans cell density and activity in mouse skin and lymph nodes affect herpes simplex type 1 (HSV-1) pathogenicity. Arch. Virol. 107, 191-205.

Spruance,S.L., Evans,T.G., McKeough,M.B., Thai,L., Araneo,B.A., Daynes,R.A., Mishkin,E.M., and Abramovitz,A.S. (1995). Th1/Th2-like immunity and resistance to herpes simplex labialis. Antiviral Research 28, 39-55.

Srivastava,R.K., Sasaki,C.Y., Hardwick,J.M., and Longo,D.L. (1999). Bcl-2-mediated drug resistance: inhibition of apoptosis by blocking nuclear factor of activated T lymphocytes (NFAT)-induced Fas ligand transcription. J Exp. Med. *190*, 253-265.

Stamminger, T., Fickenscher, H., and Fleckenstein, B. (1990). Cell type-specific induction of the major immediate early enhancer of human cytomegalovirus by cyclic AMP. J Gen Virol 71 (Pt 1), 105-113.

Stanberry,L.R., Spruance,S.L., Cunningham,A.L., Bernstein,D.I., Mindel,A., Sacks,S., Tyring,S., Aoki,F.Y., Slaoui,M., Denis,M., Vandepapeliere,P., and Dubin,G. (2002). Glycoprotein-D-adjuvant vaccine to prevent genital herpes. N. Engl. J. Med. *347*, 1652-1661.

Steinman, R.M. and Cohn, Z.A. (1973). Identification of a novel cell type in peripheral lymphoid organs of mice. I. Morphology, quantitation, tissue distribution. J. Exp. Med. *137*, 1142-1162.

Steinman, R.M., Turley, S., Mellman, I., and Inaba, K. (2000). The Induction of Tolerance by Dendritic Cells That Have Captured Apoptotic Cells. J. Exp. Med. 191, 411-416.

Stevens, J.G., Wagner, E.K., Devi-Rao, G.B., Cook, M.L., and Feldman, L.T. (1987). RNA complementary to a herpesvirus alpha gene mRNA is prominent in latently infected neurons. Science. 235, 1056-1059.

Stingley,S.W., Ramirez,J.J., Aguilar,S.A., Simmen,K., Sandri-Goldin,R.M., Ghazal,P., and Wagner,E.K. (2000). Global analysis of herpes simplex virus type 1 transcription using an oligonucleotide-based DNA microarray. J Virol. 74, 9916-9927.

Stock, A.T., Mueller, S.N., van Lint, A.L., Heath, W.R., and Carbone, F.R. (2004). Cutting edge: prolonged antigen presentation after herpes simplex virus-1 skin infection. J Immunol *173*, 2241-2244.

Stoll,S., Delon,J., Brotz,T.M., and Germain,R.N. (2002). Dynamic imaging of T cell-dendritic cell interactions in lymph nodes. Science. *296*, 1873-1876.

Struyf,F., Posavad,C.M., Keyaerts,E., Van Ranst,M., Corey,L., and Spear,P.G. (2002). Search for polymorphisms in the genes for herpesvirus entry mediator, nectin-1, and nectin-2 in immune seronegative individuals. J Infect. Dis. *185*, 36-44.

Suvas,S., Kumaraguru,U., Pack,C.D., Lee,S., and Rouse,B.T. (2003). CD4+CD25+ T cells regulate virus-specific primary and memory CD8+ T cell responses. J. Exp. Med. *198*, 889-901.

Swetman, C.A., Leverrier, Y., Garg, R., Gan, C.H., Ridley, A.J., Katz, D.R., and Chain, B.M. (2002). Extension, retraction and contraction in the formation of a dendritic cell dendrite: distinct roles for Rho GTPases. Eur. J Immunol *32*, 2074-2083.

Tabeta,K., Georgel,P., Janssen,E., Du,X., Hoebe,K., Crozat,K., Mudd,S., Shamel,L., Sovath,S., Goode,J., Alexopoulou,L., Flavell,R.A., and Beutler,B. (2004). Toll-like receptors 9 and 3 as essential components of innate immune defense against mouse cytomegalovirus infection. Proc. Natl. Acad. Sci. U. S. A *101*, 3516-3521.

Tabi,Z., Moutaftsi,M., and Borysiewicz,L.K. (2001). Human cytomegalovirus pp65- and immediate early 1 antigen-specific HLA class I-restricted cytotoxic T cell responses induced by cross-presentation of viral antigens. J Immunol *166*, 5695-5703.

Tachibana, N., Okayama, A., Ishizaki, J., Yokota, T., Shishime, E., Murai, K., Shioiri, S., Tsuda, K., Essex, M., and Mueller, N. (1988). Suppression of tuberculin skin reaction in healthy HTLV-I carriers from Japan. Int. J Cancer *42*, 829-831.

Tajima,K. (1990). The 4th nation-wide study of adult T-cell leukemia/lymphoma (ATL) in Japan: estimates of risk of ATL and its geographical and clinical features. The T- and B-cell Malignancy Study Group. Int. J Cancer 45, 237-243.

Takai,Y. and Nakanishi,H. (2003). Nectin and afadin: novel organizers of intercellular junctions. J Cell Sci 116, 17-27.

Takeuchi,O., Hoshino,K., and Akira,S. (2000). Cutting Edge: TLR2-Deficient and MyD88-Deficient Mice Are Highly Susceptible to Staphylococcus aureus Infection. J Immunol *165*, 5392-5396.

Takeuchi,O., Kawai,T., Muhlradt,P.F., Morr,M., Radolf,J.D., Zychlinsky,A., Takeda,K., and Akira,S. (2001). Discrimination of bacterial lipoproteins by Toll-like receptor 6. Int. Immunol. *13*, 933-940.

Takeuchi,O., Sato,S., Horiuchi,T., Hoshino,K., Takeda,K., Dong,Z., Modlin,R.L., and Akira,S. (2002). Cutting Edge: Role of Toll-Like Receptor 1 in Mediating Immune Response to Microbial Lipoproteins. J Immunol *169*, 10-14.

Tal,G., Mandelberg,A., Dalal,I., Cesar,K., Somekh,E., Tal,A., Oron,A., Itskovich,S., Ballin,A., Houri,S., Beigelman,A., Lider,O., Rechavi,G., and Amariglio,N. (2004). Association between common Toll-like receptor 4 mutations and severe respiratory syncytial virus disease. J Infect. Dis. *189*, 2057-2063.

Tanabe, M., Kurita-Taniguchi, M., Takeuchi, K., Takeda, M., Ayata, M., Ogura, H., Matsumoto, M., and Seya, T. (2003). Mechanism of up-regulation of human Toll-like receptor 3 secondary to infection of measles virus-attenuated strains. Biochemical and Biophysical Research Communications *311*, 39-48.

Taniguchi, T. and Takaoka, A. (2002). The interferon-alpha/beta system in antiviral responses: a multimodal machinery of gene regulation by the IRF family of transcription factors. Curr. Opin. Immunol 14, 111-116.

Tassaneetrithep,B., Burgess,T.H., Granelli-Piperno,A., Trumpfheller,C., Finke,J., Sun,W., Eller,M.A., Pattanapanyasat,K., Sarasombath,S., Birx,D.L., Steinman,R.M., Schlesinger,S., and Marovich,M.A. (2003). DC-SIGN (CD209) mediates dengue virus infection of human dendritic cells. J. Exp. Med. *197*, 823-829.

Taub,D.D., Conlon,K., Lloyd,A.R., Oppenheim,J.J., and Kelvin,D.J. (1993). Preferential migration of activated CD4+ and CD8+ T cells in response to MIP-1 alpha and MIP-1 beta. Science 260, 355-358.

Termeer, C., Benedix, F., Sleeman, J., Fieber, C., Voith, U., Ahrens, T., Miyake, K., Freudenberg, M., Galanos, C., and Simon, J.C. (2002). Oligosaccharides of Hyaluronan Activate Dendritic Cells via Tolllike Receptor 4. J. Exp. Med. *195*, 99-111.

Thebeau, L.G. and Morrison, L.A. (2002). B7 costimulation plays an important role in protection from herpes simplex virus type 2-mediated pathology. J Virol 76, 2563-2566.

Theil,D., Derfuss,T., Paripovic,I., Herberger,S., Meinl,E., Schueler,O., Strupp,M., Arbusow,V., and Brandt,T. (2003). Latent herpesvirus infection in human trigeminal ganglia causes chronic immune response. Am. J. Pathol. *163*, 2179-2184.

Thierfelder, W.E., van Deursen, J.M., Yamamoto, K., Tripp, R.A., Sarawar, S.R., Carson, R.T., Sangster, M.Y., Vignali, D.A., Doherty, P.C., Grosveld, G.C., and Ihle, J.N. (1996). Requirement for Stat4 in interleukin-12-mediated responses of natural killer and T cells. Nature. *382*, 171-174.

Tigges,M.A., Leng,S., Johnson,D.C., and Burke,R.L. (1996). Human herpes simplex virus (HSV)specific CD8+ CTL clones recognize HSV-2-infected fibroblasts after treatment with IFN-gamma or when virion host shutoff functions are disabled. J Immunol *156*, 3901-3910.

Tikhomirov,O. and Carpenter,G. (2004). Ligand-induced, p38-dependent apoptosis in cells expressing high levels of epidermal growth factor receptor and ErbB-2. J Biol. Chem. 279, 12988-12996.

Toka,F.N., Gierynska,M., and Rouse,B.T. (2003). Codelivery of CCR7 ligands as molecular adjuvants enhances the protective immune response against herpes simplex virus type 1. J. Virol. 77, 12742-12752.

Tomazin,R., Boname,J., Hegde,N.R., Lewinsohn,D.M., Altschuler,Y., Jones,T.R., Cresswell,P., Nelson,J.A., Riddell,S.R., and Johnson,D.C. (1999). Cytomegalovirus US2 destroys two components of the MHC class II pathway, preventing recognition by CD4+ T cells. Nat. Med. *5*, 1039-1043.

Tomazin, R., Hill, A.B., Jugovic, P., York, I., van Endert, P., Ploegh, H.L., Andrews, D.W., and Johnson, D.C. (1996). Stable binding of the herpes simplex virus ICP47 protein to the peptide binding site of TAP. EMBO J. 15, 3256-3266.

Tomazin,R., van Schoot,N.E., Goldsmith,K., Jugovic,P., Sempe,P., Fruh,K., and Johnson,D.C. (1998). Herpes simplex virus type 2 ICP47 inhibits human TAP but not mouse TAP. J. Virol. 72, 2560-2563.

Tong,L., Pav,S., White,D.M., Rogers,S., Crane,K.M., Cywin,C.L., Brown,M.L., and Pargellis,C.A. (1997). A highly specific inhibitor of human p38 MAP kinase binds in the ATP pocket. Nat. Struct. Biol. 4, 311-316.

Torcia, M., De Chiara, G., Nencioni, L., Ammendola, S., Labardi, D., Lucibello, M., Rosini, P., Marlier, L.N., Bonini, P., Dello, S.P., Palamara, A.T., Zambrano, N., Russo, T., Garaci, E., and Cozzolino, F. (2001). Nerve growth factor inhibits apoptosis in memory B lymphocytes via inactivation of p38 MAPK, prevention of Bcl-2 phosphorylation, and cytochrome c release. J Biol. Chem. *276*, 39027-39036.

Tornaletti,S. and Pfeifer,G.P. (1996). UV damage and repair mechanisms in mammalian cells. Bioessays. 18, 221-228.

Tortorella,D., Gewurz,B.E., Furman,M.H., Schust,D.J., and Ploegh,H.L. (2000). Viral subversion of the immune system. Annu. Rev. Immunol. *18*, 861-926.

Traver, D., Akashi, K., Manz, M., Merad, M., Miyamoto, T., Engleman, E.G., and Weissman, I.L. (2000). Development of CD8{alpha}-Positive Dendritic Cells from a Common Myeloid Progenitor. Science 290, 2152-2154.

Trevejo,J.M., Marino,M.W., Philpott,N., Josien,R., Richards,E.C., Elkon,K.B., and Falck-Pedersen,E. (2001). TNF-alpha -dependent maturation of local dendritic cells is critical for activating the adaptive immune response to virus infection. Proc. Natl. Acad. Sci. U. S. A *98*, 12162-12167.

Trinchieri,G. (2003). Interleukin-12 and the regulation of innate resistance and adaptive immunity. Nat. Rev. Immunol. *3*, 133-146.

Tropea, F., Troiano, L., Monti, D., Lovato, E., Malorni, W., Rainaldi, G., Mattana, P., Viscomi, G., Ingletti, M.C., Portolani, M., and (1995). Sendai virus and herpes virus type 1 induce apoptosis in human peripheral blood mononuclear cells. Exp. Cell Res. *218*, 63-70.

Trottein, F., Pavelka, N., Vizzardelli, C., Angeli, V., Zouain, C.S., Pelizzola, M., Capozzoli, M., Urbano, M., Capron, M., Belardelli, F., Granucci, F., and Ricciardi-Castagnoli, P. (2004). A type I IFN-dependent pathway induced by Schistosoma mansoni eggs in mouse myeloid dendritic cells generates an inflammatory signature. J Immunol *172*, 3011-3017.

Tsujimura,H., Tamura,T., and Ozato*,K. (2003). Cutting Edge: IFN Consensus Sequence Binding Protein/IFN Regulatory Factor 8 Drives the Development of Type I IFN-Producing Plasmacytoid Dendritic Cells. J Immunol 170, 1131-1135.

Tuettenberg, A., Jonuleit, H., Tuting, T., Bruck, J., Biermann, V., Kochanek, S., Knop, J., and Enk, A.H. (2004). Early adenoviral gene expression mediates immunosuppression by transduced dendritic cell (DC): implications for immunotherapy using genetically modified DC. J Immunol *172*, 1524-1530.

Turner, E.E., Fedtsova, N., and Rosenfeld, M.G. (1996). POU-domain factor expression in the trigeminal ganglion and implications for herpes virus regulation. Neuroreport. 7, 2829-2832.

Uehori, J., Matsumoto, M., Tsuji, S., Akazawa, T., Takeuchi, O., Akira, S., Kawata, T., Azuma, I., Toyoshima, K., and Seya, T. (2003). Simultaneous Blocking of Human Toll-Like Receptors 2 and 4 Suppresses Myeloid Dendritic Cell Activation Induced by Mycobacterium bovis Bacillus Calmette-Guerin Peptidoglycan. Infect. Immun. 71, 4238-4249.

Vabulas, R.M., Braedel, S., Hilf, N., Singh-Jasuja, H., Herter, S., Ahmad-Nejad, P., Kirschning, C.J., da Costa, C., Rammensee, H.G., Wagner, H., and Schild, H. (2002). The Endoplasmic Reticulum-resident Heat Shock Protein Gp96 Activates Dendritic Cells via the Toll-like Receptor 2/4 Pathway. J. Biol. Chem. 277, 20847-20853.

van Lint, A., Ayers, M., Brooks, A.G., Coles, R.M., Heath, W.R., and Carbone, F.R. (2004). Herpes simplex virus-specific CD8+ T cells can clear established lytic infections from skin and nerves and can partially limit the early spread of virus after cutaneous inoculation. J. Immunol. *172*, 392-397.

van Seventer, J.M., Nagai, T., and van Seventer, G.A. (2002). Interferon-beta differentially regulates expression of the IL-12 family members p35, p40, p19 and EBI3 in activated human dendritic cells. J Neuroimmunol. *133*, 60-71.

Van Vliet,K.E., Graaf-Miltenburg,L.A., Verhoef,J., and Van Strijp,J.A. (1992). Direct evidence for antibody bipolar bridging on herpes simplex virus-infected cells. Immunology 77, 109-115.

Verhasselt, V., Vanden Berghe, W., Vanderheyde, N., Willems, F., Haegeman, G., and Goldman, M. (1999). N-Acetyl-L-Cysteine Inhibits Primary Human T Cell Responses at the Dendritic Cell Level: Association with NF-{kappa}B Inhibition. J Immunol *162*, 2569-2574.

Verschoor, A., Brockman, M.A., Gadjeva, M., Knipe, D.M., and Carroll, M.C. (2003). Myeloid C3 determines induction of humoral responses to peripheral herpes simplex virus infection. J. Immunol. *171*, 5363-5371.

Verschoor, A., Brockman, M.A., Knipe, D.M., and Carroll, M.C. (2001). Cutting edge: myeloid complement c3 enhances the humoral response to peripheral viral infection. J Immunol *167*, 2446-2451.

Vidalain, P.O., Azocar, O., Lamouille, B., Astier, A., Rabourdin-Combe, C., and Servet-Delprat, C. (2000a). Measles virus induces functional TRAIL production by human dendritic cells. J Virol. 74, 556-559.

Vidalain, P.O., Azocar, O., Servet-Delprat, C., Rabourdin-Combe, C., Gerlier, D., and Manie, S. (2000b). CD40 signaling in human dendritic cells is initiated within membrane rafts. EMBO J. 19, 3304-3313.

Visintin, A., Mazzoni, A., Spitzer, J.H., Wyllie, D.H., Dower, S.K., and Segal, D.M. (2001). Regulation of Toll-like receptors in human monocytes and dendritic cells. J Immunol *166*, 249-255.

Vollstedt,S., Arnold,S., Schwerdel,C., Franchini,M., Alber,G., Di Santo,J.P., Ackermann,M., and Suter,M. (2004). Interplay between alpha/beta and gamma interferons with B, T, and natural killer cells in the defense against herpes simplex virus type 1. J. Virol. *78*, 3846-3850.

Vuckovic,S., Gardiner,D., Field,K., Chapman,G.V., Khalil,D., Gill,D., Marlton,P., Taylor,K., Wright,S., Pinzon-Charry,A., Pyke,C.M., Rodwell,R., Hockey,R.L., Gleeson,M., Tepes,S., True,D., Cotterill,A., and Hart,D.N. (2004). Monitoring dendritic cells in clinical practice using a new whole blood single-platform TruCOUNT assay. J. Immunol. Methods. 284, 73-87.

Wang,C., Al Omar,H.M., Radvanyi,L., Banerjee,A., Bouman,D., Squire,J., and Messner,H.A. (1999a). Clonal heterogeneity of dendritic cells derived from patients with chronic myeloid leukemia and enhancement of their T-cells stimulatory activity by IFN-[alpha]. Experimental Hematology *27*, 1176-1184.

Wang,S.W., Pawlowski,J., Wathen,S.T., Kinney,S.D., Lichenstein,H.S., and Manthey,C.L. (1999b). Cytokine mRNA decay is accelerated by an inhibitor of p38-mitogen-activated protein kinase. Inflamm. Res. 48, 533-538.

Wang,X., Huong,S.M., Chiu,M.L., Raab-Traub,N., and Huang,E.S. (2003). Epidermal growth factor receptor is a cellular receptor for human cytomegalovirus. Nature 424, 456-461.

Ward,P.L., Barker,D.E., and Roizman,B. (1996). A novel herpes simplex virus 1 gene, UL43.5, maps antisense to the UL43 gene and encodes a protein which colocalizes in nuclear structures with capsid proteins. J Virol. 70, 2684-2690.

Warner, M.S., Geraghty, R.J., Martinez, W.M., Montgomery, R.I., Whitbeck, J.C., Xu, R., Eisenberg, R.J., Cohen, G.H., and Spear, P.G. (1998). A cell surface protein with herpesvirus entry activity (HveB) confers susceptibility to infection by mutants of herpes simplex virus type 1, herpes simplex virus type 2, and pseudorabies virus. Virology. *246*, 179-189.

Weigt,H., Muhlradt,P.F., Emmendorffer,A., Krug,N., and Braun,A. (2003). Synthetic mycoplasmaderived lipopeptide MALP-2 induces maturation and function of dendritic cells. Immunobiology. 207, 223-233. Weinmann,A.S., Plevy,S.E., and Smale,S.T. (1999). Rapid and selective remodeling of a positioned nucleosome during the induction of IL-12 p40 transcription. Immunity. *11*, 665-675.

Weir, J.P. (2001). Regulation of herpes simplex virus gene expression. Gene 271, 117-130.

Welstead,G.G., Hsu,E.C., Iorio,C., Bolotin,S., and Richardson,C.D. (2004). Mechanism of CD150 (SLAM) Down Regulation from the Host Cell Surface by Measles Virus Hemagglutinin Protein. J. Virol. 78, 9666-9674.

West,M.A., Prescott,A.R., Eskelinen,E.L., Ridley,A.J., and Watts,C. (2000). Rac is required for constitutive macropinocytosis by dendritic cells but does not control its downregulation. Curr. Biol. 10, 839-848.

Whitbeck,J.C., Peng,C., Lou,H., Xu,R., Willis,S.H., Ponce,D.L., Peng,T., Nicola,A.V., Montgomery,R.I., Warner,M.S., Soulika,A.M., Spruce,L.A., Moore,W.T., Lambris,J.D., Spear,P.G., Cohen,G.H., and Eisenberg,R.J. (1997). Glycoprotein D of herpes simplex virus (HSV) binds directly to HVEM, a member of the tumor necrosis factor receptor superfamily and a mediator of HSV entry. J. Virol. *71*, 6083-6093.

Whitley, R. (2004). Neonatal herpes simplex virus infection. Curr. Opin. Infect. Dis. 17, 243-246.

Wilcox,R.A., Chapoval,A.I., Gorski,K.S., Otsuji,M., Shin,T., Flies,D.B., Tamada,K., Mittler,R.S., Tsuchiya,H., Pardoll,D.M., and Chen,L. (2002). Cutting edge: Expression of functional CD137 receptor by dendritic cells. J Immunol *168*, 4262-4267.

Wilkin,F., Stordeur,P., Goldman,M., Boeynaems,J.M., and Robaye,B. (2002). Extracellular adenine nucleotides modulate cytokine production by human monocyte-derived dendritic cells: dual effect on IL-12 and stimulation of IL-10. Eur. J Immunol *32*, 2409-2417.

Wittels, M. and Spear, P.G. (1991). Penetration of cells by herpes simplex virus does not require a low pH-dependent endocytic pathway. Virus. Res. 18, 271-290.

Wollenberg, A., Wagner, M., Gunther, S., Towarowski, A., Tuma, E., Moderer, M., Rothenfusser, S., Wetzel, S., Endres, S., and Hartmann, G. (2002). Plasmacytoid dendritic cells: a new cutaneous dendritic cell subset with distinct role in inflammatory skin diseases. J. Invest Dermatol. *119*, 1096-1102.

Wood, J.N., Lillycrop, K.A., Dent, C.L., Ninkina, N.N., Beech, M.M., Willoughby, J.J., Winter, J., and Latchman, D.S. (1992). Regulation of expression of the neuronal POU protein Oct-2 by nerve growth factor. J. Biol. Chem. *267*, 17787-17791.

Woodhead, V.E., Binks, M.H., Chain, B.M., and Katz, D.R. (1998). From sentinel to messenger: an extended phenotypic analysis of the monocyte to dendritic cell transition. Immunology *94*, 552-559.

Wooten, R.M., Ma, Y., Yoder, R.A., Brown, J.P., Weis, J.H., Zachary, J.F., Kirschning, C.J., and Weis, J.J. (2002). Toll-Like Receptor 2 Is Required for Innate, But Not Acquired, Host Defense to Borrelia burgdorferi. J Immunol *168*, 348-355.

WuDunn,D. and Spear,P.G. (1989). Initial interaction of herpes simplex virus with cells is binding to heparan sulfate. J Virol. 63, 52-58.

Wysocka, J. and Herr, W. (2003). The herpes simplex virus VP16-induced complex: the makings of a regulatory switch. Trends. Biochem. Sci. 28, 294-304.

Xia,C.Q. and Kao,K.J. (2003). Suppression of interleukin-12 production through endogenously secreted interleukin-10 in activated dendritic cells: involvement of activation of extracellular signal-regulated protein kinase. Scand. J Immunol *58*, 23-32.

Xie, J., Qian, J., Wang, S., Freeman, M.E., III, Epstein, J., and Yi, Q. (2003). Novel and Detrimental Effects of Lipopolysaccharide on In Vitro Generation of Immature Dendritic Cells: Involvement of Mitogen-Activated Protein Kinase p38. J Immunol *171*, 4792-4800.

Yamamoto, M., Sato, S., Hemmi, H., Sanjo, H., Uematsu, S., Kaisho, T., Hoshino, K., Takeuchi, O., Kobayashi, M., Fujita, T., Takeda, K., and Akira, S. (2002). Essential role for TIRAP in activation of the signalling cascade shared by TLR2 and TLR4. Nature. *420*, 324-329.

Yamamoto, M., Sato, S., Hemmi, H., Uematsu, S., Hoshino, K., Kaisho, T., Takeuchi, O., Takeda, K., and Akira, S. (2003). TRAM is specifically involved in the Toll-like receptor 4-mediated MyD88-independent signaling pathway. Nat. Immunol 4, 1144-1150.

Yanagawa,Y. and Onoe,K. (2002). CCL19 induces rapid dendritic extension of murine dendritic cells. Blood. 100, 1948-1956.

Yanagawa,Y. and Onoe,K. (2003). CCR7 ligands induce rapid endocytosis in mature dendritic cells with concomitant up-regulation of Cdc42 and Rac activities. Blood *101*, 4923-4929.

Yang,C.H., Murti,A., Pfeffer,S.R., Basu,L., Kim,J.G., and Pfeffer,L.M. (2000). IFNalpha/beta promotes cell survival by activating NF-kappa B. Proc. Natl. Acad. Sci U. S. A. 97, 13631-13636.

Yang,C.H., Murti,A., Pfeffer,S.R., Kim,J.G., Donner,D.B., and Pfeffer,L.M. (2001). Interferon alpha /beta promotes cell survival by activating nuclear factor kappa B through phosphatidylinositol 3-kinase and Akt. J Biol. Chem. 276, 13756-13761.

Yasumoto, S., Okabe, N., and Mori, R. (1986). Role of epidermal Langerhans cells in resistance to herpes simplex virus infection. Arch. Virol. *90*, 261-271.

Ye,G.J., Vaughan,K.T., Vallee,R.B., and Roizman,B. (2000). The herpes simplex virus 1 U(L)34 protein interacts with a cytoplasmic dynein intermediate chain and targets nuclear membrane. J. Virol. 74, 1355-1363.

Yokota,S., Yokosawa,N., Kubota,T., Suzutani,T., Yoshida,I., Miura,S., Jimbow,K., and Fujii,N. (2001). Herpes simplex virus type 1 suppresses the interferon signaling pathway by inhibiting phosphorylation of STATs and janus kinases during an early infection stage. Virology *286*, 119-124.

Yokota,S., Yokosawa,N., Okabayashi,T., Suzutani,T., Miura,S., Jimbow,K., and Fujii,N. (2004). Induction of suppressor of cytokine signaling-3 by herpes simplex virus type 1 contributes to inhibition of the interferon signaling pathway. J Virol *78*, 6282-6286.

Yoneyama,H., Matsuno,K., Zhang,Y., Nishiwaki,T., Kitabatake,M., Ueha,S., Narumi,S., Morikawa,S., Ezaki,T., Lu,B., Gerard,C., Ishikawa,S., and Matsushima,K. (2004). Evidence for recruitment of plasmacytoid dendritic cell precursors to inflamed lymph nodes through high endothelial venules. Int. Immunol *16*, 915-928.

Yonezawa, A., Morita, R., Takaori-Kondo, A., Kadowaki, N., Kitawaki, T., Hori, T., and Uchiyama, T. (2003). Natural Alpha Interferon-Producing Cells Respond to Human Immunodeficiency Virus Type 1 with Alpha Interferon Production and Maturation into Dendritic Cells. J. Virol. 77, 3777-3784.

Yoon, M. and Spear, P.G. (2002). Disruption of adherens junctions liberates nectin-1 to serve as receptor for herpes simplex virus and pseudorabies virus entry. J. Virol. 76, 7203-7208.

York,I.A., Roop,C., Andrews,D.W., Riddell,S.R., Graham,F.L., and Johnson,D.C. (1994). A cytosolic herpes simplex virus protein inhibits antigen presentation to CD8+ T lymphocytes. Cell *77*, 525-535.

Yoshimura, S., Bondeson, J., Foxwell, B.M., Brennan, F.M., and Feldmann, M. (2001). Effective antigen presentation by dendritic cells is NF-kappaB dependent: coordinate regulation of MHC, co-stimulatory molecules and cytokines. Int. Immunol *13*, 675-683.

Youssef, R., Shaker, O., Sobeih, S., Mashaly, H., and Mostafa, W.Z. (2002). Detection of herpes simplex virus DNA in serum and oral secretions during acute recurrent herpes labialis. J. Dermatol. 29, 404-410.

Yu,Q., Kovacs,C., Yue,F.Y., and Ostrowski,M.A. (2004). The role of the p38 mitogen-activated protein kinase, extracellular signal-regulated kinase, and phosphoinositide-3-OH kinase signal transduction pathways in CD40 ligand-induced dendritic cell activation and expansion of virus-specific CD8+ T cell memory responses. J Immunol *172*, 6047-6056.

Yurochko, A.D. and Huang, E.S. (1999). Human cytomegalovirus binding to human monocytes induces immunoregulatory gene expression. J. Immunol. *162*, 4806-4816.

Yurochko,A.D., Hwang,E.S., Rasmussen,L., Keay,S., Pereira,L., and Huang,E.S. (1997). The human cytomegalovirus UL55 (gB) and UL75 (gH) glycoprotein ligands initiate the rapid activation of Sp1 and NF-kappaB during infection. J Virol 71, 5051-5059.

Yurochko,A.D., Kowalik,T.F., Huong,S.M., and Huang,E.S. (1995). Human cytomegalovirus upregulates NF-kappa B activity by transactivating the NF-kappa B p105/p50 and p65 promoters. J. Virol. 69, 5391-5400.

Zachos,G., Clements,B., and Conner,J. (1999). Herpes simplex virus type 1 infection stimulates p38/c-Jun N-terminal mitogen-activated protein kinase pathways and activates transcription factor AP-1. J. Biol. Chem. 274, 5097-5103.

Zamanian-Daryoush, M., Mogensen, T.H., DiDonato, J.A., and Williams, B.R.G. (2000). NF-kappa B Activation by Double-Stranded-RNA-Activated Protein Kinase (PKR) Is Mediated through NF-kappa B-Inducing Kinase and Ikappa B Kinase. Mol. Cell. Biol. 20, 1278-1290.

Zehn,D., Cohen,C.J., Reiter,Y., and Walden,P. (2004). Extended presentation of specific MHC-peptide complexes by mature dendritic cells compared to other types of antigen-presenting cells. Eur. J Immunol *34*, 1551-1560.

Zeller, J.C., Panoskaltsis-Mortari, A., Murphy, W.J., Ruscetti, F.W., Narula, S., Roncarolo, M.G., and Blazar, B.R. (1999). Induction of CD4+ T Cell Alloantigen-Specific Hyporesponsiveness by IL-10 and TGF-{beta}. J Immunol *163*, 3684-3691.

Zhang, D., Zhang, G., Hayden, M.S., Greenblatt, M.B., Bussey, C., Flavell, R.A., and Ghosh, S. (2004). A toll-like receptor that prevents infection by uropathogenic bacteria. Science *303*, 1522-1526.

Zhao,X., Deak,E., Soderberg,K., Linehan,M., Spezzano,D., Zhu,J., Knipe,D.M., and Iwasaki,A. (2003). Vaginal submucosal dendritic cells, but not Langerhans cells, induce protective Th1 responses to herpes simplex virus-2. J. Exp. Med. *197*, 153-162.

Zhou,G., Galvan,V., Campadelli-Fiume,G., and Roizman,B. (2000). Glycoprotein D or J delivered in trans blocks apoptosis in SK-N-SH cells induced by a herpes simplex virus 1 mutant lacking intact genes expressing both glycoproteins. J. Virol. 74, 11782-11791.

Zhou,G. and Roizman,B. (2001). The domains of glycoprotein D required to block apoptosis depend on whether glycoprotein D is present in the virions carrying herpes simplex virus 1 genome lacking the gene encoding the glycoprotein. J Virol 75, 6166-6172.

Zhou,G. and Roizman,B. (2002). Cation-independent mannose 6-phosphate receptor blocks apoptosis induced by herpes simplex virus 1 mutants lacking glycoprotein D and is likely the target of antiapoptotic activity of the glycoprotein. J. Virol. *76*, 6197-6204.

Zhu,C., Rao,K., Xiong,H., Gagnidze,K., Li,F., Horvath,C., and Plevy,S. (2003). Activation of the Murine Interleukin-12 p40 Promoter by Functional Interactions between NFAT and ICSBP. J. Biol. Chem. 278, 39372-39382.

~

.

Zhu,W., Downey,J.S., Gu,J., Di Padova,F., Gram,H., and Han,J. (2000). Regulation of TNF expression by multiple mitogen-activated protein kinase pathways. J Immunol *164*, 6349-6358.

Zhu,Z., Gershon,M.D., Ambron,R., Gabel,C., and Gershon,A.A. (1995). Infection of cells by varicella zoster virus: inhibition of viral entry by mannose 6-phosphate and heparin. Proc. Natl. Acad. Sci. U. S. A. 92, 3546-3550.